#### UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

### **FORM 10-Q**

(Mark One)

**☑ QUARTERLY REPORT PURSUANT TO SECTION 13 or 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934** 

For the quarterly period ended September 30, 2024

 $\square$  TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

> For the transition period from to Commission File Number: 1-4018



(Exact name of registrant as specified in its charter)

**Delaware** 

(State or other jurisdiction of incorporation or organization) (I.R.S. Employer Identification No.)

3005 Highland Parkway **Downers Grove, Illinois** (Address of principal executive offices)

60515

53-0257888

(Zip Code)

(630) 541-1540

(Registrant's telephone number, including area code)

Securities registered pursuant to Section 12(b) of the Act:

Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Common Stock	DOV	New York Stock Exchange
1.250% Notes due 2026	DOV 26	New York Stock Exchange
0.750% Notes due 2027	DOV 27	New York Stock Exchange

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 19	34
during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such file	ng
requirements for the past 90 days.	

Yes ☑ No □

Indicate by check mark whether the registrant has submitted electronically every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes ☑ No □

2	the definiti	ons of "large accelerated filer," "accelerated	,	on-accelerated filer, a smaller reporting co "smaller reporting company," and "eme	1 2/
Large Accelerated Filer	<b></b>	Accelerated Filer		Emerging Growth Company	
Non-Accelerated Filer	Ш	Smaller Reporting Company	Ц		
	-	check mark if the registrant has elected no ided pursuant to Section 13(a) of the Exch.		extended transition period for complying	with any new
Indicate by check mark whether th	ne registrant	is a shell company (as defined in Rule 12b	-2 of the E	xchange Act). Yes □ No ☑	

The number of shares outstanding of the Registrant's common stock as of October 18, 2024 was 137,191,966.

#### Dover Corporation Form 10-Q Table of Contents

<u>PART I —</u>	<u>- FINANCIAL INFORMATION</u>	Page
Item 1.	<u>Financial Statements (unaudited)</u>	
	Condensed Consolidated Statements of Earnings for the three and nine months ended September 30, 2024 and 2023	<u>1</u>
	Condensed Consolidated Statements of Comprehensive Earnings for the three and nine months ended September 30, 2024 and 2023	<u>2</u>
	Condensed Consolidated Balance Sheets at September 30, 2024 and December 31, 2023	<u>3</u>
	Condensed Consolidated Statements of Stockholders' Equity for the three and nine months ended September 30, 2024 and 2023	<u>4</u>
	Condensed Consolidated Statements of Cash Flows for the nine months ended September 30, 2024 and 2023	<u>6</u>
	Note 1 — Basis of Presentation	1 2 3 4 6 7 7 9
	Note 2 — Revenue	7
	Note 3 — Acquisitions	9
	Note 4 — Discontinued and Disposed Operations	<u>11</u>
	Note 5 — Inventories, net	12
	Note 6 — Property, Plant and Equipment, net	12
	Note 7 — Credit Losses	12
	Note 8 — Goodwill and Other Intangible Assets	<u>13</u>
	Note 9 — Restructuring Activities	12 12 12 13 14
	Note 10 — Borrowings	<u>14</u>
	Note 11 — Financial Instruments	<u>16</u>
	Note 12 — Income Taxes	<u>17</u>
	Note 13 — Equity Incentive Program	<u>17</u> <u>18</u>
	Note 14 — Commitments and Contingent Liabilities	<u>19</u>
	Note 15 — Other Comprehensive Earnings	<u>20</u>
	Note 16 — Segment Information	<u>20</u>
	Note 17 — Stockholders' Equity	20 22 23 23 24 25 43 44
	Note 18 — Earnings per Share	<u>23</u>
	Note 19 — Recent Accounting Pronouncements	<u>23</u>
	Note 20 — Subsequent Events	<u>24</u>
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>25</u>
Item 3.	Quantitative and Qualitative Disclosures About Market Risk	<u>43</u>
Item 4.	Controls and Procedures	<u>44</u>
PART II –	OTHER INFORMATION	
Item 1.	Legal Proceedings	<u>45</u>
Item 1A.	Risk Factors	45
Item 2.	Unregistered Sales of Equity Securities and Use of Proceeds	45
Item 3.	Defaults Upon Senior Securities	45
Item 4.	Mine Safety Disclosures	45 45 45 45 45 45 46
Item 5.	Other Information	45
Item 6.	Exhibits	46
		_
SIGNATU	IRES .	<u>47</u>

#### **Item 1. Financial Statements**

# DOVER CORPORATION CONDENSED CONSOLIDATED STATEMENTS OF EARNINGS (In thousands, except per share data) (Unaudited)

		Three Months En	ded	September 30,	Nine Months End	led S	September 30,
	2024			2023	2024		2023
Revenue	\$	1,983,542	\$	1,958,428	\$ 5,816,043	\$	5,779,664
Cost of goods and services		1,220,355		1,219,047	3,603,146		3,631,023
Gross profit		763,187		739,381	2,212,897		2,148,641
Selling, general and administrative expenses		429,570		402,838	1,301,606		1,234,223
Operating earnings		333,617		336,543	911,291		914,418
Interest expense		34,128		32,390	102,867		100,407
Interest income		(5,176)		(3,808)	(14,013)		(8,552)
Gain on dispositions		(68,633)		_	(597,913)		_
Other income, net		(13,032)		(10,274)	(33,016)		(20,758)
Earnings before provision for income taxes		386,330		318,235	1,453,366		843,321
Provision for income taxes		73,434		56,252	291,781		157,636
Earnings from continuing operations		312,896		261,983	1,161,585		685,685
Earnings from discontinued operations, net		34,204		27,770	99,558		74,881
Net earnings	\$	347,100	\$	289,753	\$ 1,261,143	\$	760,566
Earnings per share from continuing operations:							
Basic	\$		\$	1.87	\$ 	\$	4.90
Diluted	\$	2.26	\$	1.86	\$ 8.37	\$	4.88
Earnings per share from discontinued operations:							
Basic	\$	0.25	\$	0.20	\$ 0.72	\$	0.54
Diluted	\$	0.25	\$	0.20	\$ 0.72	\$	0.53
Net earnings per share:							
Basic	\$	2.53	\$	2.07	\$ 9.14	\$	5.44
Diluted	\$	2.51	\$	2.06	\$ 9.08	\$	5.41
Weighted average shares outstanding:							
Basic		137,251		139,878	137,913		139,833
Diluted		138,223		140,615	138,830		140,603

# DOVER CORPORATION CONDENSED CONSOLIDATED STATEMENTS OF COMPREHENSIVE EARNINGS (In thousands) (Unaudited)

	Thi		End 0,	ed September	N	September		
		2024		2023		2024		2023
Net earnings	\$	347,100	\$	289,753	\$	1,261,143	\$	760,566
Other comprehensive earnings (loss), net of tax				_		_		
Foreign currency translation adjustments:								
Foreign currency translation gain (loss)		55,121		(54,114)		13,176		(16,207)
Reclassification of foreign currency translation losses to earnings		_		_		13,931		_
Total foreign currency translation adjustments (net of \$10,468, \$(7,274), \$3,008 and \$(58) tax benefit (provision), respectively)		55,121		(54,114)		27,107		(16,207)
Pension and other post-retirement benefit plans:				_				
Amortization of actuarial gain included in net periodic pension cost		(363)		(516)		(1,099)		(1,578)
Amortization of prior service (credits) costs included in net periodic pension cost		(160)		245		(472)		764
Total pension and other post-retirement benefit plans (net of \$138, \$82, \$415 and \$247 tax benefit, respectively)		(523)		(271)		(1,571)		(814)
Changes in fair value of cash flow hedges:								
Unrealized net (loss) gain arising during period		(1,107)		587		(246)		246
Net loss (gain) reclassified into earnings		39		369		(665)		2,067
Total cash flow hedges (net of \$315, \$(273), \$269 and \$(660) tax benefit (provision), respectively)		(1,068)		956		(911)		2,313
Other comprehensive earnings (loss), net of tax	, <u> </u>	53,530		(53,429)		24,625		(14,708)
Comprehensive earnings	\$	400,630	\$	236,324	\$	1,285,768	\$	745,858

# DOVER CORPORATION CONDENSED CONSOLIDATED BALANCE SHEETS

(In thousands) (Unaudited)

	Sept	ember 30, 2024	Dec	ember 31, 2023
ASSETS				
Current assets:				
Cash and cash equivalents	\$	386,766	\$	398,561
Receivables, net		1,428,961		1,321,107
Inventories, net		1,214,268		1,144,089
Prepaid and other current assets		132,745		139,348
Assets held for sale - current		583,161		387,130
Total current assets		3,745,901		3,390,235
Property, plant and equipment, net		995,766		978,472
Goodwill		4,974,447		4,637,564
Intangible assets, net		1,645,379		1,445,204
Other assets and deferred charges		551,479		555,084
Assets held for sale - non-current		_		341,954
Total assets	\$	11,912,972	\$	11,348,513
LIABILITIES AND STOCKHOLDERS' EQUITY				
Current liabilities:				
Short-term borrowings	\$	379,295	\$	468,282
Accounts payable		865,188		854,465
Accrued compensation and employee benefits		250,017		258,786
Deferred revenue		185,055		194,798
Accrued insurance		92,543		86,085
Other accrued expenses		334,617		296,895
Federal and other income taxes		62,000		36,878
Liabilities held for sale - current		218,488		217,581
Total current liabilities		2,387,203		2,413,770
Long-term debt		3,007,820		2,991,759
Deferred income taxes		353,609		346,383
Non-current income tax payable		6,158		28,024
Other liabilities		460,183		426,914
Liabilities held for sale - non-current				35,058
Stockholders' equity:				
Total stockholders' equity		5,697,999		5,106,605
Total liabilities and stockholders' equity	\$	11,912,972	\$	11,348,513

# DOVER CORPORATION CONDENSED CONSOLIDATED STATEMENTS OF STOCKHOLDERS' EQUITY (In thousands, except per share data)

(Unaudited)

	Common ock \$1 par value	p	Additional aid-in capital	Retained comprehensive earnings (loss) T			reasury stock	Total stockholders' equity	
Balance at July 1, 2024	\$ 259,971	\$	829,335	\$ 11,768,023	\$	(266,771)	\$	(7,226,935)	\$ 5,363,623
Net earnings	_			347,100		_		_	347,100
Dividends paid (\$0.515 per share)	_		_	(70,723)		_		_	(70,723)
Common stock issued for the exercise of share-based awards	35		(3,856)	_		_		_	(3,821)
Stock-based compensation expense	_		8,290	_		_		_	8,290
Common stock acquired, including accelerated share repurchase program and excise tax	_		53,229	_		_		(53,229)	_
Other comprehensive earnings, net of tax	_		_	_		53,530		_	53,530
Balance at September 30, 2024	\$ 260,006	\$	886,998	\$ 12,044,400	\$	(213,241)	\$	(7,280,164)	\$ 5,697,999

	Common ock \$1 par value	 dditional d-in capital	Retained earnings	 ccumulated other mprehensive loss	Tr	easury stock	s	Total stockholders' equity
Balance at July 1, 2023	\$ 259,818	\$ 875,041	\$ 10,552,433	\$ (227,502)	\$	(6,797,685)	\$	4,662,105
Net earnings	_	_	289,753	_				289,753
Dividends paid (\$0.51 per share)	_	_	(71,408)	_		_		(71,408)
Common stock issued for the exercise of share-based awards	17	(1,283)	_	_		_		(1,266)
Stock-based compensation expense	_	6,745	_	_		_		6,745
Other comprehensive loss, net of tax	_	_		(53,429)				(53,429)
Balance at September 30, 2023	\$ 259,835	\$ 880,503	\$ 10,770,778	\$ (280,931)	\$	(6,797,685)	\$	4,832,500

# DOVER CORPORATION CONDENSED CONSOLIDATED STATEMENTS OF STOCKHOLDERS' EQUITY

(In thousands, except per share data) (Unaudited)

Retained

earnings

Additional paid-in capital

Accumulated other

comprehensive

earnings (loss)

Total stockholders'

equity

Treasury stock

Common stock \$1 par

value

Net earnings						0 ( /	_		<u> </u>
Dividends paid (\$1.54 per share)	Balance at January 1, 2024	\$ 259,842	\$	886,690	\$ 10,995,624	\$ (237,866)	\$	(6,797,685)	\$ 5,106,605
Common stock issued for the exercise of share-based awards         164         (10,890)         —         —         —         —         (10,80)           Stock-based compensation expense         —         32,969         —         —         —         33           Common stock acquired, including accelerated share repurchase program and excise tax         —         (21,771)         —         —         (482,479)         (50-00)           Other comprehensive earnings, net of tax         —         —         —         24,625         —         —         2           Balance at September 30, 2024         \$ 260,006         \$ 886,998         \$ 12,044,400         \$ (213,241)         \$ (7,280,164)         \$ 5,69           Balance at January 1, 2023         \$ 259,644         \$ 867,560         \$ 10,223,070         \$ (266,223)         \$ (6,797,685)         \$ 4,28           Net earnings         —         —         —         760,566         —         —         —         76           Dividends paid (\$1.52 per share)         —         —         (212,882)         —         —         —         (21:00)           Common stock issued for the exercise of share-based awards         191         (12,525)         —         —         —         —         —         —	Net earnings	_		_	1,261,143	_		_	1,261,143
share-based awards         164         (10,890)         —<	Dividends paid (\$1.54 per share)	_		_	(212,367)	_		_	(212,367)
Common stock acquired, including accelerated share repurchase program and excise tax         —         (21,771)         —         —         (482,479)         (500)           Other comprehensive earnings, net of tax         —         —         —         24,625         —         —         2           Balance at September 30, 2024         \$ 260,006         \$ 886,998         \$ 12,044,400         \$ (213,241)         \$ (7,280,164)         \$ 5,69           Common stock \$1 par value         Additional paid-in capital paid-in capital value         Retained carnings         Accumulated other comprehensive loss         Treasury stock required to the carnings         Treasury stock         \$ 4,28           Net earnings         —		164		(10,890)	_	_		_	(10,726)
Common stock   Salance at January 1, 2023   Section   Section   Stock   Section   Se	Stock-based compensation expense	_		32,969	_	_		_	32,969
Other comprehensive earnings, net of tax         —         —         —         24,625         —         2           Balance at September 30, 2024         \$ 260,006         \$ 886,998         \$ 12,044,400         \$ (213,241)         \$ (7,280,164)         \$ 5,69           Common stock \$1 par value         Additional paid-in capital         Retained earnings         Accumulated other comprehensive loss         Treasury stock         stockhold equity           Balance at January 1, 2023         \$ 259,644         \$ 867,560         \$ 10,223,070         \$ (266,223)         \$ (6,797,685)         \$ 4,28           Net earnings         —         —         —         760,566         —         —         —         76           Dividends paid (\$1.52 per share)         —         —         (212,882)         —         —         —         (215,622)         —         —         —         —         (215,622)         —         —         —         —         (215,622)         —	accelerated share repurchase program and	_		(21,771)	_	_		(482,479)	(504,250)
Common stock \$1 par value         Additional paid-in capital paid-in capital         Retained earnings         Accumulated other comprehensive loss         Treasury stock reasury stock         Total stockhold equity           Net earnings         —         —         —         760,566         —         —         —         76           Dividends paid (\$1.52 per share)         —         —         (212,882)         —         —         —         (212,882)         —         —         —         (212,882)         —         —         —         —         (212,882)         —	Other comprehensive earnings, net of tax	_		` _	_	24,625			24,625
Common stock \$1 par value         Additional paid-in capital paid-in capital         Retained earnings         Accumulated other comprehensive loss         Treasury stock reasury stock         Total stockhold equity           Net earnings         —         —         —         760,566         —         —         —         76           Dividends paid (\$1.52 per share)         —         —         (212,882)         —         —         —         (212,882)         —         —         —         (212,882)         —         —         —         —         (212,882)         —	Balance at Sentember 30, 2024	\$ 260,006	\$	886,998	\$ 12,044,400	\$ (213,241)	\$	(7,280,164)	\$ 5,697,999
Balance at January 1, 2023\$ 259,644\$ 867,560\$ 10,223,070\$ (266,223)\$ (6,797,685)\$ 4,28Net earnings———760,566———76Dividends paid (\$1.52 per share)———(212,882)———(212,882)Common stock issued for the exercise of share-based awards191(12,525)——————(12,525)Stock-based compensation expense—25,468—————2						 			
Net earnings       —       —       760,566       —       —       76         Dividends paid (\$1.52 per share)       —       —       (212,882)       —       —       (212,882)         Common stock issued for the exercise of share-based awards       191       (12,525)       —       —       —       —       (12,525)         Stock-based compensation expense       —       25,468       —       —       —       2		ck \$1 par	_			 	Tì	reasury stock	Total stockholders' equity
Dividends paid (\$1.52 per share)       —       —       (212,882)       —       —       (212,882)         Common stock issued for the exercise of share-based awards       191       (12,525)       —       —       —       —       (12,525)       —       —       —       —       —       22,468       —       —       —       2	Balance at January 1, 2023	\$ 259,644	\$	867,560	\$ 10,223,070	\$ (266,223)	\$	(6,797,685)	\$ 4,286,366
Common stock issued for the exercise of share-based awards  191 (12,525) — — — (12,525) — — 25,468 — — 2	Net earnings	_		_	760,566	_		_	760,566
share-based awards       191       (12,525)       —       —       —       —       (12,525)       —       —       —       —       —       —       25,468       —       —       —       —       2	Dividends paid (\$1.52 per share)	_		_	(212,882)	_		_	(212,882)
		191		(12,525)	_	_		_	(12,334)
04	• •	_		25,468	_	_		_	25,468
Other comprehensive loss, net of tax — — — $(14,708)$ — $(14,708)$	Other comprehensive loss, net of tax	_			_	(14,708)		_	(14,708)
Other, net — — 24 — —	0.1					( ))			( ) /
	Other, net			_	24			_	24

#### DOVER CORPORATION CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (In thousands) (Unaudited)

Nine Months Ended September 30, 2024 2023 **Operating Activities:** \$ Net earnings 1,261,143 760,566 Adjustments to reconcile net earnings to cash provided by operating activities: Earnings from discontinued operations, net (99.558)(74.881)Depreciation and amortization 251,179 227,519 Stock-based compensation expense 32,297 24,836 Gain on dispositions (597,913)Other, net (36,033)(1,407)Cash effect of changes in assets and liabilities: Accounts receivable, net (59,082)(5,960)Inventories 42,553 (36,936)Prepaid expenses and other assets (19,780)2,068 Accounts payable (60,549)(82,364)Accrued compensation and employee benefits (30,322)(40,718)Accrued expenses and other liabilities 31,174 (14,884)Accrued and deferred taxes, net (63,224)(39,861)720,982 Net cash provided by operating activities 648,881 **Investing Activities:** Additions to property, plant and equipment (126, 131)(113,626)Acquisitions, net of cash and cash equivalents acquired (602,654)(7,166)Proceeds from dispositions, net of cash transferred 767,689 Other 11,710 2,225 Net cash provided by (used in) investing activities 63,119 (131,072) **Financing Activities:** Repurchase of common stock, including accelerated share repurchase program (500,000) Change in commercial paper and other short-term borrowings, net (528,816) (88,987)Dividends paid to stockholders (212,367)(212,882)Payments to settle employee tax obligations on exercise of share-based awards (13,731)(12,334)Other (3,360)(3,173)Net cash used in financing activities (818,445) (757,205) **Cash Flows from Discontinued Operations:** Net cash provided by operating activities of discontinued operations 108,284 99,281 Net cash used in investing activities of discontinued operations (14,359)(5,451)Net cash provided by discontinued operations

Cash and cash equivalents at beginning of period, including cash held for sale		415,861		380,868
Cash and cash equivalents at end of period, including cash held for sale	\$	396,766	\$	301,098
	Septe	mber 30, 2024	Dece	mber 31, 2023
Cash and cash equivalents	\$	386,766	\$	398,561
Cash and cash equivalents held for sale		10,000		17,300
Cash and cash equivalents, including cash held for sale	\$	396,766	\$	415,861

Effect of exchange rate changes on cash and cash equivalents

Net decrease in cash and cash equivalents, including cash held for sale

93,925

(6,575)

(19,095)

93,830

(6,305)

(79,770)

(Amounts in thousands except share data and where otherwise indicated) (Unaudited)

#### 1. Basis of Presentation

The accompanying unaudited interim condensed consolidated financial statements have been prepared pursuant to the rules and regulations of the Securities and Exchange Commission ("SEC") for interim periods and do not include all of the information and note disclosures required by accounting principles generally accepted in the United States of America ("GAAP") for complete financial statements. These unaudited interim condensed consolidated financial statements should therefore be read in conjunction with the Consolidated Financial Statements and Notes for Dover Corporation ("Dover" or the "Company") for the year ended December 31, 2023, included in the Company's Annual Report on Form 10-K filed with the SEC on February 9, 2024. The year-end condensed consolidated balance sheet was derived from audited financial statements. Certain amounts in the prior year have been reclassified to conform to the current year presentation.

The Environmental Solutions Group ("ESG") business, an operating company within the Engineered Products segment, was held for sale and reported as discontinued operations as of September 30, 2024. Therefore, the Company has classified the results of operations as discontinued operations in the condensed consolidated statements of earnings and the condensed consolidated statements of cash flows and classified the assets and liabilities as held for sale in the condensed consolidated balance sheets for all periods presented. The discussion in the notes to these condensed consolidated financial statements, unless otherwise noted, relates solely to our continuing operations. See Note 4 — Discontinued and Disposed Operations for further details.

The accompanying unaudited interim condensed consolidated financial statements have been prepared in accordance with U.S. GAAP. The preparation of these financial statements requires us to make estimates and assumptions that affect amounts reported in the condensed consolidated financial statements and accompanying disclosures. Although these estimates are based on management's knowledge of current events and expectations about actions that the Company may undertake in the future, actual results may differ from those estimates. Our interim condensed consolidated financial statements are unaudited but reflect all adjustments of a normal, recurring nature that are, in the opinion of management, necessary for a fair statement of results for these interim periods. The results of operations of any interim period are not necessarily indicative of the results of operations for the full year.

#### 2. Revenue

#### Revenue from Contracts with Customers

A majority of the Company's revenue is short cycle in nature with shipments within one year from order. A small portion of the Company's revenue derives from contracts extending over one year. The Company's payment terms generally range between 30 to 90 days and vary by the location of businesses, the type of products manufactured to be sold and the volume of products sold, among other factors.

#### Disaggregation of Revenue

Revenue from contracts with customers is disaggregated by segment and geographic location, as they best depict the nature and amount of the Company's revenue. See Note 16 — Segment Information for further details.

#### **Performance Obligations**

Approximately 95% of the Company's revenue is recognized at a point in time, rather than over time as the Company completes its performance obligations. Specifically, revenue is recognized when control transfers to the customer, typically upon shipment or completion of installation, testing, certification, or other substantive acceptance provisions required under the contract. Approximately 5% of the Company's revenue is recognized over time and relates to the sale of equipment or services, including software solutions and services, in which the Company transfers control of a good or service over time and the customer simultaneously receives and consumes the benefits provided by the Company's performance as the Company performs, or the Company's performance creates or enhances an asset the customer controls as the asset is created or enhanced, or the Company's performance does not create an asset with an alternative use to the Company and the Company has an enforceable right to payment for its performance to date plus a reasonable margin.

A majority of the Company's contracts have a single performance obligation which represents, in most cases, the equipment or product being sold to the customer. Some contracts include multiple performance obligations such as a product and the related

(Amounts in thousands except share data and where otherwise indicated) (Unaudited)

installation, extended warranty, software and digital solutions, and/or maintenance services. For contracts with multiple performance obligations, the Company allocates the total transaction price to each performance obligation in an amount based on the estimated relative standalone selling prices of the promised goods or services underlying each performance obligation.

At September 30, 2024, we estimated that \$180,209 in revenue is expected to be recognized in the future related to performance obligations that are unsatisfied (or partially unsatisfied) at the end of the reporting period. We expect to recognize approximately 65.7% of the Company's unsatisfied (or partially unsatisfied) performance obligations as revenue through 2025, 17.8% in 2026, with the remaining balance to be recognized in 2027 and thereafter.

The Company applied the standard's practical expedient that permits the omission of unsatisfied performance obligations for (i) contracts with an original expected length of one year or less and (ii) contracts for which the Company recognizes revenue at the amount to which the Company has the right to invoice for services performed.

#### Contract Balances

Contract assets primarily relate to the Company's right to consideration for work completed but not billed at the reporting date. Contract liabilities relate to advance consideration received from customers or advance billings for which revenue has not been recognized and are reduced when the associated revenue from the contract is recognized.

The following table provides information about contract assets and contract liabilities from contracts with customers:

	Septen	nber 30, 2024	Decen	nber 31, 2023	<b>December 31, 2022</b>
Contract assets - current	\$	12,852	\$	19,561	\$ 11,074
Contract liabilities - current		185,055		194,798	241,595
Contract liabilities - non-current		15,449		7,098	6,417

The revenue recognized during the nine months ended September 30, 2024 and 2023 that was included in contract liabilities at the beginning of the period amounted to \$162,670 and \$201,395, respectively.

(Amounts in thousands except share data and where otherwise indicated) (Unaudited)

#### 3. Acquisitions

#### 2024 Acquisitions

During the nine months ended September 30, 2024, the Company acquired seven businesses in separate transactions for total consideration of \$636,390, net of cash acquired and inclusive of contingent consideration of \$33,736 (a non-cash financing activity). These businesses were acquired to complement and expand upon existing operations within the Clean Energy & Fueling, Engineered Products and Imaging & Identification segments. The goodwill recorded as a result of these acquisitions represents the economic benefits expected to be derived from product line expansions and operational synergies and is non-deductible for income tax purposes.

On July 19, 2024, the Company acquired 100% of the equity interests in the Marshall Excelsior Company ("MEC"), a supplier of highly-engineered flow control components for transportation, storage, and use in liquified petroleum gas and other industrial gases, for \$392,142, net of cash acquired. The MEC acquisition expands the Company's critical flow control capabilities in the Clean Energy & Fueling segment. In connection with this acquisition, the Company recorded goodwill of \$183,878 and intangible assets of \$194,100, primarily related to customer intangibles. The fair value for customer intangibles at the acquisition date was determined using the multi-period excess earnings method under the income approach. The fair value measurements of intangible assets are based on significant unobservable inputs, and thus represent Level 3 inputs. Significant assumptions used in assessing the fair values of intangible assets include discounted future cash flows, customer attrition rates and discount rates. The fair values of the assets acquired and liabilities assumed, and the related tax balances, are based on preliminary estimates and assumptions. These preliminary estimates and assumptions could change significantly during the measurement period as the Company finalizes the valuations of the assets acquired and liabilities assumed, and the related tax balances.

The following presents the preliminary allocation of purchase price to the assets acquired and liabilities assumed, for the MEC 2024 acquisition, based on the estimated fair values at acquisition date:

	Total
Current assets, net of cash acquired	\$ 58,637
Property, plant and equipment	10,300
Goodwill	183,878
Intangible assets	194,100
Other assets and deferred charges	5,602
Current liabilities	(17,896)
Non-current liabilities	 (42,479)
Net assets acquired	\$ 392,142

On January 17, 2024, the Company acquired 100% of the equity interests in the Transchem Group ("Transchem"), a supplier of car wash chemicals and associated solutions, for \$48,241, net of cash acquired and inclusive of contingent consideration. The Transchem acquisition expands the Company's chemical product offerings in the Clean Energy & Fueling segment, specializing in wash performance and water reclaim technology that reduces water usage and lowers car wash operators' cost. In connection with this acquisition, the Company recorded goodwill of \$24,712 and intangible assets of \$26,309, primarily related to customer intangibles.

On January 31, 2024, the Company acquired 100% of the equity interests in Bulloch Technologies, Inc. ("Bulloch"), a provider of point-of-sale ("POS"), forecourt controller and electronic payment server solutions to the convenience retail industry, for \$121,917, net of cash acquired and inclusive of contingent consideration. The acquisition of Bulloch expands the Company's offering in North America with highly complementary POS and forecourt solutions within the Clean Energy & Fueling segment. In connection with this acquisition, the Company recorded goodwill of \$73,850 and intangible assets of \$62,417, primarily related to customer intangibles.

On July 18, 2024, the Company acquired 100% of the equity interests in Demaco Holland B.V. ("Demaco"), a provider of critical flow control components for cryogenic applications used in a wide range of end markets, for \$42,556, net of cash acquired and inclusive of contingent consideration. The acquisition of Demaco expands the Company's offering within the Clean Energy & Fueling segment. In connection with this acquisition, the Company recorded goodwill of \$22,698 and intangible assets of \$20,159, primarily related to customer intangibles.

(Amounts in thousands except share data and where otherwise indicated) (Unaudited)

On August 9, 2024, the Company acquired 100% of the equity interest in Criteria Labs, Inc. ("Criteria Labs"), a provider of radio frequency devices and microelectronic engineering solutions tailored for high-reliability applications, for \$14,914, net of cash acquired and inclusive of contingent consideration. The acquisition of Criteria Labs expands the Company's offerings within the Engineered Products segment. In connection with this acquisition, the Company recorded goodwill of \$7,622 and intangible assets of \$7,900, primarily related to unpatented technologies.

On August 9, 2024, the Company acquired 100% of the equity interest in SPS Cryogenics B.V. and Special Gas Systems B.V ("SPS Cryogenics"), a designer, manufacturer, and supplier of vacuum-insulated piping systems for a wide variety of liquified gases, for \$10,924, net of cash acquired. The acquisition of SPS Cryogenics expands the Company's presence in Europe with highly complementary offerings within the Clean Energy & Fueling segment. In connection with this acquisition, the Company recorded goodwill of \$4,779 and intangible assets of \$5,677, primarily related to customer intangibles.

One other immaterial acquisition was completed during the nine months ended September 30, 2024, within the Imaging & Identification segment. The acquisition is highly complementary to our existing track and trace solutions business, grows our presence in the European market and adds complementary offerings to our portfolio.

The following presents, for the six acquisitions other than MEC, the preliminary allocation of purchase price to the assets acquired and liabilities assumed, based on their estimated fair values at acquisition date:

	Total
Current assets, net of cash acquired	\$ 25,873
Property, plant and equipment	5,038
Goodwill	133,661
Intangible assets	126,358
Other assets and deferred charges	9,520
Current liabilities	(14,934)
Non-current liabilities	 (41,268)
Net assets acquired	\$ 244,248

The amounts assigned to goodwill and major intangible asset classifications for all 2024 acquisitions were as follows:

	Amou	int allocated	Useful life (in years)
Goodwill - non-deductible	\$	317,539	na
Customer intangibles		273,625	9 - 15
Unpatented technologies		29,021	5 - 8
Trademarks		17,812	15
	\$	637,997	

#### 2023 Acquisitions

On August 28, 2023, the Company acquired 100% of the equity interests in the Arc Pacific group ("Arc Pacific"), a global supplier of can washers, dry-off, pin and internal bake ovens for the metal packaging industry, for \$8,833, net of cash acquired and including contingent consideration. The Arc Pacific acquisition extends the Company's reach into can processing equipment production within the Climate & Sustainability Technologies segment. In connection with this acquisition, the Company recorded goodwill of \$2,990 and intangible assets of \$7,670, primarily related to customer intangibles. During the nine months ended September 30, 2024, the Company recorded measurement period adjustments resulting in an increase to goodwill of \$371 and an increase in purchase price of \$250.

(Amounts in thousands except share data and where otherwise indicated) (Unaudited)

#### 4. Discontinued and Disposed Operations

#### **Discontinued Operations**

On July 21, 2024, the Company signed a definitive agreement to sell its ESG business, which is part of the Company's Engineered Products segment, to Terex Corporation for total consideration, net of cash transferred, of \$2.0 billion, subject to post-closing adjustments. The ESG business was classified as held for sale as of September 30, 2024, and qualifies for discontinued operations reporting because its disposal represents a strategic shift that will have a major effect on the Company's operations and financial results. As a result, the Company has classified the results of operations as discontinued operations in the condensed consolidated statements of earnings and the condensed consolidated statements of cash flows and classified the assets and liabilities as held for sale in the condensed consolidated balance sheets for all periods presented.

Summarized results of the Company's discontinued operations are as follows:

	Three Months En	ded	l September 30,	Nine Months End	led September 30,			
	 2024		2023	2024		2023		
Revenue	\$ 231,777	\$	194,840	\$ 671,479	\$	552,713		
Cost of goods and services	164,115		141,206	474,705		402,484		
Gross profit	67,662		53,634	196,774		150,229		
Selling, general and administrative expenses	24,318		17,407	67,599		52,776		
Operating earnings	43,344		36,227	129,175		97,453		
Other (income) expense, net	(1)		_	696		(1)		
Earnings from discontinued operations before	 _							
provision for income taxes	43,345		36,227	128,479		97,454		
Provision for income taxes	9,141		8,457	28,921		22,573		
Earnings from discontinued operations, net	\$ 34,204	\$	27,770	\$ 99,558	\$	74,881		

Assets and liabilities that were held for sale are summarized below:

	September 30, 2024			December 31, 2023
Assets:				
Cash and cash equivalents	\$	10,000	\$	_
Receivables, net		138,531		110,933
Inventories, net		79,986		81,363
Prepaid and other current assets		2,156		2,190
Property, plant and equipment, net		69,301		53,344
Goodwill		244,123		244,123
Intangible assets, net		35,536		38,709
Other assets and deferred charges		3,528		5,778
Total assets held for sale	\$	583,161	\$	536,440
Liabilities:				
Accounts payable	\$	128,886	\$	104,077
Other current liabilities		55,217		48,936
Other non-current liabilities		34,385		35,058
Total liabilities held for sale	\$	218,488	\$	188,071

For the period ended September 30, 2024, all assets and liabilities held for sale are presented as current in the condensed consolidated balance sheets due to the timing of the transaction close. For the period ended December 31, 2023, current assets and liabilities held for sale of \$194,486 and \$153,013, respectively, and non-current assets and liabilities held for sale of \$341,954 and \$35,058, respectively, are presented in the condensed consolidated balance sheets.

(Amounts in thousands except share data and where otherwise indicated) (Unaudited)

#### 2024 Dispositions

On March 31, 2024, the Company completed the sale of the De-Sta-Co business, an operating company within the Engineered Products segment, for total consideration, net of cash transferred, of \$674,727. Of the total consideration, \$63,000 was received upon finalization of closing activities in India and China, which occurred during the second quarter. This sale resulted in a preliminary pre-tax gain on disposition of \$529,201 (\$414,372 after-tax) included within the condensed consolidated statements of earnings for the nine months ended September 30, 2024. The total consideration and pre-tax gain on disposition are preliminary and subject to standard post-closing adjustments. The sale did not meet the criteria to be classified as a discontinued operation, as it did not represent a strategic shift that would have a major effect on operations and financial results.

De-Sta-Co met the criteria to be classified as held for sale beginning September 30, 2023. For the period ended December 31, 2023, current assets and liabilities held for sale of \$192,644 and \$64,568, respectively, are presented in the condensed consolidated balance sheets.

On September 30, 2024, a minority owned equity method investment held within the Climate & Sustainability Technologies segment was sold and the Company received its proportionate share of the proceeds amounting to \$92,962. The sale resulted in a preliminary pre-tax gain of \$68,712, subject to customary post-closing adjustments and included within the condensed consolidated statements of earnings for the three and nine months ended September 30, 2024.

#### 2023 Dispositions

There were no dispositions during the nine months ended September 30, 2023.

#### 5. Inventories, net

	Septem	ber 30, 2024	Dece	mber 31, 2023
Raw materials	\$	685,728	\$	648,896
Work in progress		254,568		214,934
Finished goods		405,035		395,617
Subtotal		1,345,331		1,259,447
Less reserves		(131,063)		(115,358)
Total	\$	1,214,268	\$	1,144,089

#### 6. Property, Plant and Equipment, net

	<b>September 30, 2024</b>	<b>December 31, 2023</b>
Land	\$ 65,164	\$ 64,722
Buildings and improvements	631,262	592,136
Machinery, equipment and other	1,960,607	1,860,315
Property, plant and equipment, gross	2,657,033	2,517,173
Accumulated depreciation	(1,661,267)	(1,538,701)
Property, plant and equipment, net	\$ 995,766	\$ 978,472

Depreciation expense totaled \$38,830 and \$39,177 for the three months ended September 30, 2024 and 2023, respectively. For the nine months ended September 30, 2024 and 2023, depreciation expense totaled \$115,210 and \$113,620, respectively.

#### 7. Credit Losses

The Company is exposed to credit losses primarily through sales of products and services. Due to the short-term nature of such receivables, the estimate of the amount of accounts receivable that may not be collected is based on the aging of the accounts receivable balances and other historical and forward-looking information on the financial condition of customers. Balances are written off when determined to be uncollectible.

(Amounts in thousands except share data and where otherwise indicated) (Unaudited)

The following table provides a rollforward of the allowance for credit losses deducted from accounts receivable that represent the net amount expected to be collected.

	2024	2023
Balance at January 1	\$ 30,679 \$	38,504
Provision for expected credit losses, net of recoveries	4,039	636
Amounts written off charged against the allowance	(4,559)	(4,848)
Other, including foreign currency translation	 (63)	(1,429)
Balance at September 30	\$ 30,096 \$	32,863

#### 8. Goodwill and Other Intangible Assets

The changes in the carrying value of goodwill by reportable operating segments were as follows:

	Engineered Products		Clean Energy & Fueling		Imaging & Identification		Pumps & Process Solutions		Climate & Sustainability Technologies	Total
Balance at January 1, 2024	\$	415,258	\$	1,409,302	\$ 1,092,960	\$	1,208,571	\$	511,473	\$ 4,637,564
Acquisitions		7,622		309,917	_				_	317,539
Measurement period adjustments		_		_	_		227		371	598
Foreign currency translation		1,606		9,374	3,564		3,860		342	18,746
Balance at September 30, 2024	\$	424,486	\$	1,728,593	\$ 1,096,524	\$	1,212,658	\$	512,186	\$ 4,974,447

During the nine months ended September 30, 2024, the Company recognized additions of \$317,539 to goodwill as a result of the acquisitions discussed in Note 3 — Acquisitions, and disposed of \$58,663 of goodwill that was previously classified as held for sale as of December 31, 2023. See Note 4 — Discontinued and Disposed Operations for further details.

The Company's definite-lived and indefinite-lived intangible assets by major asset class were as follows:

	<b>September 30, 2024</b>					December 31, 2023							
	Gr	Gross Carrying Amount				Net Carrying Amount		Gross Carrying Amount		Accumulated Amortization	Net Carrying Amount		
Amortized intangible assets:	'												
Customer intangibles	\$	2,378,879	\$	1,165,273	\$	1,213,606	\$	2,097,985	\$	1,064,609	\$	1,033,376	
Trademarks		287,546		155,554		131,992		268,605		142,505		126,100	
Patents		205,000		146,652		58,348		204,591		140,438		64,153	
Unpatented technologies		274,712		167,721		106,991		244,042		151,944		92,098	
Distributor relationships		83,450		67,976		15,474		82,031		63,343		18,688	
Other		35,034		12,705		22,329		24,211		10,053		14,158	
Total		3,264,621		1,715,881		1,548,740		2,921,465		1,572,892		1,348,573	
Unamortized intangible assets:													
Trademarks		96,639		_		96,639		96,631		_		96,631	
Total intangible assets, net	\$	3,361,260	\$	1,715,881	\$	1,645,379	\$	3,018,096	\$	1,572,892	\$	1,445,204	

For the three months ended September 30, 2024 and 2023, amortization expense was \$47,838 and \$38,205, respectively. For the nine months ended September 30, 2024 and 2023, amortization expense was \$135,969 and \$113,899, respectively. Amortization expense is primarily comprised of acquisition-related intangible amortization.

During the nine months ended September 30, 2024, the Company acquired \$320,458 of intangible assets through acquisitions. These assets were classified as customer intangibles, unpatented technologies and trademarks and included in the Clean Energy & Fueling, Engineered Products and Imaging & Identification segments. See Note 3 — Acquisitions for further details.

(Amounts in thousands except share data and where otherwise indicated) (Unaudited)

#### 9. Restructuring Activities

The Company's restructuring charges by segment were as follows:

	Thre	e Months En	ded S	September 30,	Nine Months Ended September 30				
		2024		2023		2024		2023	
Engineered Products	\$	991	\$	938	\$	2,969	\$	5,415	
Clean Energy & Fueling		8,544		(37)		15,434		15,954	
Imaging & Identification		1,804		233		4,645		1,437	
Pumps & Process Solutions		964		1,637		3,929		6,266	
Climate & Sustainability Technologies		1,238		1,138		14,261		2,585	
Corporate		296		476		391		1,603	
Total	\$	13,837	\$	4,385	\$	41,629	\$	33,260	
These amounts are classified in the condensed consolidated statement	ts of ear	nings as follo	ws:						
Cost of goods and services	\$	7,768	\$	1,198	\$	26,908	\$	10,353	
Selling, general and administrative expenses		6,069		3,187		14,721		22,907	
Total	\$	13,837	\$	4,385	\$	41,629	\$	33,260	

The restructuring expenses of \$13,837 incurred during the three months ended September 30, 2024 were primarily related to exit costs and headcount reductions in the Clean Energy & Fueling segment. The restructuring expenses of \$41,629 incurred during the nine months ended September 30, 2024 were primarily related to product line exit costs and headcount reductions in the Clean Energy & Fueling and Climate & Sustainability Technologies segments. These restructuring programs were initiated in 2023 and 2024 and the Company will continue to make proactive adjustments to its cost structure to align with current demand trends.

The Company's severance and exit accrual activities were as follows:

	Severance	Exit	Total
Balance at January 1, 2024	\$ 18,646	\$ 3,113	\$ 21,759
Restructuring charges	15,745	25,884 (1)	41,629
Payments	(23,973)	(7,625)	(31,598)
Other, including foreign currency translation	(178)	$(16,828)^{(1)}$	(17,006)
Balance at September 30, 2024	\$ 10,240	\$ 4,544	\$ 14,784

<sup>(1)</sup> Exit reserves activity includes non-cash asset charges related to a product line exit within the Climate & Sustainability Technologies segment.

#### 10. Borrowings

Borrowings consist of the following:

	Septer	mber 30, 2024	De	cember 31, 2023
Short-term Short-term			'	
Commercial paper	\$	378,600	\$	467,600
Other		695		682
Short-term borrowings	\$	379,295	\$	468,282

During the nine months ended September 30, 2024, commercial paper borrowings decreased \$89,000. The borrowings outstanding under the commercial paper program had a weighted average annual interest rate of 4.94% and 5.51% as of September 30, 2024 and December 31, 2023.

(Amounts in thousands except share data and where otherwise indicated) (Unaudited)

				Carrying a	ımount <sup>(1)</sup>		
		Principal	Se	ptember 30, 2024	Г	December 31, 2023	
Long-term	<u></u>						
3.15% 10-year notes due November 15, 2025	\$	400,000	\$	399,242	\$	398,737	
1.25% 10-year notes due November 9, 2026 (euro-denominated)	€	600,000		665,676		657,628	
0.750% 8-year notes due November 4, 2027 (euro-denominated)	€	500,000		553,996		547,342	
6.65% 30-year debentures due June 1, 2028	\$	200,000		199,632		199,557	
2.950% 10-year notes due November 4, 2029	\$	300,000		298,071		297,787	
5.375% 30-year debentures due October 15, 2035	\$	300,000		297,245		297,058	
6.60% 30-year notes due March 15, 2038	\$	250,000		248,477		248,392	
5.375% 30-year notes due March 1, 2041	\$	350,000		345,465		345,258	
Other				16		_	
Total long-term debt			\$	3,007,820	\$	2,991,759	

<sup>(1)</sup> Carrying amount is net of unamortized debt discount and deferred debt issuance costs. Total unamortized debt discounts were \$9.5 million and \$10.9 million as of September 30, 2024 and December 31, 2023, respectively. Total deferred debt issuance costs were \$7.5 million and \$8.9 million as of September 30, 2024 and December 31, 2023, respectively.

The discounts are being amortized to interest expense using the effective interest method over the life of the issuances. The deferred issuance costs are amortized on a straight-line basis over the life of the debt, as this approximates the effective interest method.

On April 6, 2023, the Company entered into a \$1.0 billion five-year unsecured revolving credit facility and on April 4, 2024, the Company entered into a new \$500.0 million 364-day unsecured revolving credit facility (together, the "Credit Agreements") with a syndicate of banks. The current 364-day credit facility replaced the previous \$500.0 million 364-day credit facility, which expired on April 4, 2024. The lenders' commitments under the Credit Agreements will terminate and any outstanding loans under the Credit Agreements will mature on April 6, 2028 and April 3, 2025, respectively. The Company may elect to extend the maturity date of any loans under the new 364-day credit facility until April 3, 2026, subject to conditions specified therein. The Credit Agreements are designated as a liquidity back-stop for the Company's commercial paper program and also are available for general corporate purposes. At the Company's election, loans under the Credit Agreements will bear interest at a base rate plus an applicable margin. The Credit Agreements require the Company to pay facility fees and impose various restrictions on the Company such as, among other things, a requirement to maintain a minimum interest coverage ratio of consolidated EBITDA to consolidated net interest expense of not less than 3.0 to 1. As of September 30, 2024 and December 31, 2023, there were no outstanding borrowings under the five-year, current or previous 364-day credit facilities.

The Company was in compliance with all covenants in the Credit Agreements and other long-term debt covenants at September 30, 2024 and had an interest coverage ratio of consolidated EBITDA to consolidated net interest expense of 20.7 to 1.

#### Letters of Credit and other Guarantees

As of September 30, 2024, the Company had approximately \$160.0 million outstanding in letters of credit, surety bonds, and performance and other guarantees which primarily expire on various dates through 2035. These letters of credit and bonds are primarily issued as security for insurance, warranty and other performance obligations. In general, we would only be liable for the amount of these guarantees in the event of default in the performance of our obligations, the probability of which is believed to be remote.

(Amounts in thousands except share data and where otherwise indicated) (Unaudited)

#### 11. Financial Instruments

#### **Derivatives**

The Company is exposed to market risk for changes in foreign currency exchange rates due to the global nature of its operations and certain commodity risks. In order to manage these risks, the Company has hedged portions of its forecasted sales and purchases which occur within the next twelve months that are denominated in non-functional currencies, with currency forward contracts designated as cash flow hedges. At September 30, 2024 and December 31, 2023, the Company had contracts with total notional amounts of \$153,054 and \$171,425, respectively, to exchange currencies, principally euro, pound sterling, Swedish krona, Canadian dollar, Chinese yuan, and Swiss franc. The Company believes it is probable that all forecasted cash flow transactions will occur.

In addition, the Company had outstanding contracts with a total notional amount of \$91,892 and \$84,867 as of September 30, 2024 and December 31, 2023, respectively, that are not designated as hedging instruments. These instruments are used to reduce the Company's exposure for operating receivables and payables that are denominated in non-functional currencies. Gains and losses on these contracts are recorded in other income, net in the condensed consolidated statements of earnings.

The following table sets forth the fair values of derivative instruments held by the Company as of September 30, 2024 and December 31, 2023 and the balance sheet lines in which they are recorded:

	Fair Value As	sset (Liability)	
	September 30, 2024	December 31, 2023	<b>Balance Sheet Caption</b>
Foreign currency forward	\$ 1,188	\$ 1,675	Prepaid and other current assets
Foreign currency forward	(1,531)	(874)	Other accrued expenses

For a cash flow hedge, the change in estimated fair value of a hedging instrument is recorded in accumulated other comprehensive earnings (loss), net of tax as a separate component of the condensed consolidated statements of stockholders' equity and is reclassified into revenues or cost of goods and services in the condensed consolidated statements of earnings during the period in which the hedged transaction is settled. The amount of gains or losses from hedging activity recorded in earnings is not significant, and the amount of unrealized gains and losses from cash flow hedges that are expected to be reclassified to earnings in the next twelve months is not significant; therefore, additional tabular disclosures are not presented. There are no amounts excluded from the assessment of hedge effectiveness, and the Company's derivative instruments that are subject to credit risk contingent features were not significant.

The Company is exposed to credit loss in the event of nonperformance by counterparties to the financial instrument contracts held by the Company; however, nonperformance by these counterparties is considered unlikely as the Company's policy is to contract with highly-rated, diversified counterparties.

The Company has designated the €600,000 and €500,000 of euro-denominated notes issued November 9, 2016 and November 4, 2019, respectively, as hedges of a portion of its net investment in euro-denominated operations. Changes in the value of the euro-denominated debt are recognized in foreign currency translation adjustments within other comprehensive earnings (loss) of the condensed consolidated statements of comprehensive earnings to offset changes in the value of the net investment in euro-denominated operations. Changes in the value of the euro-denominated debt resulting from exchange rate differences are offset by changes in the net investment due to the high degree of effectiveness between the hedging instruments and the exposure being hedged.

Amounts recognized in other comprehensive earnings for the gains (losses) on net investment hedges were as follows:

	TI	Three Months Ended September 30,				Nine Months Ended September 30,			
		2024		2023		2024		2023	
Gain (loss) on euro-denominated debt	\$	(45,956)	\$	32,773	\$	(13,201)	\$	262	
Tax (expense) benefit		10,468		(7,274)		3,008		(58)	
Net gain (loss) on net investment hedges, net of tax	\$	(35,488)	\$	25,499	\$	(10,193)	\$	204	

(Amounts in thousands except share data and where otherwise indicated) (Unaudited)

#### Fair Value Measurements

ASC 820, Fair Value Measurements and Disclosures, establishes a fair value hierarchy that requires the Company to maximize the use of observable inputs and minimize the use of unobservable inputs when measuring fair value. A financial instrument's categorization within the hierarchy is based on the lowest level of input that is significant to the fair value measurement. ASC 820 establishes three levels of inputs that may be used to measure fair value as follows:

Level 1 inputs are quoted prices (unadjusted) in active markets for identical assets or liabilities.

Level 2 inputs include inputs other than Level 1 that are observable, either directly or indirectly, such as quoted prices in active markets for similar assets and liabilities, quoted prices for identical or similar assets or liabilities in markets that are not active, or other inputs that are observable or can be corroborated by observable market data for substantially the full term of assets or liabilities.

Level 3 inputs are unobservable inputs in which little or no market data exists, therefore requiring an entity to develop its own assumptions.

The following table presents the Company's assets and liabilities measured at fair value on a recurring basis as of September 30, 2024 and December 31, 2023:

	Septemb	er 30, 2024	<b>December 31, 2023</b>		
	Le	evel 2	Le	vel 2	
Assets:					
Foreign currency cash flow hedges	\$	1,188	\$	1,675	
Liabilities:					
Foreign currency cash flow hedges		1,531		874	

The derivative contracts are measured at fair value using models based on observable market inputs such as foreign currency exchange rates and interest rates; therefore, they are classified within Level 2 of the fair value hierarchy.

In addition to fair value disclosure requirements related to financial instruments carried at fair value, accounting standards require disclosures regarding the fair value of all of the Company's financial instruments.

The estimated fair value of long-term debt at September 30, 2024 and December 31, 2023, was \$3,025,107 and \$2,950,401, respectively. The estimated fair value of long-term debt is based on quoted market prices for similar instruments and is, therefore, classified as Level 2 within the fair value hierarchy.

The carrying values of cash and cash equivalents, trade receivables, accounts payable and short-term borrowings approximate their fair values as of September 30, 2024 and December 31, 2023 due to the short-term nature of these instruments.

#### 12. Income Taxes

The effective tax rates for the three months ended September 30, 2024 and 2023 were 19.0% and 17.7%, respectively. The increase in the effective tax rate for the three months ended September 30, 2024 relative to the prior year comparable period was primarily driven by a gain on disposition.

The effective tax rates for the nine months ended September 30, 2024 and 2023 were 20.1% and 18.7%, respectively. The increase in the effective tax rate for the nine months ended September 30, 2024 relative to the prior year comparable period was primarily driven by gains on dispositions.

Dover and its subsidiaries file tax returns in the U.S., including various state and local returns, and in other foreign jurisdictions. We believe adequate provision has been made for all income tax uncertainties. The Company is routinely audited by taxing authorities in its filing jurisdictions, and a number of these audits are currently underway. The Company believes that within the next twelve months uncertain tax positions may be resolved and statutes of limitations will expire, which could result in a decrease in the gross amount of unrecognized tax benefits of approximately \$0 to \$3,751.

(Amounts in thousands except share data and where otherwise indicated) (Unaudited)

#### 13. Equity Incentive Program

The Company typically makes its annual grants of equity awards pursuant to actions taken by the Compensation Committee of the Board of Directors at its regularly scheduled first quarter meeting. During the nine months ended September 30, 2024, the Company issued stock-settled appreciation rights ("SARs") covering 345,354 shares, performance share awards ("PSAs") of 43,602 and restricted stock units ("RSUs") of 87,755. During the nine months ended September 30, 2023, the Company issued SARs covering 349,491 shares, PSAs of 43,656 and RSUs of 85,053.

The Company uses the Black-Scholes option pricing model to determine the fair value of each SAR on the date of grant. Expected volatilities are based on Dover's stock price history, including implied volatilities from traded options on Dover stock. The Company uses historical data to estimate SAR exercise and employee termination patterns within the valuation model. The expected life of SARs granted is derived from the output of the option valuation model and represents the average period of time that SARs granted are expected to be outstanding. The interest rate for periods within the contractual life of the awards is based on the U.S. Treasury yield curve in effect at the time of grant.

The assumptions used in determining the fair value of the SARs awarded during the respective periods were as follows:

	SAF	Rs
	2024	2023
Risk-free interest rate	4.13 %	3.91 %
Dividend yield	1.28 %	1.32 %
Expected life (years)	5.5	5.4
Volatility	31.32 %	30.65 %
Grant price	\$160.11	\$153.25
Fair value per share at date of grant	\$51.17	\$47.27

The PSAs granted in 2024 vest based on the attainment of two equally weighted measures: (i) Dover's performance relative to established internal metrics (performance condition) and (ii) Dover's performance relative to its peer group (companies listed under the S&P 500 Industrials sector; market condition).

The grant date fair value of the performance condition portion is determined using Dover's closing stock price at the date of grant and the amount of expense recognized over the vesting period is subject to adjustment based on the expected attainment of the performance condition. The fair value per share at the date of grant for the 2024 performance condition portion is \$177.19.

The grant date fair value of the 2024 market condition portion, and all 2023 PSAs, is determined using the Monte Carlo simulation model. The amount of expense recognized over the vesting period is not subject to change based on future market conditions. The assumptions used in the Monte Carlo model to determine the fair value of the PSAs granted in the respective periods were as follows:

	PSAs	
	2024	2023
Risk-free interest rate	4.37 %	4.28 %
Dividend yield	1.15 %	1.32 %
Expected life (years)	2.8	2.9
Volatility	23.30 %	27.30 %
Grant price	\$177.19	\$153.25
Fair value per share at date of grant	\$287.62	\$249.48

The performance and vesting periods for all 2024 and 2023 PSAs is three years.

The Company also has granted RSUs, and the fair value of these awards was determined using Dover's closing stock price on the date of grant, which was \$160.11 and \$153.25 for RSUs granted in 2024 and 2023, respectively.

(Amounts in thousands except share data and where otherwise indicated) (Unaudited)

Stock-based compensation is reported within selling, general and administrative expenses in the condensed consolidated statements of earnings. The following table summarizes the Company's compensation expense relating to all stock-based incentive plans:

	Three Months Ended September 30,				Nine Months Ended September 30			
		2024		2023		2024		2023
Pre-tax stock-based compensation expense	\$	8,187	\$	6,687	\$	32,297	\$	24,836
Tax benefit		(729)		(692)		(3,141)		(2,536)
Total stock-based compensation expense, net of tax	\$	7,458	\$	5,995	\$	29,156	\$	22,300

Pre-tax stock-based compensation expense attributable to discontinued operations was \$103 and \$58 for the three months ended September 30, 2024 and 2023, respectively, and \$672 and \$632 for the nine months ended September 30, 2024 and 2023, respectively. These expenses are included within stock-based compensation expense in the condensed consolidated statements of stockholders' equity. See Note 4 — Discontinued and Disposed Operations for further details

#### 14. Commitments and Contingent Liabilities

#### Litigation

A few of the Company's subsidiaries are involved in legal proceedings relating to the cleanup of waste disposal sites identified under federal and state statutes which provide for the allocation of such costs among "potentially responsible parties." In each instance, the extent of the Company's liability appears to be relatively insignificant in relation to the total projected expenditures and the number of other "potentially responsible parties" involved and is anticipated to be immaterial to the Company. In addition, a few of the Company's subsidiaries are involved in ongoing remedial activities at certain current and former plant sites, in cooperation with regulatory agencies, and appropriate estimated liabilities have been established. At September 30, 2024 and December 31, 2023, these estimated liabilities for environmental and other matters, including private party claims for exposure to hazardous substances that are probable and estimable, were not significant.

The Company and some of its subsidiaries are also parties to a number of other legal proceedings incidental to their businesses. These proceedings primarily involve claims by private parties alleging injury arising out of use of the Company's products, patent infringement, employment matters and commercial disputes. Management and legal counsel, at least quarterly, review the probable outcome of such proceedings, the costs and expenses reasonably expected to be incurred and currently accrued to-date and consider the availability and extent of insurance coverage. The Company has estimated liabilities for these other legal matters that are probable and estimable, and at September 30, 2024 and December 31, 2023, these estimated liabilities were immaterial. While it is not possible at this time to predict the outcome of these legal actions, in the opinion of management, based on the aforementioned reviews, the Company is not currently involved in any legal proceedings which, individually or in the aggregate, could have a material effect on its financial position, results of operations, or cash flows.

#### **Warranty Accruals**

Estimated warranty program claims are provided for at the time of sale of the Company's products. Amounts provided for are based on historical costs and adjusted for new claims and are included within other accrued expenses and other liabilities in the condensed consolidated balance sheets. The changes in the carrying amount of product warranties through September 30, 2024 and 2023, were as follows:

	20	024	2023	3
Balance at January 1	\$	42,243	\$	43,056
Provision for warranties		42,686		36,208
Settlements made		(40,482)	(:	37,588)
Other adjustments, including acquisitions and currency translation		(1,260)		(332)
Balance at September 30	\$	43,187	\$	41,344

(Amounts in thousands except share data and where otherwise indicated) (Unaudited)

#### 15. Other Comprehensive Earnings

Amounts reclassified from accumulated other comprehensive earnings (loss) to earnings during the three and nine months ended September 30, 2024 and 2023 were as follows:

	Three Months Ended September 30,			Nine Months Ended September 30,				
	2024		2023	2024			2023	
Foreign currency translation:			_					
Reclassification of foreign currency translation losses to earnings	\$ _	\$	_	\$	13,931	\$	_	
Tax benefit	_		_		_		_	
Net of tax	\$ _	\$	_	\$	13,931	\$	_	
Pension plans:								
Amortization of actuarial gain	\$ (468)	\$	(632)	\$	(1,418)	\$	(1,912)	
Amortization of prior service (credits) costs	(193)		279		(568)		851	
Total before tax	(661)		(353)		(1,986)		(1,061)	
Tax provision	138		82		415		247	
Net of tax	\$ (523)	\$	(271)	\$	(1,571)	\$	(814)	
Cash flow hedges:								
Net loss (gain) reclassified into earnings	\$ 24	\$	450	\$	(854)	\$	2,568	
Tax provision (benefit)	15		(81)		189		(501)	
Net of tax	\$ 39	\$	369	\$	(665)	\$	2,067	

The Company recognizes the amortization of net actuarial gains and losses and prior service costs and credits in other income, net within the condensed consolidated statements of earnings.

Cash flow hedges consist mainly of foreign currency forward contracts. The Company recognizes the realized gains and losses on its cash flow hedges in the same line item as the hedged transaction, such as revenue, cost of goods and services, or selling, general and administrative expenses.

#### 16. Segment Information

The Company categorizes its operating companies into five reportable segments as follows:

- Engineered Products segment provides a wide range of equipment, components, software, solutions and services to the vehicle aftermarket, aerospace and defense, industrial winch and hoist, and fluid dispensing end-markets.
- Clean Energy & Fueling segment provides components, equipment, software solutions and services enabling safe and reliable storage, transport and dispensing of traditional and clean fuels (including liquefied natural gas, hydrogen, and electric vehicle charging), cryogenic gases, and other hazardous substances along the supply chain, and safe and efficient operation of convenience retail, retail fueling and vehicle wash establishments.
- Imaging & Identification segment supplies precision marking and coding, product traceability, brand protection and digital textile printing equipment, as well as related consumables, software and services to the global packaged and consumer goods, pharmaceutical, industrial manufacturing, textile and other end-markets.
- Pumps & Process Solutions segment manufactures specialty pumps and flow meters, fluid transfer connectors, highly engineered precision
  components, instruments and digital controls for rotating and reciprocating machines, polymer processing equipment, serving single-use
  biopharmaceutical production, diversified industrial manufacturing applications, chemical production, plastics and polymer processing, midstream
  and downstream oil and gas, clean energy markets, thermal management, food and beverage, semiconductor production and medical applications
  and other end-markets.

(Amounts in thousands except share data and where otherwise indicated) (Unaudited)

 Climate & Sustainability Technologies segment is a provider of innovative and energy-efficient equipment, components and parts for the commercial refrigeration, heating and cooling and beverage can-making equipment end-markets.

Management uses segment earnings to evaluate segment performance and allocate resources. Segment earnings is defined as earnings before purchase accounting expenses, restructuring and other costs (benefits), loss (gain) on disposition, disposition costs, corporate expenses/other, interest expense, interest income and provision for income taxes.

Segment financial information and a reconciliation of segment results to consolidated results were as follows:

	Three Months Ended September 30,				Nine Months Ended September 30,			
		2024		2023		2024		2023
Revenue:								
Engineered Products	\$	296,117	\$	309,431	\$	914,234	\$	922,794
Clean Energy & Fueling		500,685		466,959		1,408,752		1,338,854
Imaging & Identification		283,966		276,179		848,365		831,202
Pumps & Process Solutions		472,463		431,373		1,415,431		1,310,880
Climate & Sustainability Technologies		431,127		475,911		1,232,125		1,380,237
Intersegment eliminations		(816)		(1,425)		(2,864)		(4,303)
Total consolidated revenue	\$	1,983,542	\$	1,958,428	\$	5,816,043	\$	5,779,664
Earnings from continuing operations:	·			_				
Segment earnings:								
Engineered Products	\$	56,621	\$	63,525	\$	171,248	\$	156,461
Clean Energy & Fueling		99,536		92,483		256,747		249,704
Imaging & Identification		77,247		70,316		222,992		199,967
Pumps & Process Solutions		138,277		117,907		394,231		362,488
Climate & Sustainability Technologies		76,015		84,060		205,901		233,912
Total segment earnings		447,696		428,291		1,251,119		1,202,532
Purchase accounting expenses (1)		48,356		38,956		136,875		118,203
Restructuring and other costs (2)		16,581		11,581		52,142		43,777
Gain on dispositions (3)		(68,633)		_		(597,913)		_
Corporate expense / other (4)		36,110		30,937		117,795		105,376
Interest expense		34,128		32,390		102,867		100,407
Interest income		(5,176)		(3,808)		(14,013)		(8,552)
Earnings before provision for income taxes		386,330		318,235		1,453,366		843,321
Provision for income taxes		73,434		56,252		291,781		157,636
Earnings from continuing operations	\$	312,896	\$	261,983	\$	1,161,585	\$	685,685

<sup>(1)</sup> Purchase accounting expenses are primarily comprised of amortization of acquired intangible assets.

<sup>(2)</sup> Restructuring and other costs relate to actions taken for headcount reductions, facility consolidations and site closures, product line exits, and other asset charges. Restructuring and other costs consist of the following:

		Three Months Ended September 30,				Nine Months Ended September 30,			
	_	2024		2023		2024		2023	
Restructuring	\$	13,837	\$	4,385	\$	41,629	\$	33,260	
Other costs, net		2,744		7,196		10,513		10,517	
Restructuring and other costs	\$	16,581	\$	11,581	\$	52,142	\$	43,777	

<sup>(3)</sup> Gain on dispositions including post-closing adjustments, see Note 4 — Discontinued and Disposed Operations for further details.

<sup>(4)</sup> Certain expenses are maintained at the corporate level and not allocated to the segments. These expenses include executive and functional compensation costs, non-service pension costs, non-operating insurance expenses, shared business services and digital overhead costs, deal related expenses and various administrative expenses relating to the corporate headquarters.

(Amounts in thousands except share data and where otherwise indicated) (Unaudited)

The following table presents revenue disaggregated by geography based on the location of the Company's customers:

	Th	ree Months En	ded	September 30,	Nine Months Ended September 30,					
Revenue by geography		2024 2023				2024		2023		
United States	\$	1,099,830	\$	991,661	\$	3,185,188	\$	2,972,261		
Europe		406,556		433,066		1,250,813		1,312,165		
Asia		206,548		235,184		609,848		680,539		
Other Americas		190,829		182,808		551,839		518,908		
Other		79,779		115,709		218,355		295,791		
Total	\$	1,983,542	\$	1,958,428	\$	5,816,043	\$	5,779,664		

#### 17. Stockholders' Equity

#### Share Repurchases

In August 2023, the Company's Board of Directors approved a new standing share repurchase authorization whereby the Company may repurchase up to 20 million shares beginning on January 1, 2024 through December 31, 2026.

On February 29, 2024, the Company entered into a \$500,000 accelerated share repurchase agreement (the "ASR Agreement") with Citibank, N.A. ("Citibank") to repurchase its shares in an accelerated share repurchase program (the "ASR Program"). The ASR Program is classified as equity, initially recorded at fair value with no subsequent remeasurement. The Company conducted the ASR Program under the current share repurchase authorization. The Company funded the ASR Program with proceeds from commercial paper.

Under the terms of the ASR Agreement, the Company paid Citibank \$500,000 on March 1, 2024 and on that date received initial delivery of 2,569,839 shares, representing a substantial majority of the shares expected to be retired over the course of the ASR Agreement. In July 2024, Citibank delivered 299,443 additional shares which completed the ASR Program. During 2024, the Company received a total of 2,869,282 shares upon completion of the ASR Agreement. The total number of shares ultimately repurchased under the ASR Agreement was based on the volume-weighted average share price of Dover's common stock during the calculation period of the ASR Program, less a discount, which was \$174.26 over the term of the ASR Program.

In the three and nine months ended September 30, 2024 and 2023, exclusive of the ASR Program, there were no share repurchases. As of September 30, 2024, 17,130,718 shares remain authorized for repurchase under the August 2023 share repurchase authorization.

(Amounts in thousands except share data and where otherwise indicated) (Unaudited)

#### 18. Earnings per Share

The following table sets forth a reconciliation of the information used in computing basic and diluted earnings per share:

	T	hree Months En	ded S	September 30,	ľ	Nine Months End	September 30,	
		2024		2023		2024		2023
Earnings from continuing operations	\$	312,896	\$	261,983	\$	1,161,585	\$	685,685
Earnings from discontinued operations, net		34,204		27,770		99,558		74,881
Net earnings	\$	347,100	\$	289,753	\$	1,261,143	\$	760,566
Basic earnings per common share:								
Earnings from continuing operations	\$	2.28	\$	1.87	\$	8.42	\$	4.90
Earnings from discontinued operations, net	\$	0.25	\$	0.20	\$	0.72	\$	0.54
Net earnings	\$	2.53	\$	2.07	\$	9.14	\$	5.44
Weighted average shares outstanding		137,251,000		139,878,000		137,913,000		139,833,000
Diluted earnings per common share:								
Earnings from continuing operations	\$	2.26	\$	1.86	\$	8.37	\$	4.88
Earnings from discontinued operations, net	\$	0.25	\$	0.20	\$	0.72	\$	0.53
Net earnings	\$	2.51	\$	2.06	\$	9.08	\$	5.41
Weighted average shares outstanding		138,223,000		140,615,000		138,830,000		140,603,000

The following table is a reconciliation of the share amounts used in computing earnings per share:

	Three Months End	ed September 30,	Nine Months End	ed September 30,
	2024	2023	2024	2023
Weighted average shares outstanding - basic	137,251,000	139,878,000	137,913,000	139,833,000
Dilutive effect of assumed exercise of SARs and vesting of performance shares and RSUs	972,000	737,000	917,000	770,000
Weighted average shares outstanding - diluted	138,223,000	140,615,000	138,830,000	140,603,000

Diluted earnings per share amounts are computed using the weighted average number of common shares outstanding and, if dilutive, potential common shares outstanding during the period. Potential common shares consist of the incremental common shares issuable upon the exercise of SARs and vesting of performance shares and RSUs, as determined using the treasury stock method.

The weighted average number of anti-dilutive potential common shares excluded from the calculation above were approximately 44,000 and 39,000 for the three months ended September 30, 2024 and 2023, respectively, and 63,000 and 54,000 for the nine months ended September 30, 2024 and 2023, respectively.

#### 19. Recent Accounting Pronouncements

#### Recently Issued Accounting Standards

In December 2023, the FASB issued ASU No. 2023-09 Income Taxes (Topic 740): Improvements to Income Tax Disclosures, which expands the disclosures required in an entity's income tax rate reconciliation table and requires disclosure of income taxes paid both in U.S. and foreign jurisdictions. The amendments are effective for fiscal years beginning after December 15, 2024. Early adoption is permitted. The Company is currently evaluating this ASU to determine its impact on the Company's disclosures.

In November 2023, the FASB issued ASU No. 2023-07, Segment Reporting (Topic 280): Improvements to Reportable Segment Disclosures. The amendment requires disclosure of significant segment expenses that are regularly provided to the chief operating decision maker and included within each reported measure of segment profit or loss, an amount and description of its composition for other segment items, and interim disclosures of a reportable segment's profit or loss and assets. The amendments are effective for fiscal years beginning after December 15, 2023, and for interim periods within fiscal years beginning after December 15, 2024. Early adoption is permitted. The Company is currently evaluating this ASU to determine its impact on the Company's disclosures.

(Amounts in thousands except share data and where otherwise indicated) (Unaudited)

#### Recently Adopted Accounting Standard

In September 2022, the FASB issued ASU No. 2022-04 Liabilities-Supplier Finance Programs (Topic 405-50): Disclosure of Supplier Finance Program Obligations. The amendments in this update require a buyer in a supplier finance program to disclose information about the program's nature, activity during the period, changes from period to period, and potential magnitude. The Company adopted the guidance when it became effective on January 1, 2023, except for the rollforward requirement, which was adopted when it became effective January 1, 2024. The adoption did not have a material impact on the Company's condensed consolidated financial statements.

The Company facilitates the opportunity for suppliers to participate in a voluntary supply chain financing ("SCF") program with a third-party financial institution. Participating suppliers are paid directly by the SCF financial institution and, in addition, may elect to sell receivables due from the Company to the SCF financial institution for early payment. Thus, participating suppliers have additional potential flexibility in managing their liquidity by accelerating, at their option and cost, collection of receivables due from the Company.

The Company and its suppliers agree on commercial terms, including payment terms, for the goods and services the Company procures, regardless of whether the supplier participates in SCF. For participating suppliers, the Company's responsibility is limited to making all payments to the SCF financial institution on the terms originally negotiated with the supplier, irrespective of whether the supplier elects to sell receivables to the SCF financial institution. The Company does not determine the terms or conditions of the arrangement between the SCF financial institution and the Company's suppliers. The SCF financial institution pays the supplier on the invoice due date for any invoices that were not previously sold by the supplier. The agreement between the Company and the SCF financial institution does not require the Company to provide assets pledged as security or other forms of guarantees.

Outstanding payments related to the SCF program are recorded within accounts payable in our condensed consolidated balance sheets. As of the September 30, 2024 and December 31, 2023 amounts due to the SCF financial institution were approximately \$131,534 and \$156,245, respectively.

#### 20. Subsequent Events

On October 8, 2024, the Company completed the previously announced sale of the ESG business, an operating company within the Engineered Products segment, for total consideration, net of cash transferred, of \$2.0 billion, subject to customary post-closing adjustments. The initial accounting for the disposition is incomplete as a result of the timing of the transaction. Accordingly, it is impracticable for us to disclose the estimated gain on disposition. See Note 4 — Discontinued and Disposed Operations for further details.

#### Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

Refer to the section below entitled "Special Note Regarding Forward-Looking Statements" for a discussion of factors that could cause our actual results to differ from the forward-looking statements contained below and throughout this quarterly report.

Throughout this Management's Discussion and Analysis of Financial Condition and Results of Operations ("MD&A"), we refer to measures used by management to evaluate performance, including a number of financial measures that are not defined under accounting principles generally accepted in the United States of America ("GAAP"). Please see "Non-GAAP Disclosures" at the end of this Item 2 for further detail on these financial measures. We believe these measures provide investors with important information that is useful in understanding our business results and trends. Reconciliations within this MD&A provide more details on the use and derivation of these measures.

#### **OVERVIEW**

Dover is a diversified global manufacturer and solutions provider delivering innovative equipment and components, consumable supplies, aftermarket parts, software and digital solutions, and support services through five operating segments: Engineered Products, Clean Energy & Fueling, Imaging & Identification, Pumps & Process Solutions, and Climate & Sustainability Technologies. The Company's entrepreneurial business model encourages, promotes and fosters deep customer engagement and collaboration, which has led to Dover's well-established and valued reputation for providing superior customer service and industry-leading product innovation. Unless the context indicates otherwise, references herein to "Dover," "the Company," and words such as "we," "us," or "our" include Dover Corporation and its consolidated subsidiaries.

Dover's five operating segments are as follows:

- Our Engineered Products segment provides a wide range of equipment, components, software, solutions and services to the vehicle aftermarket, aerospace and defense, industrial winch and hoist, and fluid dispensing end-markets.
- Our Clean Energy & Fueling segment provides components, equipment, software solutions and services enabling safe and reliable storage, transport and dispensing of traditional and clean fuels (including liquefied natural gas, hydrogen, and electric vehicle charging), cryogenic gases, and other hazardous substances along the supply chain, and safe and efficient operation of convenience retail, retail fueling and vehicle wash establishments.
- Our Imaging & Identification segment supplies precision marking and coding, product traceability, brand protection and digital textile printing equipment, as well as related consumables, software and services to the global packaged and consumer goods, pharmaceutical, industrial manufacturing, textile and other end-markets.
- Our Pumps & Process Solutions segment manufactures specialty pumps and flow meters, fluid transfer connectors, highly engineered precision
  components, instruments and digital controls for rotating and reciprocating machines, polymer processing equipment, serving single-use
  biopharmaceutical production, diversified industrial manufacturing applications, chemical production, plastics and polymer processing, midstream
  and downstream oil and gas, clean energy markets, thermal management, food and beverage, semiconductor production and medical applications
  and other end-markets.
- Our Climate & Sustainability Technologies segment is a provider of innovative and energy-efficient equipment, components and parts for the commercial refrigeration, heating and cooling and beverage can-making equipment end-markets.

In the third quarter of 2024, revenue was \$2.0 billion, which increased \$25.1 million, or 1.3%, as compared to the third quarter of 2023. This was driven by acquisition-related revenue growth of 3.8% and organic revenue growth of 0.3%, partially offset by disposition-related decline of 2.7% and an unfavorable impact from foreign currency translation of 0.1%. The results were driven by solid demand across most end markets and strategic pricing initiatives.

The 0.3% organic revenue growth for the third quarter of 2024 was driven by our Engineered Products, Imaging & Identification and Pumps & Process Solutions segments which grew 12.1%, 3.3%, and 1.9%, respectively. The growth was partially offset by the Climate & Sustainability Technologies and Clean Energy & Fueling segments which declined 9.4% and 1.2%, respectively. For further information, see "Segment Results of Operations" within this Item 2.

#### Table of Contents

From a geographic perspective, organic revenue for the U.S., our largest market, increased 8.4% in the third quarter of 2024 compared to the prior year comparable quarter, driven by broad-based growth across all segments. Organic revenue increased for Other Americas by 1.7%, and decreased for Asia and Europe by 9.5% and 4.6%, respectively.

Bookings were \$1.9 billion for the three months ended September 30, 2024, an increase of \$0.1 billion, or 5.6% compared to the prior year comparable quarter. Included in this result was organic growth of 5.1% and acquisition-related growth of 3.5%, partially offset by disposition-related decline of 2.9% and an unfavorable impact from foreign currency translation of 0.1%. The organic bookings growth was primarily driven by positive demand trends and order timing.

Restructuring and other costs for the three months ended September 30, 2024 were \$16.6 million which included restructuring charges of \$13.8 million and other costs of \$2.7 million. Restructuring and other costs were generally related to exit costs and headcount reductions in the Clean Energy & Fueling segment. For further discussion related to our restructuring and other costs, see "Restructuring and Other Costs (Benefits)," within this Item 2.

During the three months ended September 30, 2024, the Company completed four business acquisitions for approximately \$460.5 million, subject to post-closing adjustments and inclusive of contingent consideration. See Note 3 — Acquisitions in the condensed consolidated financial statements in Item 1 of this Form 10-Q for further details.

On September 30, 2024, a minority owned equity method investment held within the Climate & Sustainability Technologies segment was sold and the Company received its proportionate share of the proceeds amounting to \$92,962 which resulted in a preliminary pre-tax gain of \$68,712, subject to customary post-closing adjustments. See Note 4 — Discontinued and Disposed Operations in the condensed consolidated financial statements in Item 1 of this Form 10-Q for further details.

On October 8, 2024, the Company completed the previously announced sale of the Environmental Solutions Group ("ESG") business, an operating company within the Engineered Products segment, for total consideration, net of cash transferred, of \$2.0 billion, subject to customary post-closing adjustments. For the three and nine months ended September 30, 2024 and 2023, the results of ESG are presented as discontinued operations as the sale represents a strategic shift in operations with a major impact on our operations and financial results. See Note 4 — Discontinued and Disposed Operations in the condensed consolidated financial statements in Item 1 of this Form 10-Q for further details. The discussion in this MD&A, unless otherwise noted, relates solely to our continuing operations.

During the nine months ended September 30, 2024, the Company received a total of 2,869,282 shares upon completion of the accelerated repurchase agreement (the ASR Agreement"). The total number of shares ultimately repurchased under the ASR Agreement was based on the volume-weighted average share price of Dover's common stock during the calculation period of the accelerated share repurchase program (the "ASR Program"), less a discount, which was \$174.26 over the term of the ASR Program.

In the three and nine months ended September 30, 2024 and 2023, exclusive of the ASR Program, there were no share repurchases. As of September 30, 2024, 17,130,718 shares remain authorized for repurchase under the August 2023 share repurchase authorization.

#### **CONSOLIDATED RESULTS OF OPERATIONS**

	Three Mo	onth	s Ended Septemb	er 30,	Nine Months Ended September 30,					
(dollars in thousands, except per share figures)	2024		2023	% / Point Change	2024		2023	% / Point Change		
Revenue	\$ 1,983,542	\$	1,958,428	1.3 %	\$ 5,816,043	\$	5,779,664	0.6 %		
Cost of goods and services	1,220,355		1,219,047	0.1 %	3,603,146		3,631,023	(0.8)%		
Gross profit	763,187		739,381	3.2 %	2,212,897		2,148,641	3.0 %		
Gross profit margin	38.5 %		37.8 %	0.7	38.0 %		37.2 %	0.8		
Selling, general and administrative expenses	429,570		402,838	6.6 %	1,301,606		1,234,223	5.5 %		
Selling, general and administrative expenses as a percent of revenue	21.7 %		20.6 %	1.1	22.4 %		21.4 %	1.0		
Operating earnings	333,617		336,543	(0.9)%	911,291		914,418	(0.3)%		
Interest expense	34,128		32,390	5.4 %	102,867		100,407	2.5 %		
Interest income	(5,176)		(3,808)	35.9 %	(14,013)		(8,552)	63.9 %		
Gain on dispositions	(68,633)		_	nm*	(597,913)		_	nm*		
Other income, net	(13,032)		(10,274)	nm*	(33,016)		(20,758)	nm*		
Earnings before provision for income taxes	386,330		318,235	21.4 %	1,453,366		843,321	72.3 %		
Provision for income taxes	73,434		56,252	30.5 %	291,781		157,636	85.1 %		
Effective tax rate	19.0 %		17.7 %	1.3	20.1 %		18.7 %	1.4		
Earnings from continuing operations	\$ 312,896	\$	261,983	19.4 %	\$ 1,161,585	\$	685,685	69.4 %		
Earnings from discontinued operations, net	34,204		27,770	nm*	99,558		74,881	nm*		
Net earnings	\$ 347,100	\$	289,753	19.8 %	\$ 1,261,143	\$	760,566	65.8 %		
Earnings per common share from continuing operations - diluted	\$ 2.26	\$	1.86	21.5 %	\$ 8.37	\$	4.88	71.5 %		

<sup>\*</sup> nm - not meaningful

#### Revenue

Revenue for the three months ended September 30, 2024 increased \$25.1 million, or 1.3%, from the prior year comparable quarter. The increase in revenue was driven by acquisition-related growth of 3.8% and organic revenue growth of 0.3%, partially offset by disposition-related decline of 2.7% and an unfavorable impact from foreign currency translation of 0.1%. Customer pricing favorably impacted revenue by approximately 1.6% in the third quarter of 2024 and by 3.2% in the prior year comparable quarter.

Revenue for the nine months ended September 30, 2024 increased \$36.4 million, or 0.6%, from the prior year comparable period. The increase primarily reflects an acquisition-related growth of 2.8% and organic revenue remained flat. This increase was partially offset by a disposition-related decline of 1.8% and an unfavorable impact from foreign currency translation of 0.4%. Customer pricing favorably impacted revenue by approximately 1.6% for the nine months ended September 30, 2024, and by 4.3% in the prior year comparable period.

#### Gross Profit

Gross profit for the three months ended September 30, 2024 increased \$23.8 million, or 3.2%, and gross profit margin increased 70 basis points to 38.5%, versus the prior year comparable quarter. The gross profit margin increase was driven by positive product mix and productivity actions.

Gross profit for the nine months ended September 30, 2024 increased \$64.3 million, or 3.0%, and gross profit margin increased by 80 basis points to 38.0%, from the prior year comparable period. Gross profit margin increased driven by positive product mix and productivity actions.

#### Selling, General and Administrative Expenses

Selling, general and administrative expenses for the three months ended September 30, 2024 increased \$26.7 million, or 6.6%, from the prior year comparable quarter, primarily driven by increased employee compensation and benefits and acquisition-related amortization costs. As a percentage of revenue, selling, general and administrative expenses increased 110 basis points as compared to the prior year comparable quarter to 21.7%.

Selling, general and administrative expenses for the nine months ended September 30, 2024 increased \$67.4 million, or 5.5%, from the prior year comparable period, primarily driven by increased employee compensation and benefits, acquisition-related amortization costs and computer software costs, partially offset by lower restructuring costs. Selling, general and administrative expenses as a percentage of revenue increased 100 basis points as compared to the prior year comparable period to 22.4%.

Research and development costs, including qualifying engineering costs, are expensed when incurred and amounted to \$35.6 million and \$34.4 million for the three months ended September 30, 2024 and 2023, respectively, and \$108.0 million and \$105.1 million, for the nine months ended September 30, 2024 and 2023, respectively. The costs as a percentage of revenue are 1.8% and 1.9% for the three and nine months ended September 30, 2024, respectively, and 1.8% for the three and nine months ended September 30, 2023.

#### Gain on Dispositions

Gain on dispositions of \$597.9 million for the nine months ended September 30, 2024 was primarily due to the sale of the De-Sta-Co business on March 31, 2024 and the sale of a minority owned equity investment on September 30, 2024. See Note 4 — Discontinued and Disposed Operations in the condensed consolidated financial statements in Item 1 of this Form 10-Q for further details.

#### Other Income, net

Other income, net includes non-service pension benefit, deferred compensation plan investments gain or loss, earnings or charges from equity method investments, foreign exchange gain or loss, and various other items. Other income, net for the three and nine months ended September 30, 2024 increased \$2.8 million and \$12.3 million, respectively, from the comparable prior period primarily driven by an increase in equity earnings and gain on deferred compensation plan investments.

#### Income Taxes

The effective tax rates for the three months ended September 30, 2024 and 2023 were 19.0% and 17.7%, respectively. The increase in the effective tax rate for the three months ended September 30, 2024 relative to the prior year comparable quarter was primarily driven by a gain on disposition.

The effective tax rates for the nine months ended September 30, 2024 and 2023 were 20.1% and 18.7%, respectively. The increase in the effective tax rate for the nine months ended September 30, 2024 relative to the prior year comparable period was primarily driven by the gains on dispositions.

The Company is monitoring the changes in tax laws resulting from the Organization for Economic Cooperation and Development's multi-jurisdictional plan of action to address base erosion and profit shifting. We do not expect this to have a material impact on our effective tax rate.

See Note 12 — Income Taxes in the condensed consolidated financial statements in Item 1 of this Form 10-Q for additional details.

#### Earnings from Continuing Operations

Earnings from continuing operations for the three months ended September 30, 2024 increased 19.4% to \$312.9 million, or \$2.26 diluted earnings per share from continuing operations, from \$262.0 million, or \$1.86 diluted earnings per share from continuing operations, in the prior year comparable quarter. The increase in earnings from continuing operations is driven by the after-tax gain on the sale of the minority owned equity investment, partially offset by higher selling, general and administrative expense.

Earnings from continuing operations for the nine months ended September 30, 2024 increased 65.8% to \$1.2 billion, or \$8.37 diluted earnings per share from continuing operations, from \$685.7 million, or \$4.88 diluted earnings per share from continuing operations, in the prior year comparable period. The increase in earnings from continuing operations is primarily driven by the after-tax gains on the sale of De-Sta-Co and a minority owned equity method investment totaling \$464.2 million.

#### **Discontinued Operations**

For the three and nine months ended September 30, 2024 and 2023, the historical results of ESG were presented as discontinued operations as the sale represents a strategic shift that will have a major impact on our operations and financial results. For the three and nine months ended September 30, 2024, earnings from discontinued operations, net were \$34.2 million and \$99.6 million, respectively. For the three and nine months ended September 30, 2023, earnings from discontinued operations, net were \$27.8 million and \$74.9 million, respectively.

#### SEGMENT RESULTS OF OPERATIONS

The summary that follows provides a discussion of the results of operations of each of our five reportable operating segments (Engineered Products, Clean Energy & Fueling, Imaging & Identification, Pumps & Process Solutions, and Climate & Sustainability Technologies). Each of these segments is comprised of various product and service offerings that serve multiple markets. We evaluate our operating segment performance based on segment earnings as defined in Note 16 — Segment Information in the condensed consolidated financial statements in Item 1 of this Form 10-Q.

We report organic revenue growth, which excludes the impact of foreign currency exchange rates and the impact of acquisitions and divestitures. See "Non-GAAP Disclosures" at the end of this Item 2.

Additionally, we use the following operational metrics in monitoring the performance of the business. We believe the operational metrics are useful to investors and other users of our financial information in assessing the performance of our segments:

- Bookings represent total orders received from customers in the current reporting period and exclude de-bookings related to orders received in prior
  periods, if any. This metric is an important measure of performance and an indicator of order trends.
- Organic bookings represent bookings excluding the impact of foreign currency exchange rates and the impact of acquisitions and dispositions. This metric is an important measure of performance and an indicator of revenue order trends.
- Book-to-bill is a ratio of the amount of bookings received from customers during a period divided by the amount of revenue recorded during that same period. This metric is a useful indicator of demand.

#### **Engineered Products**

Our Engineered Products segment provides a wide range of equipment, components, software, solutions and services to the vehicle aftermarket, aerospace and defense, industrial winch and hoist, and fluid dispensing end-markets.

	Three M	[ont]	hs Ended Septe	ember 30,	Nine Months Ended September 30,					
(dollars in thousands)	 2024		2023	% Change	2024		2023	% Change		
Revenue	\$ 296,117	\$	309,431	(4.3)%	\$ 914,234	\$	922,794	(0.9)%		
Segment earnings	\$ 56,621	\$	63,525	(10.9)%	\$ 171,248	\$	156,461	9.5 %		
Segment margin	19.1 %		20.5 %		18.7 %		17.0 %			
Operational metrics:										
Bookings	\$ 284,823	\$	330,566	(13.8)%	\$ 895,290	\$	957,233	(6.5)%		
Components of revenue decline:										
Organic growth				12.1 %				10.5 %		
Acquisitions				0.3 %				0.1 %		
Dispositions				(17.0)%				(11.5)%		
Foreign currency translation				0.3 %				— %		
Total revenue decline			•	(4.3)%			-	(0.9)%		

#### Third Quarter 2024 Compared to the Third Quarter 2023

Engineered Products revenue for the third quarter of 2024 decreased \$13.3 million, or 4.3%, as compared to the third quarter of 2023, due to a disposition-related decline of 17.0%, partially offset by organic growth of 12.1%, acquisition-related growth of 0.3% and a favorable impact from foreign currency translation of 0.3%. The disposition-related decline was due to the divestiture of De-Sta-Co in the first quarter of 2024. Customer pricing favorably impacted revenue by approximately 1.1% in the third quarter of 2024 and 1.2% in the prior year comparable quarter.

The organic revenue growth was primarily driven by our vehicle service business, which saw increased demand in Europe and improved production performance in North America, along with favorable demand trends in our industrial winch and hoist business. We expect positive organic growth trends to continue into the fourth quarter on the back of improved demand conditions and production performance in our vehicle service business.

Engineered Products segment earnings decreased \$6.9 million, or 10.9%, compared to the third quarter of 2023. The decrease was primarily due to the disposition of De-Sta-Co and shipment timing of aerospace and defense components, which was partially offset by organic volume increases and favorable price versus cost dynamics. Segment margin decreased to 19.1% from 20.5% as compared to the prior year comparable quarter.

Bookings decreased 13.8% for the segment, due primarily to a disposition-related decline of 15.1%, partially offset by acquisition-related growth of 1.1% and a favorable impact from foreign currency translation of 0.2%. The disposition-related bookings decline was due to the divestiture of De-Sta-Co in the first quarter of 2024. Segment book-to-bill was 0.96.

#### Nine Months Ended September 30, 2024 Compared to the Nine Months Ended September 30, 2023

Engineered Products revenue for the nine months ended September 30, 2024 decreased \$8.6 million, or 0.9%, compared to the prior year comparable period. This was comprised of a disposition-related decline of 11.5%, partially offset by organic revenue growth of 10.5% and acquisition-related growth of 0.1%. Organic revenue growth was driven by improved production performance and increased demand in our vehicle service business, along with solid demand trends in both our aerospace and defense business, and in our industrial winch and hoist business. Customer pricing favorably impacted revenue by approximately 0.8% and by 2.0% in the prior year comparable period.

Segment earnings for the nine months ended September 30, 2024 increased \$14.8 million, or 9.5%, as compared to the 2023 comparable period. The increase was primarily driven by organic volume increases and favorable price versus cost dynamics, partially offset by disposition impacts. Segment margin increased to 18.7% from 17.0% as compared to the prior year comparable period.

#### Clean Energy & Fueling

Our Clean Energy & Fueling segment provides components, equipment, software solutions and services enabling safe and reliable storage, transport and dispensing of traditional and clean fuels (including liquefied natural gas, hydrogen, and electric vehicle charging), cryogenic gases, and other hazardous substances along the supply chain, and safe and efficient operation of convenience retail, retail fueling and vehicle wash establishments.

	Three M	[ont]	hs Ended Sept	ember 30,	Nine Months Ended September 30,					
(dollars in thousands)	 2024		2023	% Change		2024		2023	% Change	
Revenue	\$ 500,685	\$	466,959	7.2 %	\$	1,408,752	\$	1,338,854	5.2 %	
Segment earnings	\$ 99,536	\$	92,483	7.6 %	\$	256,747	\$	249,704	2.8 %	
Segment margin	19.9 %		19.8 %			18.2 %		18.7 %		
Operational metrics:										
Bookings	\$ 507,329	\$	449,663	12.8 %	\$	1,421,025	\$	1,344,326	5.7 %	
Components of revenue growth:										
Organic (decline) growth				(1.2)%					0.8 %	
Acquisitions				8.8 %					4.7 %	
Foreign currency translation				(0.4)%					(0.3)%	
Total revenue growth				7.2 %				-	5.2 %	

#### Third Quarter 2024 Compared to the Third Quarter 2023

Clean Energy & Fueling revenue for the third quarter of 2024 increased \$33.7 million, or 7.2%, as compared to the third quarter of 2023, driven by acquisition-related growth of 8.8%, partially offset by an organic decline of 1.2% and an unfavorable foreign currency translation impact of 0.4%. Customer pricing favorably impacted revenue in the third quarter of 2024 by approximately 2.0% and by 4.0% in the prior year comparable quarter.

The organic revenue decline was primarily due to lower shipments in Europe for above-ground retail fueling projects and reduced year-over-year demand in vehicle wash solutions, partially offset by pricing initiatives, clean energy solutions and strong demand in North America above-ground retail fueling equipment. We expect organic growth in the fourth quarter driven by strong demand in North America above and below-ground retail fueling, as well as the clean energy solutions business.

Clean Energy & Fueling segment earnings increased \$7.1 million, or 7.6%, over the prior year comparable quarter. The increase was primarily driven by strategic pricing and the favorable impact of acquisitions, partially offset by the negative impact from lower volumes and integration costs in clean energy solutions. Segment margin remained flat compared to prior year comparable quarter.

Overall bookings increased 12.8% as compared to the prior year comparable quarter, driven by organic growth of 7.0% and acquisition growth of 6.1%, partially offset by an unfavorable impact from foreign currency translation of 0.3%. The organic bookings growth was primarily driven by North America above and below-ground retail fueling equipment and clean energy platform. Segment book-to-bill was 1.01.

#### Nine Months Ended September 30, 2024 Compared to the Nine Months Ended September 30, 2023

Clean Energy & Fueling segment revenue increased \$69.9 million, or 5.2%, as compared to the nine months ended September 30, 2023, attributable to acquisition-related growth of 4.7% and organic growth of 0.8%, partially offset by an unfavorable impact from foreign currency translation of 0.3%. Organic revenue growth was driven by pricing actions and above-ground retail fueling equipment, partially offset by lower volume in fluid transfer solutions and below-ground retail fueling businesses. Customer pricing favorably impacted revenue by approximately 2.6% and by approximately 4.4% in the prior year comparable period.

Clean Energy & Fueling segment earnings increased \$7.0 million or 2.8%, for the nine months ended September 30, 2024. Pricing actions and productivity initiatives and the favorable impact of acquisitions were partially offset by lower volume and inflationary costs. Segment margin decreased to 18.2% from 18.7% in the prior year comparable period, primarily due to product mix.

#### **Imaging & Identification**

Our Imaging & Identification segment supplies precision marking and coding, product traceability, brand protection and digital textile printing equipment, as well as related consumables, software and services to the global packaged and consumer goods, pharmaceutical, industrial manufacturing, textile and other end-markets.

	Three M	lont	hs Ended Sept	ember 30,	Nine Months Ended September 30,					
(dollars in thousands)	 2024		2023	% Change		2024		2023	% Change	
Revenue	\$ 283,966	\$	276,179	2.8 %	\$	848,365	\$	831,202	2.1 %	
Segment earnings	\$ 77,247	\$	70,316	9.9 %	\$	222,992	\$	199,967	11.5 %	
Segment margin	27.2 %		25.5 %			26.3 %		24.1 %		
Operational metrics:										
Bookings	\$ 281,289	\$	271,113	3.8 %	\$	848,363	\$	823,917	3.0 %	
Components of revenue growth:										
Organic growth				3.3 %					2.8 %	
Acquisitions				0.6 %					0.5 %	
Foreign currency translation				(1.1)%					(1.2)%	
Total revenue growth				2.8 %				_	2.1 %	

#### Third Quarter 2024 Compared to the Third Quarter 2023

Imaging & Identification revenue for the third quarter of 2024 increased \$7.8 million, or 2.8%, as compared to the third quarter of 2023, comprised of organic growth of 3.3% and acquisition-related growth of 0.6%, partially offset by an unfavorable impact from foreign currency translation of 1.1%. Customer pricing favorably impacted revenue in the third quarter of 2024 by approximately 2.6% and by approximately 5.3% in the prior year comparable quarter.

The organic revenue growth was primarily driven by pricing actions and increased demand for marking and coding equipment and consumables, partly offset by softer digital textile printing equipment and consumables. We expect continued favorable organic growth in the fourth quarter primarily driven by continued strength in marking and coding and favorable pricing.

Imaging & Identification segment earnings increased \$6.9 million, or 9.9%, over the prior year comparable quarter. The increase was primarily driven by higher volume, pricing and productivity initiatives, partially offset by an unfavorable impact from foreign currency translation. Segment margin increased to 27.2% from 25.5% in the prior year comparable quarter.

Overall bookings increased 3.8% as compared to the prior year comparable quarter, reflecting organic growth of 4.3% and acquisition-related growth of 0.5%, partially offset by an unfavorable impact from foreign currency translation of 1.0%. The organic bookings growth was primarily driven by serialization software order timing and improved marking and coding demand. Segment book-to-bill was 0.99.

#### Nine Months Ended September 30, 2024 Compared to the Nine Months Ended September 30, 2023

Imaging & Identification segment revenue increased \$17.2 million, or 2.1%, as compared to the nine months ended September 30, 2023, attributable to organic growth of 2.8% and acquisition-related growth of 0.5%, partially offset by an unfavorable impact from foreign currency translation of 1.2%. The organic revenue growth was primarily driven by pricing initiatives, partially offset by weaker demand in our digital textile printing business. Customer pricing favorably impacted revenue by approximately 3.0% and by approximately 6.0% in the prior year comparable period.

Imaging & Identification segment earnings increased \$23.0 million, or 11.5%, for the nine months ended September 30, 2024 over the prior year comparable period. The increase was primarily driven by favorable product mix, pricing initiatives and cost controls, partially offset by the unfavorable impact of foreign currency translation. Segment margin increased to 26.3% from 24.1% in the prior year comparable period.

#### **Pumps & Process Solutions**

Our Pumps & Process Solutions segment manufactures specialty pumps and flow meters, fluid transfer connectors, highly engineered precision components, instruments and digital controls for rotating and reciprocating machines, polymer processing equipment, serving single-use biopharmaceutical production, diversified industrial manufacturing applications, chemical production, plastics and polymer processing, midstream and downstream oil and gas, clean energy markets, thermal management, food and beverage, semiconductor production and medical applications and other end-markets.

	Three M	ont	ns Ended Septe	ember 30,	Nine Months Ended September 30,						
(dollars in thousands)	 2024		2023	% Change	2024		2023	% Change			
Revenue	\$ 472,463	\$	431,373	9.5 %	\$ 1,415,431	\$	1,310,880	8.0 %			
Segment earnings	\$ 138,277	\$	117,907	17.3 %	\$ 394,231	\$	362,488	8.8 %			
Segment margin	29.3 %		27.3 %		27.9 %		27.7 %				
Operational metrics:											
Bookings	\$ 448,074	\$	363,111	23.4 %	\$ 1,383,132	\$	1,221,725	13.2 %			
Components of revenue growth:											
Organic growth				1.9 %				1.0 %			
Acquisitions				7.0 %				6.8 %			
Foreign currency translation				0.6 %				0.2 %			
Total revenue growth			_	9.5 %			_	8.0 %			

#### Third Quarter 2024 Compared to the Third Quarter 2023

Pumps & Process Solutions revenue for the third quarter of 2024 increased \$41.1 million, or 9.5%, as compared to the third quarter of 2023, driven by acquisition-related growth of 7.0%, organic growth of 1.9% and a favorable impact from foreign currency translation of 0.6%. Acquisition-related growth was driven by the acquisition of FW Murphy Production Controls business ("FW Murphy") in the fourth quarter of 2023. Customer pricing favorably impacted revenue in the third quarter of 2024 by approximately 1.7% and by approximately 3.6% in the prior year comparable quarter.

The organic revenue growth was primarily driven by increased revenue in our precision components business and increased demand for connectors used in bioprocessing and high performance computing and data center applications, partially offset by decreased revenue in our plastics and polymer processing solutions business. We expect overall revenue to trend favorably in the fourth quarter driven by the FW Murphy acquisition, positive demand trends in bioprocessing and thermal connectors in liquid cooling for high performance computing applications. We expect this growth to be partially offset by lower shipments in our polymer processing equipment business.

Pumps & Process Solutions segment earnings increased \$20.4 million, or 17.3%, over the prior year comparable quarter. The increase was driven by the favorable impact from the FW Murphy acquisition, along with the positive impact of product line mix, pricing initiatives and productivity actions. Segment margin increased to 29.3% from 27.3% in the prior year comparable quarter driven by productivity initiatives and favorable portfolio mix.

Overall bookings increased 23.4% as compared to the prior year comparable quarter with organic growth of 15.1% and acquisition-related growth of 8.3%. The organic bookings growth was primarily driven by positive demand trends in connectors, supported by improving customer sentiment in bioprocessing and high performance computing applications. Segment book-to-bill was 0.95.

#### Nine Months Ended September 30, 2024 Compared to the Nine Months Ended September 30, 2023

Pumps & Process Solutions segment revenue increased \$104.6 million, or 8.0%, as compared to the nine months ended September 30, 2023, attributable to acquisition-related growth of 6.8% for the acquisition of FW Murphy, organic growth of 1.0% and a favorable impact from foreign currency translation of 0.2%. The organic growth was primarily driven by pricing initiatives, increased revenue in precision components and single-use components used in biopharmaceutical manufacturing, partially offset by our polymer processing equipment business. Customer pricing favorably impacted revenue by approximately 1.6% and by approximately 4.6% in the prior year comparable period.

Pumps & Process Solutions segment earnings increased \$31.7 million, or 8.8%, for the nine months ended September 30, 2024 over the prior year comparable period. The increase was driven by the favorable impact from the FW Murphy acquisition, product line mix, pricing initiatives and productivity, partially offset by lower volumes. Segment margin increased to 27.9% from 27.7% from the prior year comparable period.

#### Climate & Sustainability Technologies

Our Climate & Sustainability Technologies segment is a provider of innovative and energy-efficient equipment, components and parts for the commercial refrigeration, heating and cooling and beverage can-making equipment end-markets.

		Three M	ontl	hs Ended Sept	ember 30,	Nine Months Ended September 30,						
(dollars in thousands)	<u> </u>	2024		2023	% Change	2024		2023	% Change			
Revenue	\$	431,127	\$	475,911	(9.4)%	\$ 1,232,125	\$	1,380,237	(10.7)%			
Segment earnings	\$	76,015	\$	84,060	(9.6)%	\$ 205,901	\$	233,912	(12.0)%			
Segment margin		17.6 %		17.7 %		16.7 %		16.9 %				
Operational metrics:												
Bookings	\$	332,503	\$	340,474	(2.3)%	\$ 1,191,858	\$	1,023,028	16.5 %			
Components of revenue decline:												
Organic decline					(9.4)%				(10.7)%			
Acquisitions					<b>—</b> %				0.3 %			
Foreign currency translation					<u> </u>				(0.3)%			
Total revenue decline					(9.4)%			- -	(10.7)%			

#### Third Quarter 2024 Compared to the Third Quarter 2023

Climate & Sustainability Technologies revenue decreased \$44.8 million, or 9.4%, as compared to the third quarter of 2023, reflecting organic revenue decline of 9.4%. Customer pricing favorably impacted revenue in the third quarter of 2024 by approximately 1.0% and by approximately 2.4% in the prior year comparable quarter.

#### Table of Contents

The organic revenue decline was primarily due to near-term, transient slowing in heat exchanger demand in Europe due to HVAC OEMs efforts to reduce component inventories, as well as continued headwinds in new beverage can-making equipment sales as customers pivot from new equipment investment to scaling production and expanding utilization of recent capacity additions. This headwind was partially offset by increased demand for retail refrigeration equipment and services, including the growing demand for low-global warming potential ("GWP") CO<sub>2</sub> refrigerant systems. We expect organic revenue declines to continue into the fourth quarter, primarily due to continued year-over-year reductions in the sales of heat exchangers in Europe and in beverage can-making equipment. We expect to begin to see sequential improvements in demand for both of these product lines in 2025.

Climate & Sustainability Technologies segment earnings decreased \$8.0 million, or 9.6%, as compared to the third quarter of 2023. The segment earnings decrease was primarily due to the negative impact from lower volumes in heat exchangers and beverage can-making equipment, partially offset by increased volumes in retail refrigeration, along with productivity initiatives and cost actions across the segment. Segment margin decreased to 17.6% from 17.7% in the prior year comparable quarter.

Bookings in the third quarter of 2024 decreased 2.3% from the prior year comparable quarter, reflecting an organic decline of 2.1% and an unfavorable impact from foreign currency translation of 0.2%. The organic bookings decline was principally due to order timing in retail refrigeration. Segment book-to-bill was 0.77.

# Nine Months Ended September 30, 2024 Compared to the Nine Months Ended September 30, 2023

Climate & Sustainability Technologies segment revenue decreased \$148.1 million, or 10.7%, compared to the nine months ended September 30, 2023, reflecting an organic revenue decline of 10.7%, and an unfavorable foreign currency translation impact of 0.3%, partially offset by acquisition-related growth of 0.3%. The organic revenue decline for the nine months ended September 30, 2024 was due to near-term, transient slowing in heat exchanger demand in Europe due to HVAC OEMs efforts to reduce component inventories, as well as continued headwinds in new beverage can-making equipment sales as customers pivot from new equipment investment to scaling production and expanding utilization of recent capacity additions. This was partially offset by increased demand for retail refrigeration equipment and services, including the growing demand for low-GWP CO<sub>2</sub> refrigerant systems. Customer pricing favorably impacted revenue by approximately 0.2%, and by approximately 4.4% in the prior year comparable period.

Climate & Sustainability Technologies segment earnings decreased \$28.0 million, or 12.0%, for the nine months ended September 30, 2024, as compared to the prior year comparable period. Segment margin decreased to 16.7% from 16.9% in the prior year comparable period. The earnings decrease was primarily due to lower volumes in heat exchangers and beverage can-making equipment, partially offset by increased retail refrigeration volumes, productivity initiatives and cost reduction actions.

# Reconciliation of Segment Earnings to Earnings from Continuing Operations

	Three Months Ended September 30,					Nine Months Ended September 30,				
(in thousands)		2024		2023		2024		2023		
<b>Earnings from Continuing Operations:</b>										
Segment earnings:										
Engineered Products	\$	56,621	\$	63,525	\$	171,248	\$	156,461		
Clean Energy & Fueling		99,536		92,483		256,747		249,704		
Imaging & Identification		77,247		70,316		222,992		199,967		
Pumps & Process Solutions		138,277		117,907		394,231		362,488		
Climate & Sustainability Technologies		76,015		84,060		205,901		233,912		
Total segment earnings	<u></u>	447,696		428,291		1,251,119		1,202,532		
Purchase accounting expenses (1)		48,356		38,956		136,875		118,203		
Restructuring and other costs (2)		16,581		11,581		52,142		43,777		
Gain on dispositions (3)		(68,633)		_		(597,913)		_		
Corporate expense / other (4)		36,110		30,937		117,795		105,376		
Interest expense		34,128		32,390		102,867		100,407		
Interest income		(5,176)		(3,808)		(14,013)		(8,552)		
Earnings before provision for income taxes		386,330		318,235		1,453,366		843,321		
Provision for income taxes		73,434		56,252		291,781		157,636		
Earnings from continuing operations	\$	312,896	\$	261,983	\$	1,161,585	\$	685,685		

<sup>(1)</sup> Purchase accounting expenses are primarily comprised of amortization of acquired intangible assets.

# Restructuring and Other Costs (Benefits)

Restructuring and other costs are not presented in our segment earnings because these costs are excluded from the segment operating performance measure reviewed by management. During the three and nine months ended September 30, 2024, we incurred restructuring charges of \$13.8 million and \$41.6 million and other costs, net of \$2.7 million and \$10.5 million. Restructuring charges for the three months ended September 30, 2024 were primarily related to exit costs and headcount reductions in the Clean Energy & Fueling segment. Restructuring charges for the nine months ended September 30, 2024 were primarily related to product line exit costs and headcount reductions in the Clean Energy & Fueling, and the Climate & Sustainability Technologies segments. These restructuring programs were initiated in 2023 and 2024 and the Company will continue to make proactive adjustments to its cost structure to align with current demand trends. Other costs, net of \$10.5 million for the nine months ended September 30, 2024, were primarily due to non-cash asset impairment charges and reorganization costs in the Climate & Sustainability Technologies and Imaging & Identification segments, respectively. These restructuring and other charges were recorded in cost of goods and services and selling, general and administrative expenses in the condensed consolidated statement of earnings. Additional programs beyond the scope of the announced programs may be implemented during 2024 with related restructuring charges.

<sup>(2)</sup> Restructuring and other costs relate to actions taken for headcount reductions, facility consolidations and site closures, product line exits, and other asset charges.

<sup>(3)</sup> Gain on dispositions, including post-closing adjustments; see Note 4 — Discontinued and Disposed Operations in the condensed consolidated financial statements in Item 1 of this Form 10-Q for further details.

<sup>(4)</sup> Certain expenses are maintained at the corporate level and not allocated to the segments. These expenses include executive and functional compensation costs, non-service pension costs, non-operating insurance expenses, shared business services and digital overhead costs, deal related expenses and various administrative expenses relating to the corporate headquarters.

We recorded the following restructuring and other costs for the three and nine months ended September 30, 2024:

Thron	Months	Ended	Sentember	30	2024
i nree	VIONINS	ranaea	Sentember	.70.	20124

(in thousands)	ngineered Products	Clean Energy & Fueling	Imaging & Identification	Pumps & Process Solutions	Climate & Sustainability Technologies	Corporate	Total
Restructuring	\$ 991	\$ 8,544	\$ 1,804	\$ 964	\$ 1,238	\$ 296	\$ 13,837
Other (benefits) costs	(4)	438	1,545	14	320	431	2,744
Restructuring and other costs	\$ 987	\$ 8,982	\$ 3,349	\$ 978	\$ 1,558	\$ 727	\$ 16,581

# Nine Months Ended September 30, 2024

(in thousands)	Engineered Products	(	Clean Energy & Fueling	Imaging & Identification	Pumps & Process Solutions	Climate & Sustainability Technologies	(	Corporate	Total
Restructuring	\$ 2,969	\$	15,434	\$ 4,645	\$ 3,929	\$ 14,261	\$	391	\$ 41,629
Other costs, net	12		1,779	2,773	66	4,208		1,675	10,513
Restructuring and other costs	\$ 2,981	\$	17,213	\$ 7,418	\$ 3,995	\$ 18,469	\$	2,066	\$ 52,142

Restructuring and other costs for the three and nine months ended September 30, 2023 included restructuring charges of \$4.4 million and \$33.3 million, respectively and other costs, net of \$7.2 million and \$10.5 million, respectively. Restructuring charges for the three months ended September 30, 2023 primarily related to headcount reductions and exit costs in the Pumps & Process Solutions, Climate & Sustainability Technologies and Engineered Products segments. Restructuring charges for the nine months ended September 30, 2023 primarily related to headcount reductions and exit costs in the Clean Energy & Fueling, Pumps & Process Solutions, Engineered Products and Climate & Sustainability Technologies segments. These restructuring programs were initiated in 2022 and 2023 and were undertaken in light of current market conditions. Other costs, net of \$7.2 million and \$10.5 million for the three and nine months ended September 30, 2023, were primarily due to an asset impairment in our Climate & Sustainability Technologies segment and product line rationalization and footprint reduction in our Clean Energy & Fueling segment. These restructuring and other charges were recorded in cost of goods and services and selling, general and administrative expenses in the condensed consolidated statement of earnings.

We recorded the following restructuring and other costs for the three and nine months ended September 30, 2023:

#### Three Months Ended September 30, 2023

(in thousands)	1	Engineered Products		ean Energy & Fueling		Imaging & lentification		Pumps & Process Solutions		Climate & Sustainability Technologies	(	Corporate		Total
Restructuring	\$	938	\$	(37)	\$	233	\$	1,637	\$	1,138	\$	476	\$	4,385
Other costs, net		37		1,182		328		113		3,939		1,597		7,196
Restructuring and other costs	\$	975	\$	1,145	\$	561	\$	1,750	\$	5,077	\$	2,073	\$	11,581

# Nine Months Ended September 30, 2023

(in thousands)	]	Engineered Products	(	Clean Energy & Fueling	Imaging & Identification	Pumps & Process Solutions	Climate & Sustainability Technologies	(	Corporate	Total
Restructuring	\$	5,415	\$	15,954	\$ 1,437	\$ 6,266	\$ 2,585	\$	1,603	\$ 33,260
Other costs, net		71		3,141	824	110	4,642		1,729	10,517
Restructuring and other costs	\$	5,486	\$	19,095	\$ 2,261	\$ 6,376	\$ 7,227	\$	3,332	\$ 43,777

# Purchase Accounting Expenses

Purchase accounting expenses primarily relate to amortization of acquired intangible assets. These expenses are not presented in our segment earnings because they are excluded from the segment operating performance measure reviewed by management. These expenses reconcile to segment earnings as follows:

	Three Months Ended September 30,				Nine Months Ended September 30,			
(in thousands)		2024		2023		2024	2023	
Purchase Accounting Expenses								
Engineered Products	\$	2,806	\$	3,441	\$	8,049	\$	10,604
Clean Energy & Fueling		24,928		19,775		67,231		58,882
Imaging & Identification		5,914		5,803		17,320		17,354
Pumps & Process Solutions		9,658		5,111		29,131		16,888
Climate & Sustainability Technologies		5,050		4,826		15,144		14,475
Total	\$	48,356	\$	38,956	\$	136,875	\$	118,203

#### FINANCIAL CONDITION

We assess our liquidity in terms of our ability to generate cash to fund our operating, investing and financing activities. Significant factors affecting liquidity are cash flows generated from operating activities, capital expenditures, acquisitions, dispositions, dividends, repurchase of outstanding shares, adequacy of available commercial paper and bank lines of credit and the ability to attract long-term capital with satisfactory terms. We generate substantial cash from the operations of our businesses and remain in a strong financial position, with sufficient liquidity available for reinvestment in existing businesses and strategic acquisitions.

#### **Cash Flow Summary**

The following table is derived from our condensed consolidated statements of cash flows:

	Nine	Nine Months Ended September 30,								
Cash Flows from Operations (in thousands)	203	24		2023						
Net cash flows provided by (used in):										
Operating activities	\$	648,881	\$	720,982						
Investing activities		63,119		(131,072)						
Financing activities		(818,445)		(757,205)						

# **Operating Activities**

Cash flow from operating activities for the nine months ended September 30, 2024 decreased by \$72.1 million compared to September 30, 2023, primarily due to tax payments related to the gain from the De-Sta-Co divestiture.

Adjusted Working Capital: We believe adjusted working capital (a non-GAAP measure calculated as receivables, plus inventory, less accounts payable) provides a meaningful measure of liquidity by showing changes caused by operational results. The following table provides a calculation of adjusted working capital:

Adjusted Working Capital (in thousands)	Septembe	I	December 31, 2023		
Receivables, net	\$	1,428,961	\$	1,321,107	
Inventories, net		1,214,268		1,144,089	
Less: Accounts payable		865,188		854,465	
Adjusted working capital	\$	1,778,041	\$	1,610,731	

Adjusted working capital increased by \$167.3 million, or 10.4%, in the nine months ended September 30, 2024, which reflected an increase of \$107.9 million in receivables, net, an increase of \$70.2 million in inventory, net and an increase in accounts payable of \$10.7 million. These amounts include the effects of acquisitions, dispositions and foreign currency translation. The increase in adjusted working capital versus year-end 2023 is primarily a result of timing of cash flows, with the fourth quarter traditionally representing our highest cash flow quarter.

# **Investing Activities**

Cash flow from investing activities is derived from cash inflows from proceeds from dispositions, offset by cash outflows for acquisitions and capital expenditures. The majority of the activity in investing activities was comprised of the following:

- *Proceeds from dispositions:* During the nine months ended September 30, 2024, we received net proceeds of \$767.7 million from the sales of De-Sta-Co, an operating company within the Engineered Products segment, and a minority owned equity method investment within Climate & Sustainability Technologies segment. There were no proceeds from disposition during the nine months ended September 30, 2023. See Note 4 Discontinued and Disposed Operations in the condensed consolidated financial statements in Item 1 of this Form 10-Q for further details.
- Acquisitions: During the nine months ended September 30, 2024, we deployed approximately \$602.7 million, net, to acquire seven businesses. In comparison, during the nine months ended September 30, 2023, we deployed approximately \$7.2 million, net to acquire one business. See Note 3

   Acquisitions in the condensed consolidated financial statements in Item 1 of this Form 10-Q for further details.

• Capital spending: Capital expenditures decreased \$12.5 million during the nine months ended September 30, 2024, compared to the nine months ended September 30, 2023, in line with our planned capital expenditures for the year.

We anticipate that capital expenditures and any additional acquisitions we make through the remainder of 2024 will be funded from available cash and internally generated funds and, if necessary, through the issuance of commercial paper, or by accessing the public debt or equity markets. We estimate capital expenditures in 2024 to range from \$145.0 million to \$155.0 million.

# Financing Activities

Cash flow from financing activities generally relates to the use of cash for purchases of our common stock and payment of dividends, offset by net borrowing activity. The majority of financing activity was attributed to the following:

- \* Repurchase of common stock, including accelerated share repurchase program: During the nine months ended September 30, 2024, the Company received a total of 2,869,282 shares upon completion the ASR Agreement for \$500.0 million. During the nine months ended September 30, 2023, we repurchased no shares. See Note 17 Stockholders' Equity in the condensed consolidated financial statements in Item 1 of this Form 10-Q for further details.
- Commercial paper and other short-term borrowings, net: During the nine months ended September 30, 2024 and 2023, we used \$89.0 million and \$528.8 million, respectively, to pay off commercial paper borrowings.
- Dividend payments: Total dividend payments to common shareholders were \$212.4 million during the nine months ended September 30, 2024, as compared to \$212.9 million during the same period in 2023. Our dividends paid per common share increased 1.0% to \$1.54 during the nine months ended September 30, 2024 compared to \$1.52 during the same period in 2023. The number of common shares outstanding decreased from September 30, 2023 to September 30, 2024, as share repurchases exceeded share issuances.

#### Cash Flows from Discontinued Operations

Our cash flows from discontinued operations for the nine months ended September 30, 2024 and 2023, generated \$93.9 million and \$93.8 million, respectively, representing the operating results of ESG during the periods presented. Cash flows from discontinued operations generated for the nine months ended September 30, 2024, primarily relate to cash provided by operations of approximately \$108.3 million, partially offset by cash used in investing activities of \$14.4 million, which comprised capital expenditures partially offset by proceeds from a sale of investment. Cash flows generated for the nine months ended September 30, 2023, primarily relate to cash provided by operating activities of \$99.3 million, partially offset by capital expenditures of \$5.5 million.

# **Liquidity and Capital Resources**

#### Free Cash Flow

In addition to measuring our cash flow generation and usage based upon the operating, investing and financing classifications included in the condensed consolidated statements of cash flows, we also measure free cash flow (a non-GAAP measure) which represents net cash provided by operating activities minus capital expenditures. We believe that free cash flow is an important measure of liquidity because it provides management and investors a measurement of cash generated from operations that may be available for mandatory payment obligations and investment opportunities, such as funding acquisitions, paying dividends, repaying debt and repurchasing our common stock.

The following table reconciles our free cash flow to cash flow provided by operating activities:

		Nine Months End	led Sej	ptember 30,
Free Cash Flow (dollars in thousands)		2024		2023
Cash flow provided by operating activities	\$	648,881	\$	720,982
Less: Capital expenditures		(113,626)		(126,131)
Free cash flow	\$	535,255	\$	594,851
Cash flow from operating activities as a percentage of revenue	<u> </u>	11.2 %		12.5 %
Cash flow from operating activities as a percentage of earnings from continuing operations		55.9 %		105.1 %
Free cash flow as a percentage of revenue		9.2 %		10.3 %
Free cash flow as a percentage of earnings from continuing operations		46.1 %		86.8 %

For the nine months ended September 30, 2024, we generated free cash flow of \$535.3 million, representing 9.2% of revenue and 46.1% of earnings from continuing operations. Free cash flow for the nine months ended September 30, 2024, decreased \$59.6 million, compared to September 30, 2023, primarily due to tax payments on the De-Sta-Co gain of \$80.0 million, partially offset by lower capital expenditures. The remainder of the tax payments on the De-Sta-Co gain will be paid in the fourth quarter of 2024. The decreases in cash flow from operating activities and free cash flow as percentages of earnings from continuing operations are due primarily to the gain on sale of De-Sta-Co. See Note 4 — Discontinued and Disposed Operations in the condensed consolidated financial statements in Item 1 of this Form 10-Q for further details.

#### Capitalization

We use commercial paper borrowings for general corporate purposes, including the funding of acquisitions and the repurchase of our common stock. As of September 30, 2024, we maintained \$1.0 billion five-year and \$500.0 million 364-day unsecured revolving credit facilities (together, the "Credit Agreements") with a syndicate of banks which expire April 6, 2028 and April 3, 2025, respectively. The Credit Agreements are designated as a liquidity back-stop for the Company's commercial paper program and also are available for general corporate purposes.

At the Company's election, loans under the Credit Agreements will bear interest at a base rate plus an applicable margin. The Credit Agreements require the Company to pay facility fees and impose various restrictions on the Company such as, among other things, a requirement to maintain an interest coverage ratio of consolidated EBITDA to consolidated net interest expense of not less than 3.0 to 1.0. The Company was in compliance with all covenants in the Credit Agreements and other long-term debt covenants at September 30, 2024 and had an interest coverage ratio of consolidated EBITDA to consolidated net interest expense of 20.7 to 1. We are not aware of any potential impairment to our liquidity and expect to remain in compliance with all of our debt covenants. Additionally, our earliest long-term debt maturity is in 2025.

We also have a current shelf registration statement filed with the Securities and Exchange Commission that allows for the issuance of additional debt securities that may be utilized in one or more offerings on terms to be determined at the time of the offering. Net proceeds of any offering would be used for general corporate purposes, including repayment of existing indebtedness, capital expenditures and acquisitions.

At September 30, 2024, our cash and cash equivalents, including cash held for sale, totaled \$396.8 million, of which approximately \$251.9 million was held outside the United States. At December 31, 2023, our cash and cash equivalents, including cash held for sale, totaled \$415.9 million, of which approximately \$286.9 million was held outside the United States. Cash and cash equivalents are held primarily in bank deposits with highly rated banks. We regularly hold cash in excess of near-term requirements in bank deposits or invest the funds in government money market instruments or short-term investments, which consist of investment grade time deposits with original maturity dates at the time of purchase of no greater than three months.

On March 31, 2024, the Company completed the sale of the De-Sta-Co business for total consideration, net of cash transferred, of \$674.7 million. On September 30, 2024, a minority owned equity method investment held within the Climate & Sustainability Technologies segment was sold and the Company received its proportionate share of the proceeds amounting to \$93.0 million. On October 8, the Company completed the previously announced sale of the ESG business, an operating company in the Engineered Products segment, for total consideration, net of cash transferred, of \$2.0 billion, subject to

customary post-closing adjustments. See Note 4 — Discontinued and Disposed Operations in the condensed consolidated financial statements in Item 1 of this Form 10-Q for further details.

During the nine months ended September 30, 2024, the Company completed seven business acquisitions for total consideration of \$636.4 million, subject to post-closing adjustments and inclusive of contingent consideration. See Note 3 — Acquisitions in the condensed consolidated financial statements in Item 1 of this Form 10-Q for further details.

We utilize the net debt to net capitalization calculation (a non-GAAP measure) to assess our overall financial leverage and capacity and believe the calculation is useful to investors for the same reason. Net debt represents total debt minus cash and cash equivalents, including cash held for sale. Net capitalization represents net debt plus stockholders' equity. The following table provides a calculation of net debt to net capitalization from the most directly comparable GAAP measures:

**Net Debt to Net Capitalization Ratio** 

(dollars in thousands)	Sept	ember 30, 2024	Dec	ember 31, 2023
Commercial paper	\$	378,600	\$	467,600
Other		695		682
Total short-term borrowings		379,295		468,282
Long-term debt		3,007,820		2,991,759
Total debt		3,387,115		3,460,041
Less: Cash and cash equivalents, including cash held for sale		(396,766)		(415,861)
Net debt		2,990,349		3,044,180
Add: Stockholders' equity		5,697,999		5,106,605
Net capitalization	\$	8,688,348	\$	8,150,785
Net debt to net capitalization		34.4 %		37.3 %

Our net debt to net capitalization ratio decreased to 34.4% at September 30, 2024 compared to 37.3% at December 31, 2023. Net debt decreased \$53.8 million during the period primarily due to net proceeds from the sale of De-Sta-Co that were used to reduce our commercial paper borrowings. Stockholders' equity increased for the period as a result of current earnings of \$1,261.1 million, partially offset by share repurchases under the ASR program and dividends paid during the period.

Operating cash flow and access to capital markets are expected to satisfy our various cash flow requirements, including acquisitions, capital expenditures, purchase obligations, and lease obligations. Acquisition spending and/or share repurchases could potentially increase our debt.

We believe that existing sources of liquidity are adequate to meet anticipated funding needs at current risk-based interest rates for the foreseeable future.

# **Critical Accounting Estimates**

Our condensed consolidated financial statements and related public financial information are based on the application of GAAP which requires the use of estimates, assumptions, judgments and subjective interpretations of accounting principles that have an impact on the assets, liabilities, revenue and expense amounts reported. These estimates can also affect supplemental information contained in our public disclosures, including information regarding contingencies, risk and our financial condition. We believe our use of estimates and underlying accounting assumptions conform to GAAP and are consistently applied. We review valuations based on estimates for reasonableness on a consistent basis.

# **Recent Accounting Standards**

See Note 19 — Recent Accounting Pronouncements in the condensed consolidated financial statements in Item 1 of this Form 10-Q. The adoption of recent accounting standards as included in Note 19 — Recent Accounting Pronouncements in the condensed consolidated financial statements has not had, and is not expected to have, a significant impact on our revenue, earnings or liquidity.

# **Special Note Regarding Forward-Looking Statements**

This Quarterly Report on Form 10-Q, especially MD&A, contains "forward-looking" statements within the meaning of the Private Securities Litigation Reform Act of 1995, as amended. All statements in this document other than statements of historical fact are statements that are, or could be deemed, "forward-looking" statements. Some of these statements may be indicated by words such as "may", "anticipate", "expect", "believe", "intend", "continue", "guidance", "estimates", "suggest", "will", "plan", "should", "would", "forecast" and other words and terms that use the future tense or have a similar meaning. Forward-looking statements are based on current expectations and are subject to numerous important risks, uncertainties, and assumptions, including those described in our Annual Report on Form 10-K for the year ended December 31, 2023. Factors that could cause actual results to differ materially from current expectations include, among other things: general economic conditions and conditions in the particular markets in which we operate; supply chain constraints and labor shortages that could result in production stoppages, inflation in material input costs and freight logistics; the impacts of natural or human induced disasters, acts of war, terrorism, international conflicts, and public health crises or other future pandemics on the global economy and on our customers, suppliers, employees, business and cash flows; changes in customer demand and capital spending; competitive factors and pricing pressures; our ability to develop and launch new products in a cost-effective manner; changes in law, including the effect of tax laws and developments with respect to trade policy and tariffs; our ability to identify and complete acquisitions and integrate and realize synergies from newly acquired businesses; the impact of interest rate and currency exchange rate fluctuations; capital allocation plans and changes in those plans, including with respect to dividends, share repurchases, investments in research and development, capital expenditures and acquisitions; our ability to derive expected benefits from restructurings, productivity initiatives and other cost reduction actions; the impact of legal compliance risks and litigation, including with respect to product quality and safety, cybersecurity and privacy; and our ability to capture and protect intellectual property rights, and various other factors that are described in our periodic reports filed with or furnished to the Securities and Exchange Commission, including our Annual Report on Form 10-K for the year ended December 31, 2023. The Company undertakes no obligation to publicly update any forward-looking statement, whether as a result of new information, future events or otherwise, except as required by law. Given these risks and uncertainties, readers are cautioned not to place undue reliance on such forward-looking statements.

The Company may, from time to time, post financial or other information on its website, www.dovercorporation.com. The website is for informational purposes only and is not intended for use as a hyperlink. The Company is not incorporating any material on its website into this report.

#### **Non-GAAP Disclosures**

In an effort to provide investors with additional information regarding our results as determined by GAAP, we also disclose non-GAAP information, which we believe provides useful information to investors. Free cash flow, free cash flow as a percentage of revenue, free cash flow as a percentage of earnings from continuing operations, net debt, net capitalization, net debt to net capitalization ratio, adjusted working capital, and organic revenue growth are not financial measures under GAAP and should not be considered as a substitute for cash flows from operating activities, debt or equity, working capital or revenue as determined in accordance with GAAP, and they may not be comparable to similarly titled measures reported by other companies. We believe the net debt to net capitalization ratio and free cash flow are important measures of liquidity. Net debt to net capitalization is helpful in evaluating our capital structure and the amount of leverage we employ. Free cash flow and free cash flow ratios provide both management and investors a measurement of cash generated from operations that is available to fund acquisitions, pay dividends, repay debt and repurchase our common stock. Free cash flow as a percentage of revenue equals free cash flow divided by revenue. Free cash flow as a percentage of earnings from continuing operations equals free cash flow divided by earnings from continuing operations. We believe that reporting adjusted working capital provides a meaningful measure of liquidity by showing changes caused by operational results. We believe that reporting organic revenue growth provides a useful comparison of our revenue performance and trends between periods.

Reconciliations and comparisons to non-GAAP measures can be found above in this Item 2, Management's Discussion and Analysis of Financial Condition and Results of Operations.

# Item 3. Quantitative and Qualitative Disclosures About Market Risk

There has been no significant change in our exposure to market risk during the nine months ended September 30, 2024. For a discussion of our exposure to market risk, refer to Item 7A, "Quantitative and Qualitative Disclosures about Market Risk," contained in our Annual Report on Form 10-K for the fiscal year ended December 31, 2023.

# **Item 4. Controls and Procedures**

At the end of the period covered by this report, the Company carried out an evaluation, under the supervision and with the participation of the Company's management, including the Company's Chief Executive Officer and Chief Financial Officer, of the effectiveness of the design and operation of the Company's disclosure controls and procedures pursuant to Exchange Act Rule 13a-15(e). Based upon that evaluation, the Chief Executive Officer and Chief Financial Officer concluded that the Company's disclosure controls and procedures were effective as of September 30, 2024.

During the third quarter of 2024, there were no changes in the Company's internal control over financial reporting that materially affected, or are reasonably likely to materially affect, the Company's internal control over financial reporting.

# PART II — OTHER INFORMATION

## Item 1. Legal Proceedings

See Note 14 — Commitments and Contingent Liabilities in the condensed consolidated financial statements in Item 1 of this Form 10-Q.

#### Item 1A. Risk Factors

There have been no material changes from the risk factors previously disclosed in our Annual Report on Form 10-K for the year ended December 31, 2023.

# Item 2. Unregistered Sales of Equity Securities, Use of Proceeds and Issuer Purchases of Equity Securities

- Not applicable.
- b. Not applicable.
- c. The below table presents shares of Dover stock that we acquired during the quarter.

Period	Total Number of Shares Purchased	Average Price Paid per Share (2)	Total Number of Shares Purchased as Part of Publicly Announced Plans or Programs	Maximum Number (or Approximate Dollar Value) of Shares that May Yet Be Purchased under the Plans or Programs (1)
July 1 to July 31	299,443	\$ 174.26	299,443	17,130,718
August 1 to August 31	_	_	_	17,130,718
September 1 to September 30				17,130,718
For the Third Quarter	299,443	\$ 174.26	299,443	17,130,718

<sup>(1)</sup> In August 2023, the Company's Board of Directors approved a new standing share repurchase authorization whereby the Company may repurchase up to 20 million shares beginning on January 1, 2024 through December 31, 2026. As of September 30, 2024, the number of shares still available for repurchase under the current share repurchase authorization was 17,130,718.

# Item 3. Defaults Upon Senior Securities

Not applicable.

# **Item 4. Mine Safety Disclosures**

Not applicable.

# **Item 5. Other Information**

a.- b. None.

c. During the nine months ended September 30, 2024, no director or Section 16 officer adopted or terminated any Rule 10b5-1 trading arrangements or non-Rule 10b5-1 trading arrangements as defined in Item 408 of Regulation S-K.

<sup>(2)</sup> Under the terms of the ASR Agreement, the Company paid Citibank \$500.0 million on March 1, 2024 and on that date received initial delivery of 2,569,839 shares, representing a substantial majority of the shares expected to be retired over the course of the ASR Agreement. In July, Citibank delivered 299,443 additional shares which completed the ASR Program. During 2024, the Company received a total of 2,869,282 shares upon completion of the ASR Agreement. The total number of shares ultimately repurchased under the ASR Agreement was based on the volume-weighted average share price of Dover's common stock during the calculation period of the ASR Program, less a discount, which was \$174.26 over the term of the ASR Program.

# Item 6. Exhibits

- 2.1\*+ Transaction Agreement, dated as of July, 21 2024, by and between Dover Corporation and Terex Corporation.
- 2.2\* First Amendment to Transaction Agreement, by and between Dover Corporation and Terex Corporation, dated as of October 8, 2024.
- 31.1 Certificate pursuant to Rule 13a-14(a) of the Securities Exchange Act of 1934, as amended, signed and dated by Brad M. Cerepak.
- 31.2 Certificate pursuant to Rule 13a-14(a) of the Securities Exchange Act of 1934, as amended, signed and dated by Richard J. Tobin.
  - 32 <u>Certification pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, signed and dated by Richard J. Tobin and Brad M. Cerepak.</u>
- 101 The following materials from Dover Corporation's Quarterly Report on Form 10-Q for the quarter ended September 30, 2024 formatted in iXBRL (Inline eXtensible Business Reporting Language): (i) the Condensed Consolidated Statements of Earnings, (ii) the Condensed Consolidated Statements of Comprehensive Earnings, (iii) the Condensed Consolidated Balance Sheets, (iv) the Condensed Consolidated Statements of Stockholders' Equity, (v) the Condensed Consolidated Statements of Cash Flows, and (vi) Notes to the Condensed Consolidated Financial Statements.
- 104 Cover Page formatted in Inline XBRL and contained in Exhibit 101.
  - \* Certain schedules, annexes or exhibits have been omitted pursuant to item 601(a)(5) of Regulation S-K, but will be furnished supplementally to the SEC upon request.
  - + Portions of Exhibit 2.1 have been redacted in accordance with Item601(b)(2)(ii) of Regulation S-K.

# Signatures

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this Report on Form 10-Q to be signed on its behalf by the undersigned thereunto duly authorized.

DOVER CORPORATION

Date: October 24, 2024 /s/ Brad M. Cerepak

Brad M. Cerepak

Senior Vice President & Chief Financial Officer

(Principal Financial Officer)

Date: October 24, 2024 /s/ Ryan W. Paulson

Ryan W. Paulson

Vice President, Controller (Principal Accounting Officer)

# CERTAIN CONFIDENTIAL PORTIONS OF THIS EXHIBIT MARKED WITH "[redacted]" HAVE BEEN REDACTED IN ACCORDANCE WITH ITEM 601(b)(2)(ii) OF REGULATION S-K.

TRANSACTION AGREEMENT

BY AND BETWEEN

TEREX CORPORATION

**AND** 

**DOVER CORPORATION** 

Dated as of July 21, 2024

# TABLE OF CONTENTS

# Article I

	Thitlete 1	
1.1	Definitions	2
	Article II	
0.1	THE TRANSACTIONS	•
2.1	Sales and Purchases	23
2.2	Purchase Price	23
2.3	Closing Purchase Price	23
2.4	Post-Closing Adjustment	24 29
2.5	Withholding Article III	29
	CLOSING AND CLOSING DELIVERIES	
3.1	Closing; Time and Place	30
3.2	Deliveries by Seller	30
3.3	Deliveries by Purchaser	31
3.4	Payment Mechanics	31
	Article IV	
	REPRESENTATIONS AND WARRANTIES OF SELLER	
4.1	Authority; Enforceability	32
4.2	Non-Contravention; Consents	33
4.3	Organization; Acquired Companies	33
4.4	Title; Shares	35
4.5	Financial Information; Liabilities	35
4.6	Absence of Certain Changes	37
4.7	Compliance with Legal Requirements	37
4.8	Material Contracts	38
4.9	Litigation	40
4.1	Intellectual Property; Information Technology	31
4.11	Real Property	44
4.12	Labor Matters	46
4.13	Employee Benefits	47
4.14	Taxes	49
4.15	Sufficiency of Assets	41
4.16	Environmental Matters	52
4.17	Anti-Bribery Matters	53
4.18	Brokers	53
4.19	Related Party Transactions	54
4.2	Key Customers and Key Suppliers	52
4.21	Accounts Payable; Accounts Receivable	54
4.22	Product Warranties; Product Liability	55
4.23	Government Contracts	55
4 24	Insurance	57

4.25	Trade Controls	57
4.26	No Other Representations	58
	Article V	
	REPRESENTATIONS AND WARRANTIES OF PURCHASER	
5.1	Authority; Enforceability	57
5.2	Non-Contravention; Consents	59
5.3	Organization	60
5.4	Litigation	60
5.5	Securities Matters	60
5.6	Financing	60
5.7	Solvency	62
5.8	Brokers	52
5.9	R&W Insurance Policy	63
5.1	Pending Transactions	63
5.11	Investigation	63
5.12	Disclaimer Regarding Projections	64
5.13	No Other Representations	64
	Article VI	
	COVENANTS OF THE PARTIES	
6.1	Conduct of the Business Prior to Closing	65
6.2	Pre-Closing Access to Information	69
6.3	Cooperation	70
6.4	Consents; Termination of Intercompany Agreements	71
6.5	Confidentiality	72
6.6	Reasonable Best Efforts; Regulatory Filings	73
6.7	Financing Cooperation	76
6.8	Financing Obligation	79
6.9	Financial Statements	82
6.1	Insurance	82
6.11	R&W Insurance Policy	84
6.12	Registered Office Addresses	84
6.13	Exclusivity	85
	Article VII	
	ADDITIONAL COVENANTS OF THE PARTIES	
7.1	Transitional Trademark Rights	85
7.2	IP Licenses	86
7.3	Post-Closing Access to Information	87
7.4	D&O Insurance	87
7.5	Non-Competition; Non-Solicitation	89

7.6	Further Assurances	91
7.7	Assignment of Right to Enforce Restrictive Covenants	92
	Article VIII	
	TAX MATTERS	
8.1	Section 338(h)(10) Election; Purchase Price Allocation	92
8.2	Tax Returns	84
8.3	Tax Sharing Agreements	95
8.4	Transfer Taxes	95
8.5	Straddle Periods	95
8.6	Cooperation	96
8.7	Prohibited Actions	96
8.8	Survival	97
0.0	Article IX	91
	EMPLOYEES	
9.1	Transfer of Business Employees	97
9.2	Continuation Period	97
9.3	Annual Cash Bonuses	98
9.4	Vacation and Paid Time Off	98
9.5	Communications	99
9.6	Immigration Compliance	99
9.7	COBRA	99
9.8	No Third-Party Beneficiaries	99
	Article X	
	CONDITIONS TO CLOSING	
10.1	Conditions of Purchaser	99
10.2	Conditions of Seller	100
10.3	Mutual Conditions	101
10.4	Failure of Conditions	101
10.5	Waiver of Conditions	101
	Article XI	
	TERMINATION	
11.1	Termination	101
11.2	Notice of Termination	103
11.3	Effect of Termination	103
11.0	Article XII	100
	SURVIVAL; INDEMNIFICATION	
12.1	Survival Survival	103
12.1	Indemnification by Seller	103
14.4	machinina di	107

12.3	Indemnification by Purchaser	105
12.4	Procedures	105
12.5	Exclusive Remedy	109
12.6	Additional Indemnification Provisions	109
12.7	Limitation on Liability	109
12.8	Tax Treatment of Indemnity Payments and Post-Closing Adjustments	110
12.9	Indemnification Obligations Net of Insurance Proceeds and Other Amounts	110
	Article XIII	
	MISCELLANEOUS PROVISIONS	
13.1	Expenses	111
13.2	Interpretation	111
13.3	Entire Agreement	112
13.4	Amendment and Waivers	112
13.5	Successors and Assigns	112
13.6	Governing Law	113
13.7	Jurisdiction; Venue; Service of Process	113
13.8	Waiver of Jury Trial	114
13.9	Specific Performance	115
13.1	Severability	115
13.11	Certain Releases	116
13.12	The Seller Disclosure Schedule, Exhibits and Schedules	117
13.13	Notices	118
13.14	No Third-Party Beneficiaries	119
13.15	Provision Regarding Legal Representation	119
13.16	No Other Duties	120
13.17	Reliance on Counsel and Other Advisors	120
13.18	Public Announcements	120
13.19	Counterparts	120

# **Exhibits**

Exhibit A: IP Seller IP

Exhibit B: Form of Transition Services Agreement

Exhibit C: Intellectual Property Assignment Agreement

Exhibit D: R&W Insurance Policy

**Schedules** 

Schedule A: Accounting Principles

# INDEX

Section	
Accounting Principles	1.1
Accrued PTO	9.4
Acquired Companies	1.1
Acquired Company	1.1
Acquired Company Benefit Plan	1.1
Acquired Company IP	1.1
Acquired Company IT Assets	1.1
Acquired Company Registered IP	4.10(a)
Acquired Company Source Code	4.10(h)
Acquired Entities	Recitals
Acquired Entity	Recitals
Acquired Entity 1	Recitals
Acquisition Proposal	1.1
Affiliate	1.1
Agreement	1.1
Allocation	8.1(e)(i)
Alternative Debt Financing	6.8(c)
Alternative Debt Financing Commitment	6.8(c)
Anti-Bribery Laws	4.17
Antitrust Law	1.1
Assumed Incentive Amount	9.3
Assumed Liabilities	1.1
Audited Financial Information	6.9(a)
Available Insurance Policies	6.10(c)
Bankruptcy and Equity Exception	1.1
Benchmark Time	1.1
Benefit Plan	1.1
Business	1.1
Business Day	1.1
Business Employee	1.1
Canadian Business Employee	1.1
CARES Act	1.1
Cash	1.1
Claim Notice	12.4(a)
Closing	3.1
Closing Conditions	1.1
Closing Date	3.1
Closing Purchase Price	2.3(a)
Closing Transaction Expenses	1.1
Code	1.1
Collective Bargaining Agreement	1.1
Competing Business Activities	1.1
Compliant	1.1

Condition Satisfaction Date	1.
Confidentiality Agreement	1.
Consent	1.
Consolidated Return	1.
Consultation Period	2.4(e
Contagion Event	ì.
Continuation Period	9.2(a
Contract	1.
Copyrights	1.
Corporate Functions	1.
COVID-19	1.
Credit Agreement Amendment	5.6(a
Current Insurance	7.4(b
D&O Indemnitee	7.4(a
Data Security Requirements	4.10(k
Debt Commitment Documents	5.6(a
Debt Commitment Letter	5.6(a
Debt Financing	1.
Debt Financing Commitment	5.6(a
Definitive Agreements	6.8(a
Disputed Items	2.4(e
DOJ	6.6(a
eCommerce Platform Software	1.
Employment Commencement Date	1.
Encumbrance	1.
Environmental Authorizations	4.16(a
Environmental Law	1.
ERISA	1.
Estimated Closing Statement	2.3(a
Exchange Act	4.2(a
Final Closing Indebtedness	2.4(f
Final Closing Net Working Capital	2.4(f
Final Closing Statement	2.4(f
Final Closing Transaction Expenses	2.4(f
Final Overage	2.4(g
Final Purchase Price	2.4(1
Final Underage	2.4(h
Financing Entities	1.
Financing Related Action	13.7(c
Financing Source	1.
Former Business Employee	1.
Fraud	1.
FTC	6.6(a
CAAD	1

1.1

Government Contract

Governmental Approvals	6.6(a)
Governmental Authority	1,1
Hazardous Materials	1,1
HSR Act	1.1
Income Tax	1,1
Indebtedness	1.1
Indemnified Party	1.1
Indemnified Party Defense Matter	12.4(c
Indemnifying Party	1,1
Indemnity Payment	12.9
Initial Deductible	12.2(b
Initial Marketing Period	1,1
Insurance Policies	4.24
Insurance Proceeds	12.9
Insured Persons	7.4(b
Intellectual Property	1.1
Intellectual Property Assignment Agreement	3.2(c
Intercompany Agreements	6.4(b
Investment Screening Laws	1.1
IP Contract	1,1
IP Seller	1.1
IP Seller IP	1,1
IRS	1.1
IT Assets	1.1
Key Customer	1.1
Key Supplier	1.1
Latest Balance Sheet	4.5(a
Leased Real Property	1.1
Legal Requirement	1.1
Liabilities	1.1
Losses	1.1
Malicious Code	1,1
Marketing Periods	1,1
Material Adverse Effect	1,1
Material Contracts	4.8(a
Net Working Capital	1.1
Net Working Capital Overage	1,1
Net Working Capital Underage	1.1
Non-Business Proprietary Information	6.5(a
Notice Period	12.4(a
Open Source Software	1.1
Option	4.11(a
Options	4.11(a
Order	1.1

11.1(d)

Outside Date

Owned Real Property	4.11(a)
Patents	1.1
Pension Plan	4.13(c)
Permit	1.1
Permitted Encumbrances	1.1
Person	1.1
Personal Information	1.1
Post-Closing Statement	2.4(a)
Post-Closing Straddle Period	1.1
Post-Closing Tax Period	1.1
Pre-Closing Claims	6.10(c)
Pre-Closing Covenants	12.1(a)
Pre-Closing Straddle Period	1.1
Pre-Closing Tax Period	1.1
Preliminary Cash	2.4(a)
Preliminary Closing Purchase Price	2.4(a)
Preliminary Closing Transaction Expenses	2.4(a)
Preliminary Indebtedness	2.4(a)
Preliminary Net Working Capital	2.4(a)
Privacy Laws	1.1
Privileged Communications	13.15
Proceeding	1.1
Processing	1.1
Prohibited Modification	6.8(b)
Proprietary Software	4.10(h)
Purchase Price	2.2
Purchaser	Preamble
Purchaser Disclosure Schedule	1.1
Purchaser Fundamental Representations	1.1
Purchaser Indemnified Parties	12.2(a)
Purchaser Material Adverse Effect	1.1
Purchaser Releasee	13.11(b)
Purchaser's Draft Allocation	8.1(e)(i)
Purchaser's Knowledge	1.1
Purchaser-Filed Tax Return	8.2(b)
R&W Binder Agreement	5.9
R&W Insurance Policy	5.9
Real Property	1.1
Real Property Lease	4.11(b)
Real Property Leases	4.11(b)
Reference Date	4.5(a)
Related Person	4.19
Release	1.1
Representatives	1.1

5.6(b)

Required Amount

Required Information	1.1
Restricted Cash	1.1
Restricted Person	1.1
Retained Liabilities	1.1
Review Period	2.4(d)
Sanctioned Jurisdiction	1.1
Sanctioned Person	1.1
Sanctions	1.1
SEC	1.1
Second Deductible	12.2(b)
Secondary Marketing Period	1.1
Section 338(h)(10) Election	8.1(a)
Section 338(h)(10) Election Taxes	8.1(f)
Section 338(h)(10) Forms	8.1(b)
Securities Act	1.1
Security Plan	4.10(m)
Seller	Preamble
Seller Affiliated Group	1.1
Seller Benefit Plan	1.1
Seller Disclosure Schedule	1.1
Seller Fundamental Representations	1.1
Seller Group	1.1
Seller Group Marks	1.1
Seller Indemnified Parties	12.3
Seller Parties	5.9
Seller Party	1.1
Seller Post-Closing Statement	2.4(c)
Seller Releasee	13.11(a)
Seller's Allocation Notice	8.1(e)(i)
Seller's Knowledge	1.1
Seller-Filed Tax Return	8.2(a)
Senior Management of the Business	7.5(a)(ii)
Settlement	12.4(b)
Settlement Accountant	2.4(e)
Share Seller	Recitals
Shared Contracts	1.1
Shares	1.1
Skadden	3.1
Software	1.1
Specified Material Contracts	4.8(a)
Specified Matter	12.4(a)
Statement of Objections	2.4(d)
Straddle Period	1.1
Submission Deadline	2.4(a)

1.1

Subsidiary

Target Net Working Capital Lower Amount	1.1
Target Net Working Capital Upper Amount	1.1
Tax	1.1
Tax Authority	1.1
Tax Return	1.1
Third Party Proceeds	12.9
Third-Party Claim	12.4(a)
Trade Controls	1.1
Trade Materials	7.1(c
Trade Secrets	1.1
Trademarks	1.1
Transaction Documents	1.1
Transaction Tax Treatment	8.1(d
Transactions	1.1
Transfer Taxes	1.1
Transferred Employee	9.1
Transition Period	7.1(c
Transition Services Agreement	3.2(b)
Treasury Regulations	1.1
U.S. Foreign National Employees	9.6
Unaudited Financial Information	4.5(a)
Unpaid Income Taxes	1.1
Willful Breach	1 1

# TRANSACTION AGREEMENT

THIS TRANSACTION AGREEMENT is dated as of July 21, 2024, by and between:

- (A) Terex Corporation, a Delaware corporation ("Purchaser"); and
- **(B)** Dover Corporation, a Delaware corporation ("Seller").

#### RECITALS

WHEREAS, Seller holds, indirectly through Dover Climate & Sustainability Technologies Segment, Inc., a Delaware corporation and an indirect wholly-owned subsidiary of Seller (the "Share Seller"), all of the Shares (as defined below), which shares constitute all of the issued and outstanding capital stock of each of The Heil Co., a Delaware corporation ("Acquired Entity 1") and Marathon Equipment Company (Delaware), a Delaware corporation (together with Acquired Entity 1, the "Acquired Entities" and each, an "Acquired Entity"), as of the date of this Agreement;

WHEREAS, Seller owns, indirectly through IP Seller, the IP Seller IP;

WHEREAS, the Acquired Companies are engaged in the Business;

WHEREAS, Seller desires to cause (a) the Share Seller to sell, transfer, convey, assign and deliver to Purchaser, and Purchaser desires to purchase from the Share Seller, all of the Shares and (b) IP Seller to sell, transfer, convey, assign and deliver to Purchaser, and Purchaser desires to purchase from IP Seller, all of IP Seller's right, title and interest in and to the IP Seller IP, in each case, subject to the terms and the conditions set forth in this Agreement;

**WHEREAS**, the parties hereto desire that Purchaser and its Affiliates assume the Assumed Liabilities and that Seller and its Subsidiaries retain the Retained Liabilities, in each case, subject to the terms and conditions set forth in this Agreement;

**WHEREAS**, Seller will, and will cause its Affiliates to, and Purchaser will, and will cause its Affiliates to, at or prior to the Closing, execute and deliver each of the Transaction Documents (other than this Agreement) to which they are a party; and

WHEREAS, the parties hereto desire to make certain representations, warranties, covenants and agreements in connection with this Agreement.

**NOW, THEREFORE**, in consideration of the foregoing recitals and the mutual representations, warranties, covenants and promises contained herein and other good and valuable consideration, the adequacy and sufficiency of which are hereby acknowledged, and intending to be legally bound, the parties hereto agree as follows:

# ARTICLE I

# **DEFINITIONS**

- 1.1 <u>Definitions.</u> Capitalized terms used in this Agreement shall have the meanings set forth in this Agreement. The following terms, whenever used herein, shall have the following meanings for all purposes of this Agreement:
- "Accounting Principles" shall mean the accounting principles, policies, treatments, categorizations, practices, methods, bases and estimation techniques set forth on Schedule A attached hereto.
- "Acquired Companies" shall mean each of the Acquired Entities and their respective Subsidiaries, and "Acquired Company" shall mean any of them.
- "Acquired Company Benefit Plan" shall mean each Benefit Plan that is sponsored, maintained or contributed to solely by one or more Acquired Companies.
- "Acquired Company IP" shall mean any and all Intellectual Property that is (a) both (i) owned by the Acquired Companies as of the date of this Agreement and (ii) used in the conduct of the Business, and (b) the IP Seller IP, excluding, in each case of the foregoing clauses (a) and (b), any portion thereof that are Seller Group Marks.
- "<u>Acquired Company IT Assets</u>" shall mean all IT Assets used in the operation of the Business, to the extent owned by, or licensed or leased to, the Acquired Companies pursuant to a Contract entered into between an Acquired Company and a third party, including pursuant to outsourced or cloud computing arrangements.
- "Acquisition Proposal" shall mean, other than the transactions contemplated hereby, any inquiry, proposal or offer for, or any indication of interest in any direct or indirect (a) sale, lease, exchange, transfer, license or other disposition, in one or a series of related transactions, of all or a majority of the assets of the Business other than the sale of the Businesses' products or services in the ordinary course of business and otherwise in accordance with Section 6.1, or (b) acquisition, whether by share exchange, merger or in any other manner, by any Person or group of related Persons, in one or a series of related transactions, of any equity securities of any of the Acquired Companies.

"Affiliate" as to any Person, shall mean any other Person that, directly or indirectly, is in control of, is controlled by, or is under common control with, such Person, through one or more intermediaries or otherwise. For purposes of this definition, "control" of a Person shall mean the power, directly or indirectly, to direct or cause the direction of the management and policies of such Person, whether by ownership of equity interests, by contract or otherwise. The Acquired Companies shall be deemed, for purposes of this Agreement, Affiliates of Seller prior to the Closing and Affiliates of Purchaser from and after the Closing.

"Agreement" shall mean this Transaction Agreement (including the Seller Disclosure Schedule and all other schedules and exhibits attached hereto), as it may be amended, restated or otherwise modified from time to time.

"Antitrust Law" shall mean any Legal Requirements applicable to Purchaser, Seller or any Acquired Company under any applicable jurisdiction that are designed to prohibit, restrict or regulate actions having the purpose or effect of monopolization or restraint of trade.

"Assumed Liabilities" shall mean (a)(i) any and all Liabilities of the Acquired Companies and (ii) any and all Liabilities arising out of, resulting from or to the extent relating to the Business, including the IP Seller IP, and in the case of each of the forgoing clauses (i) and (ii), including, for the avoidance of doubt, the matters set forth on Schedule 1.1(a) of the Seller Disclosure Schedule and (b) any Taxes allocated to Purchaser pursuant to Section 8.2(c), in each of the foregoing clauses (a) and (b), regardless of (A) when or where such Liabilities arose or arise (whether arising prior to, at or after the Closing), (B) where or against whom such Liabilities are asserted or determined, (C) whether arising from or alleged to arise from negligence, gross negligence, recklessness, violation of Legal Requirement, fraud or misrepresentation by Seller and its Affiliates (including the Acquired Companies), as the case may be, or any of their past or present respective directors, officers, employees, agents, Subsidiaries or Affiliates and (D) which entity is named in any Proceeding associated with any Liability; provided that Assumed Liabilities shall not include any Liabilities (1) for Taxes for which any member of the Seller Group is responsible under Article VIII or Article XII or (2) arising out of or resulting from any Seller Benefit Plans.

"Bankruptcy and Equity Exception" shall mean the effect on enforceability of (a) any applicable Legal Requirement relating to bankruptcy, reorganization, insolvency, moratorium, fraudulent conveyance or preferential transfers, or similar Legal Requirement relating to or affecting creditors' rights generally, and (b) general principles of equity (regardless of whether enforceability is considered in a Proceeding in equity or at law).

"Benchmark Time" shall mean 12:01 a.m., Central Standard Time, on the Closing Date.

"Benefit Plan" shall mean each (a) "employee benefit plan" within the meaning of Section 3(3) of ERISA, whether or not subject to ERISA and (b) each employment, individual consulting, severance or similar contract, plan, agreement, arrangement or policy and each other benefit or compensation plan, policy, program, practice, agreement or arrangement (written or oral), including any equity, equity-based, phantom equity, stock option or other stock related rights, retirement, profit sharing, bonus, incentive, severance, post-retirement, separation, change in control, retention, employment, deferred compensation, fringe benefit, vacation, paid time off, medical, health, welfare, dental, life or disability plan, program, policy or arrangement; *provided* that "Benefit Plan" excludes any such plan, program or arrangement that is maintained, sponsored or otherwise required by a Governmental Authority to which any Person is required to contribute without discretion as to the level of benefits.

"Business" shall mean the business of researching, developing, designing, manufacturing, providing, promoting, marketing, selling, supplying and distributing refuse collection vehicles

(RCVs), stationary compaction and bailing equipment, parts (including after-market parts), associated components and services, onboard technology and software solutions used in the North American solid waste and recycling collection industry as conducted by Seller and its Affiliates (including the Acquired Companies) as of immediately prior to the Closing; *provided* that the "Business" shall exclude (a) general corporate functions provided by the Seller Group either directly or indirectly as of immediately prior to the Closing that constitute an Excluded Service (as defined in the Transition Services Agreement) or that are finance, accounting, tax, human resources, insurance and risk management, treasury, operational, environmental health and safety, legal, information technology, supply chain management shared services provided by the Seller Group (the functions referred to herein as the "Corporate Functions") and (b) the eCommerce Platform Software.

"Business Day" shall mean any day other than (a) a Saturday or a Sunday or (b) a day on which banking and savings and loan institutions are authorized or required to be closed in either (i) New York, New York or (ii) Chicago, Illinois.

"Business Employee" shall mean (i) each individual who is employed by an Acquired Company, including each such employee who is on a leave of absence (including medical leave, personal leave, military leave, workers' compensation leave, short-term disability and long-term disability) and paid or unpaid time off, and (ii) the Canadian Business Employee.

"Canadian Business Employee" shall mean the individual listed on Schedule 1.1(b) of the Seller Disclosure Schedule by employee identification number.

"CARES Act" shall mean the Coronavirus Aid, Relief, and Economic Security Act of 2020, as amended and any similar or successor Legal Requirement or any administrative or other guidance published with respect thereto by any Governmental Authority.

"Cash" shall mean, as of the time of determination, an amount equal to, without duplication, the aggregate amount of all cash and cash equivalents, investment accounts and marketable investments of the Acquired Companies (excluding Restricted Cash) plus (a) the aggregate amount of checks, wires, transfers and drafts for cash or cash equivalents (other than Restricted Cash) received or deposited for the account of the Acquired Companies, but only to the extent the associated receivable is not included as a current asset in the calculation of Net Working Capital, minus (b) the aggregate amount of uncleared checks, wires, transfers and drafts issued by the Acquired Companies, but only to the extent the associated payable is not included as a current liability in the calculation of Net Working Capital, in each case, determined in accordance with the Accounting Principles. "Restricted Cash" shall mean any (i) cash or cash equivalents held in escrow, security or similar deposits, (ii) customer deposits or other cash or cash equivalents held on behalf of third parties, (iii) cash or cash equivalents, investment accounts or marketable investments that are subject to any restriction or local exchange control, Tax or other requirements, to the extent that the amount of such cash or cash equivalents cannot, within ninety (90) days, be accessed, or be used by the Acquired Companies for the repayment of Indebtedness or (iv) any cash equivalents, investment accounts or marketable investments to the extent not convertible into cash within ninety (90) days.

"Closing Conditions" shall mean the conditions to the respective obligations of the parties hereto to consummate the Transactions, as set forth in Article X.

"Closing Transaction Expenses" shall mean, except as otherwise incurred by the Seller Group or the Acquired Companies in compliance with Section 6.7, the aggregate amount of all out-of-pocket fees and expenses (whether or not vet invoiced), incurred by the Seller Group or the Acquired Companies, or otherwise on behalf of the Acquired Companies, and, in each case, to be paid by the Acquired Companies or that constitutes an Assumed Liability to be paid by Purchaser, in connection with the sale process for the Business (or any portion thereof), or otherwise relating to the negotiation, preparation or execution of this Agreement or any documents or agreements contemplated hereby or the performance or consummation of the transactions contemplated hereby or thereby, in each case, solely to the extent unpaid as of the Benchmark Time and incurred by the Seller Group or the Acquired Companies, or otherwise on behalf of the Acquired Companies, and, in each case, payable by an Acquired Company or that constitutes an Assumed Liability to be paid by Purchaser, including: (a) fees and expenses of counsel, advisors and consultants, accountants and auditors and experts engaged in connection with the transactions contemplated by this Agreement or the sale process for the Business, in each case, to be paid by the Acquired Companies or that constitutes an Assumed Liability to be paid by Purchaser and (b) all transaction-related bonuses, severance payments and other similar payments in connection with the consummation of the transactions contemplated hereby (and including, in each case, the employer portion of any corresponding unemployment, social security, and payroll Taxes incurred under applicable Legal Requirements) (excluding (i) any severance payments arising as a result of the occurrence of one or more additional post-Closing events under so-called "double-trigger" severance provisions contained in any Acquired Company Benefit Plan, (ii) any payments or benefits made or provided pursuant to an arrangement implemented by, or severance payments or similar benefits that are triggered as a result of post-Closing actions taken by, Purchaser or any of its Affiliates (including, post-Closing, the Acquired Companies), and (iii) any retention bonuses or similar compensation that are payable based on continued employment post-Closing that were disclosed in accordance with Section 4.13(a) of this Agreement to Purchaser prior to the date hereof), in each case, to be paid by the Acquired Companies or that constitutes an Assumed Liability to be paid by Purchaser. It is agreed and understood that any item that is included in Closing Transaction Expenses shall not be included as a current liability for purposes of calculating Net Working Capital as of the Benchmark Time or as an item of Indebtedness as of the Benchmark Time.

"Code" shall mean the Internal Revenue Code of 1986.

"Collective Bargaining Agreement" shall mean any collective bargaining agreement or other labor-related agreement with any labor or trade union, works council, employee representative or association or other labor organization, representing Business Employees.

"Competing Business Activities" shall mean the research, development, design, manufacture, provision, marketing, promotion, sale, supply or distribution of any product, process or service that has the same or substantially similar purpose or use as, or is otherwise competitive with, a product, process or service researched, developed, designed, manufactured.

promoted, marketed, sold or offered for sale by the Business as of immediately prior to the Closing.

"Compliant" shall mean, as of any time of determination, with respect to the Required Information, that (a) such Required Information, taken as a whole, at such time does not contain any untrue statement of a material fact with respect to the Acquired Companies or omit to state any material fact with respect to the Acquired Companies required to be stated therein or necessary in order to make the Required Information, in the light of the circumstances under which the statements contained therein are made, not misleading with respect to the Acquired Companies, (b) the independent auditors of the Acquired Companies have not objected to the use of, withdrawn or otherwise modified any audit opinion with respect to the financial statements contained in the Required Information and have confirmed they are prepared to issue a customary comfort letter, including as to customary negative assurances and change period, in order to consummate any Debt Financing (subject to their completion of customary procedures) on any day during the applicable Marketing Period, (c) with respect to any interim financial statements, such interim financial statements have been SAS 100 reviewed by the independent auditors of the Acquired Companies, (d) such Required Information complies in all material respects with all applicable requirements of Regulation S-K and Regulation S-X under the Securities Act for a registered public offering of securities on Form S-1 (other than financial statements and other information required by Rule 3-09, 3-10, 3-16, 13-01 or 13-02 of Regulation S-X (and in each case any successor thereto), the compensation discussion and analysis or other information required by Item 402 of Regulation S-K or the executive compensation and related person disclosure rules related to SEC Release Nos. 33-8732A, 34-54302A and IC-27444A, and other information or financial data customarily excluded from a Rule 144A offering memorandum), and (e) the financial statements and other financial information included in such Required Information would not be deemed stale or otherwise be unusable under customary practices for offerings and private placements of debt securities under Rule 144A of the Securities Act and are, and remain throughout the applicable Marketing Period, sufficient to permit the Acquired Companies' independent accountants to issue comfort letters, including as to customary negative assurances and change period, in order to consummate any offering of debt securities on any day during the applicable Marketing Period, which such accountants have confirmed they are prepared to issue (it being understood and agreed by the parties hereto that any financial statements required pursuant to the terms hereof shall be in substantially the same scope and level of detail as the Required Information, except that only annual financial statements will be subject to audit procedures, with any interim periods subject to SAS 100 review).

"Condition Satisfaction Date" shall mean the date on which the Closing Conditions, other than those conditions which, by their terms or nature, are to be satisfied at the Closing, are satisfied or, to the extent permitted by applicable Legal Requirements, so waived in writing.

"Confidentiality Agreement" shall mean the confidentiality agreement between Purchaser and Seller, dated December 8, 2023.

"Consent" shall mean any consent, approval or authorization.

"Consolidated Return" shall mean any consolidated, combined, unitary or similar Tax Return that includes Seller or any of its Affiliates (other than the Acquired Companies), on the one hand, and any Acquired Company, on the other hand.

"Contagion Event" shall mean the outbreak and ongoing effects of a contagious disease, epidemic or pandemic (including COVID-19).

"Contract" shall mean any agreement, contract, instrument, obligation, promise, understanding, arrangement, commitment or undertaking of any nature that is legally binding on any Person or entity or any part of its property under applicable Legal Requirements, other than a Permit.

"COVID-19" shall mean SARS-CoV-2 or COVID-19 (and all related mutations, variations, strains and sequences), and any future intensification, resurgence, variants, evolutions or mutations thereof or related or associated health conditions, epidemics, pandemics, disease outbreaks or public health emergencies.

"<u>Debt Financing</u>" shall mean the debt financing contemplated by the Debt Commitment Documents, including any offering or private placement of debt securities pursuant to the capital markets debt financing engagement referred to therein, for the purpose of financing the Transactions and related fees and expenses contemplated by this Agreement.

"eCommerce Platform Software" shall mean the proprietary software (including the object code and source code) that is owned by Seller or any of its Affiliates (other than the Acquired Companies) that is related to the eCommerce platform made available to end users of any of the Acquired Companies, which shall include, the business solutions known as Parts Central, Partable and any aftermarket parts punch-out catalog. "eCommerce Platform Software" shall not mean or include any historical or future data or meta data related to the Business, Acquired Companies or their respective end users, including, but not limited to, data pertaining to parts, weights, dimensions, or photographs.

"Employment Commencement Date" shall mean the Closing Date.

"Encumbrance" shall mean any lien, pledge, claim, hypothecation, charge, mortgage, deed of trust, easement, servitude, purchase agreement, option, security interest or similar encumbrance.

"Environmental Law" shall mean any applicable Legal Requirement relating to pollution, protection of the environment or natural resources, or human health or safety (to the extent relating to Hazardous Materials), including the use, handling, transportation, treatment, storage, disposal, Release or threat of Release, distribution, sale, marketing, or labeling of, or exposure to, Hazardous Materials.

"ERISA" shall mean the Employee Retirement Income Security Act of 1974.

"<u>Financing Source</u>" shall mean the Persons that have committed to provide or arrange the Debt Financing, including the parties who have committed to provide Debt Financing pursuant to the Debt Commitment Documents (such persons, the "<u>Financing Entities</u>"), and, in each case, their respective former, current and future Representatives and other Affiliates; *provided* that neither Purchaser nor any Affiliate thereof shall be a Financing Source.

"Former Business Employee" shall mean each individual who was employed by an Acquired Company whose employment terminated for any reason prior to the Closing.

"Fraud" shall mean, with respect to any party hereto, an actual and intentional act of common law fraud by such party in the making of a representation or warranty expressly stated in this Agreement; which satisfies each of the following conditions: (a) such representation or warranty was materially false or materially inaccurate at the time such representation or warranty was made; (b) the party making such representation or warranty had actual knowledge (and not imputed or constructive knowledge), without any duty of inquiry or investigation, that such representation or warranty was materially false or materially inaccurate when made; (c) such party had the specific intent to deceive another party hereto and induce such other party to enter into this Agreement; and (d) such other party reasonably relied on such false or inaccurate representation or warranty in entering into this Agreement, the other Transaction Documents or the certificates required to be delivered hereunder. "Fraud" shall not include any cause of action under law or equity, including for fraud, based on constructive or imputed knowledge, negligence or recklessness, and only the Persons who committed Fraud shall be responsible for such Fraud and only to the party established to have suffered Loss as a proximate result of such Fraud.

"GAAP" shall mean the generally accepted accounting principles in the United States in effect from time to time, consistently applied.

"Government Contract" shall mean any Contract, including purchase, task, or delivery orders, between any Acquired Company or, to the extent related to the Business, Seller or any of its Subsidiaries, on the one hand, and (a) any Governmental Authority and (b) any prime contractor of any Governmental Authority in its capacity as a prime contractor.

"Governmental Authority" shall mean any United States federal, state or local, or any supra-national or other non-United States government, political subdivision, governmental, regulatory or administrative authority, instrumentality, service cooperative, agency, body or commission, self-regulatory organization or any court, tribunal or judicial or arbitral body (public or private), in each case, exercising executive, legislative, judicial, regulatory, taxing or administrative functions.

"<u>Hazardous Materials</u>" shall mean (a) petroleum, petroleum products, by-products or breakdown products, radioactive materials, asbestos and asbestos-containing materials, lead and lead-based paint, per-and polyfluoroalkyl substances, toxic mold in quantities or concentrations that exceed applicable health standards or guidelines, radon, or polychlorinated biphenyls and (b) any chemical, material or substance defined or regulated as hazardous, toxic, a pollutant or a

contaminant, or words of similar meaning or regulatory effect or for which liability or standards of conduct may be imposed under any applicable Environmental Law.

"HSR Act" shall mean the Hart-Scott-Rodino Antitrust Improvements Act of 1976.

"Income Tax" shall mean any Taxes imposed on or determined with reference to gross or net income, profits or receipts (including any franchise or withholding Taxes imposed in lieu thereof).

"Indebtedness" shall mean, without duplication with respect to the Acquired Companies (or Assumed Liabilities, but excluding Retained Liabilities): (a) the unpaid principal amount of and accrued interest in respect of indebtedness for borrowed money, whether secured or unsecured, (b) all obligations that are evidenced by a note, bond, draft, mortgage, debenture or similar debt instrument or security, (c) all obligations under or pursuant to leases which are capitalized or in accordance with GAAP are required to be capitalized (which, for the avoidance of doubt, excludes any obligations under any operating leases that may be required to be recorded on a balance sheet pursuant to Accounting Standards Codification Topic 840 or 842, including those items set forth on Schedule 1.1(c) of the Seller Disclosure Schedule), (d) all obligations in respect of letters of credit, bankers' acceptances, surety and performance bonds and guarantees, and similar obligations, in each case, solely to the extent drawn or called, (e) all indebtedness secured by a purchase money mortgage or other Encumbrance on assets of the Acquired Companies to secure all or part of the purchase price of property subject to such Encumbrance. (f) all obligations created or arising under any conditional sale or other title retention agreement with respect to property acquired, (g) all obligations to purchase, redeem, retire, defease or otherwise acquire for value any debt or equity securities, (h) all obligations under derivative financial instruments, including hedges, currency and interest rate swaps and similar instruments and arrangements, (i) any Unpaid Income Taxes, (j) any payment obligation with respect to the termination of the Intercompany Agreements and other intercompany accounts or obligations in accordance with Section 6.4(a), (k) any accrued or earned but unpaid severance, restructuring costs, pension. unfunded or underfunded pension obligations, or any other similar obligations (and any employer portion of unemployment, social security, payroll or similar Taxes payable in connection therewith) in respect of any Business Employees, Former Business Employees or current or former officers, employees, directors, consultants or other service providers of the Business or any Acquired Company, (1) any guaranty (or, pursuant to any Contract or other arrangement, that has the economic effect of a guaranty) by any Acquired Company of obligations of the types described in the foregoing clauses (a) through (k), (m) all accrued and unpaid interest, together with prepayment premiums or penalties or accrued interest, prepayment penalties, makewhole payments and termination or breakage costs or other exit fees with respect to any indebtedness of a type described in the foregoing clauses (a) through (l); provided, however, that, in no event will Indebtedness include: (i) any indebtedness incurred by any Acquired Company, on the one hand, that is owed to any other Acquired Company, on the other hand, (ii) any amounts under existing letters of credit, lines of credit and revolving credit facilities to the extent not drawn or called, or (iii) any amount that is included as a Closing Transaction Expense. Indebtedness shall not include any fees or expenses of, or expenses initiated at the written request of, Purchaser or any of its Affiliates and not otherwise explicitly

required to be incurred by Seller or any of its Affiliates under the Agreement, whether related to the Debt Financing, the Transactions or otherwise. It is agreed and understood that any item that is included in Indebtedness as of the Benchmark Time shall not be included as a current liability for purposes of calculating Net Working Capital as of the Benchmark Time.

"Indemnified Party" shall mean a party entitled to seek indemnification under Article XII.

"Indemnifying Party" shall mean a party hereto from whom indemnification or recovery is sought under Article XII.

"Initial Marketing Period" shall mean the first period of 16 consecutive Business Days after the date on which Seller shall have delivered to Purchaser the Required Information and throughout and at the end of which the Required Information delivered to Purchaser prior to the beginning of such period remains Compliant; provided that (x) the Initial Marketing Period shall either be completed on or prior to August 16, 2024 or commence no earlier than September 3, 2024, (v) the days from November 27, 2024 through and including November 29, 2024 shall not be included when counting such 16 consecutive Business Day period (and the Initial Marketing Period need not be consecutive to the extent it would have otherwise included any of those days) and (z) the Initial Marketing Period shall either be completed on or prior to December 20, 2024 or commence no earlier than January 6, 2025. Notwithstanding anything in the preceding sentence of this definition to the contrary, the Initial Marketing Period shall not commence or be deemed to have commenced if, after the date hereof and prior to the completion of such 16 consecutive Business Day period: (i) (A) Seller or any of its Affiliates has an intention to restate in any material respect any of the financial statements contained in the Required Information, or (B) any such restatement is under active consideration, in which case Seller shall promptly disclose such intention or reconsideration and the material facts and circumstances relating thereto to Purchaser and the Business's independent auditor and the Initial Marketing Period shall not commence unless and until such restatement has been completed, the applicable Required Information has been amended and, to the extent such financial statements had previously been audited, an "unqualified" audit opinion is issued with respect to such restated financial statements, or Seller has certified in writing to Purchaser and the Business's independent auditor that no restatement is required; or (ii) the Business's independent auditor shall have withdrawn any audit opinion with respect to any of the financial statements contained in the Required Information, in which case the Initial Marketing Period shall not be deemed to commence unless and until a new "unqualified" audit opinion is issued with respect to such financial statements or any restatement thereof. Notwithstanding anything in this definition to the contrary, the Initial Marketing Period shall be deemed to have ended on the date on which the Debt Financing has been consummated. If at any time Seller shall reasonably and in good faith believe that it has provided the Required Information, it may deliver to Purchaser a written notice to that effect (stating when it believes it completed such delivery), in which case, the requirement to deliver the Required Information will be deemed to have been satisfied as of the date of such notice, unless Purchaser in good faith reasonably believes that Seller has not completed the delivery of the Required Information on such date and, within three (3) Business Days after the date of delivery of such notice, delivers a written notice to Seller to that effect (stating with specificity which Required Information Seller has not delivered) and, following delivery of such

Required Information specified in such notice, the Initial Marketing Period will commence; *provided* that such written notice from Purchaser to Seller will not prejudice Seller's right to assert that the Required Information was, in fact, delivered.

"Intellectual Property" shall mean all: (a) patents, patent applications and all related divisionals, continuations, continuations-in-part, reissues, extensions, substitutions and reexaminations ("Patents"), (b) trade secrets and other confidential and proprietary know-how, information, ideas, inventions, processes, formulae, models and methodologies ("Trade Secrets"), (c) trademarks, service marks, trade names, Internet domain names, logos, trade dress, and other similar designations of source or origin, and all registrations and applications for registration of the foregoing, together with the goodwill symbolized by and arising out of the use of any of the foregoing ("Trademarks"), (d) copyrights and copyrightable subject matter (whether registered or unregistered), all registrations and applications for registration of such copyrights, and all issuances, extensions and renewals of such registrations and applications ("Copyrights"), (e) rights in Software and (f) any and all other proprietary rights.

"Investment Screening Laws" shall mean any applicable United States or non-United States Legal Requirement intended to screen, prohibit or regulate foreign investments on public interest or national security grounds.

"IP Contract" shall mean any (a) Contract pursuant to which any Acquired Company or IP Seller has granted to any Person a license to or right to use any Acquired Company IP, other than (i) licenses of Acquired Company IP granted by an Acquired Company or IP Seller on a non-exclusive basis to a customer in connection with business or sales Contracts entered into in the ordinary course of business or (ii) other Contracts in which grants of rights to use Intellectual Property are non-exclusive, incidental and not material to the performance under such Contract; (b) Contract pursuant to which any Acquired Company has been granted a license to or right to use any Intellectual Property, other than (i) licenses of commercially available off-the-shelf software that have not been modified or customized for the Acquired Companies in any material respect and is licensed to an Acquired Company for a one-time or annual fee of five-hundred thousand dollars (\$500,000) or less and (ii) other Contracts in which grants of rights to use Intellectual Property are non-exclusive, incidental and not material to the performance under such Contract and (c) consent, settlement and coexistence agreements and Orders governing the use, validity or enforceability of Acquired Company IP.

"IP Seller" shall mean Delaware Capital Formation, Inc., a Delaware corporation and an indirect wholly-owned subsidiary of Seller.

"IP Seller IP" shall mean the Intellectual Property set forth on Exhibit A.

"IRS" shall mean the United States Internal Revenue Service.

"IT Assets" shall mean all computers, computer software, systems, websites, hardware, firmware, middleware, networks, servers, workstations, routers, hubs, switches, data communications lines and all other information technology equipment, and all associated documentation.

"Key Customer" shall mean each of the top twenty (20) customers of the Business who receive goods or services of the Business through Contracts directly with Seller or any of its Subsidiaries, determined on the basis of aggregate revenue of the Business for the twelve (12) month period ended December 31, 2023, and each of the top twenty (20) customers of the Business who receive goods or services of the Business through Contracts directly with Seller or any of its Affiliates, determined on the basis of aggregate revenue of the Business for the five (5) month period ended May 31, 2024, in each case as set forth on Schedule 1.1(d) of the Seller Disclosure Schedule.

"Key Supplier" shall mean each of the top twenty (20) suppliers or vendors of the Business who provide goods or services to the Business through Contracts directly with Seller or any of its Subsidiaries, determined on the basis of the aggregate spend of the Business for the twelve (12) month period ended December 31, 2023, and each of the top twenty (20) suppliers or vendors of the Business who provide goods or services to the Business through Contracts directly with Seller or any of its Affiliates, determined on the basis of the aggregate spend of the Business for the five (5) month period ended May 31, 2024, in each case as set forth on Schedule 1.1(e) of the Seller Disclosure Schedule.

"<u>Leased Real Property</u>" shall mean all real property leased, licensed or subleased by any Acquired Company pursuant to the Real Property Leases, together with all rights and interests (including if applicable easements) of the lessee appurtenant thereto.

"<u>Legal Requirement</u>" shall mean any statute, law (including international conventions, protocols and treaties), ordinance, regulation, rule, code, Order or enforceable or binding requirement of any Governmental Authority or any other requirement or rule of law (including common law) promulgated by a Governmental Authority.

"<u>Liabilities</u>" shall mean any and all Indebtedness, liabilities, guarantees, assurances, commitments and obligations of any kind or nature, whether fixed, contingent or absolute, asserted or unasserted, matured or unmatured, liquidated or unliquidated, accrued or not accrued, known or unknown, due or to become due, reflected on a balance sheet or otherwise, whenever or however arising, including, but not limited to, those arising under any Legal Requirement or judgment, Contract or tort based on negligence or strict liability.

"Losses" shall mean all losses, damages, judgments, costs, expenses and Liabilities actually suffered or incurred (including reasonable attorneys' fees).

"Malicious Code" shall mean any (i) virus, malware, ransomware, Trojan horse, worm, back door, time bomb, drop dead device, spyware, trackware or adware and (ii) similar program, routine, instruction, device, code, contaminant, logic or effect designed or intended to disable, disrupt, erase, harm, or otherwise impede the operation of, or enable any Person to access without authorization, or otherwise materially and adversely affect the functionality of, any IT Asset (or portion thereof).

"Marketing Periods" shall mean each of the Initial Marketing Period and Secondary Marketing Period.

"Material Adverse Effect" shall mean any event, change, development or effect that, individually or in the aggregate, has had or would be reasonably expected to have a material adverse effect on the business, financial condition, assets and liabilities, operations or results of operations of the Business or the Acquired Companies, taken as a whole; provided that, no event, change, development or effect resulting or arising from or in connection with any of the following matters shall be deemed, either alone or in combination, to constitute or contribute to, or be taken into account in determining whether there has been or would reasonably be expected to be a Material Adverse Effect: (a) any national, international, foreign, domestic or regional economic, financial, social or political conditions (including changes therein), including (i) hostilities, acts of war, protests, riots, unrest, sabotage, terrorism, or military actions or any escalation or worsening of any of the same, (ii) changes in any financial, debt, credit, capital or banking markets or conditions (including the prices or availability of commodities, raw materials or energy supply used in the Business, any increase in operating costs or capital expenses, inflation or any changes in the rate, increase or decrease of inflation or any disruption thereof), and (iii) changes in interest, currency or exchange rates or tariffs or any trade wars, (b) any act of God, hurricane, flood, tornado, fire, explosion, weather event, earthquake, landslide, other natural disaster, any Contagion Event or other outbreak of illness or public health event (whether human or animal), (c) changes in legal or regulatory conditions, including changes in Legal Requirements (or standards, interpretations or enforcement thereof and including guidelines and directives of industry groups), whether or not related to a Contagion Event or other public health emergency, (d) changes in GAAP or other accounting practices, policies or requirements, or standards, interpretations, including in the interpretation or enforcement thereof, (e) changes in the industries or markets in which the Business operates, including competition in any of the geographic areas in which the Business operates, (f) the failure of the Business to meet any internal or published projections, estimates or forecasts of revenues, goals, earnings or other measures of financial or operating performance for any period (it being understood and agreed that any underlying facts giving rise or contributing to such failure that are not otherwise expressly excluded from this definition of "Material Adverse Effect" may be taken into account in determining whether a Material Adverse Effect has occurred or could reasonably be expected to occur), (g) changes in credit ratings or changes in the trading price or volume of Seller's common stock (it being understood and agreed that any underlying facts giving rise or contributing to such changes that are not otherwise expressly excluded from this definition of "Material Adverse Effect" may be taken into account in determining whether a Material Adverse Effect has occurred or could reasonably be expected to occur). (h) any effect resulting from the negotiation, execution, pendency, announcement or consummation of the Transactions or compliance with the terms of any Transaction Document (provided that the exception in this clause (h) shall not apply with respect to representations and warranties in Section 4.2 and, to the extent related to such representation and warranty, the condition set forth in Section 10.1(a)), (i) any actions taken to which Purchaser has consented in writing or that are taken after the date hereof at the written request of Purchaser, (j) the effect of any action taken or omission to act by Purchaser with respect to the Business, the Acquired Companies or the Transactions, including any communication or disclosure by Purchaser or any of its Affiliates of its plans or intentions with respect to the Business or the Acquired Companies, including Losses or threatened Losses of, or any adverse change in the relationship with, employees, customers, suppliers, vendors, resellers, distributors, financing sources, licensors, licensees, Governmental Authorities or others

having relationships with the Business, or (k) the effect of any event or action taken or omission to act by Seller or its Affiliates with the express written consent (or at the express written request) of Purchaser; *provided* that to the extent that any event, change, development or effect in the foregoing clauses (a), (b), (c), (d) or (e) has a greater adverse impact on the Business, taken as a whole, as compared to the adverse impact such event, change, development or effect has on other Persons operating in the same industries as the Business operates, then the incremental effect of such event, change, development or effect shall be taken into account in determining whether a Material Adverse Effect has occurred or would reasonably be expected to occur.

"Net Working Capital" shall have the meaning set forth on Schedule A.

"Net Working Capital Overage" shall mean the amount, if any, by which (a) the Net Working Capital as of the Benchmark Time is greater than (b) the Target Net Working Capital Upper Amount.

"Net Working Capital Underage" shall mean the amount, if any, by which (a) the Net Working Capital as of the Benchmark Time is less than (b) the Target Net Working Capital Lower Amount.

"Open Source Software" shall mean any Software that is, contains or is derived from Software distributed as freeware, shareware or open source Software, or under similar licensing or distribution models that (a) require the licensing, disclosure or distribution of source code or any other Intellectual Property to any other Person, (b) prohibit or limit the receipt of consideration in connection with licensing or distributing any Software or other Intellectual Property, (c) allow any Person to decompile, disassemble or reverse engineer any Software or (d) otherwise are identified as open source licensing or distribution models by the Open Source Initiative at www.opensource.org or the Free Software Foundation at www.fsf.org.

"Order" shall mean any order, writ, judgment, injunction (whether temporary, preliminary, or permanent), temporary restraining order, decree, stipulation, decision, ruling, determination or award entered by or with any Governmental Authority.

"<u>Permit</u>" shall mean any permit, license, authority, registration, concession, grant, franchise, certificate, accreditation, clearance, consent, approval, identification numbers exemption, waiver, filing or other authorization issued or required by any Governmental Authority under any applicable Legal Requirement.

"<u>Permitted Encumbrances</u>" shall mean (a) Encumbrances for Taxes, assessments or other governmental charges or levies not yet delinquent or the amount or validity of which is being contested in good faith by appropriate Proceedings and, in each case for which adequate accruals or reserves have been established on the books and records of the Acquired Companies in accordance with GAAP, (b) Encumbrances of carriers, warehousemen, mechanics or suppliers, materialmen, workmen, repairmen and other similar Encumbrances imposed or permitted by Legal Requirements arising or incurred in the ordinary course of business for amounts which are not yet delinquent or the amount or validity of which is being contested in good faith by appropriate Proceedings, (c) Encumbrances incurred or deposits made in the ordinary course of

business in connection with workers' compensation, unemployment insurance or other types of social security, (d) Encumbrances, defects or imperfections of title, and other similar matters, in each case affecting title to the Real Property that do not impose material monetary obligations on the Acquired Companies and do not, and would not reasonably be expected to, individually or in the aggregate, materially impair the value of the assets to which they apply, or materially and adversely affect. or materially disrupt, the ordinary course operation of the Business, (e) zoning, entitlement, building and other generally applicable land use restrictions (including those related to the use or management of environmental resources) by a Governmental Authority (provided that such liens and restrictions are not violated in a manner that would materially and adversely impair the Company's current business operations at such location), (f) Encumbrances imposed on the underlying fee interest of any Leased Real Property not created or permitted by Seller or its Affiliates (unless permission is required under the terms of the applicable lease), (g) leases, subleases, licenses, sublicenses or other agreements (including proposed leases and renewals) for the use and/or occupancy of the Real Property, and any Encumbrances arising thereunder or created thereby, (h) Encumbrances incurred in the ordinary course of business securing liabilities that are not material to the Business, taken as a whole, (i) Encumbrances arising out of, relating to or resulting from any Transaction Document, (j) rights, terms or conditions of any Real Property Lease, (k) licenses or sublicenses of Intellectual Property granted by the Acquired Companies in the ordinary course of business, (1) Encumbrances on securities created under federal, state or foreign securities Legal Requirements, (m) deposits to secure the performance of bids, trade Contracts, leases, statutory obligations, surety and appeal bonds, performance bonds and other obligations of a like nature, in each case in the ordinary course of business, (n) statutory Encumbrances of lessors or licensors imposed by Legal Requirements as against tenant's fixtures arising or incurred in the ordinary course of business for amounts which are not yet delinquent or the amount or validity of which is being contested in good faith by appropriate Proceedings and for which adequate accruals or reserves have been established on the books and records of the Acquired Companies in accordance with GAAP, (o) Encumbrances that will be released at or prior to the Closing, (p) deed restrictions limiting the use of Real Property to the extent the same are not and would not in the future be violated by the current use of such Real Property; (q) Encumbrances created or permitted by Purchaser or any of its Affiliates in writing, (r) any set of facts that would be disclosed by an accurate and up-to-date title report, inspection or survey of the Real Property; provided that such facts do not materially interfere with the current use and operation or materially impair the value of any parcel of Real Property, individually or in the aggregate and (t) those Encumbrances set forth on Schedule 1.1(f) of the Seller Disclosure Schedule.

"Person" shall mean any individual, corporation (including any non-profit corporation), general partnership, limited partnership, limited liability partnership, joint venture, estate, trust or company (including any limited liability company or joint stock company) or other similar entity, including a Governmental Authority.

"<u>Personal Information</u>" shall mean means any information that either directly or indirectly identifies or, alone or in combination with any other information, could reasonably be used to identify, locate, or contact a natural Person, including name, street address, telephone number, email address, identification number issued by a Governmental Authority, payment card

number, bank information, customer or account number, online identifier, device identifier, IP address, browsing history, search history, or other website, application, or online activity or usage data, location data, biometric data, medical or health information, or any other information that is considered "personally identifiable information," "personal information," or "personal data" or any similar term under applicable Legal Requirement or in an Acquired Company's privacy policies, notices, or the contracts to which it is subject.

"Post-Closing Straddle Period" shall mean the portion of any Straddle Period beginning after the Closing Date.

"<u>Post-Closing Tax Period</u>" shall mean any taxable period beginning after the Closing Date and any Post-Closing Straddle Period.

"Pre-Closing Straddle Period" shall mean the portion of any Straddle Period ending on the Closing Date.

"Pre-Closing Tax Period" shall mean any taxable period ending on or before the Closing Date and any Pre-Closing Straddle Period.

"Privacy Laws" shall mean all applicable Legal Requirements or any Orders concerning the privacy, security, or Processing of Personal Information (including Legal Requirements of jurisdictions where Personal Information was collected), including, as applicable, Legal Requirements concerning data breach notification; consumer protection requirements for website and mobile application privacy policies and practices; social security number protection; and data security. Without limiting the foregoing, Privacy Laws include: the Federal Trade Commission Act, the Telephone Consumer Protection Act, the Telemarketing and Consumer Fraud and Abuse Prevention Act, the Controlling the Assault of Non-Solicited Pornography and Marketing Act of 2003, the Children's Online Privacy Protection Act, the California Consumer Privacy Act of 2018, as amended by the California Privacy Rights Act of 2020, the California Online Privacy Protection Act, the California Invasion of Privacy Act, the Virginia Consumer Data Protection Act, the Colorado Privacy Act, the Connecticut Data Privacy Act, the Video Privacy Protection Act, the Computer Fraud and Abuse Act, the Electronic Communications Privacy Act, the General Data Protection Regulation (Regulation (EU) 2016/679), the UK General Data Protection Regulation and the Data Protection Act 2018, the Privacy and Electronic Communications Directive 2002/58/EC on Privacy and Electronic Communications as amended by Directive 2009/136/EC (and any member state laws and regulations implementing it), the Privacy and Electronic Communications Regulation, the Personal Information Protection and Electronic Documents Act, and Canada's Anti-Spam Legislation (SC 2010, c 23), and all other similar international, federal, state, provincial, and local Legal Requirements.

"Proceeding" shall mean any audit, action, claim, suit, litigation, lawsuit, arbitration, complaint, investigation or proceeding (whether civil, criminal or administrative) by or before any Governmental Authority.

"<u>Processing</u>" shall mean any operation or set of operations which is performed on Personal Information or on sets of Personal Information, whether or not by automated means,

such as the receipt, access, acquisition, collection, recording, organization, compilation, structuring, storage, adaptation or alteration, retrieval, consultation, use, disclosure by transfer, transmission, dissemination or otherwise making available, alignment or combination, restriction, disposal, erasure or destruction of such Personal Information or sets of Personal Information.

"<u>Purchaser Disclosure Schedule</u>" shall mean the disclosure schedules dated as of the date of this Agreement and delivered by Purchaser to Seller in connection with the execution of this Agreement.

"<u>Purchaser Fundamental Representations</u>" shall mean the representations and warranties of Purchaser set forth in <u>Section 5.1</u> (*Authority; Enforceability*), <u>Section 5.3</u> (*Organization*), <u>Section 5.7</u> (*Solvency*) and <u>Section 5.8</u> (*Brokers*).

"<u>Purchaser Material Adverse Effect</u>" shall mean any event, change, development or effect that is or would reasonably be expected to be, individually or in the aggregate, materially adverse to the ability of Purchaser to timely perform its obligations under this Agreement or would reasonably be expected to, individually or in the aggregate, prevent, impede, impair, interfere with, hinder, delay, or adversely affect, in any material respect, the consummation by Purchaser of the Transactions.

"<u>Purchaser's Knowledge</u>" and similar phrases shall mean the knowledge of the individuals set forth on <u>Schedule 1.1(a)</u> of the Purchaser Disclosure Schedule after due inquiry of each such individual of such individual's direct reports.

"Real Property" shall mean the Leased Real Property and the Owned Real Property.

"Release" shall mean any releasing, spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, leaching, dumping, disposing, dispersing, or migrating in, into, onto or through the indoor or outdoor environment.

"Representatives" shall mean officers, directors, employees, principals, partners, members, managers, attorneys, accountants, agents, consultants, financial or other advisors, authorized representatives and Affiliates.

"Required Information" shall mean, collectively, (a) the Audited Financial Information, (b) the unaudited consolidated balance sheets and the related unaudited consolidated interim statements of income, stockholders' equity and cash flows of the Business for each subsequent interim financial period (other than any fourth fiscal quarter) ended more than 45 days prior to the Closing Date (and the corresponding period of the prior fiscal year), in each case, prepared in accordance with U.S. GAAP and (c) such other customary information regarding the Business that is reasonably necessary to complete the Debt Financing and of the type and form customarily included in a confidential offering memorandum for private placements of non-convertible high-yield bonds pursuant to Rule 144A promulgated under the Securities Act (including information necessary for Purchaser's preparation of customary pro forma financial

statements to be included therein), consistent with the information required by paragraph 12 of Exhibit D of the Debt Commitment Letter.

"Restricted Person" shall mean any Person identified on the U.S. Department of Commerce's Entity List, Denied Persons List, Unverified List or Military End User List, or the U.S. Department of State's Debarred List.

"Retained Liabilities" shall mean any and all Liabilities (other than the Assumed Liabilities) of any member of the Seller Group, whether arising prior to, on or after the Closing Date.

"Sanctioned Jurisdiction" shall mean a country or territory that is itself the subject or target of any comprehensive Sanctions (at the time of this Agreement, Cuba, Iran, North Korea, Syria, and the Crimea, the so-called Luhansk People's Republic, and the so-called Donetsk People's Republic regions of Ukraine, and the non-government-controlled areas of Ukraine in the oblasts of Kherson and Zaporizhzhia).

"Sanctioned Person" shall mean any Person that is the subject of Sanctions, including as a result of being (a) listed in any list of sanctioned or designated Persons under Sanctions; (b) located, organized, or resident in a Sanctioned Jurisdiction; or (c) directly or indirectly owned fifty percent (50%) or more or controlled, individually or in the aggregate, by one or more Persons described in the foregoing clauses (a) or (b).

"Sanctions" shall mean any economic or financial sanctions and trade embargoes, including those administered by the United States (including through U.S. Department of Treasury's Office of Foreign Assets Control and the U.S. Department of State), United Nations, the European Union or any member state, or the United Kingdom.

"SEC" shall mean the Securities and Exchange Commission.

"Secondary Marketing Period" shall mean, solely to the extent the proceeds of the Debt Financing in an amount equal to at least the Required Amount have not been obtained prior to the satisfaction of the conditions set forth in Article X (other than those conditions that by their nature are to be satisfied at the Closing), the first period of 16 consecutive Business Days after the date on which (a) Seller shall have delivered to Purchaser the Required Information and throughout and at the end of which the Required Information delivered to Purchaser prior to the beginning of such period remains Compliant and (b) the conditions set forth in Article X shall have been satisfied (other than those conditions that by their nature are to be satisfied at the Closing) as of (and throughout) the beginning of such period and nothing has occurred and no condition or state of facts exists that would cause any of the conditions set forth in Article X to fail to be satisfied assuming the Closing were to be scheduled for any time during such 16 consecutive Business Day period; provided that (x) the Secondary Marketing Period shall either be completed on or prior to August 16, 2024 or commence no earlier than September 3, 2024, (y) the days from November 27, 2024 through and including November 29, 2024 shall not be included when counting such 16 consecutive Business Day period (and the Secondary Marketing Period need not be consecutive to the extent it would have otherwise included any of those days)

and (z) the Secondary Marketing Period shall either be completed on or prior to December 20, 2024 or commence no earlier than January 6, 2025. Notwithstanding anything in the preceding sentence of this definition to the contrary, the Secondary Marketing Period shall not commence or be deemed to have commenced if, after the date hereof and prior to the completion of such 16 consecutive Business Day period: (i) (A) Seller or any of its Affiliates has an intention to restate in any material respect any of the financial statements contained in the Required Information, or (B) any such restatement is under active consideration, in which case Seller shall promptly disclose such intention or reconsideration and the material facts and circumstances relating thereto to Purchaser and the Business's independent auditor and the Secondary Marketing Period shall not commence unless and until such restatement has been completed, the applicable Required Information has been amended and, to the extent such financial statements had previously been audited, an "unqualified" audit opinion is issued with respect to such restated financial statements, or Seller has certified in writing to Purchaser and the Business's independent auditor that no restatement is required; or (ii) the Business's independent auditor shall have withdrawn any audit opinion with respect to any of the financial statements contained in the Required Information, in which case the Secondary Marketing Period shall not be deemed to commence unless and until a new "unqualified" audit opinion is issued with respect to such financial statements or any restatement thereof. Notwithstanding anything in this definition to the contrary, the Secondary Marketing Period shall be deemed to have ended on the date on which the Debt Financing has been consummated. If at any time Seller shall reasonably and in good faith believe that it has provided the Required Information, it may deliver to Purchaser a written notice to that effect (stating when it believes it completed such delivery), in which case, the requirement to deliver the Required Information will be deemed to have been satisfied as of the date of such notice, unless Purchaser in good faith reasonably believes that Seller has not completed the delivery of the Required Information on such date and, within three (3) Business Days after the date of delivery of such notice, delivers a written notice to Seller to that effect (stating with specificity which Required Information Seller has not delivered) and, following delivery of such Required Information specified in such notice, the Secondary Marketing Period will commence; provided that such written notice from Purchaser to Seller will not prejudice Seller's right to assert that the Required Information was, in fact, delivered.

"Securities Act" shall mean the United States Securities Act of 1933.

"Seller Affiliated Group" shall mean any affiliated group (within the meaning of Section 1504(a) of the Code or any similar group defined under a similar provision of state, local, or non-U.S. law) that includes Seller or any of its Affiliates (other than the Acquired Companies), on the one hand, and any Acquired Company, on the other hand.

"Seller Benefit Plan" shall mean each Benefit Plan, other than an Acquired Company Benefit Plan, that is sponsored, maintained or contributed to (or required to be contributed to) by Seller or any of its Affiliates for the benefit of any Business Employee or Former Business Employee.

"Seller Disclosure Schedule" shall mean the disclosure schedules dated as of the date of this Agreement and delivered by Seller to Purchaser in connection with the execution of this Agreement.

"Seller Fundamental Representations" shall mean the representations and warranties of Seller set forth in Section 4.1 (Authority; Enforceability), Section 4.3(a) and Section 4.3(b) (Organization; Acquired Companies), Section 4.4 (Title; Shares) and Section 4.18 (Brokers).

"Seller Group" shall mean Seller and its Affiliates (other than the Acquired Companies).

"Seller Group Marks" shall mean all Trademarks consisting of or incorporating "Dover" or any other Trademark of the Seller Group set forth on Schedule 7.1 of the Seller Disclosure Schedule, and all variations and derivatives thereof.

"Seller Party" shall mean Seller, the Share Seller and the IP Seller.

"Seller's Knowledge" and similar phrases shall mean the knowledge of the individuals set forth on Schedule 1.1(g) of the Seller Disclosure Schedule after due inquiry of each such individual of such individual's direct reports. Notwithstanding the foregoing, with respect to Intellectual Property, such inquiry is not required to include freedom to operate analyses, clearance searches, validity or noninfringement analyses or opinions, or any other similar analyses or opinions of counsel.

"Shared Contracts" shall mean any (a) Contracts under which the Business and at least one other business unit of Seller or any of its Affiliates purchases or sells goods or services on a joint basis or uses goods or services on a joint basis and (b) Contracts that directly benefit both the Business and at least one other business unit of Seller and its Affiliates; *provided* that with respect to the foregoing clauses (a) and (b), in no event shall the term "Shared Contracts" include any Contracts (i) for general corporate functions furnished by Seller or its Affiliates (other than the Acquired Companies) on an enterprise-wide basis to Seller and its Affiliates, including the Corporate Functions, to the extent such functions will be provided to Purchaser under the Transition Services Agreement or (ii) set forth on Schedule 1.1(h) of the Seller Disclosure Schedule.

"Shares" shall mean all of the issued and outstanding shares of each Acquired Entity.

"Software" shall mean (a) firmware, middleware, computer programs, applications, and code, including source code and object code; (b) data and databases, whether machine-readable or otherwise and (c) development and design tools, library functions and compilers, and graphical user interfaces, together, in each case of clauses (a) through (c), with respect to any Software in existence as of the Closing Date, all (i) bug or error fixes, patches, modifications, enhancements, updates, upgrades, corrections, replacement and successor products, new versions, new releases, and derivative works of, to or based on any of the foregoing; (ii) media, documentation, user manuals and training materials, relating to or embodying any of the foregoing or on which any of the foregoing are recorded; and (iii) copies and tangible embodiments of any of the foregoing in any form or media.

"Straddle Period" shall mean any Tax period that begins on or before the Closing Date and ends after the Closing Date.

"Subsidiary" shall mean, with respect to any Person, whether incorporated or unincorporated, any other Person of which (a) such first Person directly or indirectly owns or controls at least a majority of the securities or other interests having by their terms ordinary voting power to elect a majority of the board of directors or others performing similar functions or (b) such first Person is a general partner or managing member; *provided* that, from and after the Closing, none of the Acquired Companies shall be considered a Subsidiary of Seller or any of its Affiliates.

"Target Net Working Capital Lower Amount" shall mean [redacted] dollars (\$[redacted]).

"Target Net Working Capital Upper Amount" shall mean [redacted] dollars (\$[redacted]).

"<u>Tax</u>" shall mean (a) all forms of taxation imposed by any Governmental Authority including all United States federal, state, local or non-U.S. taxes of any kind, (including income, value added, occupation, real and personal property, social security (or similar), gross receipts, sales, use, ad valorem, franchise, profits, license, withholding, payroll, employment, unemployment, excise, severance, occupation, premium or windfall profit, estimated, alternative, add-on minimum, transfer, stamp or other taxes, charges, duties, imposts, assessments, tariffs or levies) and (b) any interest, penalties and additions to tax imposed with respect to any item described in the preceding clause.

"<u>Tax Authority</u>" shall mean a Governmental Authority responsible for the imposition, assessment or collection of any Tax.

"<u>Tax Return</u>" shall mean any report, return, statement, declaration, notice, certificate or other document filed or required to be filed with any Tax Authority in connection with the determination, assessment, collection or payment of any Tax.

"Trade Controls" shall mean any applicable Legal Requirement related to (a) the importation of merchandise, the export or re-export or transfer of products (including software, technology and services), or the ethical conduct of international business activities, including, but not limited to, the Tariff Act of 1930, the Export Administration Act of 1979, the Export Control Reform Act of 2018, the Export Administration Regulations, the International Emergency Economic Powers Act, the Trading with the Enemy Act, the Arms Export Control Act, the International Traffic in Arms Regulations, Executive Orders of the President of the United States regarding economic sanctions and restrictions on transactions with designated entities (including countries, territories, terrorists, organizations and individuals), (b) Sanctions, (c) the anti-boycott regulations administered by the United States Department of Commerce, and the anti-boycott provisions administered by the United States Department of the Treasury.

"<u>Transaction Documents</u>" shall mean this Agreement, the Transition Services Agreement and the Intellectual Property Assignment Agreement, in each case including all exhibits, annexes

and schedules thereto and all amendments thereto made in accordance with the respective terms hereof and thereof and the other documents contemplated by such agreements, including the certificates delivered pursuant thereto.

"Transactions" shall mean the transactions contemplated by this Agreement and the other Transaction Documents.

"<u>Transfer Taxes</u>" shall mean any sales, use, stock transfer, real property transfer, transfer, indirect transfer, goods and services, value-added, stamp, registration, documentary, conveyancing, recording or similar Taxes.

"Treasury Regulations" shall mean the United States Treasury regulations promulgated under the Code.

"Unpaid Income Taxes" shall mean an amount (which shall not be less than zero in the aggregate or in respect of any jurisdiction or any type of Income Tax) equal to the unpaid Income Taxes of the Acquired Companies for any Pre-Closing Tax Period for which Tax Returns have not been filed as of the end of the Closing Date solely in respect of those jurisdictions in which the applicable Acquired Company is currently filing Tax Returns for such Income Taxes; provided that the amount of such Income Taxes shall be determined: (a) on an entity-by-entity basis, such that the deductions and losses of one Acquired Company may not be used to offset the income and gain of another Acquired Company unless such offset is actually permitted by applicable Legal Requirements; (b) by including in taxable income any adjustment as a result of a change in or use of an improper method of accounting on or prior to the Closing Date and any prepaid amounts or deferred revenue that, in each case, would not otherwise be included in taxable income on or prior to the Closing Date, and (c) by disregarding or otherwise excluding any transactions or other actions entered into by Purchaser, its Affiliates or the Acquired Companies outside the ordinary course of business on the Closing Date after the Closing not otherwise contemplated by this Agreement; (d) by excluding any Liabilities for contingent Income Taxes or with respect to uncertain Tax positions for which accruals or reserves have been established in accordance with GAAP; (e) in accordance with the accounting methodologies and the past practices (including reporting positions, elections and accounting methods) of the applicable Acquired Company in preparing its Tax Returns in respect of Income Taxes; (f) by excluding any deferred Tax assets and liabilities; (g) by taking into account any estimated payments and overpayments of such Income Taxes with respect to any Pre-Closing Tax Period; and (h) by excluding any Section 338(h)(10) Election Taxes and any Taxes for any Seller-Filed Tax Returns. In the case of any Straddle Period, Unpaid Income Taxes shall include any amount of Income Taxes allocable to the portion of the Straddle Period ending on and including the Closing Date as determined by applying the conventions set forth in Section 8.5.

"Willful Breach" shall mean: (a) an action or failure to act by one of the parties hereto that constitutes a material breach or violation of this Agreement, and was taken or occurred with such party's knowledge or intention that such action or failure to act constituted a material breach or violation of this Agreement, and such breach or violation: (i) resulted in, or was a principal cause of, the failure of any of the Closing Conditions to be satisfied or (ii) resulted in, or was a principal cause of, the Closing not being consummated at the time the Closing would

have occurred pursuant to <u>Section 3.1</u>, or (b) the failure of Purchaser to deliver the full consideration payable pursuant to <u>Article III</u> at the time the Closing is required to occur pursuant to <u>Section 3.1</u>.

#### ARTICLE II

#### THE TRANSACTIONS

## 2.1 Sales and Purchases.

- (a) Subject to the terms and conditions of this Agreement, at the Closing, Seller shall cause (i) the Share Seller to sell, transfer, convey, assign and deliver to Purchaser (or Purchaser's designee), and Purchaser shall (or shall cause one or more of its designees to) purchase from the Share Seller, all of the Shares and (ii) IP Seller to sell, transfer, convey, assign and deliver to Purchaser (or one or more of Purchaser's designees), and Purchaser shall (or shall cause one or more of its designees to) purchase from IP Seller all of IP Seller's right, title and interest in and to the IP Seller IP, in each case, free and clear of all Encumbrances (other than, in the case of the Shares, those arising under applicable securities Legal Requirements and, in the case of the IP Seller IP, Permitted Encumbrances of the type described in clause (j) of the definition thereof).
- (b) Subject to the terms and conditions of this Agreement, at the Closing, Purchaser shall assume and agree to pay, perform and discharge, as and when due, any and all Assumed Liabilities to the extent such Assumed Liabilities are not then a Liability of an Acquired Company.
- 2.2 <u>Purchase Price</u>. The aggregate consideration to be paid by Purchaser (or one or more of Purchaser's designees) to the Share Seller for the purchase of the Shares and the IP Seller IP shall be an amount in cash equal to (a) two billion dollars (\$2,000,000,000.00), *plus* (b) the Net Working Capital Overage (if any), *minus* (c) the Net Working Capital Underage (if any), *plus* (d) Cash as of the Benchmark Time, *minus* (e) Indebtedness as of the Benchmark Time, *minus* (f) Closing Transaction Expenses (the amount calculated pursuant to this sentence, the "<u>Purchase Price</u>").

### 2.3 <u>Closing Purchase Price</u>

- (a) Not less than four (4) Business Days prior to the anticipated Closing Date, Seller shall prepare and deliver to Purchaser a written statement that is duly executed by an authorized officer of Seller (the "Estimated Closing Statement") setting forth Seller's good faith estimate of: (i) Cash as of the Benchmark Time, (ii) Indebtedness as of the Benchmark Time, (iii) Net Working Capital as of the Benchmark Time (and the Net Working Capital Overage (if any), or the Net Working Capital Underage (if any)), (iv) Closing Transaction Expenses and (v) the resulting calculation of the Purchase Price (such amount, the "Closing Purchase Price"), together with reasonable supporting detail.
- (b) Seller shall provide Purchaser and its Representatives with reasonable access to the work papers and other documents and information that were used in, or are relevant to, the preparation of the Estimated Closing Statement, as well as making available personnel to the extent reasonably requested for review of the Estimated Closing Statement. Purchaser may

provide Seller with comments to the Estimated Closing Statement and Seller shall consider in good faith any reasonable comments from Purchaser that are provided at least one (1) Business Day prior to the Closing Date; *provided* that (i) Seller's obligation to consider in good faith any such comments shall not delay or otherwise prevent the Closing, and (ii) nothing in this Section 2.3(b) shall obligate Seller to accept Purchaser's comments and, to the extent of any dispute regarding Purchaser's comments, Seller's computations shall be conclusive solely for purposes of determining the amounts set out in the Estimated Closing Statement and for the payments under Section 2.1.

# 2.4 <u>Post-Closing Adjustment</u>.

- As soon as practicable after the Closing Date but in no event later than one-hundred twenty (120) days after the Closing Date (the "Submission Deadline"), Purchaser shall deliver to Seller a written statement that is duly executed by an authorized officer of Purchaser (the "Post-Closing Statement") setting forth Purchaser's good faith calculation of: (i) Cash as of the Benchmark Time (the "Preliminary Cash"), (ii) Indebtedness as of the Benchmark Time (the "Preliminary Indebtedness"), (iii) Net Working Capital as of the Benchmark Time (the "Preliminary Net Working Capital"), (iv) Closing Transaction Expenses (the "Preliminary Closing Transaction Expenses") and (v) the resulting calculation of the Purchase Price (such amount, the "Preliminary Closing Purchase Price"), together with reasonable supporting detail. Following the Closing Date through the Submission Deadline, Purchaser, its Affiliates, and its and their Representatives shall be permitted to reasonably access and review the books, records, work papers and other documents of the Seller Group that are reasonably related to the calculations of Cash, Indebtedness, Net Working Capital and Closing Transaction Expenses, and Seller shall, and shall cause its Affiliates and its and their respective Representatives to, cooperate with and assist Purchaser, its Affiliates, and its and their Representatives in connection with such review, including by providing reasonable access to such books, records and other documents and making available personnel to the extent reasonably requested, in each case, upon reasonable notice and during normal business hours: provided that, if Seller or its Affiliates fail to provide such cooperation, assistance or access, the Submission Deadline shall be extended by one (1) day for each day Seller or its Affiliates fail to provide such cooperation, assistance or access. Seller agrees that, following the Closing through the date that the Final Closing Statement becomes final and binding in accordance with Section 2.4(f), it will, and will cause its Affiliates to, preserve all accounting books, records, policies or procedures that are reasonably related to the calculations of Cash, Indebtedness, Net Working Capital and Closing Transaction Expenses. In the event that Purchaser delivers the Post-Closing Statement to Seller in accordance with this Section 2.4(a) prior to the Submission Deadline, Purchaser may amend, supplement or otherwise modify the Post-Closing Statement at any time on or prior to the Submission Deadline; provided, however, that the Review Period shall restart upon receipt of an amended Post-Closing Statement.
- (b) Each component of the Estimated Closing Statement, the Post-Closing Statement and the Seller Post-Closing Statement, as applicable, shall be finally and solely determined consistent with the applicable definitions of such terms set forth herein and calculated in accordance with the Accounting Principles, as applicable (i.e., no accounting

principles, policies, treatments, categorizations, practices, methods, bases or estimation techniques inconsistent with the Accounting Principles may be used by Seller or Purchaser (or if applicable, the Settlement Accountant) in calculating or determining any such component as the sole purpose of the adjustment contemplated by Section 2.4 is to measure the difference, if any, between the estimate of an amount of an item set forth in the Estimated Closing Statement and the actual amount of such item as of the Benchmark Time or Closing, as applicable). No amount shall be included in the calculations of the items in the Post-Closing Statement or the Seller Post-Closing Statement, as applicable, for changes in assets or liabilities of the Acquired Companies as a result of purchase accounting adjustments.

- If Purchaser fails to deliver a Post-Closing Statement within the time period contemplated by Section 2.4(a) and such failure has not been cured on or prior to the date that is five (5) days from the date that Purchaser is notified in writing by Seller of such failure, then, at the election of Seller, either (i) the Estimated Closing Statement shall be final and binding on all parties hereto or (ii) not later than the date that is the sixtieth (60th) day following Submission Deadline, Seller shall prepare a written statement that is duly executed by an authorized officer of Purchaser (the "Seller Post-Closing Statement") setting forth its good faith calculation of (A) Cash as of the Benchmark Time, (B) Indebtedness as of the Benchmark Time, (C) Net Working Capital as of the Benchmark Time, (D) Closing Transaction Expenses and (E) the resulting calculation of the Purchase Price, and such resulting calculation shall be final and binding on all parties hereto; provided that so long as Purchaser delivers the Post-Closing Statement within the time period contemplated by Section 2.4(a), Seller shall have no right to prepare a Seller Post-Closing Statement on the grounds that Purchaser did not provide reasonable supporting detail for such Post-Closing Statement. If Seller elects to prepare the Seller Post-Closing Statement, then, until the delivery of the Seller Post-Closing Statement and without limiting Section 7.3 in any respect, Purchaser shall provide Seller with reasonable access to (x) any and all books, records, workpapers, Contracts and other documents of the Acquired Companies relating to Seller's preparation of the Seller Post-Closing Statement as may be reasonably requested by Seller and (y) the Representatives of the Acquired Companies, in each case, during normal business hours; provided, however, that Purchaser's accountants shall not be obligated to make any work papers available to Seller except in accordance with such accountants' normal disclosure procedures and then only after Seller has signed a customary agreement relating to such access to work papers in form and substance reasonably acceptable to such accountants.
- (d) Upon receipt of the Post-Closing Statement, Seller shall have sixty (60) days (the "Review Period") to review such Post-Closing Statement and related computations of the Preliminary Cash, the Preliminary Indebtedness, the Preliminary Net Working Capital, the Preliminary Closing Transaction Expenses and the Preliminary Closing Purchase Price. Following the Closing through the date that the Final Closing Statement becomes final and binding in accordance with Section 2.4(f), Seller, its Affiliates, and its and their Representatives retained by Seller shall be permitted to reasonably access and review the books, records, work papers and other documents of the Acquired Companies that are reasonably related to the calculations of Cash, Indebtedness, Net Working Capital and Closing Transaction Expenses, and Purchaser shall, and shall cause the Acquired Companies and their Representatives to, cooperate

with and assist Seller, its Affiliates, and its and their Representatives retained by Seller in connection with such review, including by providing reasonable access to such books, records and other documents and making available personnel to the extent reasonably requested, in each case, upon reasonable notice and during normal business hours; provided that, if the accountants of Purchaser or the Acquired Companies fail to provide such cooperation, assistance or access, the Review Period shall be extended by one (1) day for each day Purchaser the Acquired Companies fail to provide such cooperation, assistance or access. Purchaser agrees that, following the Closing through the date that the Final Closing Statement becomes final and binding in accordance with Section 2.4(f), it will preserve any accounting books, records, policies or procedures on which the Post-Closing Statement is based, or upon which the Final Closing Statement is to be based. The accountants of Purchaser or the Acquired Companies shall not be obliged to make any work papers available to Seller, its Affiliates or any of its or their Representatives except in accordance with such accountants' normal disclosure procedures and then only after the applicable Persons seeking access has signed a customary agreement relating to such access to work papers in form and substance reasonably acceptable to such accountants; provided, further, that, in exercising its rights hereunder, Seller shall conduct itself so as not to unreasonably interfere in the conduct of the business following the Closing, and that Seller acknowledges and agrees that any contact by Seller with officers, directors, employees or Representatives of Purchaser or its Affiliates following the Closing shall be arranged and supervised by designated representatives of Purchaser or its Affiliates unless Purchaser otherwise consents with respect to specific contact. Notwithstanding anything in this Agreement to the contrary, Purchaser and its Affiliates shall not be required following the Closing to disclose Seller or its representatives any information that is subject to the attorney-client privilege, attorney work product doctrine, or other applicable privilege or where any such disclosure would violate any applicable Legal Requirement or expose such Person to any liability for disclosure of any personal information or personally identifiable information; provided that Purchaser shall use commercially reasonable efforts to allow for access to the extent that doing so does not result in the loss of any such protection or violate any applicable Legal Requirement. If Seller has accepted the Post-Closing Statement in writing or has not given written notice to Purchaser setting forth any objection of Seller to such Post-Closing Statement, specifying in reasonable detail the basis for such objection and Seller's proposed modifications to the Post-Closing Statement (such notice, the "Statement of Objections") prior to the expiration of the Review Period, then such Post-Closing Statement shall be final, binding and non-appealable upon the parties hereto, and shall be deemed the Final Closing Statement for purposes of Section 2.4(f).

(e) In the event that Seller delivers a Statement of Objections to Purchaser prior to the expiration of the Review Period, Seller and Purchaser shall negotiate in good faith to resolve any such objection on Cash, Indebtedness, Net Working Capital or Closing Transaction Expenses, as the case may be, within thirty (30) days following the receipt by Purchaser of the Statement of Objections or such longer period as Purchaser and Seller may mutually agree in writing (the "Consultation Period"). If Seller and Purchaser reach an agreement in writing as to any such objections within the Consultation Period, the amounts so agreed upon shall be final, binding and non-appealable upon the parties hereto and such agreement shall be deemed to be included in the Final Closing Statement for purposes of Section 2.4(f). If Seller and Purchaser are unable to reach an agreement in writing as to any such objections within the Consultation

Period, then either Seller or Purchaser may submit such matter to BDO USA, LLP, or if BDO USA, LLP is unable or unwilling to serve in such capacity, such other internationally recognized accounting or consulting firm mutually acceptable to Seller and Purchaser (in each case, the "Settlement Accountant"), for resolution of those items on the Statement of Objections that remain in dispute (the "Disputed Items"). If, within ten (10) Business Days after the date BDO USA, LLP informs Seller and Purchaser that it is unable or unwilling to serve as the Settlement Accountant, Seller and Purchaser cannot mutually agree on an alternate independent accounting firm to serve as the Settlement Accountant, either Seller or Purchaser may request the American Arbitration Association to appoint as the Settlement Accountant, within fifteen (15) days from the date of such request or as soon as practicable thereafter, a partner in a nationally recognized accounting or consulting firm that is not the auditor, independent accounting firm or otherwise an advisor or consultant to Seller or any of its Affiliates or Purchaser or any of its Affiliates, who is a certified public accountant. If requested by the Settlement Accountant, each of Seller and Purchaser agrees that it will enter into a customary engagement letter with the Settlement Accountant and provide customary indemnities in favor of the Settlement Accountant. The Settlement Accountant shall act as an expert and not as an arbitrator, and shall only consider the Disputed Items. If any Disputed Item is referred to the Settlement Accountant, Seller, on the one hand, and Purchaser, on the other hand, shall prepare written reports addressing any Disputed Item and deliver such reports to the Settlement Accountant, and each other within fifteen (15) Business Days after the date the Settlement Accountant is retained. Seller and Purchaser, as applicable, shall have ten (10) Business Days to deliver to the Settlement Accountant, and each other one written rebuttal. Unless otherwise agreed to in writing by Purchaser and Seller, any offers to compromise between Seller and Purchaser, and their respective Representatives regarding the items set forth in the Statement of Objections during the Consultation Period shall constitute confidential settlement discussions pursuant to Rule 408 of the Federal Rules of Evidence (as in effect as of the date of this Agreement) and any applicable similar state rule and shall not be submitted to, or considered by, the Settlement Accountant. The Settlement Accountant may not assign a value to any Disputed Item greater than the greatest value for such Disputed Item claimed by either Seller or Purchaser in the Statement of Objections and Post-Closing Statement, respectively, or less than the smallest value for such Disputed Item claimed by either Purchaser or Seller in the Statement of Objections and Post-Closing Statement, respectively. Seller and Purchaser shall use their respective commercially reasonable efforts to cause the Settlement Accountant to deliver a written decision resolving all disagreements as soon as practicable and in any event within thirty (30) days after the later of the submission of the (i) written reports and (ii) written rebuttals, if any, or as soon thereafter as practicable: provided that any delay by the Settlement Accountant in delivering a written report shall not affect the validity of such report. The Settlement Accountant's review and determination shall be (A) limited only to the reports, rebuttals and materials concerning Disputed Items prepared and submitted to the Settlement Accountant by Seller and Purchaser (i.e., not on the basis of an independent review), (B) based solely on such reports, rebuttals and materials submitted by Seller and Purchaser and the basis for Seller's and Purchaser's respective positions and (C) in accordance with the terms and procedures set forth in this Agreement. including the Accounting Principles, and consistent with the definitions of Cash, Indebtedness, Net Working Capital and Closing Transaction Expenses contained herein. Neither Seller nor Purchaser shall authorize the Settlement Accountant to modify or amend any term or provision hereof or modify items

previously agreed in writing between Seller and Purchaser. During the review by the Settlement Accountant, each of Seller and Purchaser shall, and shall cause their respective Subsidiaries (including, in the case of Purchaser, the Acquired Companies) and their respective Representatives to, each make available to the Settlement Accountant such information, books, records and work papers as may be reasonably requested by the Settlement Accountant to fulfill its obligations under this Section 2.4(e); provided that the accountants of Seller or Purchaser shall not be obliged to make any work papers available to the Settlement Accountant except in accordance with such accountants' normal disclosure procedures and then only after such Settlement Accountant has signed a customary agreement relating to such access to work papers. A copy of all materials submitted to the Settlement Accountant shall be promptly provided by Seller or Purchaser, as applicable, to the other party in the dispute; provided that the accountants of Seller or Purchaser, as applicable, shall not be obliged to make any work papers available to the other party except in accordance with such accountants' normal disclosure procedures and then only after such other party has signed a customary agreement relating to such access to work papers. Neither Seller nor Purchaser may disclose to the Settlement Accountant, and the Settlement Accountant may not consider for any purpose, any settlement discussions or settlement offer(s) made by or on behalf of either Seller or Purchaser unless otherwise agreed by Seller and Purchaser. None of Seller, Purchaser or any of their respective Affiliates shall have any ex parte communications or meetings with the Settlement Accountant regarding the subject matter hereof without the other party's prior written consent. The Settlement Accountant shall have exclusive jurisdiction over, and resort to the Settlement Accountant as provided in this Section 2.4(e) shall be the only recourse and remedy of the parties hereto against one another with respect to, any disputes arising out of the calculation of, and any adjustments to, the Purchase Price; provided that, upon the determination of the Settlement Accountant, such determination may be entered and enforced in any court of competent jurisdiction in accordance with Section 13.7. The final determination with respect to all Disputed Items shall be set forth in a written statement by the Settlement Accountant delivered to Seller and Purchaser and, absent mathematical error that is not promptly corrected by the Settlement Accountant or manifest error or fraud, the resolution of the dispute by the Settlement Accountant shall be final, binding and non-appealable on the parties hereto and such determination may be entered and enforced in any court of competent jurisdiction in accordance with Section 13.7. The costs and expenses of the Settlement Accountant shall be borne by Seller and Purchaser in proportion to the difference between the Settlement Accountant's determination of the Purchase Price and the determination of the Purchase Price claimed by Seller and Purchaser. For example, if Seller claims that the Purchase Price is, in the aggregate, \$1,000 greater than the amount determined by Purchaser and if the Settlement Accountant ultimately resolves the dispute by awarding to Seller an aggregate of \$300 of the \$1,000 contested, then the costs and expenses of the Settlement Accountant will be allocated 30% to Purchaser and 70% to Seller. Notwithstanding the foregoing, Seller and Purchaser shall each be responsible for one half of any initial retainers or other fees and expenses of the Settlement Accountant, subject to reallocation by the Settlement Accountant at the conclusion of the proceeding as set forth immediately above.

(f) The "<u>Final Closing Statement</u>" shall mean (i) the Estimated Closing Statement in the event that Purchaser has failed to deliver the Post-Closing Statement in accordance with the terms of this Agreement and Seller elects for the Estimated Closing

Statement to be final and binding in accordance with Section 2.4(c), (ii) the Post-Closing Statement (A) in the event that Seller has accepted the Post-Closing Statement in writing or does not deliver a Statement of Objections prior to the expiration of the Review Period, (B) as agreed to in writing by Seller and Purchaser in accordance with Section 2.4(e), or (C) as determined by the Settlement Accountant in accordance with Section 2.4(e) or (iii) the Seller Post-Closing Statement in the event that Purchaser has failed to deliver the Post-Closing Statement in accordance with the terms of this Agreement and Seller elects to deliver a Seller Post-Closing Statement in accordance with Section 2.4(c) and (1) the Cash set forth on such Final Closing Statement shall be deemed the final Indebtedness (the "Final Closing Indebtedness"), (3) the Net Working Capital set forth on such Final Closing Statement shall be deemed the final Net Working Capital (the "Final Closing Net Working Capital"), (4) the Closing Transaction Expenses set forth on such Final Closing Statement shall be deemed the final Closing Statement shall be deemed the final Closing Transaction Expenses (the "Final Closing Transaction Expenses") and (5) the Purchase Price set forth on such Final Closing Statement shall be deemed the final Purchase Price (the "Final Purchase Price").

- (g) In the event that the Final Purchase Price is greater than the Closing Purchase Price (such excess, the "<u>Final Overage</u>"), Purchaser shall pay, or cause to be paid, within five (5) Business Days of the determination of the Final Overage and the Final Closing Statement, to the Share Seller (or its designee) by wire transfer of immediately available funds, an amount equal to such Final Overage.
- (h) In the event that the Closing Purchase Price is greater than the Final Purchase Price (such excess, the "<u>Final Underage</u>"), the Share Seller shall pay, or cause to be paid, within five (5) Business Days of the determination of the Final Underage and the Final Closing Statement, to Purchaser (or its designee) by wire transfer of immediately available funds, an amount equal to such Final Underage.
- 2.5 <u>Withholding</u>. Purchaser shall be entitled to deduct and withhold from amounts otherwise payable pursuant to this Agreement such amounts as are required to be deducted and withheld under applicable Legal Requirements. Purchaser acknowledges and agrees that, as of the date of this Agreement, no withholding Taxes are expected to be applicable to the payment of the Purchase Price (other than any obligation to withhold in respect of payments that are in the nature of compensation to employees or other services providers); *provided* that Seller delivers the documentation required pursuant to Section 3.2(e). To the extent Purchaser becomes aware of any obligation to withhold (other than any obligation to withhold in respect of payments that are in the nature of compensation or result from Seller's failure to deliver the documentation required pursuant to Section 3.2(e)), it shall use commercially reasonable efforts to provide reasonable advance notice to Seller of the amounts subject to withholding and provide Seller with a reasonable opportunity to deliver any forms, documentation or other evidence that would reduce or eliminate such withholding Tax under Legal Requirements and shall reasonably cooperate with Seller in connection therewith. To the extent any amounts are deducted and withheld by Purchaser under this Section 2.5 and paid over to the applicable Governmental Authority in accordance with the applicable Legal Requirements, such amounts shall be treated

for purposes of this Agreement as having been paid to the Person in respect of which such deduction and withholding was made.

#### ARTICLE III

#### CLOSING AND CLOSING DELIVERIES

- <u>Closing</u>; <u>Time and Place</u>. The closing of the Transactions (the "<u>Closing</u>") shall be effected by the exchange of documents and signatures by electronic transmission or, if such exchange is not practicable at the offices of Skadden, Arps, Slate, Meagher & Flom LLP ("Skadden"). One Manhattan West, New York, NY 10001 (which Closing may occur by electronic exchange of documents), at 10:00 a.m. Eastern time on (a) if the Condition Satisfaction Date is on or prior to the nineteenth (19th) date of a calendar month, the date that is four (4) Business Days following the Condition Satisfaction Date (but subject to the satisfaction of those Closing Conditions which, by their terms or nature, are to be satisfied at the Closing), (b) if the Condition Satisfaction Date is after the nineteenth (19th) date of a calendar month, at Seller's election, the later of (i) the date that is four (4) Business Days following the Condition Satisfaction Date and (ii) the first (1st) Business Day of the calendar month immediately following the calendar month in which the Condition Satisfaction Date occurs (but subject to the satisfaction of those Closing Conditions which, by their terms or nature, are to be satisfied at the Closing), or (c) at such other date, time or place as Seller and Purchaser mutually agree in writing; provided, however, in no event shall the Closing Date occur until the fifth (5<sup>th</sup>) Business Day immediately following the Business Day on which the Initial Marketing Period or, if the Debt Financing has not been obtained pursuant to the terms of this Agreement and the Debt Commitment Documents prior to the commencement of the Secondary Marketing Period, Secondary Marketing Period expires (or such earlier Business Day as may be specified by Purchaser on no less than five (5) Business Days' prior written notice to Seller). The date on which the Closing occurs is referred to herein as the "Closing Date."
  - 3.2 <u>Deliveries by Seller</u>. At the Closing, Seller shall deliver, or cause to be delivered, to Purchaser:
- (a) assignments of the Shares executed by the Share Seller, in form and substance reasonably satisfactory to Purchaser or, for Shares in certificate form, certificates evidencing the Shares, duly endorsed in blank or with stock powers or a similar instrument of transfer duly executed in proper form for transfer and with any required stock transfer stamps affixed thereto;
- (b) duly executed counterparts to a transition services agreement, substantially in the form attached hereto as Exhibit B (the "Transition Services Agreement");
- (c) duly executed counterparts to an intellectual property assignment agreement executed by IP Seller with respect to IP Seller IP, substantially in the form attached hereto as <u>Exhibit C</u> (the "<u>Intellectual Property Assignment Agreement</u>");

- (d) the certificate required to be delivered by Seller pursuant to <u>Section 10.1(d)</u>;
- (e) a duly executed and completed IRS Form W-9 of each of the Share Seller and IP Seller;
- (f) fully executed Section 338(h)(10) Forms;
- (g) resignation letters, in form and substance reasonably acceptable to Purchaser, of such members of the board of directors of each of the Acquired Companies and such officers of each of the Acquired Companies which have been requested in writing by Purchaser at least ten (10) Business Days prior to the Closing Date, such resignation letters to be effective concurrently with, and subject to the occurrence of, the Closing;
- (h) evidence, in form and substance reasonably satisfactory to Purchaser of the settlement or elimination of all intercompany accounts, and the termination of all Intercompany Agreements in accordance with <u>Section 6.4(b)</u>, with no ongoing Liabilities on the part of Purchaser or any of its Affiliates, including the Acquired Companies; and
  - (i) all forms required to be delivered by Seller pursuant to <u>Section 8.1</u>.
  - 3.3 <u>Deliveries by Purchaser</u>. At the Closing, Purchaser shall deliver, or cause to be delivered:
- (a) to the Share Seller or its designee(s), by wire transfer of immediately available funds, the Closing Purchase Price;
  - (b) to Seller, duly executed counterparts to the Transition Services Agreement;
  - (c) fully executed Section 338(h)(10) Forms;
  - (d) to Seller, duly executed counterparts to the Intellectual Property Assignment Agreement; and
  - (e) to Seller, the certificate required to be delivered by Purchaser pursuant to Section 10.2(c).
- 3.4 <u>Payment Mechanics</u>. Any payment to be made pursuant to this Agreement by Purchaser shall be made to the designee(s) and the bank account or accounts designated by Seller in writing to Purchaser on or before the third (3<sup>rd</sup>) Business Day prior to the due date for payment. Any payment to be made pursuant to this Agreement by Seller shall be made to the designee and the bank account or accounts designated by Purchaser in writing to Seller on or before the third (3<sup>rd</sup>) Business Day prior to the due date for payment. Unless otherwise agreed in writing by Seller and Purchaser, any payments by wire transfer under this Agreement shall be in immediately available funds. All payments shall be made by electronic transfer on or prior to the

due date for payment and receipt of the amount due shall be an effective discharge of the relevant payment obligation.

#### ARTICLE IV

#### REPRESENTATIONS AND WARRANTIES OF SELLER

Except as set forth in the Seller Disclosure Schedule, Seller hereby represents and warrants to Purchaser as follows:

### 4.1 <u>Authority: Enforceability.</u>

- (a) Seller has the requisite corporate power and authority to execute and deliver this Agreement, and each of the other Transaction Documents to which it is or will be a party, to perform its obligations hereunder and under each of the other Transaction Documents to which it is or will be a party, and to consummate the Transactions in accordance with the terms of this Agreement, and each of the other Transaction Documents to which it is or will be a party. Each other Seller Party has the requisite corporate power and authority to execute and deliver each of the other Transaction Documents to which it is or will be a party, to perform its obligations under each of the other Transaction Documents to which it is or will be a party. The execution, delivery and performance by Seller of this Agreement and by each Seller Party of each of the other Transaction Documents to which such Seller Party is or will be a party and the consummation of the Transactions contemplated by each of the Transaction Documents to which it is or will be a party, have been duly and validly authorized by all necessary corporate action on the part of each Seller Party and such authorization has not been subsequently modified or rescinded. No additional corporate or shareholder authorization is necessary for the execution, delivery or performance by Seller of this Agreement or by any Seller Party of any other Transaction Document to which it is or will be a party.
- (b) This Agreement and each of the other Transaction Documents to which Seller is or will be a party have been duly and validly executed and delivered by Seller, and constitutes, assuming due authorization, execution and delivery of this Agreement and the other Transaction Documents by Purchaser and any other parties thereto, a valid and binding legal obligation of Seller, enforceable against Seller in accordance with the terms hereof or thereof, as applicable, subject to the Bankruptcy and Equity Exception. Assuming due authorization, execution and delivery of each of the other Transaction Documents to which any other Seller Party is or will be a party by the other parties thereto, each of the other Transaction Documents will constitute a valid and binding legal obligation of the other Seller Parties party thereto, enforceable against such Seller Parties in accordance with the terms thereof, in each case, subject to the Bankruptcy and Equity Exception.

### 4.2 <u>Non-Contravention; Consents.</u>

- (a) The execution and delivery by Seller of this Agreement and by each Seller Party of each other Transaction Document to which it is a party does not, and the performance by Seller of this Agreement and by each Seller Party of each other Transaction Document to which such Seller Party is a party will not, require any Consent or Permit of, registration, declaration or filing with, or notification to, any Governmental Authority, except (i) under applicable Antitrust Laws or Investment Screening Laws, (ii) under the applicable requirements of the Securities Exchange Act of 1934 (the "Exchange Act"), or applicable blue sky laws, (iii) compliance with any Permits relating to the Business, (iv) for such other Consents, filings or notifications, the failure of which to make or obtain would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, and (v) those required by reasons of the regulatory status or operations of Purchaser.
- (b) The execution and delivery by Seller of this Agreement and by each Seller Party of each of the other Transaction Documents to which such Seller Party is a party does not, and the consummation of the Transactions will not, (i) conflict with, violate or constitute a breach of, or require notice, consent or waiver under any provision of the organizational documents of the applicable Seller Party or any of their respective Subsidiaries, (ii) conflict with, violate or constitute a breach of, in each case, in any material respect, or require any notice, consent or waiver under any Legal Requirement applicable to the Business, the Seller Parties or any of their respective Subsidiaries, except with respect to the required filings and approvals set forth in Sections 4.2(a)(i) and (ii) and Schedule 4.2(b)(ii) of the Seller Disclosure Schedule or (iii) result in a breach of, constitute (with or without due notice or lapse of time or both) a default under, result in the creation or acceleration (or loss of benefit from) of any rights or obligations under, or create in any party the right to, accelerate, terminate, modify or cancel, or require any notice, consent or waiver under, any Material Contract to which any Acquired Company is a party or any Shared Contract, except, in the case of the foregoing clauses (ii) or (iii), as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

### 4.3 <u>Organization; Acquired Companies</u>.

(a) Seller is duly incorporated, validly existing and in good standing under the Legal Requirements of the state of Delaware and each of the other Seller Parties is duly incorporated, validly existing and in good standing under the Legal Requirements of the state of its jurisdiction of incorporation, in each case, except as would not reasonably be expected to materially impair or materially delay Seller or any other Seller Party from consummating the Transactions or otherwise prevent Seller from performing in all material respects its obligations hereunder or any Seller Party under any other Transaction Document to which it is or will be a party. Seller has all necessary corporate power and authority to conduct its business in the manner in which it is currently being conducted, except where the absence of such power and authority to conduct its business would not reasonably be expected to materially impair or materially delay Seller from consummating the Transactions or otherwise prevent Seller from performing in all material respects its obligations hereunder.

- (b) Each of the Acquired Companies (i) is duly organized, incorporated or formed, validly existing and in good standing under the Legal Requirements of the jurisdiction of its organization, incorporation or formation, in all material respects, and (ii) has all necessary organizational or corporate power and authority to conduct its business in all material respects in the manner in which it is being conducted as of the date of this Agreement and as of the Closing.
- (c) Each of the Acquired Companies is duly qualified or licensed to do business in the jurisdictions in which the property and assets owned, leased or operated by it, or the nature of the business conducted by it, makes such qualification or licensing necessary, except where the failure to be so duly qualified or licensed would not, individually or in the aggregate, (i) reasonably be expected, individually or in the aggregate, to be material to the Acquired Companies or the Business, taken as a whole, or (ii) prevent or materially delay the consummation of the Transactions.
- (d) Schedule 4.3(d) of the Seller Disclosure Schedule sets forth a true and complete list of the names of each Subsidiary of the Acquired Entities, the jurisdiction in which each such Subsidiary of the Acquired Entities is organized and the equity ownership thereof (including the authorized capitalization, number of outstanding shares of each class of capital stock or other equity interest and the record and beneficial owners thereof), in each case, as of the date of this Agreement and as of the Closing. Except as set forth on Schedule 4.3(d) of the Seller Disclosure Schedule, the Acquired Entities do not own, directly or indirectly (including indirectly through any Acquired Company), any capital stock, shares, membership interests, other equity or ownership rights, interests or other securities or derivatives in any Person, other than the Acquired Entities' Subsidiaries.
- (e) All of the outstanding issued share capital, shares or membership interests, other equity rights, interests or other securities of each Acquired Company are and, as of the Closing shall be, duly and validly issued and outstanding, fully paid and nonassessable and legally and beneficially owned, directly or indirectly, by Seller and the Acquired Entities, as applicable, free and clear of all Encumbrances (other than Encumbrances on securities imposed by applicable Legal Requirements). All of the outstanding issued share capital, shares or membership interests, other equity rights, interests or other securities of each Acquired Company have been issued in compliance with applicable Legal Requirements and the organizational documents of such Acquired Company, and are not subject to any preemptive, subscription or similar right under any provision of any of the foregoing.
- (f) None of Seller nor any of the Acquired Companies is the subject of any bankruptcy, dissolution, liquidation, reorganization or similar proceeding.
- (g) Seller has made available to Purchaser accurate and complete copies of the organizational documents of each Acquired Company as in effect as of the date of this Agreement, and such organizational documents are in full force and effect and no Acquired Company is in violation of any such organizational document.

## 4.4 <u>Title; Shares</u>.

- (a) The Share Seller is the sole record and beneficial owner of all of the outstanding Shares, has good and valid title to the Shares and has full power and authority to transfer and deliver to Purchaser (or Purchaser's designee) the Shares free and clear of all Encumbrances (other than Encumbrances on securities imposed by applicable Legal Requirements), and, upon the payment by Purchaser of the Purchase Price at the Closing, Purchaser (or Purchaser's designee) shall be the sole record and beneficial owner of all of the outstanding Shares, free and clear of all Encumbrances (other than Encumbrances on securities imposed by applicable Legal Requirements). Except pursuant to this Agreement, there is no obligation pursuant to which the Share Seller or any of its Affiliates has granted any option, warrant or other right to any Person to acquire, receive or vote any Shares.
- (b) The Shares are duly authorized, validly issued, fully paid and nonassessable and owned by the Share Seller, free and clear of all Encumbrances (other than Encumbrances on securities imposed by applicable Legal Requirements). Except for the Shares, there are no shares of capital stock of or other voting or equity interests in any of the Acquired Entities that are issued, reserved for issuance or outstanding. There are no shares of capital stock of or other voting or equity interests in any Subsidiary of any of the Acquired Entities that are issued, reserved for issuance or outstanding that are not directly owned by an Acquired Entity.
- (c) There are no outstanding warrants, options, rights, agreements, convertible, exercisable or exchangeable securities or outstanding or authorized appreciation, phantom interest, profit participation or similar rights or other commitments (i) pursuant to which any Acquired Company is or may become obligated to issue, deliver, sell, transfer or grant any (x) shares of capital stock of or other voting or equity interests in an Acquired Company or (y) security convertible into, or exercisable or exchangeable for, any shares of capital stock of or other voting or equity interests in an Acquired Company, (ii) pursuant to which any Acquired Company is or may become obligated to issue, deliver, sell, transfer or grant any such warrant, option, right, unit, security, commitment or undertaking described in the foregoing clause (i) or (iii) that gives any Person the right to receive any benefits or rights similar to any rights enjoyed by or accruing to the holder of the Shares or any shares of capital stock of or other voting or equity interests in any Acquired Company. There are no voting trusts, proxies or other agreements or undertakings with respect to the voting, dividend rights or disposition of the Shares or any shares of capital stock of or other voting or equity interests in any Acquired Company.

### 4.5 Financial Information; Liabilities.

(a) <u>Schedule 4.5(a)</u> of the Seller Disclosure Schedule sets forth true, correct and complete copies of (i) the unaudited combined balance sheet of the Business as of March 31, 2024 (such date, the "<u>Reference Date</u>" and such balance sheet, the "<u>Latest Balance Sheet</u>"), and the related unaudited combined statement of income of the Business (taking into account the Corporate Functions used by the Business) for the three (3) month period ended March 31, 2024, and (ii) the unaudited combined balance sheet of the Business as of December 31, 2022 and December 31, 2023, and the related unaudited combined statements of income of the Business (taking into account the Corporate Functions used by the Business) for the fiscal years ended

December 31, 2022 and December 31, 2023 (such unaudited financial statements referred to in the foregoing clauses (i) and (ii), collectively, the "<u>Unaudited Financial Information</u>") (it being understood, however, that the Business has not been operating historically as a separate "standalone" entity and, therefore, the Unaudited Financial Information will reflect certain costs and other allocations made that may not reflect what would have been incurred if the Business had operated as a standalone business during the period covered by the Unaudited Financial Information and does not reflect costs and other allocations that would be allocated to the Business on a carveout basis). The Unaudited Financial Information: (A) has been prepared in good faith and, except as set forth in the notes thereto, in accordance with GAAP, consistently applied throughout the periods covered thereby, (B) fairly presents the financial condition and results of operations of the Business (taking into account the Corporate Functions used by the Business), as of the dates and for the periods therein specified, and (C) has been derived from books and records of Seller that are regularly maintained by management of Seller and are based upon group accounting guidelines and procedures of Seller used to prepare Seller and its Affiliates' consolidated financial statements in accordance with GAAP.

- (b) Seller has established and maintains a system of internal accounting controls sufficient to provide reasonable assurance that, with respect to the Business transactions (i) including transactions between the Business, on the one hand, and Seller or any of its Affiliates (other than the Business), on the other hand, are recorded as necessary based upon group accounting guidelines and procedures of Seller to permit preparation of Seller's consolidated financial statements in conformity with GAAP and to maintain asset accountability therein and (ii) involving the Business are recorded as necessary based upon group accounting guidelines and procedures of Seller to permit preparation of Seller's consolidated financial statements that are free from material misstatement, except, in the case of each of clauses (i) through (ii), for any deficiency that, individually or in the aggregate, is not material to the Acquired Companies or the Business, taken as a whole. The Acquired Companies have possession of all material books, records and other material documents (whether in paper or electronic form) pertaining to the Business, and the properties and assets of the Business. Since the date that is three (3) years from the date of this Agreement, none of Seller, its Affiliates, their independent accountant(s) or their respective board of managers or board of directors, as applicable, has received any written, or to Seller's Knowledge, oral notification of any (A) "significant deficiency" in the internal controls over financial reporting or (C) fraud, whether or not material, that involves management or other employees who have a significant role in the internal controls over financial reporting.
- (c) There are no Liabilities of the Acquired Companies or the Business of any nature, whether or not accrued, contingent or otherwise, other than such Liabilities: (i) that are specifically set forth and adequately reserved against in the Latest Balance Sheet, (ii) incurred in the ordinary course of business, (iii) arising out of, relating to or resulting from the Transactions or the announcement, negotiation, execution or performance of any Transaction Document, by one of the Acquired Companies since the date of the Latest Balance Sheet or (iv) that would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

# 4.6 <u>Absence of Certain Changes</u>.

- (b) Since December 31, 2023, there has not occurred any event, change, development or effect that has had or would reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect. For purposes of Section 10.1(a), this Section 4.6(b) shall be deemed true and correct in all respects as of the Closing as if made on the Closing Date unless any such Material Adverse Effect is continuing as of the Closing Date.
- (c) During the past twelve (12) months, none of the Acquired Companies or, to the extent related to the Business, Seller or any of its other Affiliates has failed to make, reduced or delayed any repairs or capital expenditures or commitments for capital expenditures, which would not reasonably be expected to be, individually or in the aggregate, material to the Business or the Acquired Companies, taken as a whole.

## 4.7 <u>Compliance with Legal Requirements</u>.

- (a) Since January 1, 2021, (i) the Acquired Companies and the Business have been in compliance with applicable Legal Requirements and (ii) Seller and its Affiliates have not received any written or, to Seller's Knowledge oral, notice from any Governmental Authority of any actual, alleged, alleged or potential violation by any Acquired Company or the Business of any applicable Legal Requirement, except, in the case of each of the foregoing clauses (i) and (ii) for which such instances of non-compliance which would not reasonably be expected to be, individually or in the aggregate, material to the Business or the Acquired Companies, taken as a whole. To Seller's Knowledge, no investigation or inquiry is currently being conducted, or threatened, by any Governmental Authority with respect to the Acquired Companies or the Business.
- (b) Except as would not reasonably be expected to be, individually or in the aggregate, material to the Business or the Acquired Companies, taken as a whole, (i) the Acquired Companies own, hold, possess or lawfully use in the operation of the Business all Permits, which are necessary to conduct the Business as conducted as of the date of this Agreement, (ii) all such Permits are in full force and effect and (iii) none of the Acquired Companies have been in conflict with, or in default, breach or violation of, any such Permit. There is no actual, pending or to Seller's Knowledge threatened in writing, revocation, cancellation, termination, suspension, restriction, materially adverse modification or non-renewal of any such Permit, nor any actual, pending, or to Seller's Knowledge threatened notice of material violation, order of forfeiture or complaint or investigation against any Acquired

Company relating to any conflict with, or default, breach or violation of, any such Permit in any material respect.

### 4.8 <u>Material Contracts</u>.

- (a) <u>Schedule 4.8(a)</u> of the Seller Disclosure Schedule lists all of the following Contracts to which any Acquired Company is a party or to which Seller or any of its Affiliates is a party that is a Shared Contract or primarily relates to the Business (indicating with an asterisk (\*) any such Contracts to which Seller or any of its Affiliates (other than the Acquired Companies) is a party) and that are in effect and not entirely fulfilled or performed as of the date of this Agreement (other than Benefit Plans) (the Contracts required to be listed on <u>Schedule 4.8(a)</u> of the Seller Disclosure Schedule, collectively, the "<u>Material Contracts</u>"); *provided* that (x) order forms, purchase orders, statements of work and (y) any Contracts of the type described in <u>Section 4.8(a)(iii)</u>, in each case, need not be listed on <u>Schedule 4.8(a)</u> of the Seller Disclosure Schedule (the Contracts described in clauses (x) and (y), the "<u>Specified Material Contracts</u>"), but shall otherwise constitute Material Contracts hereunder:
  - (i) any Contract with a Key Customer;
  - (ii) any Contract with a Key Supplier;
  - (iii) Contracts that (A) involve aggregate payments to the Acquired Companies, or aggregate payments by the Acquired Companies, in each case, in excess of \$3,000,000 in the prior twelve (12) months or (B) are reasonably expected to involve aggregate payments to the Acquired Companies, or aggregate payments by the Acquired Companies, in each case, in excess of \$3,000,000 in any calendar year period;
  - (iv) any Contract that requires Seller or any of its Subsidiaries (including the Acquired Companies) to deal exclusively with a third party in connection with the sale or purchase of any product or service or geographic area;
  - (v) any Contract that contains (A) "most favored nation", first refusal, right of first negotiation, first offer provisions or similar preferential terms or (B) take-or-pay or similar minimum purchase requirements, in each case, in favor of any other Person;
  - (vi) any Contract that relates to an acquisition, lease or divestiture of the equity, assets or property or business of any Person (whether by merger, sale of stock or other equity, sale of assets or otherwise) (A) with a purchase price in excess of \$3,000,000, (B) that is material to the operation of the Business, taken as a whole or (C) that contains covenants, indemnities or other obligations that remain in effect and would reasonably be likely to be material to the Business, taken as a whole;
  - (vii) any Contract relating to Indebtedness for borrowed money of the Acquired Companies or with respect to the Business;

- (viii) any Contract that creates any Encumbrance (other than any Permitted Encumbrance) upon any Owned Real Property, any Leased Real Property or any material asset of any Acquired Company or the Business;
  - (ix) any Contract that is a material IP Contract;
  - (x) any Shared Contract;
- (xi) any Government Contract involving aggregate revenue of the Business in excess of \$3,000,000 for the twelve (12) month period ended December 31, 2023;
- (xii) any Contract that provides for any joint venture, partnership, collaboration or other arrangement involving a sharing of profits or losses of any Acquired Company with any Person;
- (xiii) any Contract limiting or restraining (or purporting to limit or restrain) in any material respect Seller or any of its Subsidiaries (including the Acquired Companies) or the Business from (A) competing with any Person in any market or geographic area or in any business, (B) engaging in any type of business or (C) acquiring any entity, in each case, that relates to or affects the Business or any of the Acquired Companies;
- (xiv) any Contract involving a loan (other than transactions on credit in the ordinary course of business) or advance to (other than advances to any Business Employee extended in the ordinary course of business), or investment in, any Person or any Contract relating to the making of any such loan, advance or investment;
- (xv) any Contract involving any actual or threatened Proceeding or other dispute (A)(1) entered into since January 1, 2021 and (2) that has involved or will involve payment in an amount in excess of \$250,000 (net of third-party insurance coverage) or (B) that contains ongoing material obligations, including obligations to pay amounts, individually or in the aggregate, in excess of \$500,000 (net of third-party insurance coverage and excluding compliance with confidentiality, non-disparagement, and other similar customary provisions);
- (xvi) any Contract requiring any capital commitment or capital expenditure (or series of the capital expenditures) by Seller or any of its Subsidiaries (including the Acquired Companies) in respect of the Business in excess of \$3,000,000;
- (xvii) any Contract that contains obligations with respect to any contingent payment of any type (including under any purchase price adjustment, earn-out, deferred payment or similar provision) in excess of \$3,000,000;
  - (xviii) any Real Property Lease that is material to the Business;

- (xix) any supply or tolling Contract for the supply of raw materials, intermediates or finished goods for which there is no reasonably available alternative source as of the date of this Agreement; and
- (xx) any Contract that contains any material indemnification or contribution right or obligation, other than any such right or obligation (1) incurred in the ordinary course of business with any customer or supplier, (2) that provides for any type of customary director and officer indemnification arrangement or (3) in respect of Retained Liabilities.
- (b) Except as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, (i) each of the Material Contracts is in full force and effect, (ii) there exists no default or breach under any such Material Contracts by any Acquired Company or Seller or any of its Subsidiaries or, to Seller's Knowledge, any other party to such Material Contracts. (iii) there exists no event or circumstance with respect to any Acquired Company or Seller or any of its Subsidiaries or, to Seller's Knowledge, any other party to such Material Contracts, that (with notice or lapse of time or both) would create a default or breach under any of the Material Contracts or result in a termination right thereof or would cause or permit the acceleration of or other changes of or to any material right or obligation or the loss of any material benefit thereunder and (iv) there exists no actual or, to Seller's Knowledge, threatened termination or cancellation of any Material Contract. Seller has made available to Purchaser a complete and accurate copy of each Material Contract, other than any Specified Material Contract, that is in effect as of the date of this Agreement (together with all legally binding amendments, modifications, schedules or supplements thereto). Except as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, none of the Acquired Companies or, to the extent related to the Business, Seller or any of its other Affiliates has received any written or, to Seller's Knowledge, oral notice under any Material Contract that any counterparty to any Material Contract intends to terminate any such Material Contracts or is repudiating, not renewing, modifying, or accelerating any material obligation under any Material Contract or that it intends to do so. There have been no material disputes under any Material Contract during the period beginning three (3) years prior to the date hereof.
- 4.9 <u>Litigation</u>. Since January 1, 2021, there have been no Proceedings pending, or, to Seller's Knowledge, threatened against any Acquired Company, any of the Acquired Companies' material assets or properties or the Business or any of their present or former officers, employees, or directors in their capacity as such that, if adversely decided, would, individually or in the aggregate, reasonably be expected to result in (a) a Loss in excess of \$500,000 to the Acquired Companies or the Business for which insurance is unavailable, (b) equitable or injunctive relief, (c) any criminal or regulatory sanction or (d) seeking to or having the effect of preventing, materially impairing, or materially delaying the transactions contemplated by this Agreement, in the case of clauses (a), (b) and (c), except for such Proceedings as would not reasonably be expected to be, individually or in the aggregate, material to the Business or the Acquired Companies. Neither the Acquired Companies nor the Business is subject to any material outstanding or unsatisfied Order or settlement agreement (excluding customary confidentiality

and non-disparagement provisions) that is, or would reasonably be expected to be, material to the Business or the Acquired Companies.

## 4.10 <u>Intellectual Property; Information Technology</u>.

- (a) <u>Schedule 4.10(a)</u> of the Seller Disclosure Schedule sets forth a list of all material United States and international registrations and applications for: (i) Patents, (ii) Trademarks, and (iii) Copyrights, in each case of the foregoing clauses (i)-(iii), that constitute Acquired Company IP (excluding the IP Seller IP) as of the date hereof (such Intellectual Property, the "<u>Acquired Company Registered IP</u>"). The Acquired Companies solely and exclusively own the Acquired Company IP and IP Seller solely and exclusively owns the IP Seller IP, in each case of the foregoing clauses, free and clear of all Encumbrances (other than Permitted Encumbrances). Each item of Acquired Company Registered IP is duly registered or filed in the name of an Acquired Company, and each item of IP Seller IP is duly registered or file in the name of IP Seller, and all such Acquired Company IP is valid, subsisting and enforceable. None of the Acquired Company IP is subject to any outstanding Order adversely affecting any Acquired Company's or IP Seller's use thereof or rights thereto, or that would impair the validity or enforceability thereof.
- (b) No material Proceeding is pending, or to Seller's Knowledge, has been threatened in writing, against Seller or its Subsidiaries (including IP Seller and the Acquired Companies), since January 1, 2021, alleging (i) that the conduct of the Business is infringing, misappropriating or otherwise violating any Person's Intellectual Property, or (ii) the invalidity, misappropriation or unenforceability of or challenging their ownership or scope of any Acquired Company IP. The conduct of the Business as currently conducted does not infringe, misappropriate or otherwise violate the Intellectual Property rights of any third party, and the conduct of the Business has not done so since January 1, 2021.
- (c) No Proceeding is pending or has been threatened in writing since January 1, 2021, (i) by Seller or its Subsidiaries (including IP Seller and the Acquired Companies), alleging that any Person is infringing, misappropriating or otherwise violating any Acquired Company IP or (ii) by any Acquired Company or, with respect to the Business, Seller or its Subsidiaries, alleging the invalidity, misappropriation or unenforceability of any Intellectual Property. No third party is, in any material respect, infringing, misappropriating or otherwise violating the Acquired Company IP, nor has done so since January 1, 2021.
- (d) Except as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, Seller and its Subsidiaries, including IP Seller and the Acquired Companies, own or have a valid license to or other right to use all Intellectual Property used in the conduct of the Business; *provided* that the foregoing is not a representation or warranty with respect to infringement, misappropriation or other violation of Intellectual Property.
- (e) After the Closing, there will be no Intellectual Property or IT Asset owned by Seller or its Affiliates (other than any Seller Group Marks) that is necessary for or used by the Business as of the Closing as to which no provision is made in this Agreement or any

Transaction Document for continued use thereof after the Closing by the Acquired Companies or Purchaser.

- (f) Each Person who has invented, developed or created material Intellectual Property for or on behalf of the Business has assigned all of its rights in same to an Acquired Company, to the extent such rights do not vest in an Acquired Company by operation of law, except as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.
- (g) The Acquired Companies take and have taken commercially reasonable steps to protect and preserve the confidentiality and security of their material Trade Secrets (and Seller and its Subsidiaries have done the same as it relates to the Trade Secrets of the Business), and no such Trade Secrets have been disclosed to any Person except pursuant to non-disclosure obligations that require that Person to maintain the confidentiality of such Trade Secrets. No present or former officer, director, consultant, advisor, employee or independent contractor of any Acquired Company (or as it relates to the Business, of Seller or its other Subsidiaries) is in default or breach of any employment agreement, non-disclosure agreement, assignment of invention agreement or similar agreement relating to the protection, use, or transfer of Acquired Company IP, except as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.
- (h) <u>Schedule 4.10(h)</u> of the Seller Disclosure Schedule sets forth a list of all material software that is Acquired Company IP ("<u>Proprietary Software</u>"). Except as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, none of the Acquired Companies, Seller or its other Subsidiaries, have a duty or obligation to disclose, deliver, license or otherwise make available to any other Person any source code of or for the Proprietary Software ("<u>Acquired Company Source Code</u>") on a contingent basis pursuant to an escrow agreement or similar contractual arrangement. No Acquired Company Source Code has been disclosed to any Person (other than employees and independent contractors of the Acquired Companies that are subject to appropriate confidentiality obligations).
- Effect, the Acquired Companies, and as it relates to the Business, Seller and its other Subsidiaries, are and since January 1, 2021 have been in compliance, in each case, with all terms of all licenses governing the use or distribution of third party Software (including Open Source Software), and no use or activities with respect to Open Source Software by the Acquired Companies, and as it relates to the Business, Seller and its other Subsidiaries, (i) requires the licensing, disclosure or distribution of any Acquired Company Source Code or other Acquired Company IP to licensees or any other Person, (ii) prohibits or limits the receipt of consideration in connection with licensing or otherwise distributing any Proprietary Software, or (iii) allows any Person to decompile, disassemble or otherwise reverse-engineer any Proprietary Software. The Acquired Companies, and with respect to the Business, Seller and its other Subsidiaries maintain and comply with a policy to review the requirements associated with Open Source Software prior to the use thereof. An Acquired Company has in its possession the Acquired Company Source Code.

- (j) No funding, facilities or personnel of any Governmental Authority were used to develop or create, in whole or in part, any Acquired Company IP. The Acquired Companies, and as it relates to the Business, Seller and its other Subsidiaries, have not and have never been a member of, or contributor to, any industry standards organization, body, working group, or similar organization, and none of the Acquired Company IP is subject to any licensing, assignment, contribution, disclosure or other requirement or restriction of any industry standards organization, body, working group, or similar organization.
- Since January 1, 2021, the Acquired Companies, and with respect to the Business, Seller and its other Subsidiaries, are and have been in compliance in all material respects with (i) all applicable Privacy Laws, (ii) all applicable rules of self-regulatory organizations and codes of conduct to which any Acquired Companies has obligated itself to comply, including the Payment Card Industry Data Security Standard. (iii) contractual obligations concerning information security and data privacy (including the Processing of Personal Information) and (iv) their own written internal and external privacy and data security policies (collectively, the "Data Security Requirements"). All vendors, processors, subcontractors and other Persons acting for or on behalf of the Business in connection with the Processing of Personal Information or that otherwise have been authorized to have access to the Acquired Company IT Assets or the Personal Information in the possession or control of such Acquired Company, or, with respect to the Business, Seller and its other Subsidiaries, are subject to reasonable contractual requirements regarding the Processing of Personal Information, and to Seller's Knowledge, comply, and have since January 1, 2021, complied, in each case, in all material respect, with such contractual requirements. Neither the negotiation nor consummation of the transactions contemplated by this Agreement, nor any disclosure or transfer of information in connection therewith, will breach or otherwise cause any material violation of any Data Security Requirement. As of the date of this Agreement no Proceedings are pending or, to Seller's Knowledge, have been threatened in writing against the Acquired Companies, or with respect to the Business, Seller and its other Subsidiaries, since January 1, 2021 alleging a violation of the Data Security Requirements, except as would not reasonably be expected to have, individually or in the aggregate, a material adverse effect.
- (l) The Acquired Company IT Assets (i) are designed, implemented, and operated in a manner consistent with industry standards in all material respects and perform in a manner that permits the Acquired Companies to conduct their respective businesses as currently conducted, (ii) are free from material bugs, errors or other defects that would restrict the Acquired Companies from conducting their respective businesses as currently conducted, and (iii) do not contain any Malicious Code.
- (m) Each Acquired Company, and, with respect to the Business, Seller and its other Subsidiaries, has implemented and maintains an information security plan (the "Security Plan"), which includes commercially reasonable administrative, technical, and physical safeguards consistent with current industry standards, designed to protect the confidentiality, availability, integrity, and security of the Acquired Company IT Assets and the information and data stored therein (including Personal Information and other sensitive information) against any loss, damage, misuse, unauthorized use, access, interruption, modification, corruption, or

disclosure, including cybersecurity and malicious insider risks. The Security Plan conforms, and at all times has conformed, in each case, in all material respects, to the Data Security Requirements. Each Acquired Company, and, with respect to the Business, Seller and its other Subsidiaries have implemented commercially reasonable anti-malware, anti-virus, data backup, disaster avoidance and recovery procedures and business continuity procedures and technology that are in all material respects consistent with applicable industry standards. Except as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, since January 1, 2021 there has been no (i) unauthorized use or access or security breaches, loss or corruption or material malfunction or failure, of any of the Acquired Company IT Assets or any other IT Assets used in the Business (or any data or other information or transactions stored or contained in any Acquired Company IT Assets or such other IT Assets), and the Acquired Company IT Assets and such other IT Assets have not experienced denial of service attacks, continued substandard performance or other adverse events, (ii) unauthorized disclosure, or other breach of security of the Personal Information maintained by or on behalf of any Acquired Company (including, but not limited to, any event that would give rise to a breach or incident for which notification by such Acquired Company to individuals and/or Governmental Authorities is required under Data Security Requirements) or (iii) phishing, social engineering, or business email compromise incident; that, in each case of (i)-(iii), has resulted in a material monetary loss or that has otherwise had or would reasonably be expected to have, individually or in the aggregate, a material adverse effect on the business on the Business.

## 4.11 Real Property.

- (a) Schedule 4.11(a) of the Seller Disclosure Schedule sets forth a true and complete list of all real property owned in fee by any Acquired Company (together with all buildings, structures, improvements and fixtures located thereon, and all easements and other rights and interests appurtenant thereto, the "Owned Real Property"), including, with respect to each Owned Real Property, (i) the fee owner and (ii) physical address. The applicable Acquired Company has good and valid fee title to the applicable Owned Real Property, free and clear of all Encumbrances (other than Permitted Encumbrances). Seller has made available to Purchaser true, correct and complete copies of all deeds, title insurance policies (or if none for the applicable property, title insurance commitments or title insurance reports for such property), surveys, and zoning reports within the possession and control of Seller for each Owned Real Property. Other than as set forth on Schedule 4.11(a) of the Seller Disclosure Schedule or in any real property leases, licenses or other occupancy agreements relating to the Owned Real Property made available to Purchaser, (i) other than an Acquired Company, there are no parties who have a right to possess or are in possession of the Owned Real Property and (ii) no Acquired Company is a party to any Contract granting, or has otherwise granted to, another party any outstanding Encumbrance, option, right of first offer or first negotiation or right of first refusal or other similar right to purchase or lease the Owned Real Property or any portion thereof (each an "Option" and collectively, "Options") and to Seller's Knowledge, no Owned Real Property is subject to any such Options.
- (b) <u>Schedule 4.11(b)</u> of the Seller Disclosure Schedule sets forth a list of all leases, licenses and subleases to which an Acquired Company is party as a lessee, licensee or

sublessee (any such lease, license or sublease, individually, and together with all material amendments, modifications, extensions, renewals, guaranties and other agreements with respect thereto, a "Real Property Lease" and collectively, the "Real Property Leases"), including with respect to each such Leased Real Property, (i) the landlord, sublandlord, licensor, sublicensor or grantor, (ii) the tenant, subtenant, licensee, sublicensee, grantee or occupant, and (iii) to the extent specified in the applicable Real Property Lease, the address. Each Real Property Lease is a valid and binding obligation of the Acquired Company party thereto, subject to the Bankruptcy and Equity Exception. Seller has made available to Purchaser copies of the Real Property Leases as in effect as of the date of this Agreement which are accurate and complete in all material respects. Each of the applicable Acquired Companies has a valid and enforceable leasehold, license or subleasehold (as applicable) interest under each of the Real Property Leases to which it is a party, free and clear of all Encumbrances (other than Permitted Encumbrances).

- (c) None of Seller or any of its Affiliates has received any written notice of any default or event that, with or without notice, lapse of time or both, would constitute a material default by such Acquired Company under any of the Real Property Leases. No Acquired Company has received written notice from the landlord of any material default or breach (after the expiration of any notice or cure period) by an Acquired Company under any Real Property Lease, and no such material default or breach (after the expiration of any notice or cure period) exists. No Acquired Company is in material default or breach (after the expiration of any notice or cure period) by the lessor under any Real Property Lease. To Seller's Knowledge, no condition exists which, but for the giving of notice or the passage of time, would constitute a material breach or default by any Acquired Company (or to Seller's Knowledge, any other party thereto), or permit termination or modification by any party thereto or acceleration of rent by the landlord, sublessor or licensor of any Real Property Lease.
- (d) There is no pending, and to Seller's Knowledge, there is no threatened appropriation, condemnation, eminent domain or like proceedings relating to the Real Property. Neither the whole nor any portion of the Real Property has been materially damaged or destroyed by fire or other casualty and not restored to a condition reasonably sufficient for the operation thereof for its current use. Subject to Permitted Encumbrances, no improvements constituting a part of the Owned Real Property encroach, in any material respect, on real property owned by any Person other than the Acquired Companies or the owner of such Owned Real Property. Subject to Permitted Encumbrances, no Acquired Company has received written notice from a lender of a default which remains uncured after the expiration of any notice or cure period, nor is there any such default (after the expiration of any notice or cure period) under any mortgage or similar instrument encumbering any Owned Real Property. None of Seller's leasehold interests in the Leased Real Property is encumbered by a mortgage.
- (e) The Real Property and all material improvements thereon are in all material respects in adequate working condition and repair for the conduct of the Business thereon. Except as would not, individually or in the aggregate, reasonably be expected to be material to the Business or the Acquired Companies, taken as a whole, there are no structural or

other physical defects or deficiencies in the condition of the Owned Real Property. Other than as set forth in any documents relating to the Owned Real Property made available to Purchaser, none of Seller or any of its Affiliates has received any written notice of, and there are no material special Taxes, levies or assessments, pending, certified or, to Seller's Knowledge, contemplated, with respect to any of the Owned Real Property, in each case, subject to Permitted Encumbrances.

- (f) Seller has provided all material permits, certificates of occupancy, licenses, authorizations and approvals within Seller's possession and control relating to the use and occupancy of the Owned Real Property by the Acquired Companies.
- (g) The Real Property constitutes all of the real property occupied by the Acquired Companies for the conduct of the Business.

#### 4.12 Labor Matters.

- (a) <u>Schedule 4.12(a)</u> of the Seller Disclosure Schedule sets forth as a list, as of the date of this Agreement, of each Business Employee, and for each: (i) identification number, (ii) work location, (iii) hire date, (iv) annual base salary or hourly wage rate (as applicable), (v) exempt or non-exempt classification (as applicable), (vi) active or inactive status, and if inactive, the type of leave, (vii) full-time or part-time status and (viii) employing entity.
- (b) Solely with respect to the Business or any Business Employee, Seller and its Affiliates are, and since January 1, 2021 have been, in compliance with all applicable Legal Requirements relating to labor and employment, including those relating to wages, hours, collective bargaining, unemployment compensation, worker's compensation, equal employment opportunity, immigration, health and safety, worker classification, affirmative action, plant closings and employee layoffs, information privacy and security, payment and withholding of taxes and continuation coverage with respect to group health plans, except for any non-compliance that would not reasonably be expected to be material to the Business and the Acquired Companies, taken as a whole.
- (c) Since January 1, 2021, with respect to Business Employees, there has not been any strike, material slowdown, work stoppage or unfair labor practice claim against any Acquired Company, and none is pending nor, to Seller's Knowledge, threatened. To Seller's Knowledge, no activities or proceedings of any labor union to organize any Business Employees are pending or threatened. No labor union or works council represents any Business Employees in connection with their employment with any Acquired Company or the Business, nor are any Business Employees in connection with their employment with any Acquired Company or the Business covered by any Collective Bargaining Agreement.
- (d) To Seller's Knowledge, with respect to the Business Employees, Seller and its Affiliates have reasonably investigated all sexual harassment or other harassment, discrimination, retaliation or material policy violation allegations against officers, directors or managerial or supervisory employees. With respect to each such allegation (except those that the applicable Acquired Company or Seller or its Affiliates (as applicable) reasonably deemed to not

have merit), such Acquired Company or Seller or its Affiliates (as applicable) has taken corrective action reasonably calculated to prevent further improper action. Since January 1, 2021 there have been no such allegations of harassment or discrimination pending or, to Seller's Knowledge, threatened, that if known to the public would bring the Acquired Companies or the Business into material disrepute.

- (e) To Seller's Knowledge, no Business Employee is in any material respect in violation of any term of any employment agreement, non-disclosure agreement, common law non-disclosure obligation, fiduciary duty, non-competition agreement, restrictive covenant or other similar obligation: (i) to any of Seller and its Affiliates (solely in respect of the Business), the Business or the Acquired Companies or (ii) to a former employer of any such Business Employee relating (A) to the right of any such employee to be employed by any member of the Seller Group in connection with the Business, as the case may be or (B) to the knowledge or use of trade secrets or proprietary information.
- (f) To Seller's Knowledge, no Business Employee who is at the level of director or higher, intends to terminate his or her employment.

### 4.13 Employee Benefits.

- (a) Schedule 4.13(a) of the Seller Disclosure Schedule sets forth a list, as of the date of this Agreement, of each material Acquired Company Benefit Plan and each material Seller Benefit Plan (marked with an asterisk (\*)). With respect to each such material Acquired Company Benefit Plan, Seller has made available to Purchaser accurate and complete copies of, as applicable, the current plan documents (or, if such plan is not in writing, a written description of such plan), trust agreements, insurance contracts or other funding vehicles and all amendments thereto, in the case of any plan intended to be qualified under Section 401(a) of the Code, the most recent determination or opinion letter from the IRS, the most recently filed Form 5500 and any non-routine correspondence with a Governmental Authority since January 1, 2021. With respect to each such material Seller Benefit Plan, Seller has made available to Purchaser a copy of, or a summary of the material terms of, such Seller Benefit Plan.
- (b) (i) Each Acquired Company Benefit Plan and Seller Benefit Plan has been administered, funded and operated in compliance with applicable Legal Requirements and in accordance with its terms and (ii) no action is pending or, to Seller's Knowledge, threatened with respect to any Acquired Company Benefit Plan or Seller Benefit Plan (other than routine claims for benefits payable in the ordinary course, and appeals of denied claims), in each case of clauses (i) and (ii), except as would not reasonably be expected to be material to the Business and the Acquired Companies, taken as a whole. With respect to each Acquired Company Benefit Plan and Seller Benefit Plan, all contributions of any Acquired Company due on or before the date hereof have been made or properly accrued in all material respects.
- (c) No Acquired Company Benefit Plan or Seller Benefit Plan is and no Acquired Company has ever sponsored or contributed to in the prior six (6) years (i) an "employee pension benefit plan" (as defined in Section 3(2) of ERISA) or a "defined benefit plan" (as defined in Section 3(35) of ERISA) or any other plan that is subject to Section 302 or

Title IV of ERISA (each, a "Pension Plan"), (ii) a "multiemployer plan" (as defined in Section 3(37) of ERISA), (iii) a "multiple employer plan" (within the meaning of Section 413 of the Code) or (iv) a "multiple employer welfare arrangement" within the meaning of Section 3(40) of ERISA. None of the Acquired Companies have failed to make any minimum required contribution as defined in Section 302 of ERISA and Sections 412 and 430 of the Code or otherwise failed to comply with the minimum funding standards set forth in such sections with respect to any funded Pension Plans that are subject to such requirements.

- (d) Each Acquired Company Benefit Plan or Seller Benefit Plan intended to be qualified under Section 401(a) of the Code has received a favorable determination, advisory and/or opinion letter, as applicable from the IRS regarding the tax-qualified status of such plan and, to Seller's Knowledge, no event has occurred or condition exists that would reasonably be expected to result in the loss of such tax-qualification.
- (e) None of the Acquired Company Benefit Plans or Seller Benefit Plans obligates Seller or any of its Subsidiaries (including the Acquired Companies) to provide any Business Employee or Former Business Employee any life insurance or medical or health benefits after his or her termination of employment or service with any Acquired Company, other than as required under Part 6 of Subtitle B of Title I of ERISA, Section 4980B of the Code or any similar state Legal Requirement.
- (f) Except as otherwise provided in this Agreement, neither the execution of this Agreement nor the consummation of the Transactions will (whether alone or together with any other event) (i) result in any payment or benefit becoming due to any Business Employee or Former Business Employee, (ii) increase any payment or benefit to be paid or provided to any Business Employee or Former Business Employee, (iii) result in any acceleration of the time of payment, funding or vesting of any payments or benefits to any Business Employee or Former Business Employee, (iv) result in the forgiveness of any indebtedness of any Business Employee or Former Business Employee or (v) result in any payment to a Business Employee or Former Business Employee that could reasonably be construed, individually or in combination with any other such payment, to constitute an "excess parachute payment" (as defined in Section 280G(b)(1) of the Code).
- (g) None of the Acquired Companies is obligated to provide any gross-up, make-whole or other additional payment in respect of any Tax (including Taxes imposed under Section 4999 or 409A of the Code) or interest or penalty related thereto.
- (h) Except as would not reasonably be expected to be material to the Business and the Acquired Companies, taken as a whole, each Acquired Company Benefit Plan and Seller Benefit Plan or other compensation or benefit plan, program, policy, practice, contract, agreement or arrangement that constitutes, in whole or in part, a "nonqualified deferred compensation plan" (within the meaning of Section 409A of the Code) has been maintained, operated and administered in all respects in documentary compliance with Section 409A of the Code and applicable guidance thereunder, and no amount under any such plan has been or is reasonably expected to be subject to any interest or additional Taxes imposed under Section 409A of the Code.

(i) Each Acquired Company Benefit Plan and each Seller Benefit Plan that is subject to the laws of a jurisdiction other than the United States (i) if intended to qualify for special tax treatment, meets all requirements for such treatment and, to Seller's Knowledge, there are no existing circumstances or events that have occurred that could reasonably be expected to affect adversely the special tax treatment with respect to such plan, (ii) if a defined benefit pension plan that is intended to be funded and/or book-reserved, is funded and/or book reserved, as appropriate, in accordance with applicable Legal Requirements, based upon reasonable actuarial assumptions, and (iii) if intended or required to be qualified, approved or registered with a Governmental Authority, is and has been so qualified, approved or registered and nothing has occurred that could reasonably be expected to result in the loss of such qualification, approval or registration, as applicable, in each case of clauses (i), (ii) and (iii), except as would not reasonably be expected to be material to the Business and the Acquired Companies, taken as a whole. No Acquired Company Benefit Plan that is subject to the laws of a jurisdiction other than the United States is a defined benefit pension plan.

### 4.14 Taxes.

- (a) Each of the Acquired Companies and, with respect to the IP Seller IP, the IP Seller, has filed, or has caused to be filed on its behalf, all material Tax Returns required to be filed by it (taking into account any valid extensions of time in which to file), and all such Tax Returns are true, accurate, and complete in all respects. All material Taxes due and payable by the Acquired Companies have been paid in full.
- (b) Each of the Acquired Companies has (i) deducted, withheld and timely paid to the appropriate Tax Authority all material Taxes required to be deducted, withheld or paid by it (whether or not shown on any Tax Return), and (ii) complied with all related information reporting requirements in all respects.
- (c) All material Taxes required to have been collected and paid by each of the Acquired Companies on the sale of products or taxable services by the applicable Acquired Company (whether or not denominated as sales or use Taxes) have been properly collected and paid or have been accrued for in the Audited Financial Information, and all sales tax exemption certificates or other proof of the exempt nature of sales of such products or services have been properly collected and retained or, to the extent required, submitted to the appropriate Tax Authority.
- (d) No power of attorney has been granted with respect to any of the Acquired Companies as to any matter relating to Taxes that will remain in effect following the Closing.
- (e) None of the Acquired Companies has a permanent establishment (within the meaning of an applicable Tax treaty), or is otherwise subject to Tax in a jurisdiction other than the United States.
- (f) None of the Acquired Companies has ever been a member of an affiliated group or filed or been included in a combined, consolidated or unitary income Tax Return (other than the Seller Affiliated Group or a Consolidated Return, respectively). None of the Acquired

Companies is liable for the Taxes of another Person other than any of the Acquired Companies (i) under Treasury Regulations Section 1.1502-6 (or comparable provisions of any state, local or non-U.S. Tax Legal Requirement), other than any other Person that is or was a member of the Seller Affiliated Group or (ii) as a transferee or successor, or by Contract (other than a Contract entered into in the ordinary course of business, the primary purpose of which is not Taxes). No Acquired Company is party to or bound by any Tax allocation, Tax indemnity, Tax sharing or similar agreement (other than any agreement (x) entered into in the ordinary course of business, the primary purpose of which is not Taxes, or (y) solely among the Acquired Companies).

- (g) There is no current or pending dispute, audit, examination, claim, litigation, proposed adjustment, matter in controversy or other Proceeding concerning any material Tax liability of any Acquired Company or with respect to the IP Seller IP, and no such dispute, audit, examination, claim, litigation, proposed adjustment, matter in controversy or other Proceeding has been proposed or threatened in writing.
- (h) No extension or waiver of (i) the limitation period applicable to any Tax Return of any Acquired Company or (ii) any period of assessment or collection with respect to Taxes of any of the Acquired Companies or the Seller Affiliated Group has been granted or requested from Seller or any of the Acquired Companies which extension or waiver has not expired.
- (i) There are no outstanding Encumbrances for material Taxes (other than Permitted Encumbrances) on the IP Seller IP or assets of any Acquired Company.
- (j) None of the Acquired Companies will be required to include any item of income in, or exclude any item of deduction from, taxable income for any Post-Closing Tax Period as a result of: (i) any prepaid amount or deferred revenue received at or prior to the Closing; (ii) any change in method of accounting or use of improper method of accounting prior to the Closing Date; (iii) any "closing agreement" as described in Code Section 7121 (or any corresponding or similar provision of state, local or non-U.S. Income Tax Legal Requirements) executed prior to the Closing; (iv) any installment sale or open transaction disposition made prior to the Closing or (v) any intercompany transactions or excess loss accounts as described in the Treasury Regulations under Section 1502 of the Code (or any corresponding or similar provision of state, local or non-U.S. Income Tax Legal Requirements) arising on or prior to the Closing Date.
- (k) None of the Acquired Companies has entered into any voluntary disclosure agreements with any Tax Authority with respect to any Tax.
- (l) None of the Acquired Companies has distributed stock of another Person, nor had its shares distributed by another Person, in a transaction that was purported or intended to be governed, in whole or in part, by Section 355 or Section 361 of the Code (or any corresponding provision of state, local or non-U.S. Legal Requirements).
- (m) None of the Acquired Companies has participated in, or is currently participating in, any "listed transaction" within the meaning of Section 6707A(c) of the Code or

Treasury Regulations Section 1.6011-4(b)(2), or any transaction requiring disclosure under a corresponding provision of state, local, or non-U.S. Legal Requirements.

- (n) None of the Acquired Companies (i) is the beneficiary of any extension of time within which to file any Tax Return (or has made or had made on its behalf a request for such which is currently pending), other than automatic extensions of time or other extensions of time obtained in the ordinary course of business, (ii) has received a written claim by a Tax Authority in a jurisdiction in which it does not file Tax Returns that it is or may be subject to taxation therein or required to file a Tax Return therein, (iii) has received (or is subject to) any private letter or similar ruling or agreement from any Tax Authority, in each case, that will remain binding on the applicable Acquired Company after the Closing, or (iv) is a party to any joint venture, partnership or other arrangement that is treated as a partnership for U.S. federal income Tax purposes.
  - (o) Each of the Acquired Companies is classified as a corporation for U.S. federal income Tax purposes.
- (p) All payments by, to, or among any of the Acquired Companies and/or any Affiliate of the any of the Acquired Companies related to any Affiliate agreement have complied with all applicable transfer pricing requirements imposed by any Tax Authority in all material respects and all material documentation required by relevant transfer pricing laws has been timely prepared or obtained and, if necessary, retained by the applicable Acquired Company.
- (q) Except as set forth on Schedule 4.14 of the Seller Disclosure Schedule, none of the Acquired Companies has (a) deferred any payroll, social security, unemployment, withholding or other Tax under the CARES Act which deferrals remain outstanding or (b) applied for or received any Tax credits under the CARES Act. Each Acquired Company was eligible to receive and validly claimed any Tax credits it received under the CARES Act.
- (r) The Acquired Companies have each filed with the appropriate Governmental Authorities all material unclaimed property reports as required under applicable Legal Requirements and are otherwise in compliance in all material respects with applicable Legal Requirements relating to unclaimed property or escheat obligations.
- (s) For purposes of the Section 338(h)(10) Election, each of the Acquired Companies is a member of the affiliated group filing consolidated U.S. federal income Tax Returns of which Seller is the common parent. Neither Seller nor any of its Affiliates has taken or agreed to take any action, and, to Seller's Knowledge, no fact or circumstance exists, that would prevent or impede, or could reasonably be expected to prevent or impede, the timely making of the Section 338(h)(10) Election.
- 4.15 <u>Sufficiency of Assets</u>. Except as set forth on <u>Schedule 4.15</u> of the Seller Disclosure Schedule, and subject to the rights and covenants under <u>Section 7.1</u>, upon the receipt of any necessary third-party consents, approvals and Permits required in connection with consummation of the transactions contemplated by this Agreement, the material assets, rights, properties and interests owned, leased or licensed by the Acquired Companies, together with the

IP Seller IP transferred to Purchaser or its designees at the Closing, and the assets, rights properties and interests to be provided pursuant to the Transition Services Agreement or any other transitional arrangements between Seller, Purchaser or any of their respective Affiliates, are sufficient for, and constitute all of the material assets, rights, properties and interests used in or necessary for, the conduct of the Business immediately following the Closing in substantially the same manner as conducted as of the date of this Agreement; *provided* that this Section 4.15 shall not be deemed to be breached as a result of any action (i) that Seller or any of its Affiliates is or are expressly required to take pursuant to this Agreement, the Transition Services Agreement, or (ii) for which Purchaser has expressly consented in writing (including pursuant to Section 6.1); and *provided further* that the foregoing is not a representation or warranty with respect to infringement, misappropriation or other violation of Intellectual Property. Nothing in this Section 4.15 shall be deemed to expand the scope of any other representations or warranties made by Seller in this Article IV.

### 4.16 Environmental Matters.

Except as would not, individually or in the aggregate, reasonably be expected to be material to the (a) Business or the Acquired Companies, taken as a whole, (i) the Acquired Companies possess, and, since January 1, 2020, have possessed, all Permits required by Environmental Laws for the conduct of the Business (collectively, "Environmental Authorizations"); (ii) the Acquired Companies and the Business are and, since January 1, 2020, have been in compliance with all applicable Environmental Laws and Environmental Authorizations; (iii) there are no Orders or Proceedings pending or, to Seller's Knowledge, threatened, against or involving the Business, any Acquired Company or any of their respective predecessors alleging the violation of, non-compliance with or liability under any applicable Environmental Laws; (iv) no written notice under any Environmental Law has been received by Seller or any of its Affiliates from any Governmental Authority or other Person concerning the Release or possible Release of Hazardous Materials, or requiring an investigation for Hazardous Materials, at any location owned, operated or leased, now or in the past, by the Business or any Acquired Company or otherwise concerning violations of Environmental Law by or Liability of the Business or any Acquired Company pursuant to any Environmental Law, in each case, except for any such written notice that has been fully and finally resolved with no further Liability; (v) there have been no Releases or threatened Releases of, or exposure to, Hazardous Materials (A) at, on, about, under or migrating from any Real Property or any other real property currently owned, leased or operated by the Business or any of the Acquired Companies (including any facilities or structures located on, in, or under such property) or, to Seller's Knowledge, any real property formerly owned, leased or operated by the Business or any of the Acquired Companies or any of their respective predecessors or at any other third-party location, or (B) arising from or relating to the operations of or any products manufactured, marketed, sold or distributed by the Business or any of the Acquired Companies or any of their predecessors, in each case, that would reasonably be likely to give rise to violations of, or Liabilities under, any Environmental Laws; (vi) none of the Business or the Acquired Companies has assumed or retained, by contract or operation of law, any obligation under any Environmental Law or concerning any Hazardous Materials that could reasonably be expected to result in Liability of any of the Business or Acquired Company under any applicable Environmental Law; (vii) Seller is not aware of any

past, present or anticipated conditions, events, circumstances (including (A) anticipated regulatory requirements and (B) anticipated obligations pursuant to Environmental Law or Environmental Authorizations that are triggered by the contemplated transaction) that are expected to (1) materially increase the costs of or prevent continued compliance by any of the Acquired Companies with Environmental Laws and the requirements of Environmental Authorizations, or (2) give rise to Liability of the Business or any Acquired Company under any Environmental Laws; and (viii) since January 1, 2021, none of the Acquired Companies or any of their predecessors has been named as a defendant in or is or has been subject to any Proceeding relating to the use of or exposure to asbestos, and, to Seller's Knowledge, no Proceedings have been threatened.

- (b) Seller has made available to Purchaser copies of any Phase I or Phase II Environmental Site Assessments or any material reports, studies, analyses, tests, or monitoring and any other material documents possessed by (or initiated by and within the reasonable control of) Seller or any of its Affiliates pertaining to matters relating to any compliance or non-compliance with or Liability under Environmental Law with respect to the Business, any of the Acquired Companies or any of their predecessors (including any real property currently or formerly owned, leased or operated by the foregoing); *provided, however*, that with respect to compliance with Environmental Laws, routine filings or submittals made in the ordinary course which demonstrate compliance with Environmental Law are not considered to be material reports, studies, analyses, tests, or monitoring or other material documents.
- 4.17 Anti-Bribery Matters. Except as would not be material to the Acquired Companies or the Business taken as a whole, for the past five (5) years prior to the date of this Agreement, none of the Acquired Companies (including any of their respective officers, directors or employees) nor, to Seller's Knowledge, any agent or other Person acting on behalf of the Acquired Companies has (a) taken any action in violation of any anti-corruption or anti-bribery Legal Requirement applicable to any Acquired Company (in each case, as in effect at the time of such action), including the U.S. Foreign Corrupt Practices Act, the U.K. Bribery Act, any laws enacted pursuant to, or arising under, the OECD Convention on Combating Bribery of Foreign Public Officials in International Business Transactions, or any other applicable laws or regulations relating to bribery or corruption (collectively, the "Anti-Bribery Laws"), (b) used any corporate funds for unlawful contributions, gifts, entertainment or other unlawful expenses relating to political activity, (c) made, offered or authorized any unlawful payment to foreign or domestic government officials or employees or (d) made, offered or authorized any unlawful bribe, rebate, payoff, influence payment, kickback or other similar unlawful payment. For the five (5) years prior to the date of this Agreement, there have been no charges, reports, internal investigations or inquiries, voluntary disclosures or Proceedings pending, or, to Seller's Knowledge, threatened against the Business, or any of the Acquired Companies, or any of the Business's or Acquired Companies' officers or directors in their capacity as such pursuant to any applicable Anti-Bribery Laws. The Business and the Acquired Companies maintain policies and procedures reasonably designed to ensure compliance with all applicable Anti-Bribery Laws.
- 4.18 <u>Brokers</u>. Other than with respect to fees or commissions that will be borne solely by the Seller Group, neither Seller nor any of its Affiliates has retained any broker, finder, agent,

financial advisor, investment banker or similar intermediary who has acted on behalf of the Acquired Companies and no broker, finder, agent, financial advisor, investment banker or similar intermediary is entitled to, nor has any Acquired Company incurred any liability or obligation for, any brokerage fees, commissions, finders' fees, financial advisor fees or other similar fees, in each case, with respect to this Agreement or the Transactions.

- A.19 Related Party Transactions. Except as set forth on Schedule 4.19 of the Seller Disclosure Schedule, and other than Contracts, Benefits Plans or policies for employment and benefits provided to employees and other individual service providers in the ordinary course of business, no officer, director, manager or employee of Seller or its Affiliates (including the Acquired Companies) or Seller or any of its Affiliates (other than the Acquired Companies) (each, a "Related Person"): (a) has entered into any transaction with or is a party to any Contract with the Business or any Acquired Company; (b) has any right, title, or interest in or to, or uses, holds for use, or licenses, any of the assets or properties used in the Business, whether tangible or intangible (including any Intellectual Property); or (c) provides or causes to be provided to the Business any of the assets, properties, services or facilities used in the Business (other than those that will continue to be provided under the Transition Services Agreement), in each case, that is material to the Business and the Acquired Companies, taken as a whole. None of Seller or any of its Affiliates owns, leases or uses (or prior to the Closing will own, use or lease) any Owned Real Property or Leased Real Property or any other warehouses, distribution facility or other facility that is used by, or related to, the Business and any other business of Seller or any other Related Person. Schedule 4.19 of the Seller Disclosure Schedule sets forth a list of each Shared Contract and, with respect to each, as applicable, (i) the parties thereto and (ii) the subject matter. There have not been any transfers of personnel between the Business and the other businesses of Seller and its Affiliates other than in the ordinary course.
- 4.20 <u>Key Customers and Key Suppliers.</u> <u>Schedule 4.20</u> of the Seller Disclosure Schedules lists (a) the Key Customers and (b) the Key Suppliers. During the preceding twelve (12) months, (i) neither Seller nor any of its Affiliates (including the Acquired Companies) has engaged in any material dispute with any Key Customer or Key Supplier, nor has any Key Customer or Key Supplier provided written or, to Seller's Knowledge, oral notice to Seller or any of its Affiliates (including the Acquired Companies) that it intends to terminate, not renew or adversely alter in any material respect its relationship with such entity or the terms thereof and (ii) no Key Customer has (A) materially decreased its purchases of products or services from Seller or any of its Affiliates (including the Acquired Companies) relative to such Key Customer's purchasing history during the twelve (12) months prior to such change or (B) requested in writing reduced pricing (whether through increased credits or otherwise) from that in effect under an existing Contract.
- 4.21 <u>Accounts Payable; Accounts Receivable</u>. Except as would not reasonably be expected to be, individually or in the aggregate, material to the Acquired Companies or the Business, taken as a whole, (a) all accounts payable of the Business that are reflected in the Unaudited Financial Information (except as specifically set forth therein) and all accounts payable of the Business arising since the date of the Latest Balance Sheet are valid obligations of one of the Acquired Companies and have arisen only from bona fide arm's length transactions in

the ordinary course of business, and all such accounts payable have either been paid, are not yet due and payable in the ordinary course of business, are being contested by the Acquired Companies in good faith (and appropriate reserves established therefor), and if not paid, have been properly recorded, (b) all accounts receivable of the Business that are reflected in the Unaudited Financial Information (except as specifically set forth therein) and all accounts receivable of the Business arising since the date of the Latest Balance Sheet are held by one or more of the Acquired Companies and are valid obligations and have arisen only from bona fide arm's length transactions in the ordinary course of business and are not subject to defenses, credits, setoffs or counterclaims and (c) all of the outstanding receivables deemed uncollectible have been reserved against in the Unaudited Financial Information in accordance with GAAP.

Product Warranties; Product Liability. Except as would not reasonably be expected to be, individually or in the aggregate, material to the Acquired Companies or the Business, taken as a whole, (a) each product or service manufactured, provided, developed, sold, leased, licensed or delivered by or on behalf of any Acquired Company or, to the extent related to the Business, Seller or any of its Affiliates meets, and at all times since January 1, 2021 has met, in all material respects, all standards for quality and workmanship prescribed by applicable Legal Requirement and industry standards; (b) (i) no material product liability or similar claims or Proceedings and (ii) no material claims or Proceedings for breach of contractual warranties or due diligence or other obligations under Contracts or other material claims or Proceedings for breach of contract, in each case, relating to any product or service manufactured, provided, developed, sold, leased, licensed or delivered by or on behalf of any Acquired Company or, to the extent related to the Business, Seller or its Affiliates are pending or, to Seller's Knowledge, threatened since January 1, 2021; (c) since January 1, 2021, there have been no recalls (mandatory or voluntary), market withdrawals, market replacement or post-sale warning of any product manufactured, provided, developed, sold, leased, licensed or delivered of any Acquired Company or, to the extent related to the Business, Seller or any of its Affiliates (and there is no current plan for any recall, market withdrawal, market replacement or post-sale warning; nor is any such recall, market withdrawal, market replacement, or post-sale warning currently under consideration); and (d) no Acquired Company (or, to the extent related to the Business, Seller or any of its Affiliates) has received any written communication from any Governmental Authority regarding a recall, withdrawal, suspension or discontinuance of any such product since January 1, 2021.

### 4.23 Government Contracts.

(a) Except as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, since January 1, 2021: (i) each current Government Contract is in full force and effect, constitutes a legal, valid and binding agreement enforceable in accordance with its terms and was awarded in compliance with applicable Legal Requirements; (ii) the Acquired Companies and Seller and any of its Subsidiaries have complied, in all material respects, with the terms and conditions of each Government Contract and all Legal Requirements applicable to each Government Contract; (iii) the representations, certifications and warranties made by each Acquired Company and by Seller or any of its Subsidiaries in writing with respect to each Government Contract were accurate, in all material respects, as of

their respective effective dates and each Acquired Company and Seller or any of its Subsidiaries has made any reasonably required updates to such representations, certifications and statements; (iv) all invoices and claims for payment, reimbursement or adjustment submitted by any Acquired Company or by Seller or any of its Subsidiaries in connection with any payments from any Governmental Authority pursuant to any Government Contract were current, accurate and complete in all material respects as of their respective submission dates; (v) no terminations for default or cause, cure notices, or "show cause" notices have been issued in writing to any of the Acquired Entities with respect to any Government Contract; (vi) there are no material outstanding claims, requests for equitable adjustment, audits, or disputes relating to any Government Contract; (viii) there have been no written document requests, subpoenas, search warrants or civil investigative demands addressed to any of the Acquired Companies or Seller or any of its Subsidiaries in connection with or related to any Government Contract; and (ix) there have been no written requests by a Governmental Authority for a contract price adjustment based on a claimed disallowance by an applicable Governmental Authority or claim of defective pricing relating to any Government Contract.

- (b) Except as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, since January 1, 2021, none of the Acquired Companies or, to the extent related to the Business, Seller or any of its other Affiliates nor any of their directors, officers, agents or employees has been debarred, suspended, or proposed for debarment or otherwise excluded from participation in the award of Government Contracts by any Governmental Authority.
- Effect, since January 1, 2021, no Acquired Company nor Seller nor any of its Subsidiaries (i) has conducted or initiated any internal investigation, or made a voluntary disclosure or mandatory disclosure under the Federal Acquisition Regulation mandatory disclosure provisions (48 C.F.R. § 52.203-13) or similar regulatory requirements with respect to any alleged misstatement, or omission, arising under or relating to any Government Contract, or (ii) has received credible evidence of a violation of a federal criminal Legal Requirement involving the fraud, conflict of interest, bribery, or gratuity provisions found in Title 18 of the U.S. Code, a violation of the civil False Claims Act (31 U.S.C. §§ 3729-3733), or a significant overpayment, in connection with the award, performance, closeout, or receipt of any Government Contract.
- (d) Except as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, since January 1, 2021, (i) all material costs charged to any Government Contract have been allowable, allocable, reasonable, and reimbursable in accordance with applicable cost principles and the terms of the underlying Government Contracts; (ii) no Acquired Company nor Seller nor any of its Subsidiaries has submitted any certified cost or pricing data that was not current, accurate or complete in all material respects as of the certification date in connection with any Government Contract; (iii) no Governmental Authority has, to Seller's Knowledge, threatened to assess against any Acquired Company or Seller or any of its Subsidiaries any material penalties, credits, or other similar contractual offsets pursuant to any performance-based Government Contract that contains service level

arrangements or performance guarantees; (iv) none of the Acquired Companies nor Seller nor any of its Subsidiaries has sold a product or service to any customer at a price that would invoke any "most favored customer" pricing provision under any Government Contract, except as in accordance with the terms of such Government Contract; and (v) each of the Acquired Companies and Seller and any of its Subsidiaries has maintained adequate systems of internal controls appropriate for the operations of the Business that are in compliance in all material respects with all relevant and applicable requirements of the Government Contracts.

4.24 Insurance. Schedule 4.24 of the Seller Disclosure Schedule contains a true and complete list of all material insurance policies currently maintained by or for the benefit of the Business or the Acquired Companies (other than any insurance policy comprising a Benefit Plan) ("Insurance Policies"), true and complete copies of which have been made available to Purchaser, other than the Insurance Policies in the name of the Seller Group. The Insurance Policies are sufficient for compliance in all material respects by the Acquired Companies or, to the extent related to the Business, Seller or any of its other Affiliates with all Material Contracts. The Insurance Policies are in full force and effect; all premiums due and payable under the Insurance Policies have been paid in full; the limits of each Insurance Policy are fully in place without any exhaustion or erosion; neither Seller nor any of its Affiliates (including the Acquired Companies) has received notice of cancellation, non-renewal, termination or material modification of any terms and conditions with respect to any of the Insurance Policies; neither Seller nor any of its Affiliates (including the Acquired Companies) is in material breach or default with respect to its obligations under any of the Insurance Policies relating to the Business or the Acquired Companies, including for which the applicable insurer has questioned, disputed or denied coverage or reserved rights. To Seller's Knowledge, no event has occurred that would reasonably be expected to result in the cancellation, non-renewal, termination or material limitation of coverage under any Insurance Policy.

## 4.25 <u>Trade Controls</u>.

- (a) Each of the Acquired Companies (and, to the extent related to the Business, Seller and its other Affiliates) and their respective officers, directors, and, to Seller's Knowledge, employees, agents or other third parties acting on their behalf, are, and have been for the past five (5) years, in compliance with applicable Trade Controls in all material respects. Each of the Acquired Companies (and, to the extent related to the Business, Seller and its other Affiliates) have obtained all applicable material import and export licenses as well as all other necessary material licenses, consents, notices, waivers, approvals, orders, authorizations, and declarations, and completed all necessary material registrations and filings, required under applicable Trade Controls.
- (b) None of the Acquired Companies (or, to the extent related to the Business, Seller or any of its other Affiliates) is conducting, or has in the past five (5) years conducted any business, directly or indirectly, with, in, or involving any Sanctioned Person, Restricted Person, or Sanctioned Jurisdiction.

- (c) None of the Acquired Companies (or, to the extent related to the Business, Seller or any of its other Affiliates), or any of their respective officers, directors, employees, or agents, is, or has been for the past five (5) years, a Sanctioned Person or a Restricted Person.
- (d) In the past five (5) years, none of the Acquired Companies (or, to the extent related to the Business, Seller or any of its other Affiliates) has (i) made any voluntary or involuntary disclosure to any Governmental Authority with respect to any alleged act or omission arising under or relating to any non-compliance with any Trade Controls, (ii) been the subject of a past, current, pending or threatened investigation, inquiry or enforcement proceeding for a material violation of Trade Controls, or (iii) received any written notice, request, penalty, or citation for any actual or potential non-compliance with Trade Controls.
- No Other Representations. It is the explicit intent of each party hereto, and Purchaser expressly acknowledges and agrees, that, in connection with this Agreement, none of Seller nor any of its Affiliates has made, is making or has authorized any Person to make any representation or warranty whatsoever, express or implied, except those representations and warranties expressly set forth in this Article IV or in the other Transaction Documents. Except as expressly and specifically set forth in the representations and warranties made by Seller in this Article IV and the representations and warranties in the other Transaction Documents, each of Seller, its Affiliates, and its Representatives expressly disclaims any and all other representations and warranties, whether express or implied, and in entering into this Agreement, Purchaser expressly disclaims reliance on any and all such other representations or warranties, express or implied, including those relating to Seller, the Acquired Companies, the Business, the IP Seller IP or the Transactions, or any of their financial condition, business, operations, results of operations, properties, assets, liabilities or prospects, or any estimate, projection, prediction, data, financial information, teaser, confidential information presentation or any other materials or information provided or addressed to Purchaser, its Affiliates or its and their Representatives, including with respect to the accuracy or completeness of any such information. Except as expressly and specifically set forth in the representations and warranties made solely by Seller in this Article IV and the representations and warranties in the other Transaction Documents, each of Seller, its Affiliates, and its Representatives has not made, is not making, has not authorized anyone to make and specifically disclaims any statement, representation or warranty of merchantability, usage, suitability or fitness for any particular purpose with respect to assets of the Business (including the IP Seller IP), any part thereof, the workmanship thereof, and the absence of any defects therein, whether latent or patent, it being understood that such assets are being acquired "as is, where is" on the Closing Date, and in their present condition.

### ARTICLE V

# REPRESENTATIONS AND WARRANTIES OF PURCHASER

Purchaser hereby represents and warrants to Seller as follows:

5.1 <u>Authority; Enforceability</u>.

- (a) Purchaser has the requisite corporate power and authority to execute and deliver this Agreement, and each of the other Transaction Documents to which it is or will be a party, to perform its obligations hereunder and under each of the other Transaction Documents to which it is or will be a party, and to consummate the Transactions in accordance with the terms of this Agreement, and each of the other Transaction Documents to which it is or will be a party and performance by Purchaser of this Agreement and each of the other Transaction Documents to which it is or will be a party and the consummation of the Transactions contemplated by each of the Transaction Documents to which it is or will be a party, have been duly and validly authorized by all necessary corporate action on the part of Purchaser and such authorization has not been subsequently modified or rescinded. No additional corporate or shareholder authorization is necessary for the execution, delivery or performance by Purchaser of this Agreement or any other Transaction Document to which it is or will be a party or the consummation by Purchaser of the Transaction in accordance with the terms of this Agreement and each other Transaction Document to which it is a party.
- (b) This Agreement and each of the other Transaction Documents to which Purchaser is or will be a party have been duly and validly executed and delivered by Purchaser and constitutes, assuming due authorization, execution and delivery of this Agreement and the other Transaction Documents by Seller and the other parties thereto, a valid and binding legal obligation of Purchaser, enforceable against Purchaser in accordance with the terms hereof, subject to the Bankruptcy and Equity Exception. Assuming due authorization, execution and delivery of each of the other Transaction Documents to which Purchaser is or will be a party by the other parties thereto, each of the other Transaction Documents will constitute a valid and binding legal obligation of Purchaser, enforceable against Purchaser in accordance with the terms thereof, in each case, subject to the Bankruptcy and Equity Exception.

### 5.2 Non-Contravention; Consents.

- (a) The execution and delivery by Purchaser of this Agreement and each other Transaction Document to which it is or will be a party does not, and the performance by Purchaser of this Agreement and each other Transaction Document to which it is or will be a party will not, require any Consent or Permit of, registration, declaration or filing with, or notification to, any Governmental Authority, except (i) under applicable Antitrust Laws or Investment Screening Laws, (ii) under the applicable requirements of the Exchange Act or applicable blue sky laws, (iii) compliance with any Permits relating to the Business, (iv) for such other Consents, filings or notifications, the failure of which to make or obtain would not reasonably be expected to have, individually or in the aggregate, a Purchaser Material Adverse Effect.
- (b) The execution and delivery by Purchaser of this Agreement and each of the other Transaction Documents to which it is or will be a party does not, and the consummation of the Transactions will not, (i) conflict with, violate or constitute a breach of, or require notice, consent or waiver under any provision of the organizational documents of Purchaser, (ii) conflict with, violate or constitute a breach of, or require any notice, consent or waiver under any Legal Requirement applicable to Purchaser, except with respect to the required filings and approvals

set forth in Sections 5.2(a)(i) and (ii) and in Schedule 5.2(b)(ii) of the Purchaser Disclosure Schedule or (iii) result in a breach of, constitute (with or without due notice or lapse of time or both) a default under, result in the creation or acceleration (or loss of benefit from) of any rights or obligations under, or create in any party the right to, accelerate, terminate, modify or cancel, or require any notice, consent or waiver under, any Contract to which Purchaser is a party, except, in the case of the foregoing clauses (ii) or (iii), as would not reasonably be expected to have, individually or in the aggregate, a Purchaser Material Adverse Effect.

- 5.3 <u>Organization</u>. Purchaser is duly incorporated, validly existing and in good standing under the Legal Requirements of the state of Delaware, except as would not reasonably be expected to have, individually or in the aggregate, a Purchaser Material Adverse Effect. Purchaser has all necessary corporate power and authority to conduct its business in the manner in which it is currently being conducted, except where the absence of such power and authority to conduct its business would not reasonably be expected to materially impair or materially delay Purchaser from consummating the Transactions or otherwise prevent Purchaser from performing in all material respects its obligations hereunder or under any other Transaction Document to which it is or will be a party.
- 5.4 <u>Litigation</u>. Since January 1, 2021, there are no Proceedings pending, or, to Purchaser's Knowledge, threatened against Purchaser, any of Purchaser's material assets or properties or any of its present or former officers, employees or directors in their capacity as such that, if adversely decided, would reasonably be expected to have, individually or in the aggregate, a Purchaser Material Adverse Effect.
- 5.5 <u>Securities Matters.</u> Purchaser is an "accredited investor" (as such term is defined in Rule 501 of Regulation D under the Securities Act). The Shares are being acquired by Purchaser (or Purchaser's designee) for its own account, and not with a view to, or for the offer or sale in connection with, any public distribution or sale of the Shares or any interest in them. Purchaser has sufficient knowledge and experience in financial and business matters to be capable of evaluating the merits and risks of its investment in the Shares, and Purchaser is capable of bearing the economic risks of such investment, including a complete loss of its investment in the Shares. Purchaser acknowledges that the Shares are not registered under the Securities Act, or any applicable provincial, state and foreign securities Legal Requirement, and agrees that the Shares may not be sold, transferred, offered for sale, pledged, hypothecated or otherwise disposed of except pursuant to a registered offering in compliance with, or in a transaction exempt from, the registration requirements of the Securities Act, or any applicable provincial, state and foreign securities Legal Requirement.

## 5.6 <u>Financing</u>.

(a) Purchaser has delivered to Seller true, complete and correct fully executed copies of (i) the debt commitment letter (including all exhibits, schedules and annexes thereto and as amended or modified from time to time in accordance with its terms and to the extent permitted by Section 6.8(b), the "Debt Commitment Letter"), dated as of the date hereof, from the Financing Entities, and (ii) that certain Incremental Assumption and Amendment Agreement and Amendment, dated as of the date hereof (including all exhibits, schedules and annexes

thereto and as amended and restated, supplemented or otherwise modified from time to time in accordance with its terms and to the extent permitted by Section 6.8(b), the "Credit Agreement Amendment" and together with the Debt Commitment Letter, the "Debt Commitment Documents"), among the Purchaser, the Financing Entities and the other parties thereto, and (iii) any fee letters related to the foregoing (redacted, in the case of such fee letters, solely with respect to the amounts and percentages of the fees and other economic terms that are customarily redacted in connection with similar financings; provided that such redactions would not be reasonably expected to conceal any term that could terminate, reduce to below the Required Amount the amount of, or adversely affect the conditionality or enforceability of the Debt Financing) (such Debt Commitment Documents and each such fee letter, collectively, the "Debt Financing Commitment"), pursuant to which the applicable Financing Entities have committed, on the terms and subject to the conditions set forth therein, to provide debt financing in the amounts set forth therein to Purchaser for the purposes of, among other things, financing the Transactions and related fees and expenses.

As of the date hereof, the Debt Financing Commitment is a legal, valid, binding and enforceable obligation of Purchaser and (to Purchaser's knowledge) each of the other parties thereto (subject to the Bankruptcy and Equity Exception). As of the date hereof, the Debt Financing Commitment is in full force and effect, and the Debt Financing Commitment has not been terminated, withdrawn, rescinded or otherwise amended or modified in any respect and the commitments contained therein have not been terminated, withdrawn, rescinded, reduced or otherwise amended or modified in any respect (and no such termination, withdrawal, rescission, reduction, amendment or modification thereof is contemplated other than an amendment or modification solely to join additional Financing Sources thereto and to make certain other amendments, including to upsize the Purchaser's revolver, modify existing financial maintenance covenants and make certain other amendments and modifications relating to the Purchaser's revolver, provided that such amendments are not Prohibited Modifications). As of the date hereof, there are no side letters or other Contracts, agreements, arrangements or understandings of any kind (written or oral) to which Purchaser or (to Purchaser's knowledge) any of its Affiliates is a party that are directly or indirectly related to the Debt Financing Commitment or the Debt Financing, other than customary engagement letters and fee credit letters. Purchaser has fully paid any and all commitment fees or other fees or expenses in connection with the Debt Financing that are payable on or prior to the date hereof. As of the date hereof, there are no conditions precedent or other contingencies related to the funding of the Required Amount, other than as expressly set forth in the Debt Commitment Documents. As of the date hereof, no event has occurred as of the date hereof which, with or without notice, lapse of time or both, (i) constitutes, or could reasonably be expected to constitute, a default or breach on the part of Purchaser or, to Purchaser's Knowledge, any other party thereto under any term or condition of the Debt Financing Commitment or (ii) could reasonably be expected to (A) make any of the representations of Purchaser or, to Purchaser's Knowledge, any other party thereto set forth in the Debt Financing Commitment inaccurate in any respect, (B) result in any of the conditions in the Debt Financing Commitment not being satisfied on a timely basis or (C) otherwise result in the Debt Financing in an amount equal to at least the Required Amount not being available in accordance with the terms of the Debt Financing Commitment. As of the date hereof, no Financing Entity has notified Purchaser of its intention to terminate the Debt Financing

Commitment or not to provide all or any portion of the Debt Financing. Assuming (1) the accuracy of the representations and warranties set forth in Article IV in all material respects and (2) the performance by Seller of the covenants and agreements contained in this Agreement in all material respects, the Debt Financing, when funded in accordance with the Debt Financing Commitment, shall provide Purchaser with cash proceeds on the Closing Date in an amount that, together with cash then otherwise immediately available to Purchaser, is sufficient for the satisfaction of all of their respective obligations under the Transaction Documents and the Debt Financing Commitment, including the (i) payment of the Closing Purchase Price (as well as the Final Purchase Price) and the repayment or refinancing of any outstanding Final Closing Indebtedness contemplated or required to be repaid or otherwise satisfied in connection with the consummation of the transactions contemplated hereby, (ii) payment of any and all fees and expenses of or required to be paid by Purchaser or any of its Affiliates on or prior to the Closing Date in connection with the transactions contemplated hereby and by the Debt Financing and (iii) satisfaction of all other payment obligations of Purchaser or any of its Affiliates contemplated hereunder and under the other Transaction Documents and under the Debt Financing Commitment required to be made at or in connection with the Closing (the amounts contemplated by clauses (i) through (iii), the "Required Amount").

- (c) Purchaser acknowledges and agrees that, notwithstanding anything to the contrary in this Agreement, the consummation of the Debt Financing shall not be a condition to the obligation of Purchaser to consummate the transactions contemplated hereby.
- 5.7 Solvency. As of the Closing, after giving effect to any indebtedness being incurred on such date in connection herewith (including the funding of the full amount of the Debt Financing), and assuming (i) satisfaction of the conditions to Purchaser's obligations to consummate the Closing set forth in Article X and (ii) that the Required Information presents fairly in all material respects the consolidated financial condition and consolidated results of operations of the Acquired Companies as of the end of the periods thereby and as of the Closing Date, neither Purchaser nor the Acquired Companies, on a consolidated basis, will (a) be insolvent (either because its financial condition is such that the sum of its debts (including a reasonable estimate of the amount of all contingent liabilities) is greater than the fair value of its assets, or because the present fair saleable value of its assets will be less than the amount required to pay its probable liability on its debts as they become absolute and matured), (b) have unreasonably small capital with which to engage in its business or (c) have incurred or plan to incur debts beyond its ability to pay as they become absolute and matured. No transfer of property is being made and no obligation is being incurred in connection with the Transactions with the intent to hinder, delay or defraud either present or future creditors of Purchaser, its Affiliates or the Acquired Companies.
- 5.8 <u>Brokers</u>. Other than with respect to fees or commissions that will be borne solely by Purchaser and its Affiliates, Purchaser and its Affiliates have not retained any broker, finder, agent, financial advisor, investment banker or similar intermediary who has acted on behalf of Purchaser and its Affiliates and no broker, finder, agent, financial advisor, investment banker or similar intermediary is entitled to, nor have Purchaser and its Affiliates incurred any liability or

obligation for any brokerage fees, commissions or finders' fees, financial advisor fees or other similar fees, in each case, with respect to this Agreement or the Transactions.

- R&W Insurance Policy. Attached hereto as Exhibit D is a true, accurate and complete copy of the binder agreement (the "R&W Binder Agreement") for the buyer-side representations and warranties insurance policy(ies) (the "R&W Insurance Policy") including the form of the R&W Insurance Policy. The R&W Binder Agreement is in full force and effect and is a legal, valid, binding and enforceable obligation of Purchaser and the insurer(s) party thereto, subject to the Bankruptcy and Equity Exception. Purchaser is solely responsible for all costs to procure, maintain and make claims under the R&W Insurance Policy, including all premiums, retention amounts, Taxes, expenses and costs of any nature whatsoever. The R&W Insurance Policy: (a) names Purchaser, or an Affiliate thereof, as an insured thereunder; (b) contains a provision whereby the insurer(s) expressly waives, and agrees not to pursue, directly or indirectly, any subrogation, contribution or any other rights against Seller. its Affiliates and their respective current and former Representatives, Affiliates and fiduciaries (or the functional equivalent of any such position) (collectively, "Seller Parties") based upon, arising out of, or relating to this Agreement or the transactions contemplated hereby, or the negotiation, execution or performance of this Agreement, other than in the case of Fraud by any such Seller Party, and then only to the extent of such Fraud by such Seller Party; and (c) names the Seller Parties as express third-party beneficiaries in respect of the waiver in clause (b) of this Section 5.9. The R&W Insurance Policy does not increase the liability hereunder of the Seller Parties. As of the date of this Agreement, neither Purchaser nor any of its Affiliates has received or given any written or, to Purchaser's Knowledge, oral notice or claim of any material breach or violation of, or default under, the R&W Binder Agreement.
- 5.10 <u>Pending Transactions</u>. Neither Purchaser nor any of its Affiliates is a party to any pending transaction or contemplating any transaction, in each case, to acquire (by merging or consolidating with, by purchasing a substantial portion of the assets of or equity in, or by any other similar transaction) any Person (or business division or unit thereof), where, in the case of such contemplated transaction, the entering into of a definitive agreement relating to or, in either case, the consummation of such transaction would reasonably be expected to (a) impose any material delay in the obtaining of, or materially increase the risk of not obtaining, any Consents, Orders or Governmental Approvals necessary to consummate the Transactions or the expiration or termination of any applicable waiting period, (b) materially increase the risk of any Governmental Authority entering an Order prohibiting the consummation of the Transactions or (c) otherwise materially delay the consummation of the Transactions.
- 5.11 <u>Investigation</u>. Purchaser is an informed and sophisticated purchaser and has such knowledge and experience in financial and business matters that it is capable of evaluating the merits and risks of its purchase of the Acquired Companies and the IP Seller IP, and its consummation of the Transactions. Purchaser acknowledges and agrees that it has (a) conducted such inquiries and independent investigations into the Business, the IP Seller IP and the Acquired Companies as it has deemed sufficient to make an independent and informed decision with respect to the execution, delivery and performance of this Agreement and the consummation of the Transactions and (b) been furnished with or afforded adequate access to and the adequate

opportunity to review the books, records, facilities and personnel of Seller, IP Seller and the Acquired Companies for purposes of conducting a due diligence investigation of the Business, the IP Seller IP and the Acquired Companies. Purchaser expressly acknowledges and agrees that none of Seller, its Affiliates nor any other Person has made, makes or is authorized to make any representations or warranties to Purchaser, express or implied, relating to Seller, the Acquired Companies, the Seller IP, the Business or the Transactions other than those representations and warranties of Seller expressly set forth in Article IV or the other representations and warranties in the other Transaction Documents, and that none of Seller, its Affiliates nor any other Person shall be subject to any liability or any claim by Purchaser in respect of such other representations or warranties. In making its determination to proceed with the Transactions and acquire the Shares and the IP Seller IP, Purchaser expressly acknowledges and agrees that it has relied exclusively on its own independent investigation and the representations and warranties of Seller set forth in Article IV and the other representations and warranties in the other Transaction Documents, and that it is not relying on and expressly disclaims reliance on any other statement, representation or warranty made by Seller, its Affiliates or any other Person, whether oral or written, express or implied, including those relating to Seller, the Acquired Companies, the IP Seller IP, the Business or the Transactions, or any of their financial condition, business, operations, results of operations, properties, assets, liabilities or prospects, or any estimate, projection, prediction, data, financial information, teaser, confidential information presentation or any other materials or information provided or addressed to Purchaser, its Affiliates or its and their Representatives or any other Person, including with respect to the accuracy or completeness of any such information.

- 5.12 <u>Disclaimer Regarding Projections</u>. In connection with Purchaser's due diligence investigation of the Business, the IP Seller IP and the Acquired Companies, Purchaser has received from Seller, its Affiliates and its and its Affiliates' Representatives certain projections and other forecasts, whether written or oral, related to the Business, the IP Seller IP, Seller and the Acquired Companies. Purchaser expressly acknowledges and agrees that (a) there are uncertainties inherent in attempting to make such projections and forecasts, (b) Purchaser is familiar with such uncertainties and is taking full responsibility for making its own evaluation of the adequacy and accuracy of all forecasts, predictions, projections estimates or other materials, documents or information relating to the Business and the IP Seller IP, (c) Purchaser is not relying upon such projections and other forecasts, and (d) Purchaser shall have no claim against Seller and its Affiliates with respect to any inaccuracy, misstatement or omission with respect to any such financial forecasts, predictions or financial projections, whether written or oral, made available to Purchaser, its Affiliates or its and their Representatives or any other Person in any data room, confidential information presentation or any other materials or information provided or addressed to Purchaser, its Affiliates or its and their Representatives.
- 5.13 No Other Representations. Except as expressly provided in this Article V and the other representations and warranties made in the other Transaction Documents, Purchaser does not make any other express or implied representations or warranties to Seller, and, in entering into this Agreement, Seller expressly acknowledges and agrees that it is not relying on any statement, representation or warranty, other than those representations and warranties set forth in

this Article V and the other representations and warranties made in the other Transaction Documents.

#### ARTICLE VI

### COVENANTS OF THE PARTIES

# 6.1 <u>Conduct of the Business Prior to Closing.</u>

- Except (i) as expressly required by this Agreement or any other Transaction Document, (ii) as required by Legal Requirement, (iii) for matters set forth on Schedule 6.1(a) of the Seller Disclosure Schedule or (iv) with the written consent of Purchaser (which consent shall not be unreasonably withheld, conditioned or delayed; it being understood that the failure of Purchaser to respond to a written request for consent within five (5) Business Days thereafter shall be deemed to constitute consent for all purposes hereunder), from the date of this Agreement until the earlier of the Closing or the termination of this Agreement pursuant to its terms, Seller shall, and shall cause its Affiliates to in each case, exclusively with respect to the Acquired Companies and the Business, (A)(x) operate the Business in the ordinary course of business in all material respects and (y) use commercially reasonable efforts to (1) maintain the assets and properties of the Business, (2) preserve the current relationships of the Business with customers, suppliers, distributors, licensors, licensees, contractors, employees, Governmental Authorities and other business relations; (3) preserve the goodwill and ongoing operations of the Business and (4) comply in all material respects with all Legal Requirements (provided, however, that (1) no action or inaction by Seller or any of its Affiliates with respect to any matters specifically addressed by Section 6.1(a)(B) shall be deemed to be a breach of this Section 6.1(a)(A) unless such action or inaction would constitute a breach of Section 6.1(a)(B), (2) the failure of Seller or any of its Affiliates (including the Acquired Companies) to take any action prohibited by Section 6.1(a)(B) shall in no circumstances be deemed a breach of this Section 6.1(a)(A) and (3) Purchaser's written consent (which may be email) with respect to any specific action or matter pursuant to Section 6.1(a)(B) shall be deemed to constitute consent with respect to such action or matter for all purposes under this <u>Section 6.1(a)</u>) and (B) not do any of the following:
  - (i) modify, amend or otherwise change the organizational documents of any Acquired Company in a manner adverse to Purchaser;
  - (ii) split, reverse split, combine, subdivide or reclassify any shares of capital stock or other voting or equity interests of any Acquired Company (including the Shares) or effect any recapitalization or like change in the capitalization of any Acquired Company, or issue any other security in respect of, in lieu of or in substitution for any shares of capital stock of or other voting or equity interests in any Acquired Company (including the Shares);
  - (iii) redeem, purchase or otherwise acquire any shares of capital stock of or other voting or equity interests in any Acquired Company (including the Shares), or issue, deliver, sell, pledge, encumber, transfer or grant (A) any shares of capital stock of

or other voting or equity interests in any Acquired Company (including the Shares) or (B) any warrant, option, right, agreements, "phantom" stock right, stock appreciation right, stock-based performance unit, convertible, exercisable or exchangeable securities or rights or any other commitment or undertaking (1) pursuant to which any Acquired Company is or may become obligated to issue, deliver, sell, transfer or grant (x) any shares of capital stock of or other voting or equity interests in any Acquired Company (including the Shares) or (y) any security convertible into, or exercisable or exchangeable for, any shares of capital stock of or other voting or equity interests in any Acquired Company (including the Shares), or (2) pursuant to which any Acquired Company is or may become obligated to issue, deliver, sell, transfer or grant any such warrant, option, right, unit, security, commitment or undertaking described in the foregoing clause (1) that gives any Person the right to receive any benefit or right similar to any rights enjoyed by or accruing to the holders of any shares of capital stock of or other voting or equity interests in any Acquired Company (including the Shares);

- (iv) except for (A) indebtedness that will be settled in full or terminated or canceled at or before the Closing, (B) indebtedness for borrowed money incurred under credit facilities of any Acquired Company in existence as of the date hereof and (C) indebtedness in respect of currency or interest rate hedges entered into to hedge currency or interest rate risks arising in the ordinary course of business and not for speculative purposes, (1) incur in excess of \$1,000,000 of indebtedness for borrowed money outstanding at any time; (2) enter into any Contract involving financing or borrowing of money; or (3) assume, guarantee or endorse the obligations of any Person if, in each of the foregoing (1), (2) and (3), such obligations would be obligations of the Acquired Companies following the Closing or would be Assumed Liabilities;
- (v) permit any of the material assets of the Acquired Companies or the Business to become subjected to any Encumbrance other than (A) those Encumbrances which will be removed at or prior to the Closing or (B) Permitted Encumbrances:
- (vi) with respect to any Acquired Company, fail to maintain its corporate existence, adopt or enter into any plan of complete or partial liquidation, dissolution, restructuring, recapitalization or other reorganization or merge or consolidate with any other Person, or, with respect to any Acquired Company or the Business, enter into any joint venture or similar venture with any other Person;
- (vii) permit any Acquired Company or the Business to purchase any securities or make any investment or capital contribution in any Person, either by purchase of stock or securities, contributions to capital or asset transfers, or acquire (either by merger, consolidation or otherwise), direct or indirect control over, any Person, business, business organization or division thereof other than any such purchases, investments, capital contributions or acquisitions by an Acquired Company for which the aggregate consideration paid (A) in any individual transaction is not in excess of \$2,500,000 or (B) in the aggregate is not in excess of \$5,000,000;

- (viii) permit any Acquired Company to loan or advance any amount other than (A) to another Acquired Company or (B) loans or advances to employees of any Acquired Company for travel and business expenses in the ordinary course of business;
- (ix) except for the sale of the Businesses' products or services in the ordinary course of business, sell, transfer, lease (other than renewals of existing leases), sublease or otherwise dispose of any material properties or assets of (A) the Acquired Companies or Business having an aggregate value in excess of \$1,000,000 or (B) the Acquired Companies to any member of the Seller Group;
- (x) commit to make any capital expenditures for the Business to be made after the Closing in excess of \$1,000,000, individually, or \$2,500,000, in the aggregate, other than as set forth in the capital budget of the Business made available to Purchaser prior to the date hereof;
- (xi) waive, abandon, or otherwise dispose of any rights in or to any material Acquired Company IP, (A) other than the expiration of Acquired Company Registered IP in accordance with the applicable statutory term and (B) with respect to Acquired Company IP that Seller has determined in the ordinary course of business consistent with past practice (including its policies and procedures regarding the protection and maintenance of Intellectual Property) is no longer material:
- (xii) disclose any material Trade Secrets of the Business, other than to any Persons that are subject to a confidentiality or non-disclosure obligations protecting against further disclosure thereof;
- (xiii) declare or set aside any dividends or distributions on any shares of capital stock of or other voting or equity interests of any Acquired Company (in cash or in kind) to the extent such dividends or distributions (A) are payable at or after the Closing or (B) payable in anything other than cash;
- (xiv) initiate any Proceeding relating to the Business outside the ordinary course of business or enter into any settlement or release with respect to any Proceeding relating to the Business to the extent constituting an Assumed Liability other than (A) any settlement or release that contemplates only the payment of money of an amount less than \$250,000 individually or \$1,000,000 in the aggregate (net of any insurance coverage) and does not involve any equitable relief imposed against the Acquired Companies or the Business (excluding confidentiality, non-disparagement, and other similar customary provisions) and results in a release of the claims giving rise to such Proceeding, (B) any settlement or release involving the payment of liabilities to the extent reflected or reserved against in the reserved against in the Latest Balance Sheet or that are fully covered by insurance policies of Seller or its Subsidiaries (including the Acquired Companies) or (C) to the extent such Proceeding and any associated liabilities are Retained Liabilities;

- (xv) other than in the ordinary course of business, for any Acquired Company (A) make, change, or revoke any Tax election, (B) make any change to any Tax or accounting method or system of internal accounting control of the Business, (C) change any annual Tax accounting period, (D) file any amended Tax Return, (E) enter into any material closing agreement or settle any material Tax claim or assessment, (F) surrender any right to claim a material refund of Taxes, or (G) consent to any extension or waiver of the limitation period applicable to any material Tax claim or assessment;
  - (xvi) accelerate the collection of accounts receivable, materially outside the ordinary course of business;
- (xvii) terminate, cancel, modify, waive any right under, amend, or assign any Material Contract (other than any Specified Material Contract), other than in the ordinary course of business, or enter into any Contract that would have constituted a Material Contract (other than a Specified Material Contract) if entered into prior to the date hereof, other than new Contracts that are entered into in the ordinary course of business;
  - (xviii) modify, renew, extend, or enter into any Collective Bargaining Agreement;
- (xix) (A) acquire any material real property interest, (B) incur, grant, suffer, create or assume any Encumbrance (other than Permitted Encumbrances) with respect to Real Property of the Business, or (C) sell, lease, or transfer or dispose of any Real Property of the Business;
- (xx) except as may be required by the terms of an Acquired Company Benefit Plan, Seller Benefit Plan or Collective Bargaining Agreement (A) increase or promise to increase (i) the compensation of any Business Employee, other than, (I) for any Business Employee at or above the level of director who the Seller or the applicable Acquired Company determines, in their sole discretion, has received a competitive offer of employment from any other company, increases in the annual target compensation for any such Business Employee up to an amount equal to 125% of such Business Employee's annual target compensation, or (II) for any Business Employee below the level of director, increases in annual base salary and proportionate target bonus opportunity in the ordinary course of business not to exceed 1% in the aggregate (and proportionate target bonus opportunity) or (ii) the severance, termination pay or benefits of any Business Employee, except for increases under Seller Benefit Plans (other than any severance plan, program, practice or policy) that provide for health, welfare or retirement benefits and that are applied in a substantially similar manner to all similarly situated employees of Seller and its Affiliates who participate in such plans, (B) enter into, adopt, amend or terminate any Acquired Company Benefit Plan, other than (i) renewals for health and welfare benefits, that would not reasonably be expected to result in a material liability to Purchaser and its Affiliates (including, following the Closing, the Acquired Companies) and (ii) entering into employment agreements or offer

letters with Business Employees hired after the date hereof that provide for terms and conditions consistent with past practice, (C) hire or terminate (other than for cause) any Business Employee at or above the level of director, other than to hire a Business Employee to fill any position or role vacancies (provided, that, the annual target compensation for any such newly hired Business Employee shall not exceed 125% of the annual target compensation of the Business Employee who previously occupied the applicable position or role), (D) transfer the employment of any Business Employee outside of the Business other than where such Business Employee applied for, and was selected in a competitive process not targeted at Business Employees, a position within the Seller Group that is outside of the Business, (E) transfer the sponsorship of any Acquired Company Benefit Plan to Seller or one of its Affiliates (other than the Acquired Companies) or cause any Acquired Company to accept the transfer of the sponsorship of any Seller Benefit Plan; (F) recognize or certify any labor union, labor organization, works council, or group of employees as the bargaining representative for any Business Employees; or (G) implement or announce any employee layoffs, plant closings, or other actions covering Business Employees that could implicate the Worker Adjustment and Retraining Notification Act of 1988 and any similar Legal Requirement; or

- (xxi) enter into any legally binding commitment with respect to any of the foregoing.
- (b) Notwithstanding anything to the contrary in this Agreement, nothing in this Section 6.1 shall prohibit or otherwise restrict in any way the operation of the business of Seller or its Subsidiaries, except solely with respect to the conduct of the Business. Notwithstanding any provision herein to the contrary, prior to (but not after) the Benchmark Time, without the consent of Purchaser, Seller and its Subsidiaries will be permitted to (i) declare and pay dividends and distributions of, or otherwise transfer or advance, to Seller or any Subsidiary thereof, any Cash (including in connection with any "cash sweep" or cash management practices), (ii) make any payments under, or repay (in part or in full), refinance or replace, any Indebtedness or intercompany receivables, payables, loans and balances and (iii) take any action contemplated pursuant to Sections 6.4 or 6.8.
- (c) Nothing contained in this Agreement is intended to give Purchaser, directly or indirectly, the right to control or direct the operations of the Business or any of the Acquired Companies prior to the Closing. Prior to the Closing, Seller shall, and shall cause its Subsidiaries to, exercise, consistent with the terms and conditions of this Agreement, control and supervision over the operations of the Business and the Acquired Companies.

## 6.2 <u>Pre-Closing Access to Information</u>.

(a) Until the earlier of (i) the Closing and (ii) the termination of this Agreement pursuant to its terms, Seller shall, and shall cause IP Seller and the Acquired Companies (and to the extent related to the Business, its other Affiliates) to, permit Purchaser and the Financing Sources to have reasonable access, upon reasonable prior notice, during normal business hours in a manner so as not to interfere with the normal business operations of Seller and its Affiliates and in accordance with the reasonable procedures established in good

faith by Seller, to the books, assets, properties, Contracts and records of the Acquired Companies and the Business; provided, however, that the foregoing shall not: (1) require Seller or its Affiliates (including IP Seller and the Acquired Companies) to provide access or to disclose information where Seller reasonably believes in good faith that such access or disclosure would contravene any Legal Requirement (including those relating to data protection or privacy) or Contract, or would result in the waiver of any legal privilege or work-product protection; provided that Seller shall use commercially reasonable efforts to allow for access to the extent that doing so does not result in the loss of any such protection, (2) include any invasive investigations, sampling or testing whatsoever for or regarding any environmental matters, which may be granted or withheld in Seller's sole and absolute discretion, (3) require Seller or any of its Affiliates to provide Purchaser, its Affiliates or its and their Representatives with (A) any Consolidated Return (or copy thereof), (B) information relating to businesses of Seller or any of its Affiliates other than the Business or (C) information relating to individual performance or evaluations or medical histories, (4) require Seller or its Affiliates to provide Purchaser or its Representatives with any information related to the Transactions or Seller's or its Representatives' evaluation thereof, including projections, financial or other information related thereto other than projections, financial or other information prepared in the ordinary course of the Business without being primarily prepared for the Transactions or (5) require Seller to provide (A) confidentiality or non-disclosure agreements, letters of intent, expressions of interest or other proposals received in connection with transactions comparable to the Transactions or any information or analysis relating to any such communications or (B) financial or operating data or other information that has not previously been prepared by Seller or its Affiliates (including the Acquired Companies), or that is not otherwise prepared in the ordinary course of operating the Business. Any information disclosed under this <u>Section 6.2(a)</u> will be subject to the provisions of <u>Section 6.5</u>.

- (b) Until the earlier of the Closing and the termination of this Agreement pursuant to its terms, Purchaser shall not, and shall cause its Affiliates and its and their respective Representatives not to, communicate with any of the officers, directors, employees, customers or landlords of, or suppliers to, the Business, the Acquired Entities or their Affiliates (including the Acquired Companies), to the extent such communications are related to the Business or the Acquired Companies, without the prior written consent of Seller (which consent may not be unreasonably withheld, conditioned or delayed); *provided* that nothing in this Section 6.2 shall prohibit Purchaser, its Affiliates and its and their respective Representatives from communicating with (i) such Persons in the ordinary course of their respective business unrelated to this Agreement, the Transactions, the Acquired Companies or the Business in connection with ongoing commercial relationships or (ii) Business Employees at such a time and in such a manner as mutually agreed by the parties in advance in connection with post-Closing matters relating to such Business Employees' employment.
- 6.3 <u>Cooperation</u>. Subject to <u>Section 6.6</u>, prior to the Closing, Seller shall, and shall cause its Affiliates to, and Purchaser shall, and shall cause its Affiliates to, use reasonable best efforts to cause all Closing Conditions to be met as promptly as reasonably practicable and, in any event, prior to the Outside Date.

### 6.4 Consents; Termination of Intercompany Agreements.

- Subject in all cases to the other terms and conditions set forth in this Agreement, prior to the Closing, Seller shall, and shall cause its Affiliates and the Acquired Companies to, use commercially reasonable efforts to obtain any Consents of, and make any registrations, declarations, filings and notifications to, any Persons (other than to the extent subject to Section 6.6) that may be (i) required in connection with the Transactions and (ii) requested by Purchaser. Except with the written consent of Seller, prior to Closing, Seller or its Affiliate shall be the sole party that contacts or communicates with any counterparty to third-party Consents (other than to the extent subject to Section 6.6) concerning this Agreement, the other Transaction Documents or the Transactions; provided that, Seller and its applicable Affiliates shall consult in good faith with Purchaser and keep Purchaser reasonably informed, in each case, regarding the obtaining of any third-party Consents, and shall not agree to any adverse modifications or amendments to any Contract or Permit or to the payment of any amount in connection with obtaining or seeking any Consent, waiver, novation, modification, replication or approval applicable to the Acquired Companies or the Business following the Closing without the prior written consent of Purchaser. Any amounts payable in connection with obtaining any Consent, waiver, novation, modification, replication or approval under any Contract or Permit, and any financial penalty or fee or refund of any amounts payable in connection with, or as a result of, the Transactions under any Contract or Permit of any Acquired Company, shall be borne and paid by Purchaser. Notwithstanding anything to the contrary contained in this Agreement, neither Seller nor any of its Affiliates shall (i) be required to expend any money, commence or participate in any Proceeding, incur liabilities or offer or grant any accommodation (financial or otherwise) to any third party to obtain any Consent described in this Section 6.4(a), or (ii) have any obligations to obtain any Consent (except as expressly set forth in this <u>Section 6.4(a)</u> and <u>Section 6.6</u>).
- (b) Except for (i) the Transaction Documents (and each other agreement or instrument expressly contemplated by the Transaction Documents to be entered into by any member of the Seller Group, on the one hand, and any Acquired Company, on the other hand), (ii) any Contracts or understandings to which any third party is a party (including the Shared Contracts) and (iii) any intercompany contracts, any arrangements, financing agreements, intercompany loans, transactions, accounts, commitments and claims between the Acquired Companies, on the one hand, and Seller or any of its Affiliates (other than an Acquired Company), on the other hand (the "Intercompany Agreements"), shall be terminated (or deemed terminated without any further action on the part of any party thereto) effective no later than as of the Closing without any party having any continuing obligations or liability to the other party under the Intercompany Agreements. Seller shall, and shall cause the Acquired Companies to, settle all amounts due and payable to an Acquired Company from Seller or any of its Affiliates (other than an Acquired Company), and all amounts due and payable to Seller or any of its Affiliates (other than an Acquired Company) from an Acquired Company, shall, in each case, prior to the Closing and in a manner that does not result in an increase of any ongoing Liabilities of the Acquired Companies following the Closing.

- (c) Seller agrees to take the actions set forth on <u>Schedule 6.4(c)</u> of the Seller Disclosure Schedule.
- (d) Purchaser acknowledges and agrees that (i) the Business as presently conducted receives or benefits from the Corporate Functions and (ii) effective as of the Closing, the sole obligations of Seller and its Affiliates with respect to the provision of any such Corporate Functions to the Business shall be as set forth in the Transition Services Agreement.

## 6.5 <u>Confidentiality</u>.

- The terms of the Confidentiality Agreement are incorporated into this Agreement by reference and shall continue in full force and effect (and all obligations thereunder shall be binding upon Purchaser and its Representatives (as defined in the Confidentiality Agreement) as set forth therein) until the Closing, at which time the obligations under the Confidentiality Agreement shall terminate; provided, however, that Purchaser's confidentiality obligations shall terminate only in respect of that portion of the Proprietary Information (as defined in the Confidentiality Agreement) exclusively to the extent relating to the Acquired Companies, the IP Seller IP and the Business and, for all other Proprietary Information ("Non-Business <u>Proprietary Information</u>"), the term of the Confidentiality Agreement shall continue to apply to such Non-Business Proprietary Information until the termination or expiration of the applicable terms of the Confidentiality Agreement, and shall promptly destroy or cause to be destroyed all Non-Business Proprietary Information in its possession or in the possession of any of its Representatives (as defined in the Confidentiality Agreement) in accordance with the Confidentiality Agreement. If for any reason the Closing does not occur, the Confidentiality Agreement shall continue in full force and effect in accordance with its terms. In the event of a conflict or inconsistency between the terms expressly set forth in this Agreement (rather than incorporated by reference herein) and the Confidentiality Agreement, the terms of this Agreement will govern. Notwithstanding the foregoing, the Purchaser and its Representatives may disclose the Required Information without any obligation on the part of the Financing Sources or other recipients thereof contemplated in Section 6.7(c) hereof to comply with the terms of this Agreement or the Confidentiality Agreement.
- (b) During the four (4)-year period following the Closing, Seller shall keep confidential and refrain from using, and cause its Affiliates and its and their respective Representatives to keep confidential and refrain from using, all non-public, confidential or proprietary information concerning the Acquired Companies or the Business, except (i) as required or requested by a Governmental Authority or required pursuant to Legal Requirements or the rules or regulations of any securities exchange or listing authority or legal, administrative or judicial process (*provided* that Seller shall, to the extent permitted by Legal Requirements, promptly notify Purchaser of such requirement or request and the disclosure that is expected to be made with respect thereto and, to the extent requested by Purchaser, shall reasonably cooperate with Purchaser (at Purchaser's sole expense) to seek a protective order or other appropriate remedy to limit or obtain confidential treatment for such disclosure, and in the event no such protective order or remedy is obtained, Seller will furnish only that portion of such non-public, confidential or proprietary information which Seller is advised by counsel is required by

Legal Requirements and will exercise commercially reasonable efforts to obtain reliable assurance that confidential treatment will be accorded to such non-public, confidential or proprietary information), (ii) for information that is available, as of immediately following the Closing, generally to the public, or thereafter becomes generally available to the public, other than as a result of a breach of this Section 6.5(b), (iii) to the extent such use is reasonably necessary to enable Seller or its Affiliates to fulfill its obligations to Purchaser and the Acquired Companies under any Transaction Document, (iv) for information disclosed to Seller or any of its Affiliates following the Closing Date on a non-confidential basis by any Person not known by Seller after reasonable inquiry to be bound by an obligation of confidentiality to Purchaser or any of its Affiliates, (v) as necessary for the preparation and filing of any Tax Return or defense of any Tax Proceeding or as reasonably related to any Tax matter of any member of the Seller Group or (vi) is demonstrated by Seller or its Affiliates to have been independently developed following the Closing Date not in violation of its or its Representatives' obligations under this Section 6.5(b) and without reference to any non-public, confidential or proprietary information concerning the Acquired Companies or the Business or any information from a source that is subject to a confidentiality obligation to the Acquired Companies or the Business or is otherwise prohibited from furnishing such information to Seller, its Affiliates or their respective Representatives.

Promptly following the Closing, Seller shall deliver, or cause to be delivered, to Purchaser copies of all books, records and documents, in any form or medium, to the extent related to the Business or the Acquired Companies (in each case, including any applicable attorney-client privilege, attorney work product protection and expectation of client privilege attaching to any such books, records and documents) and in the possession of Seller and its Affiliates (other than the Acquired Companies), other than books, records and documents the provision of which is subject to the Transition Services Agreement. Seller shall promptly destroy or cause to be destroyed all duplicate or back-up copies thereof and any notes, extracts or summaries based thereon in its possession or in the possession of any of its Representatives, without retaining any copy thereof in any form or medium; provided, however, that Seller, its Affiliates and its Representatives may keep one copy of such information. including one copy of any notes, reports, summaries, analyses, compilations, forecasts, studies, interpretations, memoranda or other material based thereon, to the extent such retention is required to comply with applicable Legal Requirement or pursuant to a bona fide internal document retention policy for use solely to demonstrate compliance with such requirements (and, to the extent such information is retained electronically, ordinary access thereto shall be limited only to information technology personnel in connection with their information technology duties). Upon Purchaser's written request, Seller will certify to Purchaser in writing that all information has been so destroyed. This Section 6.5(c) shall not apply to any Consolidated Returns or other Tax Returns of the Seller Group or any books, records, documents, working papers or other information related thereto or to the Tax matters of any member of the Seller Group.

## 6.6 Reasonable Best Efforts; Regulatory Filings.

(a) Subject to the limitations set forth in this <u>Section 6.6</u>, Seller and Purchaser shall use reasonable best efforts to take, or cause to be taken, all actions, and to do, or cause to be

done, and to assist and cooperate in doing, all things necessary, proper or advisable under this Agreement and Legal Requirements to consummate and make effective the Transactions as promptly as possible following the date of this Agreement, which actions include (i) using reasonable best efforts to obtain as promptly as possible each Consent, Permit and Order of any Governmental Authority that may be, or become, necessary for the consummation of the Transactions (collectively, "Governmental Approvals"), (ii) cooperating in determining which filings are required or advisable to obtain any Governmental Approval or any exemption by any Governmental Authority, (iii) furnishing all information and documents required by or advisable under applicable Legal Requirements in connection with Governmental Approvals or filings with any Governmental Authority, (iv) filing, or causing to be filed, as promptly as practicable following the execution and delivery of this Agreement, applicable notifications with the necessary Governmental Authorities (v) using reasonable best efforts to obtain as promptly as possible the requisite clearances, approvals or expiration of any waiting period, including those under the HSR Act, and any other applicable Antitrust Laws or Investment Screening Laws, and (vi) defending any actions, whether judicial or administrative, challenging this Agreement or the consummation of the Transactions, including seeking to have any Order entered by any court or other Governmental Authority vacated or reversed. In furtherance and not in limitation of the foregoing, each party hereto agrees that it will use its reasonable best efforts to file or cause to be made as promptly as reasonably practicable, but in any event no later than fifteen (15) Business Days following the date of this Agreement, any required notification and report forms under the HSR Act with the United States Federal Trade Commission (the "FTC") and the United States Department of Justice (the "<u>DOJ</u>").

In connection with, and without limiting, the efforts referenced in Section 6.6(a), Seller and Purchaser shall (b) (i) furnish to the other, and Seller shall cause the Acquired Companies to furnish to Purchaser, such necessary information and reasonable assistance as the other may request in connection with its preparation of any filing or submission that is necessary under the HSR Act, and any other applicable Governmental Approval, (ii) permit the other party to review any filing or submission prior to forwarding to the FTC, the DOJ, and other Governmental Authorities (except where such material is confidential to a party in which case it will be provided, subject to applicable Legal Requirements, to the other party's counsel on an "external counsel" basis) and consider in good faith the other party's reasonable comments in connection therewith, (iii) keep each other apprised of the status of any material communications with, and any inquiries or requests for additional information from, any Governmental Authorities and substantially comply as promptly as reasonably practicable with any such inquiry or request and (iv) agree not to, and Seller shall cause the Acquired Companies not to, participate in any substantive meeting or discussion, either in person or by telephone or videoconference, with any Governmental Authority in connection with the Transactions, unless (A) it consults with the other party in advance and (B) gives the other party the opportunity to attend and participate; provided that a party shall not be required to give the other party the opportunity to attend and participate to the extent prohibited by such Governmental Authority. Neither Seller nor Purchaser shall commit to or agree with any Governmental Authority to stay, toll or extend any applicable waiting period or withdraw its filing under the HSR Act or any other Antitrust Law or Investment Screening Law without the prior written consent of the other party (such consent not to be unreasonably withheld, conditioned or delayed). Notwithstanding

anything to the contrary contained in this <u>Section 6.6</u> or any other provision of this Agreement, Purchaser shall, after considering in good faith Seller's views and comments, control and lead all communications, negotiations, timing, decisions, and strategy on behalf of the parties relating to the Governmental Approvals. Whether or not the Transactions are consummated, Purchaser shall be responsible for the payment of all filing fees payable to any Governmental Authority in connection with the HSR Act and any other notifications or filings required pursuant to this <u>Section 6.6</u>. Notwithstanding anything to the contrary in this <u>Section 6.6</u>, each of Seller and Purchaser may redact materials provided to the other party: (I) to remove competitively sensitive information or information concerning valuation; (II) as necessary to comply with legal or contractual arrangements; and (III) as necessary to address reasonable attorney-client privilege or other privilege or confidentiality concerns.

- (c) Seller and Purchaser will use reasonable best efforts to substantially comply with any additional requests for information, including requests for production of documents and production of witnesses for interviews or depositions by any Governmental Authority. Purchaser agrees to take any and all steps necessary or advisable to avoid or eliminate each and every impediment under any Legal Requirement that may be asserted by any Governmental Authority or any other Person so as to enable the parties hereto to expeditiously consummate the Transactions, including proposing, negotiating, committing to and consenting to any divestiture, sale, disposition, hold separate order or other structural or conduct relief, or other operational undertakings, in order to obtain any Governmental Approval; *provided, however*, that, notwithstanding anything to the contrary in this Agreement, none of Purchaser or any of its Affiliates or Subsidiaries shall be required to (and without Purchaser's prior written consent, the Acquired Companies and Seller shall not) agree to divest, sell, dispose of, encumber, limit or otherwise take any other action or agree to any remedy, whether structural, behavioral, or otherwise, that would, individually or in aggregate, reasonably be expected to have a material adverse effect on the business, operations, financial condition or results of operations of Purchaser and its Subsidiaries (including the Acquired Companies) after giving effect to the Closing, taken as a whole; *provided, further, however*, that notwithstanding anything in this Agreement to the contrary, any actions or efforts by Purchaser contemplated in this Section 6.6(c) shall be conditioned upon the consummation of the Transactions.
- (d) Purchaser shall not, directly or indirectly, acquire or agree to acquire by merging or consolidating with, or by purchasing a substantial portion of the assets of or equity in, or by any other manner, any business of any Person or other business organization or division thereof, or otherwise acquire or agree to acquire any assets if the entering into of a definitive agreement relating to, or the consummation of, such acquisition, merger or consolidation would reasonably be expected to (i) impose a material delay in the obtaining of, or materially increase the risk of not obtaining, any Governmental Approval of any Governmental Authority necessary to consummate the Transactions or the expiration or termination of any applicable waiting period under any Legal Requirement, (ii) materially increase the risk of any Governmental Authority entering an Order prohibiting the Transactions or (iii) prevent or materially delay the consummation of the Transactions.

(e) Notwithstanding anything in this <u>Section 6.6</u> to the contrary, nothing in this <u>Section 6.6</u> shall to the extent not conditioned upon the Closing, require, or be deemed to require, Seller or any of its Subsidiaries or Purchaser or any of its Subsidiaries or Affiliates to propose, negotiate, offer to commit, effect or agree to (A) any sale, divestiture, license or disposition of assets or businesses of any Acquired Company, Purchaser, or Purchaser's Subsidiaries or Affiliates or (B) any behavioral remedy of any Acquired Company, Purchaser, or Purchaser's Subsidiaries or Affiliates.

### 6.7 <u>Financing Cooperation</u>.

From the date hereof until the Closing, Seller shall use its reasonable best efforts to provide, Seller shall cause IP Seller and the Acquired Companies to use reasonable best efforts to provide, and Seller shall use reasonable best efforts to cause each of its Representatives to use reasonable best efforts to provide, at Purchaser's sole cost and expense, customary cooperation, to the extent reasonably requested by Purchaser, necessary or advisable for the arrangement of the Debt Financing (including an offering of notes in lieu of any bridge facility for the purpose of financing the Transactions and related fees and expenses contemplated by this Agreement) or any Alternative Debt Financing, including using reasonable best efforts to do the following: (i) participating (and causing appropriate members of Senior Management of the Business to participate) in a reasonable number of lender marketing meetings, presentations, drafting sessions, road shows, due diligence sessions and calls and sessions and other customary syndication activities with ratings agencies and prospective financing sources, in each case, in connection with the Debt Financing and with reasonable advance notice and at reasonable times and locations to be mutually agreed upon (it being understood that any such meetings may take place via videoconference or web conference at Seller's option), (ii) at least four (4) Business Days prior to the Closing Date, providing all information regarding the Acquired Companies and the Business required in connection with the Debt Financing by bank regulatory authorities under applicable "know your customer" and anti-money laundering rules and regulations, in each case, as reasonably requested by Purchaser in writing at least nine (9) Business Days prior to the Closing Date, (iii) furnishing Purchaser as promptly as reasonably possible with the Required Information, all of which is Compliant, (iv) furnishing Purchaser with reasonable information and materials with respect to the Acquired Companies to be used in Purchaser's preparation of customary lender and investor presentations, rating agency presentations, bank information memoranda (including a version thereof that does not contain material non-public information), business projections (it being understood that Purchaser shall have sole responsibility therefor), offering documents, prospectuses, memoranda and other similar documents for the Debt Financing, including customary authorization letters related thereto authorizing the distribution of information to prospective lenders and containing customary representations with respect to the presence or absence of material non-public information about the Acquired Companies and regarding the accuracy of the information provided by, or with respect to, the Acquired Companies; provided that any such information distributed (including any authorization letters) shall contain customary language which shall exculpate Seller and its Representatives and Affiliates with respect to any liability related to or responsibility for the contents of such information or related marketing materials by the recipients thereof, except to the extent of the representations with respect to the presence or absence of material non-public information

described above, (v) assisting with the preparation and/or filing of, and executing and delivering, any customary pledge and security documents and any other agreements, documents or certificates that facilitate the pledging of collateral as reasonably requested by Purchaser and required for the funding of the Debt Financing, including UCC termination statements, similar release documents (if any), and delivery of possessory collateral, all of which shall be effective only at or after Closing. (vi) cooperating with the Financing Sources' due diligence efforts (including the provision of "backup" support), to the extent reasonable and customary for financings similar to the Debt Financing, (vii) causing the Acquired Companies' independent auditors to provide, consistent with customary practice, customary auditors consents and customary "comfort" letters (including "negative assurance" comfort and change period comfort) as reasonably requested by the Financing Sources with respect to financial information relating to the Business included in the offering documentation for the Debt Financing (including using reasonable best efforts to obtain, to the extent applicable, consents of accountants for use of their reports in any materials relating to the Debt Financing and accountants' comfort letters, in each case as reasonably requested by Purchaser or the Financing Sources) and participate in due diligence sessions and drafting sessions with the Financing Sources and (viii) taking all corporate actions, subject to the occurrence of the Closing, reasonably requested by Purchaser that are necessary or advisable to permit the consummation of the Debt Financing and to permit the proceeds thereof to be made available on the Closing Date to consummate the transactions contemplated by this Agreement; provided that nothing in this Agreement (including this Section 6.7) will require any such cooperation or action to the extent that it could (A) require Seller or any of its Affiliates or any of their respective Subsidiaries or officers, directors, managers, employees, advisors, accountants, consultants, auditors, agents or other Representatives to pay (or agree to pay) any commitment or other fee, make any other payment, provide (or agree to provide) any indemnities, reimburse any expenses or otherwise incur any liability or other obligation in connection with the Debt Financing, (B) require Seller or any of its Affiliates or any of their respective Subsidiaries or any individual who is a member of the board of directors (or other similar governing body) of such entities to pass resolutions or consents to approve, or authorize the execution of, the Debt Financing or any Definitive Agreement (other than with respect to the Acquired Companies only, any such resolutions or consents that (x) are passed by Persons who will continue as members of the board of directors (or other similar governing body) of the Acquired Companies after the occurrence of the Closing and (y) are subject to and conditioned upon, and do not become effective until, the occurrence of the Closing), (C) require Seller or any of its Affiliates or any of their respective Subsidiaries or Representatives to enter into, execute or deliver any Contract or other documentation (other than (1) the authorization letters referred to in clause (iv) of this Section 6.7(a) and (2) with respect to the Acquired Companies only, any such documents that (x) are executed or delivered, as applicable, by Persons who will continue as officers or members of the board of directors (or other similar governing body) of the Acquired Companies after the occurrence of the Closing and (y) are subject to and conditioned upon, and do not become effective until, the occurrence of the Closing), (D) impose any personal liability on the officers, directors, managers, employees, advisors, accountants, consultants, auditors, agents or other Representatives of Seller. its Affiliates or their respective Subsidiaries, (E) unreasonably interfere with the operation of the business of Seller or any of its Affiliates or any of their respective Subsidiaries or Representatives, (F) cause any representation or warranty in this Agreement to be breached by

Seller, its Affiliates or its Subsidiaries or require any waiver or amendment of the terms of this Agreement, (G) conflict with or result in any violation of the organizational documents of Seller or its Subsidiaries or any of their respective Affiliates or any Legal Requirement, (H) result in the contravention of, or result in a violation or breach of, or a default (with or without notice, lapse of time, or both) under, any Contract to which Seller, its Subsidiaries or any of their respective Affiliates is party or by which it is bound (and, in each case, which was not entered into in contemplation of this Agreement), (I) provide access to or disclose information that Seller or its Subsidiaries reasonably determines would jeopardize any attorney-client or similar privilege or protection of Seller or any of its Affiliates or any of their respective Subsidiaries or Representatives so long as Seller or its Subsidiaries shall have used commercially reasonable efforts to disclose such information in a way that would not waive such privilege or protection, (J) require Seller or any of its Affiliates or any of their respective Subsidiaries or Representatives to provide any solvency or other similar certificate of its chief financial officer or similar Representative, or (K) require Seller or any of its Affiliates or any of their respective Subsidiaries or Representatives to provide or prepare any projections, pro forma financial statements or other forward-looking financial information or any financial information other than the Required Information. Notwithstanding anything to the contrary herein, the failure of Seller or any of its Affiliates or any of their respective Subsidiaries or Representatives to comply with this Section 6.7 shall not give rise to the failure of a condition precedent set forth in Section 10.1(b) or termination right pursuant to Section 11.1(b) unless Purchaser's failure to obtain any portion of the proceeds of the Debt Financing was a result of the material breach of the obligations of Seller to comply with its obligations under this Section 6.7.

- (b) Seller and the Acquired Companies hereby consent to the customary use of their logos in connection with the Debt Financing prior to the Closing Date; *provided* that such logos are used solely in a manner that is not intended or reasonably likely to (i) harm or disparage Seller or its Subsidiaries or their reputation, goodwill or marks or (ii) otherwise materially adversely affect Seller or any of its Subsidiaries.
- (c) Notwithstanding any other provision set forth herein, the Confidentiality Agreement or in any other agreement between Seller and Purchaser (or their respective Affiliates), Seller agrees that Purchaser and its Affiliates may share any confidential information with respect to Seller, the Acquired Companies and the Business with any Financing Sources, and that Purchaser and their respective Affiliates and such Financing Sources may share such information with potential Financing Sources in connection with any marketing efforts with respect to the Debt Financing; *provided* that the recipients of such information and any other confidential information contemplated to be provided by Seller and the Acquired Companies or any of their respective Affiliates pursuant to this <u>Section 6.7</u> are subject to customary confidentiality arrangements, including "click through" confidentiality agreements and confidentiality provisions contained in customary bank books or offering materials.
- (d) Upon written request by Seller, Purchaser will (i) upon the earlier of the Closing and termination of this Agreement in accordance with <u>Section 11.1</u>, reimburse Seller and its Subsidiaries for any reasonable and documented out-of-pocket costs or expenses (limited in the case of legal expenses, to the reasonable and documented out-of-pocket attorney's fees of one

firm of outside counsel) incurred or otherwise payable by Seller or any of its Affiliates or any of their respective Subsidiaries or Representatives in connection with their cooperation requested by Purchaser pursuant to this Section 6.7 and (ii) indemnify, defend and hold harmless Seller and its Affiliates and their respective Subsidiaries and Representatives, and the successors and assigns of each of the foregoing Persons from and against any and all liabilities, losses, damages, claims, reasonable and documented out-of-pocket costs and expenses, interest, awards, judgments and penalties suffered or incurred by them in connection with the cooperation or efforts pursuant to this Section 6.7 or otherwise in complying with their obligations in connection with the arrangement of the Debt Financing (including actions taken in accordance with this Section 6.7) or any information used or misused in connection therewith, except to the extent that any of the foregoing arises from the bad faith, gross negligence, material breach or willful misconduct of Seller or its Subsidiaries, in each case, as determined by a court of competent jurisdiction in a final and non-appealable decision.

(e) The parties hereto acknowledge and agree that the provisions contained in this Section 6.7 represent the sole obligation of Seller and its Affiliates and any of their respective Subsidiaries or Representatives with respect to cooperation in connection with the arrangement of any financing (including the Debt Financing) to be obtained by Purchaser with respect to the transactions contemplated by this Agreement (including the Debt Financing Commitment), and no other provision of this Agreement (including the Exhibits and Schedules hereto) or the Debt Financing Commitment shall be deemed to expand or modify such obligations.

## 6.8 Financing Obligation.

(a) Purchaser shall, and shall use reasonable best efforts to cause each of its Affiliates to, use its reasonable best efforts to take all actions and do all things necessary, proper or advisable to obtain the proceeds of the Debt Financing in an amount equal to at least the Required Amount on the terms and subject only to the conditions set forth in the Debt Financing Commitment at or prior to the Closing, including (i) complying with its obligations under the Debt Financing Commitment, (ii) maintaining in effect the Debt Financing Commitment and the definitive financing agreements related to the Debt Financing (the "Definitive Agreements") in accordance with the terms and conditions thereof, (iii) negotiating, entering into and delivering the Definitive Agreements on a timely basis (taking into account the Marketing Periods, as applicable) on terms and conditions (including the flex provisions) contained in the Debt Financing Commitment and without any Prohibited Modification, (iv) satisfying on a timely basis all conditions applicable to Purchaser and/or its Affiliates contained in the Debt Financing Commitment and the Definitive Agreements within their control, including the payment of any commitment, engagement or placement fees required as a condition to the Debt Financing, (v) enforcing all of its rights, and using reasonable best efforts to enforce the obligations of the other parties, under the Debt Financing Commitment and the Definitive Agreements and (vi) consummating, and obtaining the proceeds of, the Debt Financing at or prior to the Closing. Purchaser shall, upon the request of Seller, keep Seller informed on a current and timely basis and in reasonable detail of the status of its efforts to obtain the Debt Financing and of developments concerning the timing of the closing of the Debt Financing. Without limiting the

generality of the foregoing, Purchaser shall give Seller prompt written notice (A) of any actual or threatened (in writing) violation, breach or default (or, to Purchaser's Knowledge, any event or circumstance that, with or without notice, lapse of time or both, could reasonably be expected to give rise to any violation, breach or default) by any party to the Debt Financing Commitment or any Definitive Agreement, (B) of any actual or threatened (in writing) reduction, withdrawal, repudiation or termination of the Debt Financing by any party to the Debt Financing Commitment or (C) if at any time for any reason Purchaser has determined in good faith that it will not be able to obtain all or any portion of the Required Amount of the Debt Financing on the terms, in the manner or from the sources contemplated by the Debt Financing Commitment or any Definitive Agreement. As soon as reasonably practicable, but in any event within two (2) Business Days following delivery by Seller to Purchaser of written request therefor, Purchaser shall provide any information reasonably requested by Seller relating to any circumstance referred to in clauses (A) through (C) of the immediately preceding sentence.

Purchaser shall not, without the prior written consent of Seller: (i) amend, modify, supplement, or waive any of the conditions to funding contained in the Debt Financing Commitment or any Definitive Agreement or any other provision of, or remedies under, the Debt Financing Commitment or any Definitive Agreement, in each case, to the extent such amendment, modification or supplement could reasonably be expected to have the effect of (A) reducing the amount of the Debt Financing to an amount less than the Required Amount, (B) otherwise adversely affecting the ability of Purchaser in any material respect to timely consummate the transactions contemplated by this Agreement, including the ability to pay the Required Amount in full, (C) adding new or additional conditions or amending, modifying or supplementing any of the existing conditions to the Debt Financing, (D) delaying, preventing or impeding the Closing or making the timely funding of the Debt Financing (in an amount no less than the Required Amount) or satisfaction of the conditions to obtaining the Required Amount under the Debt Financing less likely to occur or (E) adversely affecting the ability of Purchaser to enforce its rights against the other parties to the Debt Financing Commitment or the Definitive Agreements (the effects described in clauses (A) through (E), collectively, the "Prohibited Modification"); or (ii) terminate the Debt Financing Commitment or any Definitive Agreement; provided, however, subject to compliance with the other provisions of this Section 6.8, Purchaser may (1) amend, modify, supplement or waive any provision of the Debt Commitment Documents to add Financing Sources that have not executed the Debt Commitment Documents as of the date hereof, (2) amend the Definitive Agreements to implement any flex provisions contained in the Debt Financing Commitment or (3) otherwise amend, modify or supplement the Debt Financing Commitment or any Definitive Agreement so long as such amendment, modification or supplement could not effect a Prohibited Modification. Purchaser shall promptly deliver to Seller copies of any such amendment, modification, supplement or replacement (with customary redactions solely to the extent consistent with the requirements under Section 5.6). In the event Purchaser amends, modifies, supplements or replaces the Debt Financing Commitment or any Definitive Agreement in accordance with this Section 6.8, references to "Debt Financing," "Financing Sources," "Definitive Agreements," and "Debt Financing Commitment" (and other like terms in this Agreement) as used in this Agreement shall be deemed to refer to the Debt Financing Commitment and/or Definitive Agreement as so amended, modified or supplemented.

In the event all conditions to the Debt Financing Commitment have been satisfied (or waived) and all of the conditions set forth in <u>Section 10.1</u> and <u>Section 10.3</u> (not including conditions which are to be satisfied by the delivery of documents, or taking of any other action, at the Closing by any party) have been satisfied (or waived), Purchaser shall use its reasonable best efforts to cause the Financing Sources to fund the Debt Financing in an amount no less than the Required Amount for purposes of consummating the transactions contemplated by this Agreement.

- If all or any portion of the Debt Financing becomes unavailable for any reason, Purchaser shall (i) notify Seller in writing of such event and the reasons giving rise to such event, as promptly as practicable following the occurrence of such event, (ii) use reasonable best efforts, and cause each of its Affiliates to use reasonable best efforts, to arrange and obtain, as promptly as possible following the occurrence of such event, alternative financing for any such unavailable portion of the Debt Financing from the same or alternative sources (it being understood that Purchaser shall not be required to pay fees, economics, or other amounts in excess of that contemplated by the Debt Commitment Documents, and the related fee letters and engagement letters, in effect on the date of this Agreement), which may include one or more of a loan financing, an offering and sale of notes, or any other financing or offer and sale of other debt securities, or any combination thereof, (A) in an amount sufficient, when added to any portion of the Debt Financing that is and will be available, that is equal to or greater than the Required Amount, (B) on terms no less favorable to Purchaser than the Debt Financing Commitment as of the date hereof (taking into account any flex provisions contained therein), (C) containing conditions that (1) are not more onerous to Purchaser than those conditions contained in the Debt Financing Commitment as of the date hereof and (2) could not reasonably be expected to delay the Closing and (D) which does not contain any Prohibited Modification (any such alternative financing, the "Alternative Debt Financing") and (iii) obtain a new financing commitment letter (together with its related term sheets and fee letters, the "Alternative Debt <u>Financing Commitment</u>") or a new definitive agreement with respect thereto that provides for such Alternative Debt Financing. In the event Purchaser obtains Alternative Debt Financing in accordance with this Section 6.8, references to "Debt Financing," "Financing Sources," and "Definitive Agreements" (and other like terms in this Agreement) as used in this Agreement shall be deemed to include any Alternative Debt Financing (and consequently the term "Debt Financing" shall include any available portion of the then-existing Debt Financing and the Alternative Debt Financing), and the term "Debt Financing Commitment" as used in this Agreement shall be deemed to include any Alternative Debt Financing Commitment.
- (d) Without limitation to any other obligation of Purchaser and its Affiliates under this Agreement, prior to, and from and after the commencement of the Initial Marketing Period, Purchaser shall use commercially reasonable efforts (taking into account market conditions, the cost of funding into escrow, the expected timing to Closing and such other factors as Purchaser in good faith deems commercially reasonable) to obtain, at or prior to the conclusion of the Initial Marketing Period, the proceeds of the Debt Financing in an amount equal to at least the Required Amount on the terms and subject only to the conditions set forth in the Debt Financing Commitment; *provided* that, the decision to obtain the Debt Financing during the Initial Marketing Period shall be the sole decision of Purchaser. During the Initial Marketing

Period, Purchaser shall consult with, and provide reasonably frequent updates on its financing efforts, to the Chief Financial Officer of Seller.

(e) Purchaser expressly acknowledges and agrees that (i) obtaining the Debt Financing is not a condition to the Closing and (ii) notwithstanding anything contained in this Agreement to the contrary, Purchaser's obligations hereunder are not conditioned in any manner upon Purchaser obtaining the Debt Financing, or any other financing.

### 6.9 Financial Statements.

- (a) Prior to the Closing, Seller shall use reasonable best efforts to cause to be prepared and delivered to Purchaser the (i) audited combined balance sheets of the Business as of December 31, 2022 and December 31, 2023 and (ii) respective related audited combined statements of income, shareholder's equity and cash flows of the Business (taking into account the Corporate Functions used by the Business), for the fiscal years ended December 31, 2022 and December 31, 2023, and the last day of any subsequent fiscal year ended at least ninety (90) calendar days before the Closing Date, in each case together with an "unqualified" audit opinion issued with respect to such audited financial statements by the Business's independent auditor (such audited financial statements referred to in the foregoing clauses (i) and (ii), collectively, the "Audited Financial Information"). The Audited Financial Information shall (A) be prepared in good faith and, except as set forth in the notes thereto, in accordance with GAAP, consistently applied throughout the periods covered thereby, (B) fairly present the financial condition and results of operations of the Business (taking into account the Corporate Functions used by the Business), as of the dates and for the periods therein specified, (C) be derived from books and records of Seller that are regularly maintained by management of Seller in accordance with GAAP and (D) have been reviewed in accordance with the applicable procedures of the American Institute of Certified Public Accountants.
- (b) During the period commencing as of the date hereof and continuing until the Closing, Seller shall use reasonable best efforts to deliver the financial information of the Acquired Companies for each calendar month in the form set forth on Schedule 6.9(b) of the Seller Disclosure Schedules to Purchaser within ten (10) Business Days of the end of such calendar month.

#### 6.10 Insurance.

- (a) Without limiting Seller's obligations under <u>Section 6.7</u>, prior to the Closing, Seller shall use commercially reasonable efforts to keep, or cause its Affiliates to use commercially reasonable efforts to keep, all insurance policies currently maintained by Seller or its Affiliates that cover the Business, or reasonably suitable replacements or renewals of at least equal coverage, in full force and effect through the Closing.
- (b) Subject to, and other than as set forth in this <u>Section 6.10(b)</u> and <u>Section 6.10(c)</u>, from and after the Closing, (i) the Acquired Companies shall cease to be insured by the Seller Group's insurance policies (other than any insurance policies held solely in the name of the Acquired Companies), including any self-insurance, fronted insurance or captive

insurance policy or program and (ii) neither Purchaser nor its Subsidiaries (including the Acquired Companies) shall have any access, right, title or interest to or in any such insurance policies (including the right to make claims or receive proceeds thereunder) to cover the Acquired Companies or any liability arising from the operation of the Business on or after the Closing. Seller or any of its Affiliates may, to be effective at the Closing, amend any insurance policies (other than any insurance policies held solely in the name of the Acquired Companies) in the manner it deems appropriate to give effect to this Section 6.10. Notwithstanding the foregoing, Seller shall direct any insurers for any third-party Insurance Policies (other than any self-insurance, fronted insurance or captive insurance policy or program) affording coverage for the Acquired Companies or the Business to continue to process any claims made thereunder by the Acquired Companies or the Business to the extent such claims were made prior to the Closing and cooperate with Purchaser following the Closing in connection therewith, and any such claims shall be further subject to Section 6.10(c) mutatis mutandis.

From and after the Closing, Seller shall, and shall cause the other members of the Seller Group to, use commercially reasonable efforts, to make any available coverage under the Seller Group's occurrence-based insurance policies other than any self-insurance, fronted insurance or captive insurance (such policies "Available Insurance Policies"), available to the Acquired Companies with respect to claims arising out of events which have occurred, or conditions which are in existence, prior to the Closing relating to the Acquired Companies, the Business Employees or former employees of the Acquired Companies (such claims "Pre-Closing Claims"), including noticing claims to and reasonably cooperating with the applicable insurer(s) and remitting proceeds to the Acquired Companies, subject to the terms and conditions of such Available Insurance Policies; provided that (i) all applicable deductibles or retentions under any such Available Insurance Policies shall be shared in the same proportion as any insurance proceeds actually received by the Seller Group, on the one hand, and the Acquired Companies, on the other hand, with respect to any one claim (or related claims) under the relevant Available Insurance Policy, (ii) except as expressly set forth in the prior clause (i). Purchaser shall exclusively bear the amount of any claims handling fees or any other amounts payable or to be retained or incurred under any such Available Insurance Policies associated with Pre-Closing Claims, (iii) Seller and its Affiliates shall not be liable for any uninsured or uncovered amounts of such Pre-Closing Claims; (iv) the Acquired Companies shall not, without the express written consent of Seller, amend, modify or waive any rights of Seller or other insureds under any such Available Insurance Policies; (v) Seller shall have the right to reasonably monitor any such Pre-Closing Claims under the Available Insurance Policies, and the Acquired Companies shall keep Seller reasonably apprised of any such Pre-Closing Claims under the Available Insurance Policies; (vi) upon Seller's request, Purchaser and the Acquired Companies shall provide Seller with reasonable documentation and information regarding any Pre-Closing Claim under an Available Insurance Policy, including with respect to the amount of the claimed loss or damage; (vii) upon Purchaser's request, Seller shall provide Purchaser with reasonable documentation and information regarding any Pre-Closing Claim or Available Insurance Policy; and (viii) the Acquired Companies shall not assign any Available Insurance Policies or any rights or claims under the Available Insurance Policies. Notwithstanding anything contained herein, (A) neither Seller nor any member of the Seller Group shall be liable to Purchaser or the Acquired Companies for any claims, or portions thereof, not reimbursed by

an insurer under any Available Insurance Policy for any reason (other than a failure to comply with this <u>Section 6.10</u>) and (B) Seller shall retain the exclusive right to control all of their insurance policies and programs, including the Available Insurance Policies, including the right to exhaust, erode, settle, release, commute, buy-back or otherwise resolve disputes with respect to any of its insurance policies and to amend, modify, terminate or waive any such insurance policies and programs or any rights thereunder; *provided* that, Seller shall not amend, modify, terminate or waive any coverage under any Available Insurance Policies with respect to periods prior to the Closing in a manner that would impair coverage thereunder available for Pre-Closing Claims or otherwise limit the rights of Purchaser under this <u>Section 6.10</u>, other than pursuant to corporate-wide or multi-business amendments, modifications, terminations, waivers or reductions of limits.

- (d) Prior to the Closing, Seller shall, and shall cause the Acquired Companies to, at Purchaser's sole cost and expense, use commercially reasonable efforts to cooperate with Purchaser in obtaining any insurance, including any "tail" insurance with respect to matters existing, occurring or arising at or prior to the Closing, related to any Acquired Company or the Business (other than the "tail" insurance contemplated in Section 7.4(b), which insurance is addressed in Section 7.4(b)); provided that failure of Purchaser to obtain any such insurance shall not delay the Closing or be the basis for any claim that the conditions in Section 10.1 have not been satisfied.
- Binder Agreement. At the Closing, or promptly thereafter, Purchaser shall deliver to Seller a copy of the R&W Insurance Policy. Purchaser acknowledges and agrees that, except in the case of Fraud, from and after the Closing, the R&W Insurance Policy (whether or not it is ultimately bound, and whether or not the R&W Insurance Policy is sufficient to cover any Losses of Purchaser or any of its Affiliates) shall be the sole and exclusive remedy of Purchaser or any of its Affiliates and its and their respective Representatives, successors and assigns of whatever kind and nature, at law, in equity or otherwise, known or unknown, which such Persons have now or may have against Seller or any of its Affiliates in the future, resulting from, arising out of, or related to any inaccuracy or breach of any representation or warranty contained in this Agreement, and none of such Persons shall have any recourse against Seller or any of its Affiliates with respect thereto. Purchaser and its Affiliates shall not consent to, amend, waive or otherwise modify the subrogation provision of the R&W Binder Agreement or the R&W Insurance Policy in any manner without Seller's prior written consent (which consent shall be in the sole and absolute discretion of Seller). Purchaser shall be solely responsible for all costs to procure, maintain and make claims under the R&W Insurance Policy, including all premiums, retention amounts, Taxes, expenses and costs of any nature whatsoever. The parties hereto acknowledge that Purchaser obtaining the R&W Insurance Policy is a material inducement to Seller entering into the transactions contemplated by this Agreement, and Seller is relying on Purchaser's covenants and obligations set forth in this Section 6.11.
- 6.12 <u>Registered Office Addresses</u>. To the extent an Acquired Company uses any facility address of Seller or any of its Affiliates (other than the Acquired Companies) as a registered office address, at Purchaser's written request, Seller shall, at Purchaser's sole cost and

expense, take any and all actions to transfer the registered office address of any such Acquired Company to the registered office address of Purchaser or any of its Affiliates effective as of the Closing.

6.13 Exclusivity. During the period from the date of this Agreement until the earlier to occur of (a) the Closing and (b) the termination of this Agreement in accordance with its terms, Seller shall not, and shall cause its Affiliates (including the Acquired Companies) and their respective Representatives not to, directly or indirectly, (i) solicit, initiate or encourage any Acquisition Proposal or (ii) institute, pursue or engage in any discussions or negotiations with, or enter into any understanding, arrangement, agreement, agreement in principle or other commitment (whether or not legally binding) with, or provide any information relating to the Business or any of member of the Seller Group to, any Person or group of Persons (other than Purchaser and its Affiliates and its and their Representatives) in furtherance of an Acquisition Proposal.

#### ARTICLE VII

## **ADDITIONAL COVENANTS OF THE PARTIES**

# 7.1 <u>Transitional Trademark Rights</u>.

- (a) The parties hereto expressly acknowledge and agree that, subject to Section 7.1(b), Section 7.1(c) and Section 7.6, (i) the Acquired Companies, as of the Closing, do not and will not have any right, title or interest (whether express or implied) in, to or under any Seller Group Mark (except for any portion thereof that is Acquired Company IP), and (ii) Purchaser and its Affiliates (including the Acquired Companies) shall have no right, title or interest in or to, or right to use, and Purchaser acknowledges and agrees that it and its Affiliates (including the Acquired Companies) will not hereafter adopt, use, apply to register or register, any such Seller Group Marks. Neither Seller nor any of its Affiliates shall, following Closing, (A) have any rights in or to any Trademarks included in the Acquired Company IP, and shall not hereafter adopt, use, apply to register or register, or authorize others to adopt, use, apply to register or register, any such Trademarks, or (B) contest the ownership or validity of any rights of the Acquired Companies or Purchaser in or to any such Trademarks.
- (b) Notwithstanding the restrictions set forth in Section 7.1(a), no later than twenty (20) days following the Closing Date, Purchaser shall cause the Acquired Companies to, and the Acquired Companies shall, change their names and cause their certificates of incorporation (or equivalent organizational documents), as applicable, to be amended to remove any reference to the Seller Group Marks.
- (c) Promptly (and in any event, within six (6) months) after the Closing Date (the "<u>Transition Period</u>") or a reasonable extension of time as the parties may otherwise agree in writing in good faith, Purchaser shall, and Purchaser shall cause its Affiliates to, cease use of the Seller Group Marks in use as of the Closing Date and take commercially reasonable actions to remove, obliterate or cover up the Seller Group Marks from the any vehicles, equipment,

business cards, purchase orders and invoices, schedules, stationary, business cards, labels, packaging materials, displays, signs, promotional materials, manuals, forms, websites, email, computer software and other materials (collectively, the "<u>Trade Materials</u>"); provided, however, that (i) Purchaser shall, and shall cause its Affiliates (including, for the avoidance of doubt, the Acquired Companies after Closing) to, comply with all reasonable instructions of Seller relating to such utilization of the Seller Group Marks that the Acquired Companies were subject to prior to the Closing Date or that otherwise apply to the Seller Group's use of the Seller Group Marks and (ii) neither Purchaser nor any of its Affiliates (including, for the avoidance of doubt, the Acquired Companies after Closing) shall develop new Trade Materials bearing the Seller Group Marks, it being agreed that Purchaser and the Acquired Companies may continue producing and manufacturing of any Trade Materials in use prior to Closing that bear the Seller Group Marks in connection with their continued operation of the Business during the Transition Period, so long as such use is consistent with the use by the Business prior to Closing (including, for clarity, with respect to the presentation of the Seller Group Marks). Any use by the Acquired Companies of any of the Seller Group Marks as permitted in this Section 7.1(c) is subject to their use of the Seller Group Marks in the same form and manner, and with the same standards of quality, of that used by the Acquired Companies immediately prior to Closing.

(d) Following the Closing Date, Purchaser shall immediately cause the Acquired Companies to cease to hold themselves out as having any affiliation with the Seller Group, except that at any and all times after Closing, Purchaser and the Acquired Companies shall be permitted to communicate to third parties that Purchaser has purchased the Acquired Companies from Seller and reference such name in such communications. Notwithstanding anything in this Agreement to the contrary, and without limiting the rights otherwise granted in Section 7.1(c), nothing shall prevent Purchaser or its Affiliates from using any Trademark (i) to refer to the historical fact that the Business was previously conducted under those Seller Group Marks under which it was conducted; (ii) to the extent already shipped to or in use by customers of Seller or its Affiliates or required or permitted under Contracts of Seller or its Affiliates as of the Closing or required or permitted under Contracts of Purchaser or its Affiliates (including, for clarity, any of the Acquired Companies); and (iii) to the extent required by or permitted as a fair use or otherwise under applicable Legal Requirement (including historical references).

## 7.2 <u>IP Licenses</u>.

(a) Effective as of the Closing, Seller, on behalf of itself and its Subsidiaries, hereby grants to each of the Acquired Companies a perpetual, irrevocable, non-exclusive, worldwide, sublicensable (including through multiple tiers for use in connection with the operation of the Business but not for the independent use by third parties), and royalty-free license to use and exploit all rights under any Intellectual Property (other than Trademarks) owned by Seller Group and its Affiliates as of the Closing that is not Acquired Company IP but was used in the Business within the twelve (12) months prior to the Closing, for use in connection with the conduct of the Business and evolutions and extensions of such Business. The foregoing license is assignable, in whole or in part, in connection with the sale or transfer of any business or assets of the Acquired Companies to which the license relates.

(b) Effective as of the Closing, each of the Acquired Companies hereby grants to Seller a perpetual, irrevocable, non-exclusive, worldwide, sublicensable (including through multiple tiers for use in connection with the operation of the businesses of Seller and its Subsidiaries (other than the Business) but not for the independent use by third parties), and royalty-free license to use and exploit all rights under any Patents, Copyrights and Trade Secrets, in each case, included in the Acquired Company IP that was used in the businesses of Seller and its Subsidiaries within the twelve (12) months prior to the Closing, for use in connection with the conduct of the businesses of Seller and its Subsidiaries (other than the Business) solely as conducted as of the Closing Date and evolutions and extensions of such businesses. The foregoing license is assignable, in whole or in part, in connection with the sale or transfer of any business or assets of Seller or its Subsidiaries to which the license relates.

## 7.3 <u>Post-Closing Access to Information</u>.

- (a) From and after the Closing, Purchaser shall, and shall cause the Acquired Companies and its other Affiliates to, afford the Seller Group and their respective Representatives, during normal business hours, upon reasonable request and advance notice, reasonable access to the employees, books, records and properties of each Acquired Company, their Affiliates and the Business, and to make copies of such books and records at Seller's expense, to the extent that such access is requested in connection with financial statements, Taxes, any potential Proceeding or investigation by or before a Governmental Authority or other Governmental Authority reporting obligations with respect to periods occurring prior to or upon the Closing; provided that the foregoing shall not apply (i) with respect to any information the disclosure of, or access to, which would waive any privilege, violate any Legal Requirement or breach any duty of confidentiality owed to any Person; provided that Purchaser shall use reasonable best efforts to allow for access to the extent that doing so does not result in the loss of any such privilege or violate or breach any such Legal Requirement or duty; or (ii) if such access is sought pursuant to this Agreement in connection with any pending, threatened, or potential claim or Proceeding against Purchaser or its Subsidiaries.
- (b) Purchaser agrees to hold, and to cause the Acquired Companies to hold, all the books and records of the Acquired Companies and the Business existing on the Closing Date and not to destroy or dispose of any such books and records for a period of seven (7) years from the Closing Date or such longer time as may be required by Legal Requirement, and thereafter, if any Acquired Company desires to destroy or dispose of such books and records, to offer first in writing at least sixty (60) days prior to such destruction or disposition to surrender them to Seller.

# 7.4 <u>D&O Insurance</u>.

(a) From and after the Closing until the date that is six (6) years after the Closing Date, Purchaser shall cause the Acquired Companies to, and the Acquired Companies shall, maintain in effect the provisions regarding indemnification, advancement of expenses and exculpation from liability as set forth in the certificate of incorporation, articles of association, bylaws or other organizational documents of the Acquired Companies that have been made available to Purchaser prior to the date hereof, which provisions shall not be amended, repealed or otherwise modified in any manner that would reasonably be expected to adversely affect the

rights thereunder (if any) of any past or present officer or director of any Acquired Company (each, a "<u>D&O Indemnitee</u>") without his/her written consent except to the extent required by applicable Legal Requirement; *provided* that, from and after the Closing, all rights of the D&O Indemnitees to and regarding such indemnification, advancement of expenses and exculpation from liability shall be mandatory rather than permissive. For the avoidance of doubt, Purchaser shall cause the Acquired Companies to maintain in effect such provisions for any claim made by a D&O Indemnitee prior to the expiration of the sixth (6<sup>th</sup>) anniversary of the Closing until such claim has been finally resolved.

- (b) At or prior to the Closing, Seller shall, at Seller's sole cost and expense, obtain six (6)-year irrevocable and non-cancellable directors' and officers' liability, employment practices liability and fiduciary liability "tail" insurance covering the D&O Indemnitees and any other natural persons who are covered by the directors' and officers' liability, employment practices liability and fiduciary liability insurance maintained by or for the benefit of the Acquired Companies in effect as of the Closing (such persons "Insured Persons" and such insurance "Current Insurance"), with respect to any actual or alleged error, misstatement, misleading statement, act, omission, circumstance, event, neglect, breach of duty or any other matter claimed against an Insured Person, in each case, that actually or allegedly existed or occurred at or prior to the Closing (including in connection with the approval or execution of any Transaction Document or the Transactions, or arising out of, relating to or resulting from any Transaction Document and the Transactions), on terms and conditions, including limits and retentions, substantially equivalent to the Current Insurance with respect to Insured Persons; provided that if such "tail" insurance is not reasonably available, then Seller, at Seller's sole cost and expense, shall obtain the most advantageous coverage that is reasonably available. Purchaser shall cause the Acquired Companies to, and the Acquired Companies shall, maintain such "tail" insurance in full force and effect from and after the Closing and shall not, and shall cause the Acquired Companies not to, amend the terms and conditions of such "tail" insurance in any manner that may be adverse to the Insured Persons.
- (c) With respect to any indemnification obligations of the Acquired Companies pursuant to this Section 7.4, Purchaser acknowledges and agrees that: (i) the Acquired Companies shall be the indemnitors of first resort with respect to all indemnification obligations of the Acquired Companies pursuant to this Section 7.4 (i.e., their obligations to an applicable D&O Indemnitee are primary, and any obligation of any other Person to advance expenses or to provide indemnification and/or insurance for the same expenses or liabilities incurred by such D&O Indemnitee are secondary) and (ii) Purchaser shall cause the Acquired Companies to, and the Acquired Companies shall, irrevocably waive, relinquish and release Seller or any of its Affiliates from any and all claims for contribution, subrogation or any other recovery of any kind in respect thereof; provided that nothing in this Agreement is intended to relieve, or shall be construed as relieving, any insurer of its obligations under any insurance policy.
- (d) If, following the Closing until the date that is six (6) years after the Closing Date, Purchaser or any Acquired Company or any of their respective successors or assigns: (i) shall consolidate with or merge into any other corporation or entity and shall not be

the continuing or surviving corporation or entity of such consolidation or merger or (ii) shall transfer all or substantially all of its properties and assets to any Person, then, in each such case, proper provision shall be made to ensure that the successors and assigns of Purchaser or such Acquired Company or any of its respective successors or assigns, as the case may be, shall assume all of the respective obligations set forth in this <u>Section 7.4</u>.

- (e) The rights of the D&O Indemnitees under this Section 7.4 shall be in addition to, and not in substitution for, any other rights such D&O Indemnitee may have under the organizational documents of the Acquired Companies, or under any applicable contracts or Legal Requirements, and Purchaser shall cause each of the Acquired Companies to, and the Acquired Companies shall, honor and perform under all indemnification agreements entered into by the Acquired Companies as in effect as of the date of this Agreement, which have been made available to Purchaser prior to the date hereof.
- (f) The obligations of Purchaser and the Acquired Companies under this <u>Section 7.4</u> shall not be terminated, amended or modified in any manner so as to adversely affect any D&O Indemnitee (including such Person's successors, heirs and legal representatives) to whom this <u>Section 7.4</u> applies without the written consent of such affected D&O Indemnitee (it being expressly agreed that the D&O Indemnitee to whom this <u>Section 7.4</u> applies shall be third-party beneficiaries of this <u>Section 7.4</u>, and this <u>Section 7.4</u> shall be enforceable by such D&O Indemnitee and their respective successors, heirs and legal representatives and shall be binding on all successors and assigns of Purchaser and each Acquired Company).

## 7.5 <u>Non-Competition; Non-Solicitation</u>.

- (a) Subject to Section 7.5(b), as a material inducement to Purchaser to enter into this Agreement, Seller shall not, and shall cause its Affiliates not to, directly or indirectly (whether by itself, through an Affiliate or in partnership or conjunction with or for, or as a member, owner, consultant or agent of, any other Person):
  - (i) for a period of four (4) years following the Closing Date, undertake, manage, participate in, carry on or be engaged in, or in any manner own, operate, or, with respect to any Competing Business Activities, advise, assist or consult with, or have an interest in, any other Person that, directly or indirectly, undertakes, manages, participates in, carries on or is engaged in, any Competing Business Activities anywhere in the world, including as a partner, shareholder, member, employee, principal, agent, trustee or consultant; and
  - (ii) (A) for a period of three (3) years following the Closing Date, solicit, entice, encourage or influence, or attempt to solicit, entice, encourage or influence, or attempt to solicit, entice, encourage or influence, any of individuals listed on Schedule 7.5(a)(ii)(A) of the Seller Disclosure Schedule (collectively, the "Senior Management of the Business") to resign or otherwise leave the employ of Purchaser or the Acquired Companies or otherwise hire, employ, engage or contract with any member of the Senior Management of the Business to perform services other than for the benefit of Purchaser or the Acquired Companies or (B) for a period of two (2) years following

the Closing Date, solicit, entice, encourage or influence, or attempt to solicit, entice, encourage or influence, or attempt to solicit, entice, encourage or influence, any Transferred Employee to resign or otherwise leave the employ of Purchaser or the Acquired Companies or otherwise hire, employ, engage or contract with any Transferred Employee to perform services other than for the benefit of Purchaser or the Acquired Companies.

- (b) Notwithstanding Section 7.5(a)(ii), Sellers and their Affiliates shall not be prohibited from or restricted in any way with respect to: (i) advertising job openings by use of newspapers, magazines, the Internet and other media not specifically directed at individual Transferred Employees or hiring any such Transferred Employees as a result thereof; (ii) hiring or soliciting any Transferred Employee who has terminated employment with Purchaser, the Acquired Companies or any Affiliate thereof, at least nine (9) months prior to the date of first contact by Seller or its Affiliates with such Transferred Employee, or whose employment has been terminated by Purchaser, the Acquired Companies or any Affiliate thereof, so long as there was no solicitation prohibited hereunder by Seller or its Affiliates prior thereto; (iii) continuing to engage in any business (other than the Business) that Seller or any of its Affiliates (other than the Acquired Companies) engages in as of the date of this Agreement, including the operations set forth on Schedule 7.5(b)(iii) of the Seller Disclosure Schedule; (iv) holding as a passive investment not more than five percent (5%) of the outstanding voting securities of any company (whether public or private) that is primarily engaged in Competing Business Activities; or (v) acquiring (in a single transaction or series of related transactions) any Person that is, or has a subsidiary, division, group, franchise or segment that is, engaged in any Competing Business Activity and, following such acquisition, actively engaging in any such Competing Business Activity, so long as for the most recent fiscal year ended prior to the date of such acquisition, the revenues derived from the Competing Business Activities were less than fifteen percent (15%) of the total consolidated revenues of such Person.
- (c) Seller expressly acknowledges that (i) each of the restrictions contained in this Section 7.5 are reasonable in all respects (including with respect to subject matter, geographical scope and time period) and such restrictions are necessary to protect Purchaser's interest in, and value of, the Acquired Companies' businesses (including the goodwill inherent therein), (ii) Seller is primarily responsible for the creation of such value, (iii) the transactions contemplated by this Agreement constitute good, valid and binding consideration for Seller's obligations, covenants and agreements contained in this Section 7.5 and (iv) Purchaser would not have entered into this Agreement or any of the transactions contemplated hereby without the restrictions contained in this Section 7.5 and this Section 7.5 being in full force and effect and binding and enforceable covenants of Seller. If a court of competent jurisdiction finds that the time period of any of the foregoing covenants is too lengthy or the geographic coverage or scope of any of the covenants is too broad, the restrictive time period will be deemed to be the longest period permissible under applicable Legal Requirement and the geographic coverage and scope will be deemed to comprise the largest coverage and scope permissible under applicable Legal Requirements. With respect to each restrictive covenant, the contemplated restricted period shall be extended during the term of any breach of such restrictive covenant.

#### 7.6 Further Assurances.

- (a) From time to time following the Closing, Seller shall, and shall cause its Affiliates to, and Purchaser shall, and shall cause its Affiliates (including the Acquired Companies) to, execute, acknowledge and deliver all reasonable further conveyances, notices, assumptions, releases and acquittances and such other instruments, and shall take such reasonable actions as may be necessary or appropriate to make effective the Transactions as may be reasonably requested by any other party hereto; provided, however, that except as otherwise provided in any Transaction Document, nothing in this Section 7.6 shall require any party hereto or any of their respective Affiliates to expend any money, commence or participate in any Proceeding, incur liabilities or offer or grant any accommodation (financial or otherwise) to any third party following the Closing.
- In furtherance, and not by way of limitation, of the foregoing, if after the Closing: (i) Seller or any of its Affiliates, or Purchaser or any of its Affiliates, discovers that any assets and properties owned, leased or licensed by any Acquired Company that are not used or held for use primarily in connection with the Business were retained by any Acquired Company, then Purchaser shall, and shall cause its Affiliates to: (A) immediately cease using such assets and properties and (B) cooperate with Seller to transfer or assign such assets and properties to Seller (or its designee) with no requirement of additional consideration to the fullest extent permitted by applicable Legal Requirements and execute and deliver any amendments or supplements to this Agreement, exhibits, schedules or the Seller Disclosure Schedule, as applicable, to transfer such assets and properties to Seller (or its designee) effective as of the Closing and (ii) Seller or any of its Affiliates, or Purchaser or any of its Affiliates, discovers that any (A) the IP Seller IP or (B) any assets or properties that are owned, leased or licensed by Seller or any of its Affiliates used or held for use primarily in connection with the Business was not retained by or transferred to any Acquired Company or, in the case of the IP Seller IP, was not transferred to Purchaser or any of its Affiliates at the Closing, then Seller shall, and shall cause its Affiliates to: (A) immediately cease using such assets and properties and (B) cooperate with Purchaser to transfer or assign such assets and properties to Purchaser (or its designee) with no requirement of additional consideration to the fullest extent permitted by applicable Legal Requirements and execute and deliver any amendments or supplements to this Agreement, exhibits, schedules or the Seller Disclosure Schedule, as applicable, to transfer such assets and properties to Purchaser effective as of the Closing. The parties hereto agree to use their reasonable best efforts to structure any transfer of assets and properties referred to in the immediately preceding sentence in a manner that minimizes Taxes.
- (c) Following the Closing, Seller shall, and shall cause its Affiliates to, promptly deliver to Purchaser (i) any mail, packages, orders, inquiries and other communications addressed to Seller or its Affiliates primarily relating to the Business and (ii) any assets or property that Seller or its Affiliates receive and that properly belongs to the Business or an Acquired Company (including any assets or property primarily related to the Business). Following the Closing, Purchaser shall, and shall cause its Affiliates to, promptly deliver to Seller (A) any mail, packages, orders, inquiries and other communications addressed to Seller or any of its Affiliates or relating to a business, product line or asset of Seller or its Affiliates other

than the Business (to the extent not primarily related to the Business) and (B) any assets or property that Purchaser or its Affiliates receive and that properly belongs to Seller or any of its Affiliates (including any assets or property of Seller or its Affiliates that are not primarily related to the Business).

- (d) If, following the Closing, (i) Seller, or any of its Affiliates, receives any funds belonging to the Business or Purchaser or its Affiliates, Seller shall, or shall cause its Affiliates to, (A) promptly advise Purchaser or its applicable Affiliate, (B) segregate and hold such funds in trust for the benefit of Purchaser or its applicable Affiliate and (C) promptly deliver such funds, together with any interest earned thereon, to an account or accounts designated in writing by Purchaser, and (ii) Purchaser or any of its Affiliates receives any funds belonging to Seller or its Affiliates in accordance with the terms of this Agreement, Purchaser shall, or shall cause its Affiliates to, (A) promptly advise Seller or its applicable Affiliate, (B) segregate and hold such funds in trust for the benefit of Seller or its applicable Affiliate and (C) promptly deliver such funds, together with any interest earned thereon, to an account or accounts designated in writing by Seller.
- 7.7 <u>Assignment of Right to Enforce Restrictive Covenants</u>. Prior to or at the Closing, Seller and each of its Affiliates (other than the Acquired Companies) shall assign to the Acquired Companies all rights under any non-disclosure, non-solicitation, non-competition or other restrictive covenants of Contracts to which Seller or any of its Affiliates (other than the Acquired Companies) is a party with any current or former employee or service provider of the Business to the extent that such provisions primarily relate to the Business.

#### ARTICLE VIII

## **TAX MATTERS**

## 8.1 Section 338(h)(10) Election; Purchase Price Allocation.

- (a) Purchaser and Seller shall join in making an election under Section 338(h)(10) of the Code (and any corresponding elections under state, local, or non-U.S. tax law) with respect to the acquisition of the Acquired Companies pursuant to this Agreement (the "Section 338(h)(10) Election"). Neither Seller nor any of its Affiliates shall take any action that would prevent or impede, or could reasonably be expected to prevent or impede, the timely making of the Section 338(h)(10) Election.
- (b) Purchaser and Seller shall (and shall cause their relevant Affiliates to) cooperate in the preparation of IRS Form 8023 and all other forms, attachments and schedules necessary to effectuate the Section 338(h)(10) Election (including any analogous forms required to effectuate the Section 338(h)(10) Election for state or local Tax purposes) (collectively, the "Section 338(h)(10) Forms"). Purchaser and Seller shall (or shall cause their relevant Affiliates to) file such Section 338(h)(10) Forms with the applicable Tax Authorities in accordance with applicable Legal Requirement. Each of Purchaser and Seller agrees that it shall not, and shall not permit any of its Affiliates to, revoke the Section 338(h)(10) Election following the filing of the Section 338(h)(10) Forms without the prior written consent of Seller and Purchaser, respectively.

- (c) Prior to the Closing, Purchaser and Seller shall agree on the form and content of, and at Closing, Purchaser and Seller shall each deliver to the other party, all duly executed Section 338(h)(10) Election Forms.
- (d) For all Tax purposes, Purchaser and Seller agree to (and agree to cause their respective Affiliates to) treat the sale of the shares of the Acquired Companies pursuant to this Agreement in accordance with Treasury Regulations Section 1.338(h)(10)-1 (the "<u>Transaction Tax Treatment</u>").

## (e) Purchase Price Allocation.

- For all Tax purposes, no later than one hundred and twenty (120) days after the Final Purchase Price is finally determined hereunder, Purchaser shall deliver to Seller a proposed allocation of the Final Purchase Price (and any amounts treated as part of the consideration for the Shares and the IP Seller IP for applicable Tax purposes), which allocation shall incorporate, reflect and be consistent with the Transaction Tax Treatment, and be determined in a manner consistent with Sections 338 and 1060 of the Code and the Treasury Regulations promulgated thereunder (the "Purchaser's Draft Allocation"). Purchaser's Draft Allocation shall include the amount allocated to the IP Seller IP. If Seller disagrees with Purchaser's Draft Allocation, Seller may, within thirty (30) days after delivery of Purchaser's Draft Allocation, deliver a notice (the "Seller's Allocation Notice") to Purchaser to such effect, specifying those items as to which Seller disagrees and setting forth Seller's proposed allocation of the Final Purchase Price (and other relevant amounts). If the Seller's Allocation Notice is duly delivered, Purchaser and Seller shall, during the twenty (20) days following such delivery, negotiate in good faith to reach agreement on the disputed items or amounts in order to determine the allocation of the Final Purchase Price (and other relevant amounts). If Purchaser and Seller are unable to reach such agreement, any such disputed items or amounts shall be resolved by the Settlement Accountant pursuant to the procedures set forth in Section 2.4(e), applied mutatis mutandis; provided that any such allocation shall incorporate, reflect and be consistent with the Transaction Tax Treatment. The allocation of the Final Purchase Price (and other relevant amounts) as prepared by Purchaser if no Seller's Allocation Notice has been timely delivered, as adjusted pursuant to any agreement reached by Purchaser and Seller, or as finally determined by the Settlement Accountant, as applicable (the "Allocation"), shall be conclusive and binding on all parties. The Allocation shall be adjusted, as necessary, to reflect any subsequent payments treated as adjustments to the Final Purchase Price. Any such adjustment shall be allocated, consistent with this Section 8.1(e)(i), to the asset, assets, share or shares (if any) to which such adjustment is attributable.
- (ii) Purchaser and Seller shall (and shall cause their Affiliates to) (A) prepare and file all Tax Returns (including the Section 338(h)(10) Forms), in a manner consistent with the Transaction Tax Treatment, the Section 338(h) (10) Election and the Allocation and (B) not take any position inconsistent therewith on any Tax Return, in connection with any Tax Proceeding or otherwise, in each case, except to the

extent otherwise required pursuant to a "determination" within the meaning of Section 1313(a) of the Code (or any similar provision of applicable state, local or non-U.S. law).

(f) Seller shall pay all Taxes attributable to the making of the Section 338(h)(10) Election (such Taxes, the "Section 338(h)(10) Election Taxes").

#### 8.2 <u>Tax Returns</u>.

- (a) Seller shall prepare and timely file, or cause to be prepared and timely filed, when due (taking into account any valid extension of a required filing date) all Tax Returns required to be filed by the Acquired Companies related to Pre-Closing Tax Periods that are due (taking into account any valid extension of a required filing date) on or before the Closing Date solely in respect of those jurisdictions in which the applicable Acquired Company is currently filing Tax Returns (each such Tax Return, a "Seller-Filed Tax Return"). Each such Seller-Filed Tax Return shall be prepared in a manner consistent with past practices of the Acquired Companies, except as otherwise required by a change in applicable Legal Requirements that is effective after the last day of the Taxable period immediately preceding the Taxable period for which the applicable Seller-Filed Tax Return will be filed. Seller shall pay, or cause to be paid, any Taxes shown as due on any Seller-Filed Tax Return at the time such Seller-Filed Tax Return is filed by Seller pursuant to this Section 8.2(a).
- (b) Purchaser (or its Affiliates) shall prepare and timely file, or cause to be prepared and timely filed, when due (taking into account any extensions of a required filing date) all Tax Returns required to be filed by the Acquired Companies related to Pre-Closing Tax Periods (including Straddle Periods) that are due after the Closing Date solely in respect of those jurisdictions in which the applicable Acquired Company is currently filing Tax Returns (each such Tax Return, a "Purchaser-Filed Tax Return shall be prepared in a manner consistent with past practices of the Acquired Companies, except as otherwise required by a change in applicable Legal Requirements that is effective after the last day of the Taxable period immediately preceding the Taxable period for which the applicable Purchaser-Filed Tax Return will be filed.
- (c) Any Purchaser-Filed Tax Return shall be provided in draft form to Seller (together with schedules, statements or other supporting documentation reasonably requested) at least twenty-five (25) Business Days (or, in the case of any Tax Return that is not an income Tax Return, as soon as reasonably practicable) prior to the due date (including any applicable valid extension) of such Purchaser-Filed Tax Return. Seller shall have the right to review and comment on such Purchaser-Filed Tax Return, and Purchaser shall consider in good faith any comments thereto that are provided by Seller to Purchaser in writing at least fifteen (15) Business Days (or, in the case of any Tax Return that is not an income Tax Return, as soon as reasonably practicable) prior to the due date (including any applicable valid extension) of such Purchaser-Filed Tax Return. Purchaser and Seller shall cooperate in good faith to resolve any disputed items with respect to any comments that were timely provided by Seller. If Purchaser and Seller fail to resolve any disputed items within five (5) Business Days following Seller's delivery of

such comments (or within such longer period as the parties may mutually agree), Purchaser and Seller shall submit such disputed items to the Settlement Accountant for resolution, and Purchaser and Seller shall instruct the Settlement Accountant to resolve such disputed items as soon as practicable prior to the due date (including any applicable valid extension) of the Purchaser-Filed Tax Return. The fees and expenses of the Settlement Accountant shall be borne in the manner contemplated by Section 2.4(e), mutatis mutandis. No later than five (5) Business Days prior to the filing of any Purchaser-Filed Tax Return pursuant to this Section 8.2(c), Seller shall pay, or caused to be paid, all Taxes shown as due on any Purchaser-Filed Tax Return at such time and as finally determined pursuant to this Section 8.2(c), except to the extent such Taxes (i) were taken into account in determining the Final Purchase Price or (ii) are allocable to a Post-Closing Straddle Period in accordance with Section 8.5, which Taxes shall be the sole responsibility of Purchaser and its Affiliates (including, after the Closing, the Acquired Companies).

- 8.3 <u>Tax Sharing Agreements</u>. To the extent relating to the Acquired Companies, Seller shall terminate, or cause to be terminated, on or before the Closing Date, the rights and obligations of the Acquired Companies pursuant to all Tax sharing agreements or similar agreements (other than the Transition Services Agreement) if any, to which any of the Acquired Companies, on the one hand, and Seller or any of its Affiliates (other than the Acquired Companies), on the one hand, nor any of the Acquired Companies, on the other hand, shall have any rights or obligations to each other after the Closing in respect of such agreements or arrangements.
- 8.4 <u>Transfer Taxes</u>. Notwithstanding anything to the contrary contained in this Agreement, any Transfer Taxes imposed as a result of the Transactions shall be borne and paid fifty percent (50%) by Purchaser and fifty percent (50%) by Seller. The party responsible under applicable Legal Requirements for filing the Tax Returns with respect to any such Transfer Taxes shall prepare and timely file such Tax Returns and promptly provide a copy of such Tax Return to the other party. The party not responsible under applicable Legal Requirements for the filing of such Tax Returns shall pay its allocable portion of such Transfer Taxes to the party filing such Tax Returns no later than five (5) Business Days prior to the due date of any such Tax Return. Seller shall, and shall cause its Affiliates to, cooperate in connection with the preparation and filing of any such Tax Returns.
- 8.5 <u>Straddle Periods</u>. In the case of any Straddle Period, the amount of any Taxes for a Pre-Closing Straddle Period shall be determined as follows: (a) in the case of any Tax based on or measured by income, receipts, or payroll, be determined based upon an interim closing of the books and hypothetical closing of the taxable year as of 11:59 p.m. on the Closing Date, and (b) in the case of any other Tax, including Taxes imposed on a periodic basis (such as real or personal property Taxes), the amount of such Taxes attributable to a Pre-Closing Straddle Period shall be equal to the amount of Taxes for such Straddle Period *multiplied by* a fraction, the numerator of which is the number of days in the Straddle Period from the beginning of the Straddle Period through and including the Closing Date and the denominator of which is the total number of days in the Straddle Period. The portion of Tax attributable to a Post-Closing Straddle

Period shall be calculated in a corresponding manner. Notwithstanding the foregoing, any Taxes attributable to actions taken outside the ordinary course of business on the Closing Date after the Closing (other than the Section 338(h)(10) Election Taxes) shall be allocated to the portion of the Straddle Period beginning after the Closing Date.

8.6 <u>Cooperation</u>. Following the Closing, Purchaser and Seller shall cooperate, as and to the extent reasonably requested by the other party in writing, in connection the filing of Consolidated Returns and other Tax Returns pursuant to this <u>Article VIII</u>. Such cooperation shall include the provision of such information in such party's possession and assistance, including access to books and records, as is reasonably relevant to any such filing and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided in accordance with this Agreement.

## 8.7 Prohibited Actions.

- (a) Other than as set forth in <u>Section 8.1</u>, Purchaser shall not take, and shall cause its Affiliates (including the Acquired Companies) not to take, any action outside the ordinary course of business after the Closing on the Closing Date (or pursuant to a plan in existence on the Closing Date) that would reasonably be expected to (i) increase any Tax liability of Seller or any of its Affiliates (other than the Acquired Companies) or of any Acquired Company (A) that would be taken into account in determining the Final Purchase Price or (B) for which Seller would otherwise be responsible pursuant to <u>Section 8.2</u> or (ii) result in a payment obligation to a Purchaser Indemnified Party pursuant to Section 12.2.
- (b) Without limiting the generality of Section 8.7(a), other than as set forth in Section 8.1, following the Closing Date, neither Purchaser nor any of its Affiliates (including the Acquired Companies) shall (i) make any election with respect to any Acquired Company or change any method of Tax accounting or any Tax accounting period of any Acquired Company, which election or change would be effective on or prior to the Closing Date, (ii) amend any previously filed Tax Returns of the Acquired Companies for a Pre-Closing Tax Period, (iii) file Tax Returns of the Acquired Companies for a Pre-Closing Tax Period in a manner inconsistent with past practice (except as otherwise provided under Section 8.2) or in a jurisdiction where the relevant Acquired Company has not historically filed Tax Returns, (iv) settle or compromise any Tax audit or similar proceeding with respect to the Acquired Companies for any Pre-Closing Tax Period, (v) initiate discussions or examinations with Tax Authorities regarding Taxes of the Acquired Companies with respect to any Pre-Closing Tax Period or (vi) make any voluntary disclosures with respect to Taxes of the Acquired Companies for Pre-Closing Tax Periods, in each case, without the prior written consent of Seller (such consent not to be unreasonably withheld, conditioned or delayed) if such action would reasonably be expected to (A) increase any Tax liability of Seller or any of its Affiliates (other than the Acquired Companies) or of any Acquired Company (I) that would be taken into account in determining the Final Purchase Price or (II) for which Seller would otherwise be responsible pursuant to Section 8.2 or (B) result in a payment obligation to a Purchaser Indemnified Party pursuant to Section 12.2.

8.8 <u>Survival</u>. This <u>Article VIII</u> shall survive the Closing until ninety (90) days after the expiration of the statute of limitations (including extensions) applicable to the relevant Tax matter.

#### ARTICLE IX

#### **EMPLOYEES**

- 9.1 <u>Transfer of Business Employees.</u> No later than five (5) Business Days prior to the Closing Date, Purchaser shall or shall cause one of its Affiliates to offer employment, in writing, to the Canadian Business Employee, effective as of 12:01 a.m. local time on the Closing Date. Such offer of employment made by Purchaser or one of its Affiliates to the Canadian Business Employee shall provide for terms and conditions consistent with this <u>Article IX</u>, employment in the same or a substantially similar position and work location (within a fifteen (15)-mile radius or such shorter distance required by applicable law in order to avoid the imposition of severance or other termination obligations), in each case, as those provided to such Canadian Business Employee immediately prior to the Closing. Purchaser shall (x) provide the Seller with a copy of the offer of employment to be made pursuant to this <u>Section 9.1</u> at least ten (10) Business Days prior to the making of such offer for the Seller's reasonable review and comment prior to the time that such offer is made and (y) promptly notify the Seller upon making such offer. Each Business Employee who is employed by any Acquired Company as of the Employment Commencement Date and, if the offer of employment with Purchaser or one of its Affiliates is accepted, the Canadian Business Employee shall be referred to herein as a "Transferred Employee."
  - 9.2 <u>Continuation Period</u>. Subject, and in addition, to requirements imposed by applicable Legal Requirements:
- (a) For the period commencing on the Employment Commencement Date and ending on the one-year anniversary of the Employment Commencement Date (the "Continuation Period"), Purchaser shall provide, or shall cause to be provided, to each Transferred Employee: (i) a base salary (or hourly base wage rate) that is at least equal to the base salary (or hourly base wage rate) provided to such Transferred Employee immediately prior to the Employment Commencement Date, (ii) a target annual cash bonus or commission opportunity provided to such Transferred Employee immediately prior to the Employment Commencement Date, (iii) long-term incentive opportunities that are at least equal to the long-term incentive opportunities provided to similarly situated employees of Purchaser, and (iv) employee health, welfare, retirement and fringe benefits and perquisites that are no less favorable in the aggregate than the employee health, welfare, retirement and fringe benefits and perquisites provided or made available to such Transferred Employee immediately prior to the Employment Commencement Date;
- (b) In the event of termination of the employment of any Transferred Employee during the Continuation Period, Purchaser shall provide, or shall cause to be provided, to such Transferred Employee severance pay and benefits no less favorable than the severance pay and benefits to which such Transferred Employee would have been entitled under any

applicable Seller Benefit Plan or Acquired Company Benefit Plan immediately prior to the Employment Commencement Date;

- (c) For purposes of vesting and eligibility, Purchaser shall, or shall cause its applicable Affiliate (including any Acquired Company) to, give each Transferred Employee full credit for all purposes under each employee benefit plan, policy or arrangement of Seller or its applicable Affiliates (other than a defined benefit plan) maintained or made available for the benefit of Transferred Employees as of and after the Employment Commencement Date by Purchaser or any of its Affiliates, for such Transferred Employee's service prior to the Employment Commencement Date with Seller and its applicable Affiliates and their respective predecessors, to the same extent and for the same purpose as such service is recognized by Seller and its applicable Affiliates immediately prior to the Employment Commencement Date; *provided* that such credit shall not be given to the extent that it would result in a duplication of benefits for the same period of service; and
- (d) Purchaser shall use commercially reasonable efforts, or shall use commercially reasonable efforts to cause its applicable Affiliates (including any Acquired Company) to: (i) waive any limitation on health and welfare coverage of such Transferred Employees due to pre-existing conditions, waiting periods, active employment requirements and requirements to show evidence of good health under any applicable health and welfare plan of Purchaser or any of its Affiliates (including any Acquired Company) and (ii) credit each such Transferred Employee with all deductible payments, co-payments and co-insurance paid by such employee under any medical plan of Seller or any of its Affiliates prior to the Employment Commencement Date during the year in which the Employment Commencement Date occurs for the purpose of determining the extent to which any such employee has satisfied any applicable deductible and whether such employee has reached the out-of-pocket maximum under any benefit plan of Purchaser or any of its Affiliates for such year.
- 9.3 <u>Annual Cash Bonuses</u>. Purchaser shall, or shall cause one of its Affiliates (including, following the Closing, the Acquired Companies) to, assume all unpaid cash incentive amounts, including cash bonuses and commissions, that are earned, credited or accrued as of the Employment Commencement Date in respect of each Transferred Employee (the "<u>Assumed Incentive Amount</u>"). Purchaser shall, or shall cause one of its Affiliates (including, following the Closing, the Acquired Companies) to, pay to the Transferred Employees the Assumed Incentive Amount in accordance with the terms of the applicable arrangements in effect immediately prior to the Employment Commencement Date at such time as such amounts would have been paid to the Transferred Employees by Seller or one of its Affiliates had the Closing not occurred; *provided* that the aggregate cash incentive amount paid to Transferred Employees for services rendered for the portion of the performance period elapsing prior to the Employment Commencement Date shall not be less than the Assumed Incentive Amount.
- 9.4 <u>Vacation and Paid Time Off</u>. Unless otherwise required under applicable Legal Requirements, a Collective Bargaining Agreement, Purchaser and its Affiliates shall cause the Acquired Companies to credit each Transferred Employee with, and assume all liabilities for, the amount of accrued but unused vacation time, paid time off and other time-off benefits ("Accrued")

<u>PTO</u>") as such Transferred Employee had with any Acquired Company, Seller or any of its Affiliates as of immediately prior to the Employment Commencement Date. Neither Seller nor its Subsidiaries shall have any liabilities for such Accrued PTO as of the Closing.

- 9.5 <u>Communications</u>. Prior to the Employment Commencement Date, except as otherwise approved in advance and in writing by Seller, Purchaser shall not make any written or oral communications pertaining to the transfer of Business Employees, any compensation or benefits matters or any redundancy and layoff plans, in each case, that may affect the Business Employees in connection with the Transactions.
- 9.6 <u>Immigration Compliance</u>. From and after the date of this Agreement and following the Employment Commencement Date, Purchaser shall, or shall cause its applicable Affiliates to, use commercially reasonable efforts to process and support visa, green card or similar applications in respect of Transferred Employees as of the Employment Commencement Date. Purchaser or its Affiliates (including, following the Closing, the Acquired Companies), as applicable, shall employ any Business Employees who are foreign nationals working in the United States in non-immigrant visa status (the "<u>U.S. Foreign National Employees</u>") under terms and conditions such that Purchaser or its Affiliates (including, following the Closing, the Acquired Companies), as applicable, qualify as a "successor employer" under applicable United States immigration laws effective as of the Employment Commencement Date. As of the Employment Commencement Date, Purchaser and its Affiliates (including, following the Closing, the Acquired Companies) agree to assume all immigration-related liabilities and responsibilities with respect to such U.S. Foreign National Employees.
- 9.7 <u>COBRA</u>. Seller and Purchaser hereby acknowledge and agree that Seller shall be solely responsible for any obligation to provide continuation coverage under Section 4980B of the Code and Part 6 of Subtitle B of Part 1 of ERISA with respect to (i) any "M&A qualified beneficiaries," as defined in Q&A 4 of Treasury Regulation Section 54.4980B-9, in connection with the Transactions and (ii) any Business Employee or Former Business Employee or their beneficiaries receiving or entitled receive to such continuation coverage as of the Closing Date.
- 9.8 No Third-Party Beneficiaries. Nothing contained in this Article IX, express or implied, is intended to confer upon any Person not a party hereto (including any Business Employee or any beneficiary thereof) any right, benefit or remedy of any nature whatsoever, including any right to employment or continued employment for any period of time by reason of this Agreement, any right to a particular term or condition of employment or any right to any specific compensation or benefits. Notwithstanding anything to the contrary contained in this Agreement, no provision of this Agreement is intended to, or does, constitute the establishment of, or an amendment to, any Benefit Plan or similar arrangement.

## ARTICLE X

#### CONDITIONS TO CLOSING

10.1 <u>Conditions of Purchaser</u>. The obligations of Purchaser to consummate the Transactions shall be subject to the satisfaction (or, if permitted by applicable Legal

Requirements, waiver by Purchaser (in its sole discretion)) of each of the following conditions at or prior to the Closing:

- (a) Representations and Warranties of Seller. (i) The representations and warranties of Seller (other than the Seller Fundamental Representations and the representation and warranty of Seller set forth in Section 4.6(b) (Absence of Certain Changes)) contained in Article IV shall be true and correct in all respects as of the Closing as if made on the Closing Date (other than any such representation and warranty that is made as of a specific date, which representation and warranty shall have been true and correct as of such date), except for breaches or inaccuracies, as the case may be, as to matters that, individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect; provided, however, that for purposes of determining the satisfaction of the condition in this clause (i), no effect shall be given to the exceptions or qualifications of "material" or "Material Adverse Effect", or similar "materiality" based exceptions or qualifications in such representations and warranties, (ii) the representation and warranty of Seller set forth in Section 4.6(b) (Absence of Certain Changes) shall be true and correct in all respects as of the Closing as if made on the Closing Date and (iii) the Seller Fundamental Representations shall be true and correct in all material respects as of the Closing as if made on the Closing Date (other than any such representation and warranty that is made as of a specific date, which representation and warranty shall have been true and correct in all material respects as of such date).
- (b) <u>Covenants of Seller</u>. The covenants contained in this Agreement required to be performed or complied with by Seller on or before the Closing shall have been performed or complied with in all material respects.
- (c) <u>No Material Adverse Effect</u>. Since the date of this Agreement, no Material Adverse Effect shall have occurred and be continuing as of the Closing Date.
- (d) <u>Certificate of Seller</u>. Purchaser shall have received a certificate signed by an authorized officer of Seller, dated as of the Closing Date, certifying that the conditions set forth in <u>Section 10.1(a)</u>, <u>Section 10.1(b)</u> and <u>Section 10.1(c)</u> have been satisfied.
- 10.2 <u>Conditions of Seller</u>. The obligations of Seller to consummate the Transactions shall be subject to the satisfaction (or, if permitted by applicable Legal Requirements, waiver by Seller (in its sole discretion)) of each of the following conditions at or prior to the Closing:
- (a) Representations and Warranties of Purchaser. (i) The representations and warranties of Purchaser (other than the Purchaser Fundamental Representations) contained in Article V shall be true and correct in all respects as of the Closing as if made on the Closing Date (other than any such representation and warranty that is made as of a specific date, which representation and warranty shall have been true and correct in all respects as of such date), except for breaches or inaccuracies, as the case may be, as to matters that, individually or in the aggregate, would not reasonably be expected to have a Purchaser Material Adverse Effect; provided, however, that for purposes of determining the satisfaction of the condition in this clause (i), no effect shall be given to the exceptions or qualifications of "material" or "Purchaser Material Adverse Effect", or similar "materiality" based exceptions or qualifications in such

representations and warranties and (ii) the Purchaser Fundamental Representations shall be true and correct in all material respects as of the Closing as if made on the Closing Date (other than any such representation and warranty that is made as of a specific date, which representation and warranty shall have been true and correct in all material respects as of such date).

- (b) <u>Covenants of Purchaser</u>. The covenants contained in this Agreement required to be performed or complied with by Purchaser on or before the Closing shall have been performed or complied with in all material respects.
- (c) <u>Certificate of Purchaser</u>. Seller shall have received a certificate signed by an authorized officer of Purchaser, dated as of the Closing Date, certifying that the conditions set forth in <u>Section 10.2(a)</u> and <u>Section 10.2(b)</u> have been satisfied.
- 10.3 <u>Mutual Conditions</u>. The respective obligations of each party to this Agreement to consummate the Transactions shall be subject to the satisfaction (or, if permitted by applicable Legal Requirements, waiver by mutual consent of Seller and Purchaser) of each of the following conditions at or prior to the Closing:
- (a) <u>Regulatory Approvals</u>. All waiting periods (and any extensions thereof) under the HSR Act applicable to the Transactions shall have expired or early termination shall have been granted.
- (b) <u>No Orders</u>. No Governmental Authority of competent authority and jurisdiction shall have enacted, issued, promulgated, enforced or entered any Order or Legal Requirement that remains in effect and makes illegal or prohibits the Transactions.
- 10.4 <u>Failure of Conditions</u>. Neither Seller nor Purchaser may rely on the failure of any condition set forth in this <u>Article X</u>, as applicable, to be satisfied if such failure was caused by such party's failure to (a) use, as required by this Agreement, its reasonable best efforts to consummate the Transactions or (b) otherwise comply with any provision of this Agreement, in all material respects.
- 10.5 <u>Waiver of Conditions</u>. The conditions set forth in <u>Section 10.1</u> may only be waived by written notice from Purchaser. The conditions set forth in <u>Section 10.2</u> may only be waived by written notice from Seller. The conditions set forth in <u>Section 10.3</u> may only be waived by written notice from both Purchaser and Seller.

#### ARTICLE XI

#### **TERMINATION**

- 11.1 <u>Termination</u>. This Agreement may be terminated at any time prior to the Closing:
  - (a) by the mutual written consent of Purchaser and Seller;
- (b) by Seller, if any representation or warranty of Purchaser shall be inaccurate or Purchaser shall have failed to comply with any covenant or agreement applicable to

Purchaser, in each case that would cause any Closing Condition set forth in Section 10.2 not to be satisfied and such (i) Closing Condition is incapable of being satisfied by the Outside Date or (ii) inaccuracy or failure to perform has not been cured on or prior to the earlier of (A) the date that is thirty (30) days from the date that Purchaser is notified in writing by Seller of such inaccuracy or failure to perform or (B) the day prior to the Outside Date; *provided, however*, that the right to terminate this Agreement under this Section 11.1(b) shall not be available to Seller if any representation or warranty of Seller shall then be inaccurate or Seller shall have failed to comply with any covenant or agreement applicable to Seller, and in each case that would cause any Closing Condition set forth in Section 10.1 not to be satisfied; *provided, further*, that the failure by Purchaser to deliver the Purchase Price at the Closing by the date the Closing is required to have occurred pursuant to this Agreement shall not be subject to cure unless otherwise expressly agreed to in writing by Seller;

- (c) by Purchaser, if any representation or warranty of Seller shall be inaccurate or Seller shall have failed to comply with any covenant or agreement applicable to Seller, in each case that would cause any Closing Condition set forth in Section 10.1 not to be satisfied and such (i) Closing Condition is incapable of being satisfied by the Outside Date or (ii) inaccuracy or failure to perform has not been cured on or prior to the earlier of (A) the date that is thirty (30) days from the date that Seller is notified in writing by Purchaser of such inaccuracy or failure to perform or (B) the day prior to the Outside Date; provided, however, that the right to terminate this Agreement under this Section 11.1(c) shall not be available to Purchaser if any representation or warranty of Purchaser shall then be inaccurate or Purchaser shall have failed to comply with any covenant or agreement applicable to Purchaser, and in each case that would cause any Closing Condition set forth in Section 10.2 not to be satisfied;
- (d) by either Seller or Purchaser, if the Closing has not occurred by April 21, 2025 (the "Outside Date"); provided, however, that the right to terminate this Agreement under this Section 11.1(d) shall not be available to any party that is then in material breach of any covenant contained in this Agreement in any manner that shall have been a principal cause of the failure of the Closing to occur by the Outside Date; provided, further, that (i) if the Condition Satisfaction Date occurs before the Outside Date, then the Outside Date will be extended to the date that is one (1) Business Day after the date on which the Closing is required to occur in accordance with Section 3.1 and (ii) if either Marketing Period has commenced before, and has not completed by, the Outside Date, then the Outside Date will be extended to the date that is one (1) Business Day after such Marketing Period expires; or
- (e) by either Seller or Purchaser (i) in the event that any Governmental Authority shall have issued an Order that permanently enjoins or otherwise makes illegal or prohibits the consummation of the Transactions and such Order shall have become final and non-appealable or (ii) if there shall be a Legal Requirement in effect that permanently makes illegal or prohibits the consummation of the Transactions; *provided*, *however*, that the right to terminate this Agreement under this Section 11.1(e) shall not be available to any party that is then in material breach of any covenant contained in this Agreement in any manner that shall have been a principal cause of the issuance of such Order or imposition of such other Legal Requirement.

- 11.2 <u>Notice of Termination</u>. If this Agreement is terminated pursuant to <u>Section 11.1</u>, written notice of such termination shall be given by the terminating party to the other party (setting forth a reasonably detailed description of the basis on which such party is terminating the Agreement).
- 11.3 Effect of Termination. If this Agreement is terminated in accordance with Section 11.1 and Section 11.2, all rights and obligations of the parties hereto shall terminate without any liability of any party or other Person; provided that (a) the rights and obligations of the parties hereto under Section 6.5 (Confidentiality), the penultimate sentence of Section 6.6(b) (Reasonable Best Efforts; Regulatory Filings), Section 6.7(d), this Section 11.3 (Effect of Termination) and Article XIII (Miscellaneous Provisions) shall survive any termination of this Agreement and (b) nothing herein shall relieve any party to this Agreement from liability for Willful Breach of any covenant or agreement contained herein occurring prior to termination or for claims of Fraud. The parties hereto may petition a court to award damages in connection with any breach by the other party hereto of the terms and conditions set forth in this Agreement, and the parties hereto agree that such damages shall not be limited to reimbursement of expenses or out-of-pocket costs, and may include the benefit of the bargain lost by, or any lost profits or opportunity or consequential damages incurred by, the other party hereto (taking into consideration all relevant matters, including other combination opportunities and the time value of money), in each case in accordance with applicable Legal Requirement.

#### ARTICLE XII

#### SURVIVAL; INDEMNIFICATION

#### 12.1 Survival.

(a) (i) The representations and warranties made by Seller in Article IV, in the certificate delivered by Seller pursuant to Section 10.1(d) shall not survive and shall terminate immediately upon the Closing; (ii) the representations and warranties made by Purchaser in Article V, in the certificate delivered by Purchaser pursuant to Section 10.2(c) shall not survive and shall terminate immediately upon the Closing; (iii) the covenants, obligations and agreements of Seller and Purchaser set forth in this Agreement that, by their terms, contemplate performance in whole prior to the Closing shall survive the Closing until the date that is twelve (12) months following the Closing (the "Pre-Closing Covenants"); and (iv) the covenants, obligations and agreements of Seller and Purchaser set forth that, by their terms, contemplate performance in whole or in part following the Closing by Seller and Purchaser shall survive the Closing in accordance with their terms and until they have been completely performed or complied with by the applicable party hereto. Notwithstanding the foregoing, the indemnification obligations of Seller pursuant to Section 12.2(a)(v) shall survive the Closing until the date that is three (3) years following the Closing. The survival periods set forth in this Section 12.1(a) shall not limit, reduce or restrict the rights or ability of any party to maintain, or recover any amounts in connection with, any claim based on Fraud. In no event shall the termination of any representation, warranty or covenant affect any claim in connection therewith

if notice is provided to the Indemnifying Party in accordance with <u>Section 12.4</u> prior to such termination.

(b) The foregoing is not intended to limit the survival periods set forth in the R&W Insurance Policy.

## 12.2 <u>Indemnification by Seller</u>.

- Subject to the provisions of this Article XII and without limitation to Section 2.1(b), effective as of and after the Closing, Seller shall indemnify, defend and hold harmless Purchaser and its Affiliates and its and their respective directors, officers, employees, agents, successors and representatives (collectively, the "Purchaser Indemnified Parties") from and against any and all Losses incurred or suffered by any of the Purchaser Indemnified Parties, to the extent arising out of, relating to or resulting from, without duplication, any: (i) breach or violation of, or default in connection with, any covenant made by or to be performed by Seller in this Agreement, (ii) Retained Liabilities, (iii) Liabilities of any of the Acquired Companies for Taxes of any member of the Seller Group (other than any of the Acquired Companies) under Treasury Regulations Sections 1.1502-6 or 1.338(h)(10)-1(d)(2) (or any similar provision of state, local, or non-U.S. law), as transferee or successor, or by reason of having been a member of a consolidated, affiliated, combined or other group for Tax purposes at any time before the Closing, (iv) Liabilities set forth on Schedule 12.2(a)(iv) of the Seller Disclosure Schedule, (v) Liabilities set forth on Schedule 12.2(a)(v) of the Seller Disclosure Schedule, and (vi) Section 338(h)(10) Election Taxes. If a Purchaser Indemnified Party's claim may be properly characterized in multiple ways in accordance with this Article XII such that such claim may or may not be subject to different time limitations and other limitations depending on such characterization, then such Purchaser Indemnified Party shall have the right to characterize such claim in a manner that maximizes the recovery and time to assert claims permitted in accordance with this Article XII; provided, however, that no Purchaser Indemnified Party shall be entitled to double recovery for any indemnifiable Losses even though such Losses may have resulted from the breach or inaccuracy of more than one of the representations, warranties and covenants in this Agreement or be subject to indemnification pursuant to multiple clauses of this Section 12.2.
- (b) Seller shall not be liable for any claim for indemnifiable Losses arising out of any claim under (i) Section 12.2(a)(i) relating to any Pre-Closing Covenant (other than the covenants in Section 7.4(b) and Section 8.2(a)) unless and until the aggregate amount of any and all such indemnifiable Losses with respect to Pre-Closing Covenants equals or exceeds the amount set forth in Schedule 12.2(b)(i) of the Seller Disclosure Schedule, in which case the Purchaser Indemnified Parties shall be entitled to recover all such indemnifiable Losses with respect to Pre-Closing Covenants from the first dollar thereof or (ii) Section 12.2(a)(iv) unless and until the aggregate amount of any and all such indemnifiable Losses equals or exceeds the amount set forth in Schedule 12.2(b)(ii) of the Seller Disclosure Schedule (the "Initial Deductible"), in which case the Purchaser Indemnified Parties shall be entitled to recover fifty percent (50%) all indemnifiable Losses arising out of any claim under Section 12.2(a)(iv) in excess of the Initial Deductible until the aggregate amount of any and all indemnifiable Losses arising out of any claim under Section 12.2(a)(iv) equals or exceeds the amount set forth in

<u>Schedule 12.2(b)(iii)</u> of the Seller Disclosure Schedule (the "<u>Second Deductible</u>"), in which case the Purchaser Indemnified Parties shall be entitled to recover all indemnifiable Losses arising out of any claim under <u>Section 12.2(a)(iv)</u> in excess of the Second Deductible.

12.3 <u>Indemnification by Purchaser</u>. Subject to the provisions of this <u>Article XII</u> and without limitation to <u>Section 2.1(b)</u>, effective as of and after the Closing, Purchaser shall indemnify, defend and hold harmless Seller and its Affiliates and its and their respective directors, officers, employees, agents, successors and representatives (collectively, the "<u>Seller Indemnified Parties</u>") from and against any and all Losses incurred or suffered by any of the Seller Indemnified Parties, to the extent arising out of, relating to or resulting from (a) any breach or violation of, or default in connection with, any covenant made by or to be performed by Purchaser in this Agreement and (b) Assumed Liabilities (other than Assumed Liabilities in respect of which Seller is obligated to indemnify Purchaser Indemnified Parties pursuant to <u>Section 12.2</u>). If a Seller Indemnified Party's claim may be properly characterized in multiple ways in accordance with this <u>Article XII</u> such that such claim may or may not be subject to different time limitations and other limitations depending on such characterization, then such Seller Indemnified Party shall have the right to characterize such claim in a manner that maximizes the recovery and time to assert claims permitted in accordance with this <u>Article XII</u>; *provided, however*, that no Seller Indemnified Party shall be entitled to double recovery for any indemnifiable Losses even though such Losses may have resulted from the breach or inaccuracy of more than one of the representations, warranties and covenants in this Agreement or be subject to indemnification pursuant to multiple clauses of this Section 12.3.

## 12.4 Procedures.

(a) In the event that any Proceeding for which an Indemnifying Party may have liability to any Indemnified Party hereunder is actually threatened, asserted against or sought to be collected from any Indemnified Party by a third party and such Indemnified Party has actual knowledge thereof (a "Third-Party Claim"), such Indemnified Party shall promptly (but no later than ten (10) Business Days after such Indemnified Party receives actual notice of such Third-Party Claim) notify the Indemnifying Party in a writing that (i) describes such Third-Party Claim in reasonable detail (including the particular sections of this Agreement pursuant to which indemnification is being sought by the Indemnified Party) and (ii) sets forth the amount or the estimated amount sought thereunder to the extent then reasonably ascertainable (which estimate shall not be conclusive of the final amount recoverable in respect of, or otherwise limit the amount of recovery the Indemnified Party may seek in respect of, such Third-Party Claim) (a "Claim Notice"); provided, however, that a delay in providing a Claim Notice in accordance with this Section 12.4(a) shall not affect the rights of an Indemnified Party hereunder, except (and only to the extent that) any such delay has a direct, material adverse and prejudicial effect on the Indemnifying Party with respect to such Third-Party Claim (in which case, the Indemnifying Party shall be relieved only of any portion of the indemnification liability hereunder that resulted from such delay); provided, further, that for any Third-Party Claims relating to the exposure or alleged exposure of any person to asbestos-containing substances or materials, the Indemnified Party shall not be required to submit a formal Claim Notice to the Indemnifying Party and may instead promptly forward a copy of any complaint, demand letter or similar

documentation to the Indemnifying Party. The Indemnifying Party shall have twenty (20) days (or such lesser number of days set forth in the Claim Notice as may be required in the event of a litigated Proceeding) after receipt of the Claim Notice (the "Notice Period") to notify the Indemnified Party whether the Indemnifying Party desires to assume the control, investigation and defense of such Third-Party Claim. For purposes of the matter set forth on Schedule 12.2(a)(iv) of the Seller Disclosure Schedule (the "Specified Matter"), the parties hereto acknowledge and agree that, notwithstanding anything to the contrary in this Agreement, Seller shall be deemed to have assumed the control, investigation and defense of the Specified Matter; provided that Seller shall (A) keep Purchaser reasonably informed of all substantive developments and events relating to the Specified Matter, (B) reasonably promptly forward copies to Purchaser of any litigation filings or substantive correspondence with other parties with respect to the Specified Matter, (C) provide Purchaser with a reasonable opportunity to review and comment on any proposed substantive litigation filings in connection with the Specified Matter and (D) notify Purchaser in advance of any settlement discussions in connection with the Specified Matter, and confer with Purchaser regarding the strategy and objectives for any such discussions. Notwithstanding anything in this Agreement to the contrary, for so long as Seller has assumed the control, investigation and defense of the Specified Matter, Seller shall be responsible for all legal defense costs (including reasonable attorneys' fees) relating to the Specified Matter.

In the event that, prior to the expiration of the Notice Period, the Indemnifying Party notifies the Indemnified Party in writing that it desires to assume the control, investigation and defense of such Third-Party Claim, subject to Section 12.4(c), (i) the Indemnifying Party shall have the right to control the investigation and defense of such Third-Party Claim at the Indemnifying Party's sole cost and expense, including the appointment, removal or replacement of counsel; provided that, other than with respect to the Specified Matter, the counsel is reasonably acceptable to the Indemnified Party; provided, further, that, other than with respect to the Specified Matter, the Indemnifying Party acknowledges in writing that it is obligated to indemnify the Indemnified Party against any Losses that may be directly or indirectly suffered, paid, incurred or sustained by the Indemnified Party that, directly or indirectly, arise out of, result from or are related to such Third-Party Claim to the extent required hereunder; (ii) the Indemnifying Party shall not settle or compromise or offer to settle or compromise ("Settlement") or consent to the entry of any Order with respect to any Third-Party Claim (including the Specified Matter) without the Indemnified Party's prior written consent (which consent shall not be unreasonably withheld, conditioned or delayed), unless (1) such Settlement or Order does not (I) include any criminal liability or injunctive or non-monetary relief against the Indemnified Party or any of its Affiliates, (II) require any admission of liability (other than with respect to the Specified Matter) or require any admission of a violation of Legal Requirement by the Indemnified Party or any of its Affiliates, or (III) other than with respect to the Specified Matter, require any admission that would have an adverse effect on other claims then pending or threatened in writing against the Indemnified Party or any of its Affiliates that have been made known to the Indemnifying Party, (2) the Indemnifying Party fully indemnifies the Indemnified Party for all Losses arising out of, resulting from or related to the Third-Party Claim that is the subject of such Settlement or Order, and (3) the settlement contains a full and unconditional release of the Indemnified Party; provided that, with respect to a Settlement of the

Specified Matter proposed by Seller, clause (ii)(2) shall be deemed to have been satisfied, and Purchaser shall be deemed to have consented to such Settlement, if Seller bears at least fifty percent (50%) of the Losses with respect to such proposed Settlement of the Specified Matter; (iii) the Indemnified Party shall reasonably cooperate with and assist the Indemnifying Party and its Representatives in the investigation, defense and Settlement of such Third-Party Claim, including by, to the extent permitted by applicable Legal Requirements, (x) furnishing documentary evidence to the extent reasonably available to the Indemnified Party or its Affiliates and (y) providing reasonable access to the Indemnified Party's Representatives, as reasonably necessary to ensure the proper and adequate defense of a Third-Party Claim and (iv) the Indemnified Party shall have the right, but not the obligation, to participate in any such investigation and defense and to employ separate counsel of its choosing (at the Indemnified Party's sole cost and expense, unless, (A) there exists a conflict of interest that would make it inappropriate for the same counsel to represent both the Indemnified Party and the Indemnifying Party, (B) there are one or more defenses available to the Indemnified Party that are not available to the Indemnifying Party or (C) the Indemnified Party assumes the defense of a Third-Party Claim after the Indemnifying Party has failed to defend in good faith a Third-Party Claim it has assumed, as provided in Sections 12.4(a) or 12.4(b), then in each case, the Indemnifying Party shall be liable for the reasonable and documented out-ofpocket fees and expenses of the Indemnified Party for one separate counsel (in addition to any necessary local counsel) to the extent such Third-Party Claim is subject to indemnification or reimbursement under this Article XII); provided that clauses (A), (B) and (C) shall not apply with respect to the Specified Matter.

- (c) Notwithstanding Section 12.4(b), other than with respect to the Specified Matter, if a Third-Party Claim (i) is a Proceeding or threatened Proceeding by a Governmental Authority, (ii) seeks injunctive or other non-monetary relief, that, if granted, would adversely affect the Indemnified Party or any of its Affiliates, (iii) seeks a finding or admission of liability or a violation of any criminal or civil regulatory Legal Requirement by the Indemnified Party or any of its Affiliates, (iv) seeks a finding or admission that would have an adverse effect on other claims actually made or threatened in writing against the Indemnified Party or any of its Affiliates or (v) would materially and adversely affect the ongoing business (including any dispute with any officers, managers, key employees, customers, suppliers, vendors and others having commercial relationships with the Indemnified Party or any of its Affiliates) of the Indemnified Party or any of its Affiliates (any such Third-Party Claim, an "Indemnified Party Defense Matter") then, in each case of the foregoing clauses (i)-(v), the Indemnified Party shall be entitled to assume the control, investigation and defense such Third-Party Claim at the sole expense of the Indemnifying Party and the Indemnifying Party shall have the right, but not the obligation, to participate in any such investigation and defense and to employ separate counsel of its choosing (at the Indemnifying Party's sole cost and expense). Notwithstanding the foregoing, the Indemnified Party shall not affect a Settlement or consent to the entry of any Order of an Indemnified Party Defense Matter, unless such Settlement or consent complies with Section 12.4(b) mutatis mutandis.
- (d) The Indemnifying Party shall give the Indemnified Party a reasonable period to review and comment upon drafts of any documentation relating to any Settlement that

the Indemnifying Party proposed to enter into or Order that the Indemnifying Party proposed to consent to, and the Indemnifying Party shall consider any such comments in good faith.

- (e) If the Indemnifying Party (i) elects not to defend the Indemnified Party against a Third-Party Claim, whether by not giving the Indemnified Party timely notice of its desire to so defend or otherwise, (ii) is not entitled to defend the Third-Party Claim as provided in Section 12.4(c), or (iii) after assuming the defense of a Third-Party Claim, fails to defend in good faith such Third-Party Claim then, in each case, the Indemnified Party shall have the right, but not the obligation, to control the investigation, defense and resolution of such Third-Party Claim, and shall reasonably consult with the Indemnifying Party regarding the strategy for investigation, defense and resolution of such Third-Party Claim, it being understood that the Indemnified Party's right to indemnification for a Third-Party Claim shall not be adversely affected by assuming the defense of such Third-Party Claim.
- (f) The Indemnified Party and the Indemnifying Party shall cooperate in order to allow for the proper and adequate investigation, defense and resolution of a Third-Party Claim, including by providing reasonable access during normal business hours to each other's relevant business records and other documents and employees. The Indemnified Party and the Indemnifying Party shall keep each other reasonably informed with respect to the status of such Third-Party Claim and shall, to the extent permitted by applicable Legal Requirements, deliver to each other copies of all material written notices and documents (including court papers) received by the other that relate to the Third-Party Claim, and the Indemnifying Party, to the extent it is controlling the investigation and defense of such Third-Party Claim, shall in good faith allow the Indemnified Party to propose comments to the materials submitted in such defense (and shall consider such comments in good faith).
- (g) In the event that any Indemnified Party has a claim against any Indemnifying Party under this Article XII for Losses not involving a Third-Party Claim that such Indemnified Party believes gives rise to a claim for indemnification or reimbursement in accordance with the terms of this Article XII, the Indemnified Party shall promptly notify the Indemnifying Party of such Losses in a writing that meets the requirements set forth in Section 12.4(a); provided, however, that a delay in providing such notification in accordance with the requirements set forth in Section 12.4(a) shall not affect the rights of an Indemnified Party hereunder, except (and only to the extent that) any such delay has a direct, material adverse and prejudicial effect on the Indemnifying Party with respect to such claim (in which case, the Indemnifying Party shall be relieved only of any portion of the indemnification obligation hereunder that resulted from such delay).
- (h) Notwithstanding anything in this <u>Section 12.4</u> to the contrary, neither Purchaser nor Seller shall be required to provide access to or disclose any information (i) that is subject to attorney-client privilege, work product protection or trade secret protection or other similar privilege or protection or (ii) if such access or disclosure would (A) or would reasonably be expected to cause material harm to such party or (B) violate applicable Legal Requirements, contravene fiduciary duty or conflict with any Contract by which Purchaser, the Acquired Companies or any of their respective Affiliates is bound; *provided* that the party not providing

access or disclosing information shall advise the other party that the party not providing access or disclosing information is withholding such information and shall use its commercially reasonable efforts to allow for such access or disclosure (or as much of it as possible) to the other party in a manner that does not violate any of the foregoing clause (i) or clause (ii).

12.5 Exclusive Remedy. Purchaser and Seller acknowledge and agree that, except with respect to Fraud and as provided for in the R&W Insurance Policy, and without limiting any right or obligation under any other documents entered into in connection with the Transactions, following the Closing, Section 8.1, Section 8.3, Section 8.4, and this Article XII and specific performance of this Agreement pursuant to Section 13.9 shall be the sole and exclusive remedies of the Seller Indemnified Parties and the Purchaser Indemnified Parties, respectively, for any Losses (including any Losses from claims for breach of contract, warranty, tortious conduct (including negligence) or otherwise and whether predicated on common law, statute, strict liability, or otherwise) that each party hereto may at any time suffer or incur, or become subject to, as a result of or in connection with this Agreement, including any failure by any party to perform or comply with any covenant or agreement that, by its terms, was to have been performed, or complied with, under this Agreement. Without limiting the generality of the foregoing, the parties hereby irrevocably waive any right of rescission they may otherwise have or to which they may become entitled.

# 12.6 Additional Indemnification Provisions.

- (a) Upon making any payment to the Indemnified Party for any indemnification claim pursuant to this <u>Article XII</u>, the Indemnifying Party shall be subrogated, to the extent of such payment, to any rights which the Indemnified Party may have against any third party insurers (other than pursuant to the R&W Insurance Policy) with respect to indemnification claim, and the Indemnified Party shall assign any such rights to the Indemnifying Party to the extent permitted by the applicable insurance policy.
- (b) Each Indemnified Party shall take, and cause its controlled Affiliates to take, commercially reasonable steps to mitigate any Loss upon becoming aware of any event or circumstance that would be reasonably expected to, or does, give rise thereto.
- Limitation on Liability. Notwithstanding anything to the contrary contained in this Agreement (including Article VIII and this Article XII), after the Closing, neither party shall be liable to the other party or its Affiliates or any other Person, whether in contract, tort (including negligence and strict liability) or otherwise, at law or in equity, and "Losses" shall not include any amounts for any special, incidental, consequential or punitive damages (except, in the case of special, incidental or consequential damages only, to the extent such damages are a reasonably foreseeable consequence of the matter giving rise to the applicable Loss); provided that nothing herein shall prevent any Purchaser Indemnified Party or Seller Indemnified Party from being indemnified pursuant to this Article XII and Article VIII for all components of awards against them in any Third-Party Claim, including components of such Third-Party Claim relating to special, incidental, consequential or punitive damages. Notwithstanding any provision herein to the contrary, no indemnity may be sought hereunder in respect of any Losses to the extent such liability (a) is specifically reserved for or otherwise specifically reflected as a

liability of the Acquired Companies in the calculations of Final Closing Indebtedness, Final Closing Net Working Capital or Final Closing Transaction Expenses as of the Closing, or (b) was otherwise actually taken into account as a liability of the Acquired Companies in determining the Final Purchase Price.

- 12.8 <u>Tax Treatment of Indemnity Payments and Post-Closing Adjustments</u>. Except to the extent otherwise required pursuant to a "determination" (within the meaning of Section 1313(a) of the Code or any similar provision of state, local or non-U.S. Legal Requirement), Seller, Purchaser and their respective Affiliates shall treat any and all payments pursuant to <u>Section 2.4</u>, <u>Article VIII</u> or this <u>Article XII</u> as an adjustment to the Final Purchase Price, for Tax purposes.
- <u>Indemnification Obligations Net of Insurance Proceeds and Other Amounts.</u> Any Losses subject to indemnification pursuant to this Article XII shall be calculated (a) net of insurance proceeds actually received by the Indemnified Party for such liability that actually reduce the amount of the Loss ("Insurance Proceeds") and (b) net of any indemnity or contribution proceeds actually received by the Indemnified Party from any third party for such liability that actually reduce the amount of the Loss ("Third Party Proceeds"); provided that for purposes of this Section 12.9, Insurance Proceeds and Third Party Proceeds shall be calculated net of any deductible, retention amount or increased insurance premiums and net of reasonable and documented costs of recovery incurred by the Indemnifying Party in obtaining such recovery. Accordingly, the amount which any Indemnifying Party is required to pay pursuant to this Article XII to any Indemnified Party pursuant to this Article XII shall be reduced by any Insurance Proceeds or Third Party Proceeds theretofore actually recovered by or on behalf of the Indemnified Party in respect of the related Loss. If an Indemnified Party receives a payment required by this Agreement from an Indemnifying Party in respect of any Loss (an "Indemnity Payment") and subsequently receives Insurance Proceeds or Third Party Proceeds, then the Indemnified Party shall pay to the Indemnifying Party an amount equal to the excess of the Indemnity Payment received over the amount of the Indemnity Payment that would have been due if the Insurance Proceeds or Third Party Proceeds had been received, realized or recovered before the Indemnity Payment was made. The Indemnified Party shall use commercially reasonable efforts to seek to collect or recover any Insurance Proceeds and any Third Party Proceeds to which the Indemnified Party is entitled in connection with any Loss for which the Indemnified Party seeks indemnification pursuant to this Article XII; provided, however, that in no event shall the Indemnified Party be required to (A) file one or more lawsuits or commence any other Proceeding or (B) seek to first recover under the R&W Insurance Policy, any insurance policy or any other third party; provided, further, that the Indemnified Party's inability to collect or recover any such Insurance Proceeds or Third Party Proceeds shall not limit the Indemnifying Party's obligations hereunder. Notwithstanding anything to the contrary provided herein, Purchaser Indemnified Parties shall have no obligation to seek to first collect or first recover any Insurance Proceeds or Third Party Proceeds prior to obtaining indemnification from the Indemnifying Party pursuant to this Article XII.

## ARTICLE XIII

## **MISCELLANEOUS PROVISIONS**

- 13.1 <u>Expenses</u>. Whether or not the Transactions are consummated, unless otherwise expressly provided herein, and except as otherwise specified in the other Transaction Documents, each party hereto shall pay its own costs and expenses in connection with this Agreement and the Transactions, including the fees and expenses of its advisers, accountants and legal counsel.
- Interpretation. Except as otherwise explicitly specified to the contrary, (a) references to a Section, Article, exhibit 13.2 or schedule means a Section or Article of, or schedule or exhibit to this Agreement, unless another agreement is specified, (b) references to a subsection or other subdivision without further reference to a Section is a reference to such subsection or subdivision as contained in the same Section in which the reference appears, (c) the word "including" (and words of similar import) means "including, without limitation," (d) references to a particular statute or regulation include all rules and regulations thereunder and any predecessor or successor statute, rules or regulation, in each case, as amended or otherwise modified from time to time, (e) words in the singular or plural form include the plural and singular form, respectively, and words of one gender shall be held to include all genders as the context requires, (f) references to the parties means the parties hereto, unless another agreement is specified, (g) references to a particular Person include such Person's successors and assigns to the extent not prohibited by this Agreement, (h) "extent" in the phrase "to the extent" means the degree to which a subject or other thing extends, and such phrase does not mean simply "if," (i) the headings contained in this Agreement, in any exhibit or schedule hereto and in the table of contents to this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement, (j) references to "\$" means United States dollars and all calculations of a number of dollars shall be rounded to the nearest whole number of cents, as applicable, with 0.5 rounded up to the next whole cent, as applicable (aggregating all payments to be made to any Person prior to such rounding), (k) "U.S." or "United States" means the United States of America, (l) the word "either," "or," "neither," "nor" and "any" are not exclusive, (m) the words "hereof," "herein," "hereby," "hereto," and derivative or similar words refer to this entire Agreement including the schedules and exhibits hereto, (n) the word "any" means "any and all," (o) the words "writing," "written" and comparable terms refer to printing, typing and other means of reproducing words (including electronic media) in a visible form, (p) no provision of this Agreement is to be construed to require, directly or indirectly, any Person to take any action, or omit to take any action, to the extent such action or omission would violate applicable Legal Requirements, (q) the term "made available to Purchaser" and words of similar import means that copies of relevant documents, instructions or materials were made available to Purchaser and Purchaser's Representatives in the data room on or prior to 11:59 p.m. Eastern time on July 20, 2024 and were so available from such time through the date hereof, (r) whenever this Agreement refers to a number of days, such number shall refer to calendar days unless Business Days are specified, and if the last day of the time period for the giving of any notice or the taking of any action required under this Agreement falls on a day that is not a Business Day, the time period for giving such notice or taking such action shall be extended

through the next Business Day following the original expiration date of such, (s) when calculating the period of time before which, within which or following which any action is to be taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded and the ending date shall be included, (t) the phrases "ordinary course" and "ordinary course of business" mean "ordinary course of business consistent with past practices" and (u) Seller and Purchaser have each participated in the negotiation and drafting of the Transaction Documents and if an ambiguity or question of interpretation should arise, the Transaction Documents shall be construed as if drafted jointly by the parties thereto and no presumption or burden of proof shall arise favoring or burdening any party by virtue of the authorship of any of the provisions in the Transaction Documents.

- 13.3 <u>Entire Agreement</u>. This Agreement and the other Transaction Documents, including the other documents, agreements, exhibits and schedules specifically referred to herein and therein, constitute the entire agreement between and among the parties hereto with regard to the subject matter hereof and supersede all prior agreements and understandings with regard to such subject matter.
- Amendment and Waivers. This Agreement shall not be amended, modified, altered or supplemented in whole or in part, except by supplemental agreement or amendment signed by Seller and Purchaser. No failure or delay by a party in exercising any right or remedy provided by Legal Requirement or under this Agreement shall impair such right or remedy or operate or be construed as a waiver or variation of it or preclude its exercise at any subsequent time and no single or partial exercise of any such right or remedy shall preclude any further exercise of it or the exercise of any other remedy. Any extension or waiver of the requirements hereunder shall be valid only if set forth in an instrument in writing by the party to be bound hereby. The waiver by any party to this Agreement of a breach of any term or provision of this Agreement shall not be construed as a waiver of any other provision or any subsequent breach. Notwithstanding the foregoing, in the event that any party seeks an amendment, modification, consent or waiver to any of (a) this sentence, (b) Section 13.14, (c) Section 13.7(c), (d) Section 13.11(c) or (e) the definition of "Financing Source" herein, in each case, that adversely affects the Financing Sources, the prior written consent of the applicable Financing Entity shall be required before any such amendment, modification, consent or waiver may become effective.
- 13.5 <u>Successors and Assigns</u>. This Agreement shall bind and inure to the benefit of the parties hereto and their respective successors and permitted assigns; *provided, however*, that no party hereto may assign any right or obligation hereunder without the prior written consent of the other parties hereto, and any assignment in violation of this <u>Section 13.5</u> shall be null and void. Notwithstanding the foregoing, (a) Seller may transfer or assign in whole or in part, its rights and interests (but not its obligations) hereunder to any of its Affiliates without Purchaser's prior written consent (but with notice to Purchaser), (b) Purchaser may transfer or assign in whole or in part, its rights and interests (but not its obligations) hereunder to any of its controlled Affiliates without Seller's prior written consent (but with notice to Seller) except to the extent that such transfer or assignment would require an amount of Taxes to be withheld under applicable Legal Requirement from any amounts otherwise payable hereunder in excess of the amount of Taxes

that would be required to be withheld absent such transfer or assignment and (c) Purchaser may transfer or assign, in whole or in part, its rights and interests (but not its obligations) hereunder to (i) any Financing Source (or any agent on behalf of all such Financing Sources) as collateral for security purposes in connection with the Debt Financing (including for purposes of creating a security interest herein or otherwise assigning as collateral in respect of such financing) or (ii) the R&W Insurance Policy provider (but solely in the case of Fraud). Notwithstanding anything to the contrary in this Section 13.5, no assignment shall relieve the assigning party of its obligations hereunder.

13.6 <u>Governing Law</u>. This Agreement, the rights of the parties and all Proceedings arising in whole or in part under or in connection herewith, will be governed by and construed and enforced in accordance with the laws of the State of Delaware, without giving effect to any choice or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the laws of any other jurisdiction.

## 13.7 <u>Jurisdiction; Venue; Service of Process</u>.

- (a) Each of the parties hereto, by its execution hereof, hereby (i) irrevocably submits to the exclusive jurisdiction of the Delaware Chancery Court (or, if the Delaware Chancery Court declines to accept jurisdiction, any United States District Court located in the State of Delaware, or if the United States District Courts located in the State of Delaware decline to accept jurisdiction, any state court of the State of Delaware) for the purpose of any Proceeding among the parties hereto relating to or arising in whole or in part under or in connection with the Transaction Documents or the Transactions, (ii) waives to the extent not prohibited by applicable Legal Requirements, and agrees not to assert, by way of motion, as a defense or otherwise, in any such Proceeding, any claim that it is not subject personally to the jurisdiction of the above-named courts, that its property is exempt or immune from attachment or execution, that any such Proceeding brought in one of the above-named courts should be dismissed on grounds of forum non conveniens, should be transferred or removed to any court other than one of the above-named courts, or should be stayed by reason of the pendency of some other Proceeding in any other court other than one of the above-named courts or that the Transaction Documents or the subject matter hereof or thereof may not be enforced in or by such court and (iii) agrees not to commence any such Proceeding other than before one of the above-named courts. Notwithstanding the previous sentence, a party may commence any Proceeding in a court other than the above-named courts solely for the purpose of enforcing an order or judgment issued by one of the above-named courts.
- (b) Each of the parties hereto (i) consents to service of process in any Proceeding among any of the parties hereto relating to or arising in whole or in part under or in connection with the Transaction Documents or the Transactions in any manner permitted by Delaware law, (ii) agrees that service of process made in accordance with the foregoing clause (i) or made by registered or certified mail, return receipt requested, at its address specified pursuant to Section 13.13, will constitute good and valid service of process in any such Proceeding and (iii) waives and agrees not to assert (by way of motion, as a defense, or otherwise) in any such Proceeding any claim that service of process made in accordance with the foregoing clause (i) or

- (ii) does not constitute good and valid service of process. This <u>Section 13.7</u> shall not apply to any dispute that is required to be decided by the Settlement Accountant.
- (c) Notwithstanding the foregoing and without limiting Sections 13.6, 13.7(a) and 13.7(b), each of Seller and the Acquired Companies (on behalf of themselves and their Affiliates) (i) agrees that it will not bring or support any action, cause of action, claim, cross-claim or third-party claim of any kind or description, whether in law or in equity, whether in contract or in tort or otherwise, against the Financing Sources in any way relating to this Agreement, the Debt Financing or any of the transactions contemplated hereby or thereby, including, but not limited to, any dispute arising out of or relating in any way to the Debt Commitment Documents or any other letter or agreement related to the Debt Financing or the performance thereof (any such action, a "Financing Related Action") in any forum other than any State or Federal court sitting in the Borough of Manhattan in the City of New York, (ii) waives any right to trial by jury in respect of any Financing Related Action and (iii) agrees that any Financing Related Action shall be governed by, and construed in accordance with, the laws of the State of New York, without regard to the conflicts of law rules of such State that would result in the application of the laws of any other State; except as otherwise expressly provided in the Debt Commitment Documents.
- Waiver of Jury Trial. TO THE EXTENT NOT PROHIBITED BY APPLICABLE LEGAL REQUIREMENTS THAT CANNOT BE WAIVED, THE PARTIES HERETO KNOWINGLY, IRREVOCABLY AND UNCONDITIONALLY WAIVE, AND COVENANT THAT THEY WILL NOT ASSERT (WHETHER AS PLAINTIFF, DEFENDANT OR OTHERWISE), ANY RIGHT TO TRIAL BY JURY IN ANY PROCEEDING ARISING IN WHOLE OR IN PART UNDER OR IN CONNECTION WITH THE TRANSACTION DOCUMENTS OR ANY OF THE TRANSACTIONS, WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER SOUNDING IN CONTRACT, TORT OR OTHERWISE. THE PARTIES HERETO AGREE THAT ANY OF THEM MAY FILE A COPY OF THIS PARAGRAPH WITH ANY COURT AS WRITTEN EVIDENCE OF THE KNOWING, VOLUNTARY AND BARGAINED-FOR AGREEMENT BETWEEN THE PARTIES HERETO UNCONDITIONALLY AND IRREVOCABLY TO WAIVE THEIR RESPECTIVE RIGHTS TO TRIAL BY JURY IN ANY PROCEEDING WHATSOEVER BETWEEN OR AMONG THEM RELATING TO THE TRANSACTION DOCUMENTS OR ANY OF THE TRANSACTIONS AND THAT SUCH PROCEEDINGS WILL INSTEAD BE TRIED IN A COURT OF COMPETENT JURISDICTION BY A JUDGE SITTING WITHOUT A JURY. EACH PARTY HERETO CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF ANY PROCEEDING, SEEK TO ENFORCE THE FOREGOING WAIVER, (B) EACH SUCH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (C) EACH SUCH PARTY MAKES THIS WAIVER VOLUNTARILY AND (D) EACH SUCH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 13.8.

## 13.9 Specific Performance.

- (a) Each party hereto acknowledges and agrees that (i) the other party will be irreparably damaged if this Agreement is not performed in accordance with its terms and (ii) any breach of this Agreement would not be adequately compensated by monetary damages, even if available. Accordingly, in addition to any other right or remedy to which a party may be entitled, at law or in equity, that party shall be entitled to equitable relief (including an injunction or injunctions) to prevent breaches or threatened breaches of this Agreement and to compel specific performance of the terms and provisions of this Agreement (including to require the other party to consummate the Closing as contemplated hereby), without (A) the need for proof of actual damages and (B) the requirement of posting any bond or other security or indemnity. Furthermore, each party hereto agrees not to raise (1) any objections to the availability of the equitable remedy of specific performance to prevent or restrain breaches of this Agreement, or to specifically enforce the terms of this Agreement (including to require the other party to consummate the Closing as contemplated hereby) (except that the party opposing specific performance may contest the existence of any breach or threatened breach), or (2) that a remedy of monetary damages would provide an adequate remedy for any such breach.
- (b) Each party hereto further agrees that (i) by seeking the remedies provided for in this Section 13.9 a party shall not in any respect waive its right to seek any other form of relief that may be available to such party under this Agreement or in the event that the remedies provided for in this Section 13.9 are not available or otherwise are not granted, and (ii) nothing set forth in this Section 13.9 shall require any party to institute any Proceeding for (or limit any party's right to institute any action for) specific performance under this Section 13.9 prior to or as a condition to exercising any termination right under Article XI, nor shall the commencement of any action pursuant to this Section 13.9 or anything set forth in this Section 13.9 restrict or limit any such party's right to terminate this Agreement in accordance with Article XI or pursue any other remedies under this Agreement that may be available then or thereafter. If any party hereto brings any Proceeding to enforce specifically the performance of the terms and conditions hereof by the other party, the party bringing such action may petition a court to extend the Outside Date (notwithstanding the termination provisions of Section 11.1(d)) (and the other party hereto shall not assert that the extension of the Outside Date is an inappropriate remedy if the other party is in material breach of any covenant contained in this Agreement in any manner that shall have been a principal cause of the failure of the Closing to occur by the Outside Date), so long as the party bringing such action is actively seeking a court order for an injunction or injunctions or to specifically enforce the terms and provisions of this Agreement.
- 13.10 <u>Severability</u>. Each of the provisions of this Agreement is severable, such that if any such provision is held to be or becomes invalid or unenforceable in any respect under the Legal Requirements of any jurisdiction, it shall have no effect in that respect in such jurisdiction and the parties hereto shall use all reasonable efforts to replace it in that respect with a valid and enforceable substitute provision the effect of which is as close to its intended effect as possible. Notwithstanding the foregoing, the parties hereto intend that the remedies and limitations thereon contained in <u>Article XI</u> and <u>Article XII</u> be construed as an integral provision of this Agreement

and that such remedies and limitations shall not be severable in any manner that increases a party's liability or obligations hereunder or under the Debt Financing Commitment.

## 13.11 Certain Releases.

- (a) Except in the case of Fraud or as otherwise expressly set forth in any Transaction Document, Purchaser, for itself and on behalf of its controlled Affiliates (including, after the Closing, the Acquired Companies) and its and their respective successors and assigns, in each case, in their capacity as such, acknowledges and agrees that, from and after the Closing, to the fullest extent permitted by Legal Requirement, any and all rights, claims, demands, obligations, liabilities, defenses, setoffs, counterclaims, actions and causes of action it has, had, may have or may have had against any Seller or any of its Affiliates (other than the Acquired Companies) or its or their Representatives (each, a "Seller Releasee"), in each case in their capacity as such, to the extent arising out of (i) Seller's ownership or operation of the Business, (ii) the negotiation, execution or performance of any Transaction Document or the Transactions, (iii) any inaccuracy or breach of any representation or warranty or the breach of any covenant, undertaking or other agreement contained in this Agreement, the Seller Disclosure Schedules and Exhibits hereto or in any certificate contemplated hereby and delivered in connection herewith or (iv) any information (whether written or oral), documents or materials furnished in connection with the Transactions, are hereby irrevocably and unconditionally waived and released and covenant not to initiate or maintain any Proceeding relating to the foregoing against any Seller Releasee; provided that nothing contained in this Agreement shall release, waive, discharge, relinquish or otherwise affect the rights or obligations of any party hereto or any of their respective Affiliates with respect to enforcing the terms of any Transaction Document.
- (b) Except in the case of Fraud or as otherwise expressly set forth in any Transaction Document, Seller, for itself and on behalf of its controlled Affiliates, and its and their respective successors and assigns, in each case, in their capacity as such, acknowledges and agrees that, from and after the Closing, to the fullest extent permitted by Legal Requirement, any and all rights, claims, demands, obligations, liabilities, defenses, setoffs, counterclaims, actions and causes of action it has, had, may have or may have had against any of (x) the Acquired Companies or their Representatives, in each case, in their capacity as such, or (y) Purchaser or any of its Affiliates (other than the Acquired Companies) or their Representatives (each of the Persons described in clauses (x) and (y), a "Purchaser Releasee"), in each case, in their capacity as such and in each case to the extent arising out of (i) solely in the case of the Acquired Companies, the operation of the Business prior to the Closing Date, (ii) the negotiation, execution or performance of any Transaction Document or the Transactions, (iii) any inaccuracy or breach of any representation or warranty or the breach of any covenant, undertaking or other agreement contained in this Agreement, the Purchaser Disclosure Schedule and Exhibits hereto or in any certificate contemplated hereby and delivered in connection herewith or (iv) any information (whether written or oral), documents or materials furnished in connection with the Transactions, are hereby irrevocably and unconditionally waived and released and covenant not to initiate or maintain any Proceeding relating to the foregoing against any Purchaser Releasee; provided that nothing contained in this Agreement shall release, waive, discharge, relinquish or

otherwise affect the rights or obligations of any party hereto or any of their respective Affiliates with respect to enforcing the terms of any Transaction Document.

- (c) Notwithstanding anything in this Agreement to the contrary, each of Purchaser and Seller acknowledge and agree that no recourse under this Agreement or any documents or instruments delivered in connection with this Agreement shall be had against any Seller Releasee (other than Seller) or any Purchaser Releasee (other than Purchaser), as applicable, whether by the enforcement of any assessment or by any legal or equitable proceeding, or by virtue of any statute, regulation or other applicable Legal Requirement, it being expressly agreed and acknowledged that no personal liability whatsoever shall attach to, be imposed on or otherwise be incurred by any Seller Releasee (other than Seller) for any obligation of Seller under this Agreement or any documents or instruments delivered in connection with this Agreement or by any Purchaser Releasee (other than Purchaser) for any obligation of Purchaser under this Agreement or any documents or instruments delivered in connection with this Agreement, for any claim based on, in respect of or by reason of such obligations or their creation.
- (d) Notwithstanding anything in this Agreement to the contrary, Seller and its Affiliates will not have any rights or claims, regardless of the legal theory under which such right or claim may be asserted, whether sounding in contract or tort, or whether at law or in equity, or otherwise under any legal or equitable theory, and will not seek any such rights or claims against any of the Financing Sources in connection with this Agreement, the Debt Commitment Documents or the Debt Financing, and no Financing Source shall have any liability to Seller or any of its Affiliates for any obligations or liabilities of the parties or for any claim (regardless of the legal theory under which such claim may be asserted, whether sounding in contract or tort, or whether at law or in equity, or otherwise under any legal or equitable theory), based on, in respect of, or by reason of, the transactions contemplated hereby, the Debt Commitment Documents or the Debt Financing. Nothing in this Section 13.11 shall in any way limit or qualify the rights and obligations of the Financing Sources and the other parties to the Debt Commitment Documents or the Debt Financing (or the Definitive Agreements) to each other thereunder or in connection therewith.
- (e) The parties hereto unconditionally and irrevocably acknowledge and agree that (i) the agreements contained in this Section 13.11 are an integral part of this Agreement and the Transactions and (ii) without the agreements set forth in this Section 13.11, the other party would not enter into this Agreement or otherwise agree to consummate the Transactions.
- 13.12 The Seller Disclosure Schedule, Exhibits and Schedules. The Seller Disclosure Schedule, schedules and exhibits attached to this Agreement shall be construed with and as an integral part of this Agreement to the same extent as if the same had been set forth verbatim herein. The representations and warranties of Seller set forth in this Agreement are made and given subject to the disclosures contained in the Seller Disclosure Schedule. Inclusion of information in the Seller Disclosure Schedule will not be construed as an admission that such information is material to the business, operations or condition (financial or otherwise) of Seller, the IP Seller IP, the Business or any Acquired Company. Disclosure of any matter in any

Section of the Seller Disclosure Schedule shall be deemed to be disclosure of such matter with respect to any other schedule of the Seller Disclosure Schedule to which such matter is specifically cross referenced or to which such matter relates to the extent it is reasonably apparent on the face of such disclosure that such disclosure applies to the relevant representation or warranty of such other Section. The headings contained in the Seller Disclosure Schedule are inserted for convenience only and shall not be considered in interpreting or construing any of the provisions contained in this Agreement.

13.13 Notices. Any notice required or permitted to be given hereunder shall be sufficient if in writing and shall be deemed to have been duly given or made: (a) when personally delivered, (b) the day of sending, if sent by email transmission prior to 9:00 p.m. Eastern time on any Business Day or the next succeeding Business Day if sent after 9:00 p.m. Eastern time on any Business Day or on any day other than a Business Day or (c) one (1) Business Day after deposit with an overnight courier service, in each case to the addresses, email addresses and attention parties indicated below (or such other address, email address or attention party as the recipient party has specified by prior notice given to the sending party in accordance with this Section 13.13):

#### To Purchaser at:

Terex Corporation
301 Merritt 7, 4th Floor
Norwalk, CT 06851
Attention: Scott Posner
Email: [Email]

With copies (which shall not constitute notice) to:

Fried, Frank, Harris, Shriver & Jacobson LLP One New York Plaza New York, NY 10004 Attention: Maxwell Yim Philip Richter Email: Maxwell.Yim@FriedFrank.com Philip.Richter@FriedFrank.com

# To Seller at:

Dover Corporation
3005 Highland Parkway
Downers Grove, IL 60515
Attention: Ivonne M. Cabrera, SVP & General Counsel;
Matthew S. Gaudette, VP & Associate Counsel, M&A
Email: [Email];
[Email]

With copies (which shall not constitute notice) to:

Skadden, Arps, Slate, Meagher & Flom LLP
One Manhattan West
New York, NY 10001
Attention: Brandon Van Dyke
Dohyun Kim
Email: Brandon.VanDyke@skadden.com
Dohyun.Kim@skadden.com

13.14 No Third-Party Beneficiaries. Except for Section 7.4 (with respect to the D&O Indemnitees and their respective successors, heirs and legal representatives), Section 13.11 (with respect to the Seller Releasees and Purchaser Releasees), Section 13.15 (with respect to Skadden) and this Section 13.14, the last sentence of Section 13.4, Section 13.7(c) and Section 13.11(c) (with respect to the Financing Sources), a Person who is not a party to this Agreement shall have no right to enforce any of its terms and this Agreement is not intended to give any Person other than the parties to this Agreement and their permitted assigns any rights hereunder.

13.15 Provision Regarding Legal Representation. It is acknowledged by each party hereto that Seller has retained Skadden to act as its counsel in connection with the Transactions and that Skadden has not acted as counsel for any other party in connection with such Transactions. The parties hereto agree that, in the event that a dispute arises after the Closing between Purchaser, any Acquired Company or their Affiliates, on the one hand, and Seller or its Affiliates, on the other hand, Skadden may represent Seller and its Affiliates in such dispute even though the interests of Seller and its Affiliates may be directly adverse to Purchaser, the Acquired Companies or their Affiliates, and even though Skadden may have represented any of the Acquired Companies or any of their Affiliates in a matter substantially related to such dispute. Purchaser further agrees that, all communications among Seller, the Acquired Companies or any of their respective Affiliates, on the one hand, and their counsel, including Skadden, on the other hand, that relate in any way to the Transactions shall be deemed attorney-client privileged communications (collectively, the "Privileged Communications") and the attorney-client privilege and the expectation of client confidence belongs to Seller and may be controlled by Seller and, notwithstanding anything to the contrary contained in this Agreement, shall not pass to or be claimed by Purchaser, any Acquired Company or any of their Affiliates. The Privileged Communications are (and upon the Closing shall remain) the property of Seller, and from and after the Closing, none of Purchaser, the Acquired Companies, their Affiliates or any Person purporting to act on behalf of or through Purchaser, the Acquired Companies or their Affiliates will seek to access, use or obtain such communications, whether by seeking a waiver of the attorney-client privilege or through other means. As to any such Privileged Communications made prior to the Closing Date, Purchaser, together with its Affiliates (including the Acquired Companies), successors and assigns, further agree that no such party may access, use or rely on any of the Privileged Communications in any Proceeding against or involving any of the parties hereto after the Closing. The Privileged Communications may be used by Seller in connection with any dispute that relates in any way to the Transactions. If Seller uses any Privileged Communications in connection with any dispute such documents and

information shall no longer be deemed to be protected by any privilege; *provided* that Purchaser and Seller reserve their rights to seek or oppose discovery relating to any other Privileged Communications. Notwithstanding the foregoing, in the event that a dispute arises between Purchaser, any Acquired Company or any of their Affiliates, on the one hand, and any other Person or Persons (other than a party to this Agreement or any of its respective Affiliates), on the other hand, after the Closing, such Acquired Company and its Affiliates may assert the attorney-client privilege to prevent disclosure of the Privileged Communications to such Person or Persons; *provided, however*, that none of the Acquired Companies nor their Affiliates may waive such privilege without the prior written consent of Seller.

- 13.16 <u>No Other Duties</u>. The only duties and obligations of the parties hereto under this Agreement are as specifically set forth in this Agreement, and no other duties or obligations shall be implied in fact, Legal Requirement or equity, or under any principle of fiduciary obligation.
- 13.17 <u>Reliance on Counsel and Other Advisors</u>. Each party hereto has consulted such legal, financial, technical or other experts as it deems necessary or desirable before entering into this Agreement.
- 13.18 Public Announcements. None of Seller or its Affiliates, Purchaser or its Affiliates or any representative of any such party shall issue or cause the publication of any press release or public announcement in respect of this Agreement or the Transactions without the prior written consent of the other party (which consent shall not be unreasonably withheld, conditioned or delayed), except (a) as may be required by Legal Requirement or stock exchange rules or as Seller or Purchaser deems necessary or advisable to comply with their respective SEC filing requirements, in which case the party seeking to publish such press release or public announcement shall use reasonable efforts to provide the other party a reasonable opportunity to comment on such press release or public announcement in advance of such publication; provided that the foregoing will not restrict or prohibit Purchaser, Seller or any of the Acquired Companies from making any announcement in compliance with the terms and conditions of this Agreement to its respective employees, customers, suppliers, vendors and other business relations (in each case, in their capacity as such) to the extent Seller or any Acquired Company reasonably determines in good faith that such announcement is necessary or advisable (but only to the extent the context of which is consistent with that of any prior announcement made in compliance with this Section 13.18), or (b) to the extent the contents of such press release or public announcement or filing have previously been released publicly by a party or are consistent in all material respects with materials or disclosures that have previously been released publicly, in each case, without violation of this Section 13.18. The parties hereto agree that the initial press release to be issued with respect to the execution of this Agreement shall be in the form agreed to by Seller and Purchaser. Notwithstanding the foregoing, this Section 13.18 shall not apply to the disclosure of the express terms of this Agreement in any public filing required by Legal Requirement, stock exchange rules or SEC filing requirements.
- 13.19 <u>Counterparts</u>. This Agreement may be signed in any number of counterparts, including via DocuSign or other electronic signature or electronic scan copies thereof delivered

by electronic mail,	, each of which	shall be deemed	an original,	, with the	same effect	as if the	signatures	thereto	and he	ereto v	were
upon the same instr	rument.										

[Signature pages follow.]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first written above.

# TEREX CORPORATION

By: /s/Simon A. Meester
Name: Simon A. Meester

Title: President and Chief Executive Officer

[Signature Page to Transaction Agreement]

# DOVER CORPORATION

By: /s/Richard J. Tobin Name: Richard J. Tobin

Title: Chairman, President & Chief Executive Officer

[Signature Page to Transaction Agreement]

IP Seller IP

A-1

Form of Transition Services Agreement

# Exhibit C

Intellectual Property Assignment Agreement

R&W Insurance Policy

D-1

Accounting Principles

Schedule A

#### FIRST AMENDMENT TO TRANSACTION AGREEMENT

THIS FIRST AMENDMENT TO TRANSACTION AGREEMENT ("Amendment") is made and entered into as of October 8, 2024, by and between Dover Corporation, a Delaware corporation ("Seller") and Terex Corporation, a Delaware corporation ("Purchaser").

### **RECITALS**:

- A. WHEREAS, Seller and Purchaser have entered into that certain Transaction Agreement, dated as of July 21, 2024 (the "<u>Transaction Agreement</u>");
- B. WHEREAS, in accordance with Section 13.4 of the Transaction Agreement, Seller and Purchaser may amend the Transaction Agreement by written agreement; and
- C. WHEREAS, Seller and Purchaser desire to amend the Transaction Agreement on the terms and conditions hereinafter set forth.

NOW, THEREFORE, in consideration of the premises, the terms and conditions contained herein, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereby covenant and agree as follows:

- 1. <u>Definitions</u>. Capitalized terms used in this Amendment without definition shall have the meanings given to them in the Transaction Agreement.
- 2. <u>Exhibit A</u>. Exhibit A attached to the Transaction Agreement is hereby deleted in its entirety and replaced with <u>Exhibit A</u> attached to this Amendment. All references in the Transaction Agreement to such Exhibit shall be deemed to refer to the Exhibit attached to this Amendment.
- 3. <u>Miscellaneous</u>. The Transaction Agreement is amended only as expressly provided in this Amendment and shall otherwise remain in full force and effect. In the event of any conflict between the terms and provisions of this Amendment and the terms and provisions of the Transaction Agreement, the terms of this Amendment shall, in all instances, control and prevail. The provisions set forth in Article XIII of the Transaction Agreement are herein incorporated by reference, *mutatis mutandis*, as if set forth in full herein.

[Remainder of Page Intentionally Left Blank; Signature Pages Follow]

IN WITNESS WHEREOF, the parties hereto have executed this Amendment as of the date first written above.

# **DOVER CORPORATION**

By: /s/Matthew S. Gaudette
Name: Matthew S. Gaudette

Title: Vice President & Assistant Secretary

[Signature Page – Amendment to Transaction Agreement]

# TEREX CORPORATION

By: <u>/s/Scott Posner</u>
Name: Scott Posner

Title: SVP, General Counsel and Secretary

[Signature Page – Amendment to Transaction Agreement]

#### Certification

### I, Brad M. Cerepak, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Dover Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(f)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures as of the end of the period covered by this report based on such evaluation; and
  - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: October 24, 2024

/s/ Brad M. Cerepak

Brad M. Cerepak Senior Vice President & Chief Financial Officer (Principal Financial Officer)

#### Certification

### I, Richard J. Tobin, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Dover Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(f)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures as of the end of the period covered by this report based on such evaluation; and
  - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: October 24, 2024

/s/ Richard J. Tobin

Richard J. Tobin
President and Chief Executive Officer
(Principal Executive Officer)

#### Certification

### Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 with Respect to the Quarterly Report on Form 10-Q for the Period ended September 30, 2024 of Dover Corporation

Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (subsections (a) and (b) of section 1350, chapter 63 of title 18, United States Code), each of the undersigned officers of Dover Corporation, a Delaware corporation (the "Company"), does hereby certify, to such officer's knowledge, that:

- 1. The Company's Quarterly Report on Form 10-Q for the period ended September 30, 2024 (the "Form 10-Q") fully complies with the requirements of Section 13(a) of the Securities Exchange Act of 1934, as amended; and
- 2. Information contained in the Form 10-Q fairly presents, in all material respects, the financial condition and results of operations of the Company.

Dated: October 24, 2024 /s/ Richard J. Tobin

Richard J. Tobin

President and Chief Executive Officer

(Principal Executive Officer)

Dated: October 24, 2024 /s/ Brad M. Cerepak

Brad M. Cerepak

Senior Vice President & Chief Financial Officer

(Principal Financial Officer)

The certification set forth above is being furnished as an exhibit solely pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 and is not being filed as part of the Form 10-Q or as a separate disclosure document of the Company or the certifying officers.